> REPAIR AND MAINTENANCE TECHNICAL MANUAL

> > FOR

# **RADIO SET**

# AN/PRC-41A

DEPARTMENT OF THE NAVY NAVAL ELECTRONIC SYSTEMS COMMAND

Published: 1 September 1969

## LIST OF EFFECTIVE PAGES

PAGE	CHANGE IN	PAGE	CHANGE IN
NUMBERS	EFFECT	NUMBERS	EFFECT
Title Page	Original	4-1 to 4-60	Original
ii to xi	Original	5-1 to 5-116	Original
1-0 to 1-18	Original	6-1 to 6-54	Original

## COLLINS RADIO COMPANY, CEDAR RAPIDS, IOWA, Contract: N00039- 69-C - 3511

Errors found in this publication (other than obvious typographical errors), which have not been corrected by means of Temporary Corrections or Permanent Changes should be reported. Such report should include the complete title of the publication and the publication number (short title); identify the page and line or figure and location of the error; and be forwarded to the Electronics Publications Section of the Bureau of Ships.

All Navy requests for NAVSHIPS electronics publications listed in the current issue of NAVSUP Publication 2002 "Requisitioning Guide and Index of Forms and Publications," Cognizance Symbol I, or in a subsequent issue of the Electronics Information Bulletin should be directed to the appropriate Forms and Publications Supply Point.

## TM 11-5820-510-35-1 NAVELEX 0967-LP-872-5021 C 1

CHANGE

HEADQUARTERS DEPARTMENT OF THE ARMY WASHINGTON, DC, *18, August 1980* 

# Repair and Maintenance Technical Manual RADIO SET AN/PRC-41A (NSN 5820-00-889-3997)

TM 11-5820-510-35-1 and NAVSHIPS 0967-872-5020, 1 September 1969, is changed as follows:

- 1. The title of the manual is changed as shown above.
- 2. A vertical bar appears opposite changed material.
- 3. Remove and insert pages as indicated in the page list below:

Remove	Insert
4-1 and 4-2	4-1 and 4-2
5-1 and 5-2	5-1 and 5-2
5-11 and 5-12	5-11 and 5-12
5-15 and 5-16	5-15 and 5-16
5-25 and 5-26	5-25 and 5-26
None	

4. File this change sheet in the front of the manual for reference purposes.

By Order of the Secretaries of the Army, the Navy, and the Air Force:

Official:

E. C. MEYER General, United States Army Chief of Staff

J. C. PENNINGTON Major General, United States Army The Adjutant General

> EARL B. FOWLER Rear Admiral, United States Navy Commander, Naval Electronic Systems Command

DISTRIBUTION:

To be distributed in accordance with DA Form 12-51, Direct and General Support Maintenance requirements for AN/PRC-41.

# TABLE OF CONTENTS

#### Paragraph

Page Paragraph

Page

#### SECTION 1 - GENERAL INFORMATION

1-1. 1-2	Scop	e	1-1 t Supplied 1-1						
1_2.	Equi	omen	t Required But Not Supplied						
1-3.	Con		In Required But Not Supplied1-1						
1-4.	Gene								
1-5.	Desc	riptio	n of Units1-4						
	a.	Gene	eral1-4						
	b.	Radio Set AN/PRC-41A1-4							
		(1) Radio Set Case CY-3883/							
			PRC-411-4						
		(2)	Radio Receiver-Transmitter						
			RT-695A/PRC-411-5						
		(3)	Receiver-Transmitter Case						
		(-)	CY-3884/PRC-41 1-7						
		(4)	Storage Battery BB-451/U 1-7						
		( ')	(a) General 1-7						
			(b) Construction 1.9						
			(b) Colloc 1-8						
			(c) Cells						
			(d) Charge Retention (Dry)1-8						
			(e) Activation1-8						
			(f) Charge Acceptance1-8						
			(g) Storage1-8						
		(5)	Antenna AS-1404/PRC-411-8						
		(6)	Handset H-33R/PT1-9						
		(7)	Rucksack Frame1-9						
		(8)	Special Purpose Electrical						
		( )	Cable Assembly CX-10831/						
			PRC-41A1-9						
	C.	Radi	o Set Accessory Kit						
	0.	M	K-706/PRC-41 1-9						
		(1)	Electronic Equipment Case						
		(1)	CV-3885/DBC-/1 1-0						
		(2)	Dowor Supply DD 2700/						
		(2)							
		$\langle 0 \rangle$	PRC-41						
		(3)	Mountings MT-2976/PRC-41						
			and MT-2977/PRC-411-10						
		(4)	Antenna AS-1405/PRC-41						
			and Accessories1-11						
		(5)	Antenna Mounting Brackets1-11						
		(6)	Radio Frequency Cable						
			Assembly CG-55G/U1-11						
		(7)	Power Electric Cable Assembly						
		( )	CX-8686/PRC-411-11						
		(8)	Power Electric Cable Assembly						
		(0)	CX-8687/PRC-41 1-11						
		(0)	Special Purpose Electrical						
		(3)	Cable Accombly CX 9699/						
		(40)	PRC-41						
		(10)	DC Adapter Cable						
		(11)	Iviaintenance Cable Kit1-12						
	_	(12)	100k Kit1-15						
1-6.	Refe	rence	9 Data1-15						
	a.	Pow	er Requirements1-15						
	b.	Freq	uency Date1-15						
	c.	Tran	smitter Data1-15						

ORGINAL
---------

SECT	ION	1 - G	ENEF	RAL INFORMATION (Cont)
	d. e.	Rece Main	eiver I Rece	Data1-15 eiver Auxiliary Output
		Data		
	т. а	Guar	a Re	ceiver Data
	y.	Data	une	
1-7.	Facto	orv or	Field	Changes 1-18
1-8.	Prep	aratic	n for	Reshipment1-19
				·
		SEC	TION	4 - TROUBLESHOOTING
4-1.	Logi	cal Tr	ouble	shooting 4-1
	a.	Histo	orical	Data Availability 4-1
		(1)	Sym	ptom Recognition
		(Z)	Sym	pioni Elaboration
		(3)	Func	tion 4-1
		(4)	Loca	lizing the Faulty Function
		(5)	Loca	lizing Trouble to the
		( )	Circu	uit
		(6)	Failu	re Analysis4-1
	b.	Gene	eral T	rouble Isolation 4-1
		(1)	Use	of Signal Generator Pad
		$\langle 0 \rangle$	CN-3	315/URM-26 4-1
10	0.00	(2) :all Eu	Iran	sistor Servicing
4 <b>-</b> 2.	over	Radi	n Sat	ΔN/PRC-41Δ 4-5
	b.	Test	Point	is
	с.	Over	all Tr	ouble Isolation
		(1)	Test	Setup 4-9
		(2)	Test	Equipment Required 4-9
			(a)	Signal Generator AN/
		USN	I-44A	4-9
			(b)	RF Wattmeter AN/
			$(\mathbf{a})$	URM-43C
			(C)	
			(d)	Signal Generator Pad
			(u)	CN-315/urm-26
			(e)	Fuse MX-1730/U 4-9
4-3.	Over	all Ma	ain Re	eceiver Functional
	Sect	ion		
	a.	Desc	riptio	n4-11
	b.	Test	Data	
		(1)	lest	Equipment Required
			(a)	Signal Generator AN/
			(h)	Signal Generator Pad
			(0)	CN-315/URM-26 4-11
			(c)	Fuse MX-1730/U 4-11
			(d)	Output Power Meter
			. ,	ME-2/U
		(2)	Test	Setup 4-11
		(3)	Func	tional Section Performance

# **TABLE OF CONTENTS (Cont)**

Paragraph

Page

#### Paragraph

Page

#### SECTION 4 - TROUBLESHOOTING (Cont)

C.	Main	Rece	iver Circuit Description4-11
	(1)	Fund	ctional Description4-11
	. ,	(a)	RF and Power Amplifier
		. ,	Module 1A84-12
			1. RF Amplifiers V1
			Through V44-12
			2. Power Amplifier V54-12
			3. AVC Amplifier Q14-12
			4. Output Control
			Relay K14-12
		(b)	1t and 2nd IF Amplifier
			Module 1A24-12
			1. IF Amplifiers Q1 and
			Q24-12
			2. Second Receiver
			Mixer Q34-12
			3. Third Receiver Mixer
			and Transmitter
			Buffer Q64-12
			4. Relay Switching4-15
		(c)	3rd IF and Squelch Module
			1A34-15
			1. Filter 1A3FL1 and
			IF Amplifiers Q1
			Q2, and Q34-15
			2. Audio Detector Q44-15
			3. AVC Detector Q64-15
			4. AVC Amplifiers Q5
			and Q74-15
			5. DC Amplifier Q10 and
			Squelch Relay
			Amplifier Q84-15
		(d)	Audio Module 1A44-16
(2)	Test [	Data.	4-16
	(a)	Test	Equipment Required4-16
		1.	Oscilloscope AN/
			USM-105A 4-16
		2.	Audio Oscillator TS
			382B/U4-16
		3.	Signal Generator AN/
			USM-44A4-16
		4.	Signal Generator AN/
		_	URM-25D4-16
		5.	Electronic Multi
			meter IS-505/U4-16
		6.	Output Power Meter
		-	ME -2/U
		1.	Signal Generator Pad
		~	CN-315/URM-26
	(1-)	8. T	MX-1/30/U4-16
	(D).	I est	Setup4-16
	(C)	iviair	
		Se	
		ISO	auon

4-4.	Ove	rall G	uard	Rec	eiver Functional
	Se	ection	۱		
	a.	Desc	criptio	on	
	b.	Test	Data	a	
		(1)	Tes	t Equ	uipment Required4-20
			(a)	Sign	al Generator AN/
				USN	1-44A4-20
			(b)	Sign	al Generator Pad
				CN-3	315/Urm-264-20
			(c)	Fuse	• MX-1730/U4-20
			(d)	Outp	but Power Meter
		$\langle 0 \rangle$	т	ME-2	2/0
		(2)	Tes	t Set	up4-20
		(3)	Fun	Clion	
	c	Gua	rd Pa	51	
	С.	Gua	ecrint	tion	4-21
		(1)	Fun	ction	al Description 4-21
		(1)	(a)	Guai	rd Receiver Module
			(u)	14	7 4-21
				1.	RF Amplifiers Q1 and
					Q2
				2.	Oscillator Q44-21
				3.	IF Amplifiers Q5, Q6,
					Q7, Q8 and Q94-21
				4.	Detector Q104-21
				5.	Audio Amplifier Q15
					and Squelch Ampli-
				_	fier Q134-21
				6.	AVC Amplifiers Q11
		$\langle 0 \rangle$	т	4 D - 4	and Q124-22
		(2)	res	t Dat	a
			(a)	1 est	Cocilloscopo AN/
				1.	
				2	Signal Generator AN/
				۷.	LISM-44A 4-22
				3	Signal Generator Pad
				0.	CN-315/URM-26
				4.	Fuse MX- 1730/U4-22
				5.	Electronic Multimeter
					TS-505/U4-22
			(b)	Test	Setup4-22
			(c)	Gu	ard Receiver Functional
				5	Section Circuit Trouble
	_		_	.	solation4-22
4-5.	Ove	erall 7	rans	smitte	er Functional
		Sectio	٥ <u>ח</u>	·····	
	а. ь	Dest	Unptio Dote	un	
	υ.	(1)	Tech	a t Ecu	upment Required 4.24
		(1)	Tee	t Set	ирпон Кединей
		$(\frac{2}{3})$	Fun	ction	al Section Performance
		(-)	Te	st	4-24
			. •		

Section.4 - TROUBLESHOOTING (Cont)

# TABLE OF CONTENTS (Cont) Page Paragraph

Paragraph

Page

## SECTION 4 - TROUBLESHOOTING (Cont)

	-	<b>T</b>			wit Description 4.04
	C.	(1)	Smitte		Description
		(1)	Func	tional	Description4-24
			(a)	1st ar	nd 2nd IF Amplifier
				Mod	lule 1A24-24
				<u>1</u> .	Third Receiver Mixer
				i	and Transmitter
					Buffer Q6 4-24
				<u>2</u> . '	Transmitter Mixer Q44-24
				<u>3</u> .	IF Amplifiers Q1 and
					Q2 and Second Trans-
					mitter Mixer 1A9CR14-27
			(b)	RF ar	nd Power Amplifier
				Mod	lule 1A84-27
			(c) N	lodula	tor Module 1A64-27
		(2)	Test	Data.	
		. ,	(a)	Test I	Equipment Required4-27
			. ,	1.	RF Wattmeter AN/
					URM-43C4-27
				2.	Electronic Multi
					meter TS-505/U 4-27
				3	Audio Oscillator TS
				38	2B/LI 4-27
			(h)	Test	Setun 4-27
			(c)	Trans	mitter Functional
			(0)	Section	on Circuit Trouble
				Isolat	ion 4-27
4-6		Ove	rall Si	hordir	nate
<b>Ξ</b> 0.		Circi	uits		4-27
		2	 חספר	rintior	4-27
		a. h		molifi	$\frac{4}{2}$
		υ.	(1)	Fund	tional Description of
				molifi	er Assembly
				(mpiin for)	4-30
			(2)	Tool	+ Data 4-30
			(2)	(2)	Tost Equipment Pequired 4.20
				(a)	1 Signal Congrator AN/
					2 Signal Consister Red
					CIN-515/URIVI-204-50
					5. Fuse MA-1750/04-50
					4. Electronic Multi-
				(1.)	meter 1 S-505/U4-30
				(D)	1 est Setup4-30
				(C)	AF Amplifier Assembly
					(Buffer) Functional
					Section Circuit, Trouble
				<u>.</u>	Isolation4-30
		c.	Main	Chas	sis 1A94-31
			(1)	Fun	ctional Description4-31
			(a)	(a)	Receive-Transmit Switch4-31
			(2)	Tes	t Data4-31
		d.	Prim	ary Po	ower Circuits4-31
			(1)	Fun	ctional Description4-31
				(a)	Storage Battery BB-451/U4-31

		(b)	Power Supply PP-3700/	
		. ,	PRC-41	4-32
	(2)	Stora	age Battery BB-451/U,	4-32
	. ,	Test	t Data	4-33
	(3)	Pow	ver Supply PP-3700/	
		PRC	C-41, Test Data	4-33
			(a) Test Equipment	
			Required	4-33
			(b) Test Setup	4-33
			(c) Power Supply PP-3700/	
			PRC-41, Circuit	
	_		Trouble Isolation	4-33
e.	Pow	er Dis	stribution Circuits	4-33
	(1)	Fund	ctional Description	4-33
		(a)	DC Power Supply Module	
			1A1	4-36
		(b)	Filament Circuits	4-36
		(C)	Receive-transmit Relays,	
			Dial Light, and Guard	
	$\langle \alpha \rangle$	<b>T</b> 4		4-39
	(2)	Test	I Data	4-39
		(a)	Test Equipment Required	4-39
		(D)	Pewer Distribution Circuit	4-39
		(C)	Trouble legistion	1 10
f	Ero	auon	nouble isolation	4-4Z
ı.	(	Circui	ite	1-12
	(1)	Fund	ctional Description Re-	2
	(1)	ceiv	ve Operation	4-42
	(2)	Fund	ctional Description	
	()	Tra	ansmit Operation	4-42
	(3)	IF Fr	requency Calculation	4-42
	(-)	(a)	Spectrum Generator	
		• •	Module	4-42
		(b)	High-Frequency Oscilla-	
		. ,	tor 1A2Q5	4-43
		(c)	Low-Frequency Oscilla-	
			tor 1A2Q7	4-43
	(4)	Test	t Data	4-44
		(a)	Test Equipment Required	4-44
		(b)	Test Setup	4-44
		(c)	Frequency Generation	
			Circuit, Trouble	
			Isolation	4-44
g.	Co	ntrol (	Circuits	4-44
	(1)	Fund	ctional Description	4-44
		(a)	Mechanical Frequency	
		<i>.</i>	Selection	4-44
		(b)	Receive Operational	
		(.)	Switching	4-44
		(C)	I ransmit Operational	4 40
	$(\mathbf{n})$	0	Switching	4-46
	(2)	Loci		
		150	iauui	+-41

SECTION 4 - TROUBLESHOOTING (Cont)

Contents

# NAVSHIPS 0967-872-5020 TM-03816B-35/2 TM11-5820-510-35-1

# 

				TABLE OF CC	NIENI	S (C	ont)				
Paragraph			Pa	age		Paragraph					
			SECTION 4 - TROUBLESHOOT	ING (Cont)							
	h.	Omr	nidirectional and Directional			α.	Fina	I Adiu	stments	5-34	
		Ante	ennas	4-47		3.	(1)	Ger	neral		
							(-)	(a)	3rd IF and Squelch Module	5-34	
		(1)	Antenna AS-1405/PRC-41	4-47	5-4.	Re	pair				
		(2)	Antenna AS-1404/PRC-41	4-47	•	a.	Rem	ioval. I	Repair, and Replacement		
4-7.	Hist	orv of	Modifications	4-47		0	f Parts	. Mod	ules, and Units	5-35	
		, , ,				b.	Mod	ule Re	emoval		
			SECTION 5 - MAINTENA	NCE		C.	Disa	sseml	blv		
							(1)	1st	and 2nd IF Amplifier		
5-1.	Failur	e, and	d Performance and Opera-				( )	Ν	Nodule	5-35	
	tion	al Rep	ports	5-1				(a)	Z1, Z2, and Z3 Slug Rack	5-35	
5-2.	Preve	entive	Maintenance	5-1				(b)	Z4, Z5, and Z6 Slug Rack	5-37	
	a.	Ма	intenance Standards	5-1				(c)	Oscillator and Switchboard		
		(1)	General	5-1				( )	and RF Coil Assemblies	5-37	
		(2)	Test Equipment and Special				(2)	RF	and Power Amplifier		
		( )	Tools	5-1			( )	Mo	dule		
		(3)	Special Procedures	5-1				(a)	Z1. Z2. and Z3 Tuned		
		(-)	(a) Junction Box	5-1				()	Circuits	5-37	
		(4)	Reference Standard					(b)	74 and 75 Tuned Circuits	5-37	
		(.)	Procedures	5-1			(3)	Sne	ectrum Generator Module	5-37	
		(5)	Preventive Maintenance				(0)	(a)	Crystal and Switch Section		
		(0)	Procedures	5-1				(4)	S2	5-37	
		(6)	Periodic Schedule Charts					(h)	Coil Turret Section	5-38	
5-3	Tunir	nd and	Adjustment	5-27				(c)	Tuning Circuit Section	5-38	
0 0.	a	Gen	eral	5-27				(0)	Radio Receiver-Transmitter		
	u.	(1)	Test Equipment	5-27				()	RT-6954/PRC-41 Main		
		(1)	Special Tools and Cables					Ċ	These is	5-38	
		(2)	Preliminary Test Setup			Ь	Ren	air	1103515	5-30 5-38	
	h		anment of High- and Low -			۵. ۵	Rea	seomh		5-30 5-38	
	ь.		requency Oscillators of 1st			с.	(1)	1 1 1	and 2nd IF Amplifier		
		21	ad 2nd IE Amplifier Module	5-28			(1)	Modu		5-38	
		(1)	Test Equipment Required					(2)	71 72 and 73 Slug Pack	5-38	
		(1)	Alignment Setup					(a) (b)	74 75 and 76 Slug Rack	5-30	
		(2)	Procedure					(0)	Oscillator and Switchboard		
	0		Flocedule					(0)	and RE Coil		
	υ.	AllQ	ior Stagos of 1st and 2nd IF							5 20	
			malifier Module	5 29			(2)	DE	and Power Amplifier		
		(1)	Tost Equipment Poquired				(2)			5 20	
		(1)	Alianment Setup					(a)	71 72 and 73 Tuned		
		(2) (3)	Procedure	<u>5-</u> 20				(a)	Circuite	5-30	
	d		Flocedule					(h)	74 and 75 Tunad Circuita		
	u.	MI		5 20			(2)	(U) Sno	24 and 25 Turied Circuits	5-39 5 40	
		(1)	Toot Equipment Dequired				(3)	Spe (a)	Criteral and Switch S2		
		(1)	Alignment Setup					(a)	Section	E 40	
		(Z)	Anynineni Selup					(h)	Coil Turrot Soction		
		(3)						(D)			
	e.	AIIQ	unitent of KF and Power	E 04			(A)	(C)		5-40	
		(1)					(4)	RaC DT			
		(1)	Alignment Setup					۲۲۱- ۲۲۱-		E 40	
		(2)	Anyninieni Selup			£	C		1886		
	1	(3)	FIUCEOUIE	5-32		Ť.	Sync		2allUII	5-41	
	T.	All(					(1)18			- 44	
	IVIOO		Jost Fauinment Descripted	F 00				IVIOC	Jule	5-41	
		(1)	Lister and Cature	5-33							
		(2)	Alignment Setup	5-33							
		(3)	Flocedure	5-34							

# **TABLE OF CONTENTS (Cont)**

Paragraph			Page	Parag	raph	Page
	SEC	CTION 5 - MAINTENANCE (Cont)	-	-	SECTION 5 - MAINTENANCE (Cont)	•
	(2)	RF and Power Amplifier		5-6.	Lubrication	5-42
		Module	5-41	5-7.	Parts Location	5-42
	(3)	Spectrum Generator				
		Module	5-41		SECTION 6 - PARTS LIST	
g.	Replac	cement 5-41				
U	(1)	Spectrum Generator		6-1.	Introduction	6-1
	( )	Module	5-41		a. Reference Designations	6-1
	(3)	RF and Power Amplifier			b. Ref Desig Prefix	6-1
	Modul	e	5-42	6-2.	List of Units	6-1
	(3)	1st and 2nd IF Amplifier		6-3.	Maintenance Parts List	6-1.1
	. ,	Module	5-42	6-4.	List of Manufacturers	6-1.1
5-5. Vol	Itage and	Resistance		6-5.	Stock Number Identification	6-1.1
Ī	Measurem	nents	5-42	6-6.	Notes	6-1.1

# LIST OF ILLUSTRATIONS

Figure	Page SECTION 1 - GENERAL INFORMATION
1-1.	Radio Set AN/PRC-41A, Operating
1-2.	Radio Set AN/PRC-41A, Equipment Contents of Radio Set Case CY-3883/PRC-411-5
1-3.	Radio Set Accessory Kit MK-706/ PRC-41, Accessory Contents of Electronic Equipment Case CY-3885/PRC -41 1-6
1-4.	Radio Receiver- Transmitter RT-695A/PRC-41, Receiver- Transmitter Case CY-3884/ PRC-41 Displaced
1-5.	Storage Battery BB-451/U, Oblique View 1-8
1-6.	Radio Set AN/PRC-41A, Man-Pack Operation
1-7.	Power Supply PP-3700/PRC-41,
1-8.	Oblique View1-10 Mounting MT-2976/PRC-41 and MT-2977/PRC-41,
1-9.	Oblique View1-11 Antenna AS-1405/PRC-41, Mast AB-777/PRC-41 and
1-10.	Accessories1-12 Antenna Mounting Bracket 1-13
1-11	Cable Assemblies 1-13
1-12.	Maintenance Cable Kit
1-13.	Contents of Tool Kit1-15

Figure		Page
	SECTION 4 - TROUBLESHOOTING	
4-1.	Radio Receiver- Transmitter, RT-695A/ PRC-41, Module and Test-Point	
	Locations	4-2
4-2.	Transistor Base Diagrams	4-5
4-3.	Radio Set AN/PRC-41A, Overall	
	Functional Block Diagram	4-6
4-4.	Radio Receiver-Transmitter	
	RT-695A/PRC-41, Receiver Func-	
	tional Section Block Diagram	4-13
4-5.	Radio Receiver-Transmitter RT-	
	695A/PRC-41, Transmitter Func-	
	tional Section Block Diagram	4-25
4-6.	Receive-Transmit Switch, Simplified	
	Schematic Diagram	4-32
4-7.	Storage Battery BB-451/U, Typical	
	Rate of Discharge	4-33
4-8.	Power Supply PP-3700/PRC-41,	
	Test Setup	4-34
4-9.	Radio Receiver- Transmitter RT-695A/	
	PRC-41, Power Distribution,	
	Simplified Schematic Diagram	4-37
4-10.	Radio Receiver-Transmitter RT-	
	695A/PRC-41, Filament Circuits	4-39
4-11.	Main Receiver and Transmitter,	
	Functional Section, Frequency	
	Scheme	4-43
4-12.	Frequency Selection, Functional	
	Diagram	4-46
4-13.	Radio Receiver-Transmitter RT-695A/	
	PRC-44, Receiver, Functional	4.40
	Switching Diagram	4-49

Page

# LIST OF ILLUSTRATIONS (Cont) Page Figure

Figure		Page
	SECTION 4 - TROUBLESHOOTING (Cont)	
4-14.	Radio Receiver- Transmitter RT- 695A/ PRC-41, Transmitter,Functional	
4-15.	Switching Diagram Antenna AS-1405/PRC-41, Azimuthal	4-51
4-16.	Plane Voltage Pattern Antenna AS-1405/PRC-41, Voltage	4-53
4-17.	Standing-Wave Ratio Antenna AS- 1404/PRC-41, Azimuthal	4-54
4-18.	Voltage Pattern Antenna AS-1404/PRC-41, Voltage	4-55
4-19.	Standing-Wave Ratio Radio Set AN/PRC-41A, Functional	4-56
	Section Servicing Block Diagram	4-57
	SECTION 5 - MAINTENANCE	
5-1. 5-2.	Junction Box, Schematic Diagram Junction Box, Suggested Layout for Eabrication	5-2 5-2
5-3.	Examples of Periodic Schedule Charts To Be Prepared for Recording	
	Preventive Maintenance Results	5-4
5-4.	Preventive Maintenance Test, D1 and D2	5-5
5-5.	Preventive Maintenance Test, W1	5-6
5-6.	Preventive Maintenance Test, M1	5-7
5-7.	Preventive Maintenance Test, M2	5-8
5-8.	Preventive Maintenance Test, M3	5-9
5-9.	Preventive Maintenance Test. M4	5-11
5-10.	Preventive Maintenance Test, M5	5-12
5-11.	Preventive Maintenance Test, M6	5-13
5-12	Preventive Maintenance Test, SA1	5-14
5-13	Preventive Maintenance Test, SA2	5-15
5-14	Preventive Maintenance Test, SA3	5-17
5-15.	Preventive Maintenance Test, SA4 and SA5	5-19
5-16.	Preventive Maintenance Test, SA6 and SA7	5-20
5-17.	Preventive Maintenance Test, SA8 and SA9	5-22
5-18.	Preventive Maintenance Test, SA10	5-24
5-19.	Preventive Maintenance Test, SA11	5-25
5-20.	Preventive Maintenance Test, SA12	5-26
5-21.	Radio Receiver-Transmitter RT- 695A/PRC-41, Standard Test	5 00
5-22.	Main Chassis Couplers in the 225-MHz Position	
5-23.	Spectrum Generator Module, Tuning Capacitor Rotor Tab Z1, Z2, Z3, Adjustment Diagram	5 20
5-24.	Spectrum Generator Module, Rotor-Stator Coincidence at 225 MHz	5-31

	SECTION 5 - MAINTENANCE (Cont)	
5-25.	RF and Power Amplifier Module,	
	Tuning Capacitor Rotor Tab Z1,	
	Z2, and Z3, Adjustment Diagram	5-32
5-26.	RF and Power Amplifier Module,	
	Tuning Capacitor Rotor Tab Z1,	
	Z4, and Z5, Adjustment Diagram	5-32
5-27.	RF and Power Amplifier Module,	
	Rotor-Stator Coincidence at 225 MHz	5-33
5-28.	1st and 2nd IF Amplifier Module,	
	Switch and Cam Positioning Diagram	5-36
5-29.	RF and Power Amplifier Module,	
	Left Side View, Lubrication Points	5-48
5-30.	RF and Power Amplifier Module,	
	Right Side View, Lubrication Points	5-48
5-31.	1st and 2nd IF Amplifier Module, Lub-	
	rication Points	5-49
5-32.	Spectrum Generator Module, Lubrica-	
	tion Points	5-49
5-33.	Radio Receiver- Transmitter RT-695A/	
	PRC-41, Right Side View, Lubrica-	
	tion Points	5-50
5-34.	Radio Receiver- Transmitter RT-695A/	
	PRC-41. Left Side View. Lubrica-	
	tion Points	5-51
5-35.	Radio Set AN/PRC-41A, Parts List	
	Unit Designation	5-52
5-36.	Radio Set Accessory Kit MK-706/	
	PRC-41. Parts List Unit	
	Designation	5-52
5-37.	Radio Receiver-Transmitter RT-695A/	
	PRC-41 and Receiver-Transmitter-Case CY-	
	3884/PRC-41	5-53
5-38.	DC Power Supply Module, Oblique	
	View. Parts Location	5-53
5-39.	DC Power Supply Module. Bottom	
0 00.	View Parts Location	5-54
5-40.	1st and 2nd IF Amplifier Module.	
	Wired Bracket, Parts Location	5-55
5-41.	3rd IF and Squelch Module. Right	
	Side View. Parts Location	5-56
5-42.	3rd IF and Squelch Module. Left Side	
•	View Parts Location	5-57
5-43	Audio Amplifier Module, Rear View	
0.01	Parts   ocation	5-58
5-44	Modulator Module, Front View	
0 11.	Parts Location	5-58
5-45	Guard Receiver Module Right Side	
0 10.	View Parts Location	5-59
5-46	Guard Receiver Module, Right Side	
	View. Parts Location.	
5-47	Guard Receiver Module. Left Side	
	View Parts Location	5-60
5-48	Guard Receiver Module, Left Side	
5 10.	View Parts Location	5-60

# LIST OF ILLUSTRATIONS (Cont)

Figure		Page
	SECTION 5 - MAINTENANCE (Cont)	
5-49.	Guard Receiver Module, Terminal Board, Parts Location	5-61
5-50.	RF and Power Amplifier Module, Rear View, Parts Location	5-62
5-51.	RF and Power Amplifier Module,	5-63
5-52.	RF and Power Amplifier Module, Right Side View Parts Location	5-63
5-53.	Radio Receiver-Transmitter RT-695A/PRC-41, Left Side View, Parts Location	5-64
5-54.	Radio Receiver-Transmitter RT 695A/PRC-41, Front Panel, Parts Location	5-65
5-55.	Receiver-Transmitter Case CY-3884/ PRC-41 Inside View Parts Location	5-66
5-56.	Power Supply PP-3700/PRC-41, Dust Cover Displaced Parts Location	5-66
5-57.	Power Supply PP-3700/PRC-41, Bottom View, Dust Cover Removed,	
5-58.	Parts Location Power Supply PP-3700/PRC-41, Side View, Dust Cover Removed, Parts	5-67
5-59.	Location Power Supply PP-3700/PRC-41, End View, Dust Cover Removed, Parts	5-67
5-60.	Power Supply PP-3700/PRC-41,	5-68
5-61.	Power Supply PP-3700/PRC-41, Side View, Dust Cover Removed, Parts Location	5-69
5-62.	Antenna AS-1405/PRC-41, Parts Location	5-69
5-63.	Mounting MT-2965/PRC-41, Parts Location	5-70
5-64.	Mounting MT-2976/PRC-41, Rear View, Parts Location	5-70
5-65.	DC Adapter Cable, Parts Location	5-71
5-66.	Cable Assemblies, Parts Location	5-71
5-67.	Mast AB-777/PRC-41, Parts Location	5-72
5-68.	Adjustable Antenna Mast Adapter, Parts Location	5-72
5-69.	Mounting MT-2977/PRC-41, Parts	5-73
5-70	Rucksack Frame and Harness	
5-71	Antenna Mounting Bracket, Parts	
	Location	5-74
5-72.	Maintenance Cable Set, Parts Location	5-75
5-73.	DC Maintenance Adapter, Parts Location	5-76

Figure		Page
	SECTION 5 - MAINTENANCE (Cont)	
5-74.	Directional Antenna Carrying Bracket,	5-76
5-75.	1st and 2nd IF Amplifier Module, Evoloted View	5-70
5-76.	1st and 2nd IF Amplifier Module, Oscillator and Switchboard	E 70
5-77.	Assembly, Exploded View	5-79
5-78.	1st and 2nd IF Amplifier Module, Amplifier Subassembly, Exploded View	5-81
5-79.	Spectrum Generator Module, Exploded	5-82
5-80.	RF and Power Amplifier Module,	5-02
5-81.	Radio Receiver-Transmitter RT- 695A/PRC-41, Main Chassis, Exploded View	5-84
5-82	AF Amplifier Assembly (Buffer)	5-85
5-83	Connector and Filter Assembly	5-85
5-84.	Receive-Transmit Switch, Exploded	5 96
5-85.	Special Purpose Electrical Cable Assembly CX-10831/PRC-41A	5-86
5-86.	Power Supply PP-3700/PRC-41,	5-87
5-87.	DC Power Supply 1A1, Schematic	5-89
5-88.	1st and 2nd IF Amplifier 1A2, Schematic Diagram	5-01
5-89.	3rd IF and Squelch 1A3, Schematic	5.02
5 00	Audio 104 Schomatic Diagram	5 05
5-91.	Spectrum Generator 1A5, Schematic	5-33
5.02	Modulator 146 Schematic Diagram	5-97
5-92. 5-93.	Guard Receiver 1A7, Schematic	5-99
5-94.	Diagram RF amd Power Amplifier 1A8, Schematic Diagram	5-101
5-95.	Radio Set AN/PRC-41A, Intercon-	5-104
5-96.	Radio Receiver-Transmitter RT- 695A/PRC-41, Main Chassis, Wiring Diagram	5-106
5-97	Electrical Wire Code	5-107
5-98	Radio Set AN/PRC-41A Schematic	0 107
E 00	Diagram	5-109
5-99.	Diagram and Schematic	5 115
	Diayiaiii	0-115

# LIST OF TABLES

	Page
SECTION 1 - GENERAL INFORMATION	
Radio Set AN/PRC-41A, Equipment	
Supplied	1-1
Radio Set Accessory Kit MK-706/ PRC-41,	
Equipment Supplied	1-2
Equipment Required But Not	
Supplied	1-3
Contents of Maintenance Cable	
Kit	1-16
Contents of Tool Kit	1-17
SECTION 4 - TROUBLESHOOTING	
How to Avoid Damage to Transistors	4-3
While Servicing Circuits	
Test Points	4-7
Overall Trouble	
Isolation	4-10
Main Receiver Functional Section,	
Trouble Isolation	4-16
Guard Receiver	
Functional Section,	
Trouble Isolation	4-22
Transmitter Functional Section,	
Trouble Isolation4-8	
AF Amplifier Assembly (Buffer) Cir-	
cuits, Trouble Isolation	4-31
Power Supply PP-3700/PRC-41,	
	SECTION 1 - GENERAL INFORMATION Radio Set AN/PRC-41A, Equipment Supplied Radio Set Accessory Kit MK-706/ PRC-41, Equipment Supplied Equipment Required But Not Supplied Contents of Maintenance Cable Kit Contents of Maintenance Cable Kit Contents of Tool Kit SECTION 4 - TROUBLESHOOTING How to Avoid Damage to Transistors While Servicing Circuits Test Points Overall Trouble Isolation Main Receiver Functional Section, Trouble Isolation Guard Receiver Functional Section, Trouble Isolation Transmitter Functional Section, Trouble Isolation4-8 AF Amplifier Assembly (Buffer) Circuits, Trouble Isolation Power Supply PP-3700/PRC-41,

	· • · • · • • • • • • • · · · · · · · ·	
	Trouble Isolation	4-35
4-9.	Power Distribution Circuits	4-40
4-10.	Frequency Generation, Circuit	
	Trouble Isolation	4-45

Та	ble Page	е
	SECTION 5 - MAINTENANCE	
5-1.	Reference Standards Procedures5-3	
5-2.	DC Power Supply Module, Voltage	
	Measurements	
5-3.	1st and 2nd IF Amplifier Module,	
	Voltage Measurements	
5-4.	3rd IF and Squeich Module, Voltage	
	Measurements	
5-5.	Audio Module, Voltage Measurements	
5-6.	Spectrum Generator Module, Voltage	
<b>- -</b>	Measurements	
5-7.	Spectrum Generator Module, Resis-	
F 0	tance Measurements	
5-6.	Moourator Module, Voltage	
5.0	Cuard Respirer Medule, Veltage	
5-9.	Moosurements 5.45	
5-10	RE and Power Amplifier Module	
5-10.	Voltage Measurements 5-45	
5-11	RE and Power Amplifier Module	
5-11.	Resistance Measurements 5-46	
5-12	Radio Receiver-Transmitter RT	
0 12.	695A/PRC-41 Main Chassis	
	Voltage Measurements 5-46	
5-13.	Power Supply PP-3700/PRC-41.	
	Normal Voltage Measurements	
5-14.	Power Supply PP-3700/PRC-41.	
	Overload Trip Voltage Measurements	
5-15.	Lubrication Instructions	

#### SECTION 6 - PARTS LIST

6-1.	List of Units	6-2
6-2.	Maintenance Parts List	6-3
6-3.	List of Manufacturers	6-51

AN/PRC-41A GENERAL INFORMATION



Figure 1-1. Radio Set AN/PRC-41A, Operating Configuration

Original

## **SECTION 1**

## **GENERAL INFORMATION**

## 1-1. SCOPE.

Information applicable to Radio Set AN/PRC-41A is contained in the Installation and Operation Technical Manual for Radio Set AN/PRC-41A NavShips 0967-872-5010 end the Repair and Maintenance Technical Manual for Radio Set AN/PRC-41A NavShips 0967-872-5020. The format and content of these manuals is in accordance with the requirements of Military Specification MIL -M - 15071 F (SHIPS) and contract N00039-69-C-3511. The installation and operation technical manual includes sections 1, 2, and 3. The repair and maintenance technical manual includes sections 1, 4, 5, and 6. These manuals are applicable to the items and accessories of Radio Set AN/PRC-41A. Section 1 contains a general description, provides reference data, lists the applicable items and accessories, and provides information required for the preparation for reshipment of the Radio Set AN/PRC-41A equipment. Section 2 provides installation information such as unpacking, power requirements for operation, site selection, inspection and adjustment, and interference reduction. Section 3 contains operating instructions. Section 4 provides theoretical description and test data of the functional sections and subordinate circuits of the equipment. Section 5 provides preventive maintenance instructions, repair information, and illustrations which are applicable to all sections such as part location, exploded views, connection diagrams, and overall equipment schematics. Section 6 lists the detail parts of the items and accessories of Radio Set AN/PRC-41A. These technical manuals are in effect upon receipt. Extracts from these publications may be made to facilitate the preparation of other Department of Defense publications.

## 1-2. EQUIPMENT SUPPLIED.

Radio Set AN/PRC-41A consists of the equipment contained in Radio Set Case CY-3883/PRC-41. Refer to table 1-1 and figure 1-2. Radio Set Accessory Kit MK-706/PRC-41 consists of the equipment contained in Electronic Equipment Case CY-3885/PRC-41. Refer to table 1-2 and figures 1-1 and 1-3.

## 1-3. EQUIPMENT REQUIRED BUT NOT SUPPLIED.

The equipment required for maintenance of Radio Set AN/PRC-41A is listed in table 1-3.

ITEM QTY		NOMENCLATURE		OVE	OVERALL DIMENSIONS (in)			WT (lb)
		NAME	DESIGNATION	Н	W	D	. ,	
1	1	Radio set case (less contents)	CY-3883/PRC-41	15-1/2	35	26-1/2	14,376	82.5
2	1	Radio receiver-transmitter (without case (CY-884/ PRC-41)	RT-695A/PRC-41	14	11-5/16	4-9/16	812	18.5
3	1	Receiver-transmitter case	CY-3884/PRC-41	12-5/8	11-3/16	4-15/32	635	3.6
4	3	Storage battery	BB-451/U	7-11/16	11-9/16	4-9/64	402	16.0
5	1	Antenna	AS-1404/PRC-41	23-1/2	1-3/4 dia			.07
6	1	Handset	H-33E/PT (GFE)	2	8	3-3/8		1.2
7	1	Rucksack frame	(GFE)	19	17	6-1/4		1.5
8	1	Harness set						3.1
9	2	Installation and operation technical manual	NavShips 0967- 872-5010 TM-03816B-12/1 TM-11-5820-510- 12-1	8-1/2	11	1-1/2		
			1_1					

## TABLE 1-1. RADIO SET AN/PRC-41A, EQUIPMENT SUPPLIED

# TABLE 1-1. (Continued)

ITEM	QTY	NOMENCLATURE			ERALL DIMENS	BIONS	VOL	WT
		NAME	DESIGNATION	Н	W	D	(in <sup>3</sup> )	(lb)
10	2	Repair and Maintenance technical manual with repair part list	NavShips 0967- 872-5020 TM-03816B-35/2 TM11-5820-510-35-1	8-1/2	11	1-3/4		
11	1	Special purpose electrical cable assembly	CX-10831/PRC-41A	24				

# TABLE 1-2. RADIO SET ACCESSORY KIT MK-706/PRC-41, EQUIPMENT SUPPLIED

		NOMENC	LATARE	OVE	ERALL DIMEN	ISIONS		
ITEM	QTY				(in)			WT (lb)
				Н	W	D	(,	()
		NAME	DESIGNATION			_		
1	1	Electronic equipment case (less contents)	CY-3885/PRC-41	15-1/2	35	26-1/2	14,376	81.0
2	1	Power Supply	PP-3700/PRC-41	7-47/64	10-15/16	4-5/32	355	17.6
3	1	Antenna	AS-1405/PRC-41	3-1/2	30-27/32	27-11/16		6.3
4	1	Mast	AB-777/PRC-41	78-1/4	3-1/2			3.0
				28-1/4	3-1/2			
5	1	Adjustable antenna mast		10-3/4	closed			1.0
6	1	Antenna mounting bracket		4-13/16	8-5/8	1-7/8		0.7
7	1	Directional antenna		1*-23/32	31-17/32	1-3/4		4.8
8	3	Guy rope accessory		10 ft				1.5
9	3	Guv stake		10	1	1		2.1
10	1	Bag (for guy stake)		12	5			0.1
11	1	Mounting	MT-2976/PRC-41	6-1/8	17-3/4	11-13/16	1230	6.6
12	1	Mounting	MT-2977/PRC-41	17/32	11-1/2	5-3/8	31	2.1
13	1	Radio frequency cable assembly	CG-55g/U	20 FT				2.4
14	1	Power electric cable	CX-8686/PRC-41	10 ft				2.8
15	1	Power electric cable assembly	CX-8687/PRC-41	50 ft				5.4
16	1	Special purpose electrical cable assembly	CX-8688/PRC-41	10 ft				1.0
17	1	Dc adapter cable		27				0.3
18	1	Maintenance cable kit		12	5			
19	1	Tool kit		7-1/4	3-1/2			0.8

\*Unless otherwise stated, all overall dimensions are in inches.

## **GENERAL INFORMATION**

# AN/PRC-41A NAVSHIPS 0967-872-5020 TM-03816B-35/2 TM11-5820-510-35-1

QTY	NOMECLATURE			
PER			REQUIRED USE	RADIO SET AN/PRC-41
EQUIP	NAME	DESIGNATION		
1	Signal generator	AN/USM-44A or AN/URM-26B	Guard and main receiver functional section trou- bleshooting and main- tenance procedures	Frequency range: 225 to 400 MHz. Type of signal: AM. Modulation: 1000 Hz internal with external provisions
1	Signal generator	AN/URM-25D	Troubleshooting and maintenance procedures	Frequency range 10 Hz to 50 MHz. Type of signal: AM Output voltage: 0.1 to 100,000 microvolts .
1	Rf wattmeter	AN/URM-43C or TS-1389	Power output measure- ments	Power output: 0 to 5 watts. Impedace: 50 ohms.
1	Oscilloscope	AN/USM-105A	Troubleshooting and maintenance procedures.	General purpose.
1	Output power meter or Output meter or Output meter	ME-2/U or ME-184 or TS-585B/U or ME-6C/U	Audio output measure- ments.	Power output: 0 to 1 watt. Impedance: 300 ohms.
1	Audio oscillator	TS-382B/U or AN/URM-127	Troubleshooting and maintenance procedures.	Frequency: 100 to 20,000 Hz. Output: 0 to 2 volts.
1	Fuse	MX-1730/U (p/o AN/ USM-44)	Attenuator protection for signal generator.	Rf fuse: 225-to 400-MHz range.
1	Multimeter	AN/PSM-4C	Troubleshooting and maintenance procedures.	Voltage ranges: 0 to 250 volts dc. 0 to 250 volts ac, ohmmeter section.
1	Electronic mul- timeter	TS-505/U or AN/USM-116	Troubleshooting and maintenance procedures	Voltage ranges: 0 to 250 volts dc. 0 to 250 volts ac, ohmmeter section.
1	Frequency counter	AN/USM-122 or CAQI-524D	Frequency measurement. to extend the frequency	
1	Electronic fre- quency con- verter	CV-394/USA-5 or CAQI-525C	Used with AN/USM-122 measuring capabilities.	Measure frequencies in 20-to 30-MHz and 225- to 400-MHz ranges.
1	Transfer osci- llator	CM-102/USM-73	Used with AN/USM-122 to extend the frequency measuring capabilities.	
1	Adjustable atten- uator	CN-318/G or CAG-874-GA	An attenuator and T- connector used for modulation and signal measurements.	Rf attenuator to provide attenu- ation for high-level outputs.

## TABLE 1-3. EQUIPMENT REQUIRED BUT NOT SUPPLIED

# TABLE 1-3. (Continued)

QTY PER	NOMENCLATURE		REQUIRED USE	RADIO SET AN/PRC-41
EQUIP	NAME	DESIGNATION		EQUIPMENT CHARACTERISTICS
1	Signal generator pad	CN-315/URM-26 (p/o AN/URM- 26B) or C BSH-50-6	Used for impedance matching between sig- nal generator and radio equipment,	Pad attenuation: 6 db. Impedance: 50-ohm input and output. Frequency range: 225 to 400 MHz.
1	Power resistor	240C	Used for loading Power Supply PP-3700/PRC-41 when performing tests.	Variable 0- to 20-ohm resistor, 100 watt.
	test set	TS-TIUGA/U	transistors.	
1	Electron tube test set	TV-7D/U	Used for checking electron tubes.	
1	Coaxial crystal detector	HP-420A (Hew- lett-Packard)		Rf crystal for detection of modu- lated rf signals.
1	Junction box	To be fabricated. Refer to section 5		
1	Battery charger	PP-3240/U, PP-4567/U, PP-6241/U.	Used for charging radio battery.	

# 1-4. GENERAL DESCRIPTION.

Radio Set AN/PRC-41A is a lightweight, portable uhf receiver-transmitter equipment. The versatility of this equipment permits man-pack, fixed station, or vehicular operation. See figure 1-1. The items and accessories of AN/PRC-41A and Accessory Kit MK-706/PRC-41A are listed in tables 1-1 and 1-2. This equipment is stored for transit in Radio Set Case CY-3883/PRC-41 and Electronic Equipment Case CY-3885/PRC-41. See figures 1-2 and 1-3. CY-3883/PRC-41 contains the items necessary for man-pack operation. CY-3885/PRC-41 contains the accessory items required for fixed station or vehicular operation, and maintenance items such as tools and extension cables. Radio Receiver-Transmitter RT-695A/PRC-41 of AN/PRC-41A may be operated on any one of 1750 channels, spaced 100kHz in the 225.0-to 399.9-MHz range. In transmit operation, the equipment uses type a3 (AM voice) emission and provides an average of at least 3 watts of power across the frequency range. All controls are an integral part of the front panel of RT-695A/PRC-41. These controls provide frequency selection, application of power, and adjustment of volume and squelch levels. Automatic relay operation in normal voice mode may be provided when two RT-695A/PRC-41 equipments are operated together. Cabling is provided for all modes of operational installations. RT-695A/PRC-41 has the capability of secure voice operation when used in conjunction with TSEC/KY-38 equipment. Primary power may be provided by either Storage Battery BB-451/U, Power Supply PP -3700/PRC -41, or vehicular battery supply. The equipment may use Antenna AS-1405/PRC-41 (directional) or Antenna AS-1404/ PRC-41 (omnidirectional).

## 1-5. DESCRIPTION OF UNITS.

**a. GENERAL.** - Radio Set AN/PRC-41A consists of Radio Set Case CY-3883/PRC-41 and its contents. Refer to figure 1-2. Radio Set Accessory Kit MK-706/PRC -41 consists of Electronic Equipment Case CY-3885/PRC-41 and its contents. Refer to figure 1-3. The following paragraphs provide physical, electrical, and functional descriptions of the case and contents of Radio Set Case CY-3885/PRC-41 and Electronic Equipment Case CY-3885/PRC-41.

## b. RADIO SET AN/PRC-41A.

(1) RADIO SET CASE CY-3883/PRC-41. Radio Set Case CY-3883/PRC-41 is an aluminum transit case with a foam rubber compartmentalized insert used for storage of the items of AN/PRC-41A

# AN/PRC-41A GENERAL INFORMATION

## NAVSHIPS 0967-872-5020 TM-03816B-35/2 TM11-5820-510-35-1



Figure 1-2. Radio Set AN/PRC-41A, Equipment Contents of Radio Set Case CY-3883/PRC-41

that are required for man-pack operation. It measures 15-1/2 by 35 by 26-1/2 inches and weighs 45 pounds. Refer to figure 1-2.

(2) RADIO RECEIVER-TRANSMITTER RT695A/PRC-41. Radio Receiver-Transmitter RT695A/PRC-41 consists of a main chassis with eight plug-in modules, a control panel, and a waterproof dust cover (Receiver-Transmitter Case CY-3884/PRC-41). Refer to figure 1-4. The modules can be removed from the main chassis by loosening captive screws on each module. All electrical connections between the modules and the main chassis are made through multipin connectors on each module to jacks mounted on the main chassis. The coaxial cables and the connectors, that are integral with the multipin connectors except for the power amplifier output, conduct the rf signals between the plug-in modules. Mechanical coupling for tuning functions is achieved by a mechanical gear train and couplers mounted on the main chassis, and the couplers of the mechanically tuned modules. The main chassis gear train is driven by the frequency selector controls on the front panel. Pressure contacts located at the rear of the main chassis, provide the electrical connections to the CY-3884/PRC-41 for supplying the primary 26.5-volt dc supply from either Storage Battery BB-451/U or Power Supply PP-3700/PRC-41. A coaxial connector, located on the front panel mates with Antenna AS-



Figure 1-3. Radio Set Accessory Kit MK-706/PRC-41, Accessory Contents of Electronic Equipment Case CY-3885/PRC-41

Paragraph 1-5b(2)

# NAVSHIPS 0967-872-5020 TM-03816B-35/2 TM11-5820-510-35-1

1404/PRC-41 (omnidirectional) or with Radio Frequency Cable Assembly CG-55G/U when using Antenna AS-1405/PRC-41 (directional). The front panel also mounts a function switch, for application of power, volume control, squelch control, three frequency selector controls, a window which indicates the frequency to which the equipment is tuned, one audio connector for Handset H-33E/PT, and one connector to provide interface with TSEC/KY-38 security equipment.

(3) RECEIVER-TRANSMITTER CASE CY-3884/PRC-41. Receiver-Transmitter Case CY-3884/PRC-41 is a dust cover for Radio Receiver-Transmitter RT-695A/PRC-41. Refer to figure 1-4.The CY-3884/PRC-41 is secured to the RT-695A/PRC-41 by four captive screws which are located at the rear of the CY-3884/PRC-41. With the CY-3884/PRC - 41 in place on the RT-695A/PRC -41, primary power is applied through pressure contacts located at the rear corners to the contacts centrally located on the inside rear of the CY-3884/PRC-41. The power is then available to the pressure contacts at the rear of the main chassis and then to the power distribution circuits of the RT-695A/PRC-41.

(4) STORAGE BATTERY BB-451/U. (Refer to figure 1-5.)

(a) GENERAL. Storage Battery BB451/U is a silver-zinc alkaline 24-volt (nominal) rechargeable storage battery. It is used to provide primary power to RT-695A/PRC-41 in a man-pack operation or where no other suitable



Figure 1-4. Radio Receiver-Transmitter RT-695A/PRC-41, Receiver-Transmitter Case CY-3884/PRC-41 Displaced

 
 Paragraph 1-5b(4)(a)
 NAVSHIPS 0967-872-5020 TM-03816B-35/2 TM11-5820-510-35-1
 AN/PRC-41A GENERAL INFORMATION

Figure 1-5. Storage Battery BB-451/U, Oblique View

is available. Storage Battery BB-451/U is secured directly on the rear of RT-695A/PRC-41 with clamps provided on either side of the battery case. Electrical connection is accomplished through pressure contacts in the top of the battery case to contacts on the rear of CY-3884/PRC-41. Three Storage Battery BB-451/U units are contained in CY-3883/PRC-41.

(b) CONSTRUCTION. - The BB-451/U is contained in a watertight fiber glass case with a stainless steel cover plate. After removing the cover plate, the cells may be removed in groups of four using ordinary hand tools. A plastic sheet for recording charging history is bonded to the underside of the cover plate. To avoid a long formation and activation period, the BB-451/U is supplied in a dry charge condition.

(c) CELLS. - The BB-451/U battery consists of 16 series-connected silver-zinc cells constructed in blocks of four cells each. Nominal open-circuit voltage of a fully charged cell is 1.86 volts per cell. The minimum voltage per cell under specified load is 1.375 volts at -11 °C (+12 °F) and above, and 1.250 volts from -10 to -25 °C (+14 to -13 °F).

(d) CHARGE RETENTION (DRY). Charge retention of a dry charged cell over the specified periods will be in accordance with the minimum values indicated below:

2 years at +30 °C (+86 °F) and below -98 percent (19.6 ampere-hours)

1 year at +50 °C (+122 °F) and below - 80 percent (16 ampere-hours)

3 months at +65 °C (+149 °F) and below - 50 percent (10 ampere-hours)

(e) ACTIVATION. - The cells of the silver-zinc battery will deliver the specified ampere-hour capacity after a 24-hour soak in the electrolyte, a 40-percent solution of potassium hydroxide and other additives furnished by the manufacturer. A booster charge is permissible if the temperature and storage limits of paragraph 1-5b(4)(d) have been exceeded.

(f) CHARGE ACCEPTANCE. The battery cells will perform according to specifications after charging at a modified constant current of 2.5 amperes average with voltage cutoff at 2.03 volts per cell. Emergency charging of the cells is possible by charging at a constant potential of 2.03 volts per cell with charge acceptance as follows: a 0.5-hour charge period replaces 50 percent of the capacity removed on previous discharge; a 4-hour charge replaces 60 percent of the capacity removed on the previous discharge. All recharging must be with Battery Charger PP-3240/U, PP-3906/U, or an exact replacement.

(g) STORAGE . - Dry charged cells are filled with an inert gas by the manufacturer and may be stored under the following conditions:

-65 to +65 °C (-85 to +149 °F)
100 percent
Any
40,000 ft

Wet charged cells should be stored in an upright position. For further detailed information concerning Storage Battery BB-451/U, see TM-04072A-15/1.

(5) ANTENNA AS-1404/PRC-41. (Refer to figure 1-6.) - Antenna AS-1404/PRC-41 is an omnidirectional antenna for use with RT-695A/ PRC-41. The AS-1404/PRC-41 is used for transmission and reception of signals in the 225- to 399.9-MHz range without electrical or physical adjustments. It may be mounted directly on antenna connector J28 on the front panel of RT-695A/PRC-41 or the connector on the antenna mounting bracket on Mounting MT-2976/ PRC-41 or, when mounted on Mast AB-777/PRC-41, it must be connected through Adapter UG-29B/U and Radio Frequency Cable Assembly CG-55G/U to the RT-695A/PRC -41.

(6) HANDSET H-33E/PT. (Refer to figure1-6.) - Handset H-33E/PT consists of a handpiece a length of cable, and a connector that mates with theH33E/PT connector J13 on the front panel of RT-695A/PRC-41. The H-33E/PT contains a carbon microphone, an electromagnetic earphone, and a press-to-talk button. In operation, pressing the press-to-talk button places the RT-695A/PRC-41 equipment in transmit operation; releasing the button returns the equipment to receive operation.

(7) RUCKSACK FRAME. (Refer to figure1-6.) - The rucksack frame is government furnished equipment. The rucksack frame and associated straps and harness provide a back-pack frame for carrying Radio Receiver-Transmitter RT-695A/PRC-41,Storage Battery BB-451/U, Handset H-33E/PT, and Antenna AS-1404/PRC-41 in man-pack operation.

(8) SPECIAL PURPOSE ELECTRICAL CABLE ASSEMBLY CX-10831/PRC-41A. (Refer to figure 1-3.) - The CX-10831/PRC-41A is a 2-footlength of 7-conductor cable with suitable mating connectors. It is used to connect RT-695A/PRC-41 to the TSEC/ KY-38 security equipment.

c. RADIO SET ACCESSORY KIT MK-706/ PRC-41.

(1) ELECTRONIC EQUIPMENT CASE CY3885/PRC-41.-Electronic Equipment





Figure 1-6. Radio Set AN/PRC-41A, Man-Pack Operation Configuration



Figure 1-7. Power Supply PP-3700/PRC-41, OBLIQUE VIEW

Case CY-3885/PRC-41 is a compartmentalized aluminum transit case used for storage of accessories of Radio Set AN/PRC-41A. The contents of CY-3885/PRC-41 permit operation of the AN/PRC-41A equipment in a fixed station or vehicular configuration, and provide extension cables and tools required for equipment maintenance. The case measures 15-1/2 by 35 by 261/2 inches and weighs 45 pounds. Refer to figure 1-3.

(2) POWER SUPPLY PP-3700/PRC-41. (Refer to figure 1-7.) - The circuits of Power Supply PP-3700/PRC-41 are contained in an aluminum case having the same dimensional configuration as Storage Battery BB-451/U. Power Supply PP-3700/PRC-41 is used to provide the 26.5-volt dc primary power for Radio Receiver-Transmitter RT-695A/PRC-41 where either 115- or 230-volt, 50- to 400-Hz power is available. Refer to paragraph 2-5b(1) of the Installation and Operation Technical Manual for Radio Set AN/PRC-41A, NavShips 0967-872-5010. This component may be clamped directly to the rear of the RT-695A/PRC -41, or its primary power may be made available by use of Power Electric Cable Assembly CX-8686/PRC-41 connected between dc connector J2 on the side of the PP-3700/PRC-41 and the rear of Mounting MT-2976/PRC-41. When operating the RT695A/PRC-41 on the test bench with Receiver-Transmitter Case CY-3884/PRC-41 removed, primary power may be applied to the RT-695A/PRC-41 through the dc maintenance adapter. Power Supply PP-3700/ PRC-41 may also be mounted on Mounting MT-2977/ PRC -41.

(3) MOUNTINGS MT-2976/PRC-41 AND MT-2977/PRC-41. (Refer to figure 1-8.) - Mounting MT-2976/PRC-41 is used to mount Radio Receiver-Transmitter RT-695A/PRC-41 in a vehicular installation. Power Electric Cable Assembly CX-8686/ PRC-41 may be connected to the input connector at the rear of MT-2976/PRC-41that contains circuits to prevent voltage polarity reversal from damaging the RT-695A/PRC-41 and filter circuits for input filtering of the primary power voltage. Primary power is taken from the input connector at the rear of the MT-2976/PRC -41, fed through the mounting circuits, and made available to the connectors at the front of the rear bracket of the MT-2976/PRC-41. These connectors mate with those at the rear of Receiver-Transmitter Case CY-3884/PRC-41. Two takeup fasteners are located at the front of the MT-2976/PRC-41 that clamp over the protection handles on the front panel of the RT-695A/PRC-41 and hold the contacts at the rear of the CY-3884/ PRC-41 to the contacts at the front of the rear bracket of the MT-2976/PRC-41 for vertical positioning of the antenna when the RT-695A/PRC -41 and MT-2976/PRC-41



Figure 1-8. Mounting MT-2976/PRC-41 and MT-2977/PRC-41, Oblique View

are installed in a horizontal position on a vehicle. Mounting MT-2977/PRC-41 is used normally in a fixed station installation for mounting Power Supply PP-3700/PRC -41.

(4) ANTENNA AS-1405/PRC-41 AND ACCESSORIES. (Refer to figure 1-9.) -Antenna AS-1405/PRC-41 is a directional antenna for use with Radio Receiver-Transmitter RT-695A/PRC-41. The AS-1405/PRC-41 is used for transmission and reception of signals in the 225- to 399.9-MHz range. The elements of AS-1405/PRC-41 are made collapsible for storage but are extended to the maximum length for use in the 225- to 399.9-MHz range. Refer to figure 1-9. The AS-1405/PRC-41 is a planar log periodic dipole array that provides unidirectional radiation characteristics, and mounts on an adjustable antenna mast adapter and Mast AB-777/PRC-41. Guy ropes and stakes permit AB-777/PRC-41 to be staked in place in a fixed station installation. The adjustable antenna mast adapter permits elevation orientation of AS-1405/PRC-41. Radio Frequency Cable Assembly CG-55G/U connects between AS-1405/PRC-41 and the antenna connector on the front panel of RT-695A/PRC-41. The AS-1405/PRC-41 may be stored in the directional antenna carrying bracket by collapsing the adjustable elements.

(5) ANTENNA MOUNTING BRACKET. (Refer to figure 1-10.) - The antenna mounting bracket permits vertical mounting of Antenna AS-1404/PRC-41 (omnidirectional) in a vehicular installation. It is normally mounted on the rear of Mounting MT-2976/ PRC-41 but may be secured to any flat vertical surface by four screws or bolts. The antenna mounting bracket has a UG-30D/U feedthrough connector. The upper part of this connector fits AS-1404/PRC-41, end the lower part mates with a connector on Radio Frequency Cable Assembly CG-55G/U.

(6) RADIO FREQUENCY CABLE ASSEMBLY CG-55G/U. (Refer to figure 1-11.) - Radio Frequency Cable Assembly CG-55G/U is a 20-foot length of RG-213/U coaxial cable with mating connectors used for connection between the antenna connector on the front panel of RT-695A/PRC-41 and AS-1405/PRC-41 or AS-1404/PRC-41 with adapter UG-29B/U.

(7) POWER ELECTRIC CABLE ASSEMBLY CX-8686/PRC-41. (Refer to figure 1-11.) - Power Electric Cable Assembly CX-8686/PRC-41 is a 10-foot length of a 3-wire electrical cable with mating connectors used for connection of Power Supply PP-3700/ PRC-41 to Mounting MT-2976/PRC-41 or to dc maintenance adapter for RT-695A/PRC-41 when CY-3884/ PRC-41 is removed.

(8) POWER ELECTRIC CABLE ASSEMBLY CX-8687/PRC-41. (Refer to figure 1-11.) - Power, Electric Cable Assembly CX-8687/PRC-41 is a 50-foot length of 3-wire cable with suitable mating connectors used to connect Power Supply PP-3700/PRC-41 to 115-volt, 50- to 400-Hz primary supply.

Paragraph 1-5c(8)

(9) SPECIAL PURPOSE ELECTRICAL CABLE ASSEMBLY CX-8688/PRC-41. (Refer to figure 1-11.) -Special Purpose Electrical Cable Assembly CX-8688/PRC-41 is a 10-foot length of 7-conductor cable with suitable mating connectors used to connect the H33E/PT connector on one Radio Receiver-Transmitter RT-695A/PRC-41 to the H33E/PT connector on another RT-695A/PRC-41 for automatic relay operation. Refer to paragraph 2-4e of the Installation and Operation Technical Manual for Radio Set AN/PRC-41A. NavShips 0967-872-5010.

(10) DC ADAPTER CABLE. (Refer to figure 1-11.) - The dc adapter cable is a 2-wire cable with a suitable connector on one end and terminal lugs on the other, and is used in a vehicular installation to connect power between the vehicle dc power source and the rear connector of Mounting MT-2976/PRC-41 through Special Purpose Electric Cable Assembly CX-8688/PRC - 41.

(11) MAINTENANCE CABLE KIT. (Refer to figure 1-12) - The maintenance cable kit is composed of five multipin cables, one Microdot cable, two Conhex cables, a BNC -to-Conhex connector, a Conhex-to-Conhex connector. and a dc maintenance adapter.



Figure 1-9. Antenna AS-1405/PRC-41, Mast AB-777/PRC-41 and Accessories







Figure 1-11. Cable Assemblies

## AN/PRC-41A GENERAL INFORMATION





The multiwire and Microdot cables are used for operating the modules of RT-695A/PRC-41 extended from the main chassis while performing maintenance procedures. The Conhex cables, BNC-to-Conhex, and Conhex-to-Conhex connectors provide adaptations for connection of test equipment. With Receiver-Transmitter Case CY-3884/PRC-41 removed from RT-695A/PRC-41, the dc maintenance adapter may be secured to the rear of the RT-695A/PRC-41. This permits application of primary power while perform ing maintenance procedures. The dc maintenance adapter is a 3-wire pressure contact to connector adapter. Refer to table 1-4.

(12) TOOL KIT. (Refer to figure 1-13.) - The tool kit contains the special tools supplied for adjustment and maintenance of RT-695A/PRC-41. Refer to table 1-5.

### **1-6. REFERENCE DATA**.

a. POWER REQUIREMENTS. - Normal power requirements are 26.5 volts dc, 10 percent (equipment will operate with an input voltage as low as 22 volts dc When using ac Power Supply PP-3700/PRC-41, the input voltage is 115 or 230 volts <u>+</u>10 percent, 50 to 400 Hz single phase. AC Power Electric Assembly Cable CX-8687/PRC-41 is normally connected for 115 volts, requiring cable fabrication for operation from a 230-volt source. Refer to paragraph 2-5b(1) of the Installation and Operation Technical Manual for Radio Set AN/PRC-41A, NavShips 0967-872-5010, for 230-volt operation.

- b. FREQUENCY DATA.
  - (1) Range: 225.0 to 399.9 MHz.
  - (2) Channels: 1750 spaced at 100-kHz intervals over the range.
  - (3) Stability: +15 kHz
  - (4) Type of frequency control: Crystal.
  - (5) Number of crystals: 39.
  - (6) Type of transmission and reception: A3 or A9.
- c. TRANSMITTER DATA.
  - (1) Power output: 3 watts or greater average unmodulated power into a 50-ohm load.
  - (2) Modulation: AM.
  - (3) Modulation sensitivity: Carbon microphone input of 0.7 volt.
  - (4) Modulation capability: 80 percent minimum (adjusted to clip between 70 to 90 percent).
  - (5) Transmitter fidelity: +4 dB, 300 Hz to 20 kHz (from 1000-Hz reference).
  - (6) Transmitter distortion: Less than 10 percent with modulation 3 dB below clipping level.
  - (7) Duty cycle: 1-minute transmit, e-minute receive, extended transmit cycle permissible.
- d. RECEIVER DATA.

(1) Sensitivity: A signal having an average level of 3 K V modulated 30 percent at 1000 Hz produces at least 7 mW at a signal-plus-noise to noise (s+n)/n ratio of 10 dB or greater.



Figure 1-13. Contents of Tool Kit

## TABLE 1-4. CONTENTS OF MAINTENANCE CABLE KIT

ITEM NUMBER	DESCRIPTION AND PURPOSE
1	Canvas bag, container for cables and adapters
2	Extension cable W2 IS an 18-inch cable terminated m Cannon DBM-F13C3P (C27) and Cannon DBMF-K13C3S(C27)-1, having 10 pins and 3 coaxial connections. Extension cable We IS used with 1st and 2nd If amplifier module.
3	Extension cable W3 IS an 18-inch cable terminated In Cannon DAM-H11C1P (C27) and Cannon DAMF-KIICIS(C27)-1, having 10 pins and 1 coaxial connection Extension cable W3 Is used with spectrum generator and quard receiver modules.
4	Extension cable W1 IS an I8-inch cable terminated m Cannon DBM-21W1P (C27) and Cannon DBM-21W1S, having 15 pins and 1 coaxial connection. Extension cable W1 IS used with rf
5	Extension cable W4 IS an 18-inch cable terminated In Cannon DA-15P-C7 and Cannon DAF-15S-C7 connectors, having 15 pins. Extension cable W4 Is used with 3rd If and squelch module.
6	Extension cable W5 IS an 18-inch cable terminated In Cannon DE-9P and Cannon DEF-9S, having nine pins Extension cable W5 IS used with do power supply, modulator, and audio modules.
7	Adapter At IS a BNC-to-Conhex adapter required for connection of test equipment to Conhex cables In RT-695A/PRC-41.
8	Adapter A2 IS a Conhex-to-Conhex adapter required for connection of Conhex cables to ex tend their length for connection of test equipment.
9	Extension cable W7 IS an 8-inch Conhex cable terminated In one Conhex 33-01 connector used for connection of test equipment
10	Extension cable W6 IS an 8-inch Conhex cable terminated In two Conhex 33-01 connectors used for extension of Conhex cables and connection of test equipment.
11	Extension cable We IS an 18-inch coaxial cable terminated In one Microdot 51-258 connector and one Microdot 3242 connector used for rf and power amplifier module.
12	Dc maintenance adapter As Is used to provide power to RT-695A/PRC-41 when operated with CY-3884/PRC-41 removed. Adapter AS secures to rear of RT-695A/PRC-41 by use of two captive screws Electrically, the do maintenance adapter has three pressure contacts like those In rear of CY-3884/PRC-41. These make contact with three pressure contacts on rear of RT-695A/PRC-41 and are connected to a 3-pin plug on opposite side of dc maintenance adapter bracket. This plug mates with CX-8686/PRC-41.
13	Adapter UG-29B/U permits connection of Radio Frequency Cable Assembly CG-55G/U to Antenna AS-1404/PRC-41 when used with Mast AB-777/PRC-41.

## AN/PRC-41A GENERAL INFORMATION

## NAVSHIPS 0967-872-5020 TM-03816B-35/2 TM11 -5820-510-35 -1

# TABLE 1-5. CONTENTS OF TOOL KIT

ITEM NUMBER	NAME	PURPOSE
1	Canvas bag	Container for tools.
2	Spanner wrench	Used for removing antenna connector J28 Handset H-33E/ PT, connector J13, and KY-38 connector J14 on front panel of Radio Recelver-Transmitter RT-695A/PRC-41, and antenna connector on antenna mounting bracket
3	Brush	Used to clean connectors
4	Screwdriver	Used for adjustment of trImpots and tuning capacitor of 1st and 2nd if amplifier, 3rd If and squelch, guard receiver, spectrum generator, modulator, and audio modules.
5	T-handle wrench	Used for loosening captive screws on modules of RT-695A/ PRC -41
6	Cross-recessed screwdriver	Used for removal and replacement of module side cover screws and for removal of main chassis back plate of R T-695A/PRC -41
7	Alignment tool	Used for adjustment of If coils m guard receiver
8	Multiple spline socket wrench # 8	Used to tighten all coupler clamps on main chassis gear tram.
9	Multiple spline socket wrench	Used for setscrews on all knobs on front panel and setscrews In gears of main chassis of RT-695A/PRC-41
10	Multiple spline socket wrench #4	Used for setscrews In main chassis gear tram, as required
11	Multiple spline socket wrench #2	Used for adjustment of rotor and grounding hub setscrews in rf and power amplifier, and spectrum generator modules.
12	Alignment tool	Used for ad]justing trimmer capacitors of rf and power amplifier module.
13	Screwdriver	Used for loosening and tightening captive screws of dust cover of Power Supply PP-3700/PRC-41, Storage Battery BB-451/U, and Recelver-Transmitter Case CY-3884/PRC-41
14	Alignment tool	Used for tracking rf and power amplifier and spectrum generator modules.

Paragraph 1-6d(2)

### NAVSHIPS 0967-872-5020 TM-03816B-35/2 TM11 -5820-510-35-1

(2) Selectivity: 6 dB, 80 kHz minimum; 60 dB, 150 kHz maximum.

(3) Images and spurious responses: 70 dB down (50 dB for signals within 500 kHz of desired channel; 45 dB for subharmonics.)

- (4) If rejection: 80 dB down.
- (5) Avc characteristics: Output within +3 dB from 10 to 100,000 uV (from 1000-microvolt reference).
- (6) Blocking: No blocking for input signals up to 0.5 volt.
- (7) Squelch operation: A change in audio output of at least 10 dB is effected by a 1-dB change in input signal.
- (8) Ultimate (s+n)/n ratio: At least 35 dB (measured at 1000 uV).
- (9) Audio output: 50 mW into a 300-ohm load with 30-percent modulation, 1000 TV.
- (10) Audio fidelity: +1, -3 dB from 300 to 3500 Hz (from 1000-Hz reference).
- (11) Audio distortion: Less than 10 percent at 50-mW output.
- e. MAIN RECEIVER AUXILIARY OUTPUT DATA.

(1) Sensitivity: Signal with average level of 3 uV, modulated 80 percent at 1000 Hz, produces at least 0.25 volt at a (s+n)/n ratio of 6 dB or greater.

- (2) Audio distortion: 15 percent maximum at 1000 uV input.
- (3) Ultimate (s+n)/n ratio: 24 dB minimum (measured at 1000 uV modulated 30 percent).
- (4) Audio fidelity: +4 dB from 300 Hz to 20 kHz (from 1000-Hz reference).
- f. GÚARD RECEIVER DATA.

(1) Sensitivity: A signal having an average level of 5 uV into the guard receiver modulated 30 percent at 1000 Hz produces at least 7 mW at a (s+n)/n ratio of 10 dB or greater.

- (2) Selectivity: 6 dB, 50 kHz minimum; 60 dB, 200 kHz maximum.
- (3) Images and spurious responses: 60 dB down.
- (4) If rejection: 80 dB down.
- (5) Avc characteristics: Output within +3dB from 10 to 100,000 uV (from 1000-uV reference).
- (6) Blocking: No blocking for input signals up to 0.5 volt.
- (7) Squelch operation: A change in audio output of at least 10 dB is effected by a 1-dB change in input signal.
- (8) Ultimate (s+n)/n ratio: At least 35 dB (measured at 1000 uV
- (9) Audio output: 50 mW into a 300-ohm load with 30-percent modulation. 1000 uV
- (10) Audio fidelity: +1, -3 dB from 300 to 3500 Hz (from 1000-Hz reference).
- (11) Audio distortion: Less than 10 percent at 50-mW output (1000-uV input, 1000 Hz 30 percent modulated).

g. GUARD RECEIVER AUXILIARY OUTPUT DATA. -A signal having a 5 -u V level into guard receiver modulated 30 percent at 1000 Hz produces at least 0.15 volt at a (s+n)/n ratio of 6 dB or greater.

## 1-7. FACTORY OR FIELD CHANGES.

Effective the date of this technical manual, there have been no factory or field changes to Radio Set AN/PRC-41A or Radio Accessory Kit MK-706/PRC-41. This manual is valid for all units of Radio Set AN/ PRC-41 modified to Radio Set AN/PRC-41A by modification kits supplied under Navships N00024-67-C1566.

## **1-8. PREPARATION FOR RESHIPMENT.**

To prepare Radio Set AN/PRC-41A and Radio Set Accessory Kit MK-706/PRC-41 for reshipment, return all items and accessories to their respective cases. See figures 2-1 and 2-2 of the Installation and Operation Technical Manual for Radio Set AN/ PRC-41A, NavShips 0967-872-5010. Replace Radio Set Case CY-3883/PRC-41 and Electronic Equipment Case CY-3885/PRC-41 in their packing boxes, and carefully nail the crate planking in place. Properly mark the packing boxes for reshipment or storage.

#### SECTION 4 TROUBLESHOOTING

#### 4-1. LOGICAL TROUBLESHOOTING.

a. HISTORICAL DATA AVAILABILITY. When adequate historical data is not available, troubleshooting procedures should be based on the following six logical steps.

(1) SYMPTOM RECOGNITION. - This is the first step in the troubleshooting procedure and is based on a complete knowledge and understanding of equipment operating characteristics. All equipment troubles are not the direct result of component failure; therefore, trouble in an equipment is not always easy to recognize since all conditions of less than peak performance are not always apparent. This type of equipment trouble is usually discovered while accomplishing preventive maintenance procedures. It is important that the not so apparent troubles, as well as the apparent troubles, be recognized.

(2) SYMPTOM ELABORATION. - After an equipment trouble has not been recognized, all the available aids designed into the equipment should be used to elaborate further on the original trouble symptom. Front panel controls should provide a better identification of the original trouble symptom; also, checking or otherwise manipulating the operating controls may eliminate the trouble.

(3) LISTING PROBABLE FAULTY FUNCTION. - The next step in logical troubleshooting is to formulate a number of logical choices as to the cause and functional section of the trouble. The logical choices are based on knowledge of the equipment operation, a full identification of the trouble symptom, and information contained in this manual. The overall functional description and its associated block diagram should be referred to when selecting possible faulty functional sections.

(4) LOCALIZING THE FAULTY FUNCTION. -(Refer to figure 4-1). For the greatest efficiency in localizing trouble, the functional sections which have been selected by the logical choice method should be tested in an order that will require the least time. This requires a choice to determine which section to test first, and should be based on the validity of the logical choice and the difficulties in making the necessary tests. If the tests do not prove that functional section to be at fault, the next selection should be tested, and so on until the faulty functional section is located. As an aid to this process, the manual contains a functional description and a servicing block diagram of the functional sections. Pertinent indications are included at significant check points on the servicing block diagram to aid in isolating the faulty section; also, test data (such as information on control settings, critical adjustments, and required test equipment) are supplied to augment the functional description and servicing block diagram of the functional.

(5) LOCALIZING TROUBLE TO THE CIRCUIT. - After the faulty functional section has been isolated, it may be necessary to make additional choices as to which circuit or group of circuits is at fault. The servicing block diagram of the functional sections and the individual functional circuit groups (where required) provide the signal flow and test location information needed to bracket and then isolate the faulty circuit. Functional descriptions, simplified schematics, and pertinent test data for individual circuits or groups of circuits of the functional section are all placed together in one area of the manual. Information which is too lengthy in nature to be included in this arrangement is readily referenced from the test data portion of the troubleshooting information.

(6) FAILURE ANALYSIS. - After the trouble (faulty component, misalignment, etc) has been located, but prior to performing corrective action, the procedures followed up to this point should be reviewed to determine exactly why the fault affected the equipment in the manner it did. This review is usually necessary to make certain that the fault discovered is actually the cause of the malfunction and not just the result of the malfunction.

b. GENERAL TROUBLE ISOLATION. - Except where noted, the Radio Receiver-Transmitter RT-695A/PRC-41 which is being tested, shall be connected according to the test setup illustrated in figure 5-21. When trouble has been isolated to a particular module of RT-695A/PRC-41, the defective module may be replaced by a like module known to be operative, and the equipment returned to tactical service. Refer to paragraph 5-4b for removal and 5-4g for replacement procedures of the modules of the RT-695A/PRC-41, as required.

(1) USE OF GENERATOR PAD CN-315/ URM-26. Test procedures of this section and section 5 make reference to the use of Signal Generator Pad CN-315/URM-26. When the CN-315 is used with the AN/USM-44A, the attenuator dial on the AN/USM-44A is read in terms of microvolts across 50 ohms and the indication must be

## Change 1 4-1

### TM 11-5820-510-35-1/NAVELEX 0967-872-5020

multiplied by two. (The CN-315/URM-26 is a 6-db pad.) When the CN-315 is not used, the attenuator dial indication (in microvolts) is unchanged.

(2) TRANSISTOR SERVICING. Techniques and precautions of servicing transistorized circuit should be pointed out to the technician. Although transistors are expected to operate indefinitely, they are subject to abuse, and unless special maintenance techniques are used, they will be ruined.





Figure 4-1. Radio Receive Transmitter RT-695A/PRC-41, Module and Test-Point Locations.

Change 1 4-2

## NAVSHIPS 0967-872-5020 TROUBLESHOOTING TM-03816B-35/2 TM11-5820-510-35-1

becoming widely used and are especially sensitive to certain kinds of overload during routine servicing. Transistors are like semiconductor diodes in this respect, except that the circuits are more complex, and because of the interaction between circuits, even more understanding is required. Parts can be burned out when measuring resistances with a multimeter even when the power supply is turned off. This means that personnel must not indiscriminately measure parts values, as has been a practice with electron tube circuits. An additional complexity results because the circuits normally use bypass and coupling capacitors designed for the lower voltages and may be damaged by the usual multimeter.

(b) Improper servicing techniques can ruin costly transistors. All personnel using or servicing transistorized equipment should know the limitations of the transistor, before applying any test signal (even a multimeter) to a transistor circuit, and should check the maximum allowable current, voltage, and power dissipation ratings of each transistor Since the resistance of a transistor changes with the magnitude and polarity of applied voltages, it is useful to consider the transistor as a simple switch or matched impedance when computing an ultrasafe maximum signal which may be applied to a particular element. The effect on associated circuits should be considered.

(c) Table 4-1, although necessarily general, provides adequate information, in most cases, for the technician who services transistorized equipment. Figure 4-2 provides an outline and locates the base, collector, and emitter leads of the various transistor types which are used in Radio Receiver-Transmitter RT-695A/PRC-41 and Power Supply PP-3700/PRC -41.

# TABLE 4-1. HOW TO AVOID DAMAGE TO TRANSISTORS WHILE SERVICING CIRCUITS

SOURCE OF ABUSE	SUGGESTIONS
Basic Failure, Voltage Breakdown, or Punch Through.	

Note

Voltage breakdown is especially critical In surface barrier types of transistors.

Cause: Safe voltage is exceeded m the nonconducting direction Allowable values for surface barrier translators are on the order of 15 volts from collector to emitter and 0 5 volt from base to emitter Sometimes 0 1 volt can be excessive.

Short-circuiting series parts, such as the load resistor,	Do not short parts with voltages present. Use
with test prods, screwdriver, or soldering Iron, permitting-	very small test prods insulate prods to the tips.
the voltage on the transistor to rise.	If a screwdriver IS used near active transistor
	circuits, It should be small and well Insulated
	Turn off power to transistors before using
	soldering Iron or uninsulated tools. Keep transistor
	away from high-voltage circuits
Using multimeter on hlgh-reslstance measuring range	Avoid use of resistance measuring circuits unless
(22.5 volts of Multimeter AN/PSM-4C, for instance,	safe. Remove batteries from multimeter
is too high for surface barrier transistors)	or use series and parallel resistors to limit
	current and voltages to safe values
Using soldering iron which connects ac from line by	Use 6-volt Iron or Isolation transformer. Always
leakage or capacitance	turn transistor circuits off and, observing
	safety precautions, connect a common ground
	before soldering
Connecting leads from ungrounded test set to transistor,	Ground all cases together using short ground
causing rf filter to connect voltage to transistor.	connections. Use all safety precautions necessary

# TABLE 4-1. (Continued)

SOURCE OF ABUSE	SUGGESTIONS
Using transformerless ac sets or test sets.	Not recommended. However, If necessary to use them, connect an electrostatically shielded isolation transformer (1 1 ratio) In the power line of the transformerless set for safety, and use common ground Check voltages before connecting test leads between equipments to assure safe values.
Using equipment with faulty power supply.	Repair power supply.
Accidentally connecting other voltages to transistor.	Check for test lead voltage that might damage transistor, and If evident, eliminate it before connecting. Ground and short probes and test leads to discharge any test set capacitors before connecting, when applicable.
Basic Failure, Burnout, or Runaway.	
Cause Allowable power dissipation is exceeded In any part of the transistor	
Shorting out, shunting, or grounding the transistor input resistor with power applied, causing Inadequate bias. Connecting the collector voltage without the proper emltter to base bias voltage.	Use extreme care to avoid shorts or shunts. Insulate test prods to the tip Do not connect test leads to transistor if ends of leads are free to short circuit. Use only insulated prods, or turn off the power Include dc isolation (suitable capacitor) between signal source and transistor Avoid connecting transistors or plugging them into sockets unless the power supply voltages are
Using multimeter (battery) on the low resistance range.	off Check on allowable currents and voltages for transistor elements Restrict resistance measuring ranges to safe ones or use limiting resistances (series and parallel as necessary).
Shorting any parts that cause excessive power to be applied- to the transistor.	Do not use a voltmeter of low resistance or other device that will radically affect circuit resistance or voltages, in either the base or collector circuits.
Using an ungrounded soldering Iron, thus connecting leakage current Into the transistor circuit.	Do not solder, connect, or disconnect with voltages on transistor. Ground iron tip (through shank) to transistor circuit ground In a safe manner, use isolation (1 1 ratio) transformer or use 6-volt iron Disconnect heated Iron before soldering if iron is large enough to hold necessary heat.
Inadvertently connecting voltages or currents (such as	Ground chassis or cases using all necessary
radio interference filter current, or leakage from the	sarety precautions Reduce stray fields (Use
voltages, or test oscillator voltages).	check test lead voltage compared to that of

## TABLE 4-1. (Continued)

SOURCE OF ABUSE	SUGGESTIONS
	transistor circuit with Electronic Multimeter AN/USM-116 to assure low enough voltage Do not connect low Impedance device across equipment voltage or current supplies or loads.
Inducing current by magnet/c field of a soldering gun (such as the transformer type)	Do not use high current carrying conductor or soldering device near wiring.
Sub]ecting transistor to power line transients	Use a suitable supply and power source.

## 4-2. OVERALL FUNCTIONAL DESCRIPTION.

a. RADIO SET AN/PRC-41A.-Radio Receiver-Transmitter RT-695A/PRC-41 is composed of eight plug-in modules and a main chassis. See figure 4-1 for relative module location and identification. The RT-695A/PRC-41 is a receiver-transmitter that provides radiotelephone (A3) communication from ground to air and ground to ground. Refer to figure 4-3, a functional block diagram. There are 1750 channels available, spaced 100 kHz apart, in the range of 225.0 to 399.9MHz. The operator can select each



Figure 4-2. Transistor Base Diagrams





Figure 4-3. Radio Set AN/PRC-41A, Overall Functional Block Diagram

channel by using three selector knobs and a frequency indicator located on the front panel of the RT-695A/ PRC-41. A second fixed-tuned receiver permits monitoring of a predetermined frequency, known as the guard channel, in the 238.0to 248.0-MHz range (usually tunee to 243.0 MHz). Primary power for operating the RT-695A/PRC-41 can be provided by Storage Battery BB-451/U, Power Supply PP-3700/ PRC-41, or by a vehicle or aircraft power source. PP-3700/PRC-41 is used in an aircraft installation if ac power is available. The choice of primary power source is dependent on the operational requirements. Reception and transmission are made and controlled by the press-to-talk switch of Handset H-33E/PT. When the AN/PRC-41A and the TSEC/ KY-38 are connected as a system, the TSEC/KY-38 is connected to the connector provided on the front panel of RT-695A/PRC-41 using Special Purpose Electrical Cable Assembly CX-10831/PRC-41A, and Handset H-33E/PT is connected to the appropriate audio jack on TSEC/KY-38. Reception and transmission is controlled by the handset. When TSEC/ KY-38 interconnect cable is connected to 1A9J14 on the RT-695A/PRC-41, transmission from 1A9J13 is inhibited but reception is possible. The RT-695A/ PRC-41 contains tripleconversion superheterodyne circuits; its main receiver and transmitter functional sections are illustrated in figures 4-4 and 4-5.

(1) Since the units and accessories of AN/ PRC-41A are not electronic in nature, with the exception of Radio Receiver-Transmitter RT-695A/ PRC -41 and Power Supply PP-3700/PRC-41, the description and test information of this section is limited to RT-695A/PRC-41 and PP-3700/PRC-41. It may be necessary to check an antenna or cable assembly for continuity; however, damage to masts, mountings, harness, cases, etc. is obvious from visual inspection. No test information for these items is contained in this section. The RT-695A/ PRC-41 functions in three ways: it provides transmission and reception on 1750 channels in the 225.0-to 399.9-MHz range, and reception on a single frequency, 243.0 MHz, known as the guard channel. The functional sections are therefore identified as the main receiver, guard receiver, and transmitter. Refer to figure 4-3. Primary power, power distribution,
AN/PRC-41A TROUBLESHOOTING Paragraph 4-2a(1)

frequency generation, control, and antenna circuits are considered subordinate and common to both the main receiver and transmitter functional sections. The guard receiver functional section contains its own frequency generation circuits. Power distribution circuits are common to all functional sections. RT-695A/PRC-41 may be used with either Antenna AS-1404/PRC-41 or Antenna AS-1405/PRC-41. The receiver output is applied to Handset H33E/PT.

(2) Paragraphs 4-3 through 4-5 provide descriptions and test data for the functional sections. Alignment and adjustment information references the procedures of section 5. Paragraph 4-6 provides descriptions and test data pertinent to the subordinate circuits. Each of the functional sections are broken down into circuits or groups of circuits and into modules. A module is defined as a plug-in subassembly. The lowest level in this breakdown process is the simplest functional block (for example, an audio amplifier stage). Only the circuits which are uncommon (those not covered in the Handbook of Electronic Circuits, NavShips 900,000.102) will be discussed. In all other cases, reference is made to RT-695A/PRC-41 schematic diagrams of section 5 and the servicing block diagram of this section.

b. TEST POINTS. -Significant test points are identified on the functional section servicing block diagram, figure 4-19, and illustrated in figure 4-1 by use of star and circle test symbols. The star test-point symbol identifies and locates test points which are used to isolate trouble to a functional section. Circle test-point symbols identify and locate test points which are used to isolate trouble to a circuit within the functional section. Unless otherwise noted, all test-point voltage measurements are with respect to ground. Refer to table 4-2 and figure 4-1.

TEST	LOCATION	FIGURE REF	ISOLATE TROUBLE TO	INDICATION
Ū.	1A9J14-J KY-38 connector on front panel of RT-695A/ PRC- 41	4-1	Main receiver functional section	Audio output shall be 50 section milliwatts minimum when measured according to procedures of paragraph 4-3b.
2	1A9J14-N KY-38 connector on front panel of RT-695A/ PRC- 41	4-1, 4-19	Main receiver functional section	Output voltage shall be 0.25 volt minimum a- cross 25-kz load.
3	1A9J14-J KY-38 connector on front panel of RT-695A/PRC- 41	4-1	Guard receiver functional section	Audio output shall be 50 milliwatts minimum when measured according to procedures of paragraph 4-3b.
4	1A9J14-S KY-38 connector on front panel of RT-695A/ PRC- 41	4-1	Guard receiver functional section	Output voltage shall be 0 15 volt minimum across 600-ohm load
6	Antenna connector 1A9J28 on front panel of RT- 695A/ PRC-41	4-1	Transmitter functional section	Transmitter power output shall be at least 3 watts average across 225- to 399 9-MHz range
	Handset H-33E/ PT connected to H33E/PT connector on front panel of RT-695A PRC- 41	4-19	Circuits of audio module	Signal should be heard at a reasonable level in H-33E/PT

#### TABLE 4-2. TEST POINTS

Table

#### NAVSHIPS 0967-872-5020

# AN/PRC-41A

### TROUBLESHOOTING

# TM-03816B-35/2 TM11 -5820-510-35-1

# TABLE 4-2. (Continued)

TEST POINT			ISOLATE TROUBLE TO	
2	1A4J2	4-1	Circuits of audio module	Normal audio signal should be observed on Oscilloscope AN/ USM-105A
3	1A2J2	4-1	Low-frequency oscillator 1A2Q7 of 1st and 2 <sup>nd</sup> if amplifier module.	Measure -2.5 to -3 5 volts dc
4	1A2J1	4-1	High frequency oscillator 1A2Q5 of 1st and 2nd if amplifier module	Measure -0 8 to -1.5 volts dc
6	1A5J2	4-1	Circuit of spectrum generator module	Measure -1.5 to -5 volts dc.
6	1A7J2	4-1	Transistor stages 1A7Q5 through 1A7Q10 of guard receiver module	Should observe a normal avc rise with an increasing signal input at 20.55 MHz.
Ō	1A7J1	4-1	Oscillator 1A7Q4 of guard receiver module.	Measure -1 to -2 volts dc.
8	1A2J3	4-1	Transistor stages 1A2Q6, 1A2Q4. 1A2Q1 and 1A2Q2 of 1st and 2nd if amplifier module.	Measure 1.7 volts dc in transmit mode
9	1A8J7	4-1	Rf amplifiers 1A8V1 through 1A8V4 of rf and power amplifier module	Measure +1.8 volts dc
0	1A9J16	4-1	Transmit-receive switch 1A9S1 of RT-695A/ PRC-41 main chassis	Set OFF-T/R-T/R/G 1A9J21 DIAL LIGHT switch to OFF with H-33E/PT keyed. Measure zero resistance between 1A9J16 and 1A9J21
(1)	1A6J2	4-1	Circuits of modulator module	Measure 10 volts ac in transmit mode with modulation signal applied.
(12	Plugs P1 and P3 Power Supply PP- 3700/PRC-41	4-8	Circuits of PP-3700/-41 PRC-41	Measure 26.5 volts dc.
13	Junction of diodes CR2 and CR4 of PP- 3700/PRC-41	4-8	Transformer T1 and diodes CR1 through CR4 of PP-3700/PRC-41	Measure 41 volts dc.

4-2

AN/PRC-41A

# TROUBLESHOOTING

Paragraph

# TM-03816B-35/2 TM11 -5820-510-35-1

# 4-2c

TABLE 4-2.	(Continued)

	Terminals 5 and 6 of transformer T1 of PP-3700/PRC-41	4-8	Transformer T1 of PP-3700/PRC-41	Measure 45 volts ac.
Ð	Red lead on inductor L1	4-8	Inductor L1, relay K1, diodes CR1 through CR4, and transformer T1 of PP-3700/PRC-41	Measure 41 volts dc.
10	Emitter of transistor Q2	4-8	Inductor L1, capacitor C1, and resistor R1 of PP-3700/PRC-41	Measure 37 volts dc.
(J)	Collector of transistor Q2	4-8	Transistors Q1 through Q4 of PP- 3700/PRC-41	Measure 27 volts dc.
ા	Junction of 1A9L1 and 1A9C1 on RT695A/PRC-41 main chassis	4-1	Power distribution circuits of RT- 695A/PRC-41	Measure 26 5 volts dc.
(J)	1A5J1	4-1	Spectrum oscillator 1A5V1 of spectrum generator module	Measure -5 to -20 volts dc.
ୖ୶	1A3J1	4-19	Squelch control circuits and if amplifier stages	Measure approximately +4 volts, no signal input, squelch relay should energize with squelch control on front panel full ccw. Observe normal avc rise with increasing signal input.
প্র	1A9J25	4-19	Filament circuits	Measure 6.3 volts dc.
22	1A9J27	4-19	Diode mixer	Measure +0.5 to +2.0 volts dc.

c. OVERALL TROUBLE ISOLATION. - If an equipment is known to be inoperative or if the preventive maintenance tests have indicated that it has less than adequate performance, perform the steps of procedure of table 4-3 to isolate the trouble to a functional subordinate section. Before beginning the trouble isolation procedures of table 4-3, inspect the units of AN/PRC-41A for loose cables, charred or discolored insulation, broken wire, improper control settings, or other evidence of equipment malfunction. Check all fuses to be certain that they have not blown, and make certain primary power is available to the equipment. Attempt operation on several channels. Check to see that the panel lights are lighted in the DIAL LIGHT position of the OFF -T/R -T/R/G-DIAL LIGHT switch. If Mounting MT-2976/PRC-41 is suspected to be at fault, use figures 5-64 and 5-99 as aids to troubleshooting.

(1) TEST SETUP. - Perform the test procedures of steps 1 through 4 of table 4-3 with Radio Receiver-Transmitter RT-695A/PRC-41 connected to a Power Supply PP -3700/PRC -41, known to be operative, with Receiver-Transmitter Case CY-3884/ PRC-41 in place on the RT-695A/PRC-41. If these steps do not reveal the source of difficulty, remove CY-3884/PRC-41, and check the frequency generation circuits according to step 5 of table 4-3.

- (2) TEST EQUIPMENT REQUIRED.
  - (a) Signal Generator AN/USM-44A.
  - (b) RF Wattmeter AN/URM-43C.
  - (c) Electronic Multimeter TS-505/U.
  - (d) Signal Generator Pad CN-315/

URM -26.

(e) Fuse MX-1730/U.

Table

#### NAVSHIPS 0967-872-5020

AN/PRC-41A

# 4-3

#### TM-03816B-35/2 TM11 -5820-510-35-1

TROUBLESHOOTING

# TABLE 4-3. OVERALL TROUBLE ISOLATION

STEP	PRELIMINARY ACTION	NORMAL INDICATION	IF INDICATION
1	Connect Handset H-33E/PT to H33E/PT connector on front panel of RT-695A/ PRC-41. Set OFF-T/R- T/R/G-DIAL LIGHT switch to DIAL LIGHT.	Dial lights light. Proceed to step 2.	IS ABNORMAL Refer to paragraph 4-6d, and check power distribution circuits.
2	Connect Signal Generator AN/USM- 44A through Signal Generator Pad CN- 315/ URM-26, and Fuse MX-1730/U to antenna connector 1A9J28 set for304.7 MHz modulated 30% with 1000 Hz at 1000 microvolts. Set RT-695A/ PRC- 41 to 304.7 MHz, and set OFF T/R- T/R/G-DIAL LIGHT switch to T/R. Rotate VOL control fully clockwise and SQUELCH control full counterclockwise	A 1000-Hz tone should be heard in H-33E/PT. Proceed to step 3.	Proceed to step 5. If performance of step 5 provides a normal indication, refer to paragraph 4-3, and check main receiver functional section. If step 5 is abnormal, refer to paragraph 4-6e, and check frequency generator circuits.
3	Set AN/USM-44A to 243 0 MHz modulated 30% with 1000 Hz at 1000 microvolts. Leave RT-695A/ PRC-41 set to 304.7 MHz. Rotate SQUELCH control fully clockwise Set OFF-T/R- T/R/G-DIAL LIGHT switch to T/R/G	A 1000-Hz tone should be heard in H-33E/PT. Proceed to step 4.	Refer to paragraph 4-4, and check the guard receiver functional section
4	Remove AN/USM-44A, CN-315/ URM- 26, and MX-1730/U from 1A8J28. Connect RF Wattmeter AN/URM-43C to antenna connector 1A9J28. Set OFF-T/R-T/R/G- Press press-to-talk switch on H-33E/PT, and measure transmitter power output at several frequencies across range.	Measure 3 watts average across frequency range	Proceed to step 5. If performance of step 5 provides a normal indication, refer to paragraph 4-5, and check transmitter functional section. If step 5 is abnormal, refer to paragraph 4-6e, and check frequency generator circuits.
5	Remove Receiver-Transmitter Case CY-3884/PRC-41 from RT-695A/ PRC- 41. Connect RT-695A/ PRC-41 into test setup shown in figure 5-21. Set OFF-T/R-T/ R/G-DIAL LIGHT switch to T/R Measure voltage at test Jacks 1A2J1 and 1A2J2 on 1st and 2nd if amplifier module and at test jack 1A5J2 on spectrum generator module.	Measure -2.5 volts dc. At 1A2J2. Measure -0.7 volt dc. at 1A2J1. Measure 1.5 volts dc. at 1A5J2	Refer to paragraph 4-6e, and check frequency generator circuits.

TROUBLESHOOTING

#### TM-03816B-35/2 TM11 -5820-510-35-1

4-3

# **4-3. OVERALL MAIN RECEIVER FUNCTIONAL**

#### SECTION.

a. DESCRIPTION. (Refer to figure 4-4.) - The main receiver functional section is composed of the circuits of the fr amplifier-receiver and transmitter module (referred to as the rf and power amplifier module), 1st and 2nd if amplifier module, 3rd if amplifier and squelch module, audio module, and main receiver buffer amplifier along with the switching and interconnecting wiring of the main chassis of the RT-695A/PRC-41. Incoming signals in the 225.0- to 399.9-MHz frequency range are amplified in the rf amplifier stages of rf and power amplifier module 1A8. The frequency scheme of this equipment actually covers an input frequency of 220.0 to 399.9 MHz; however, 225.0 MHz is the lowest frequency used. The received signals in this range are mixed with a 200- to 370-MHz first injection signal in the diode mixer. The resulting 20- to 29.9-MHz first if signal is supplied to 1st and 2nd if amplifier module 1A2, is amplified and mixed with 17.1- to 26.1-MHz second injection signal to produce the 2.9- to 3.8-MHz second if signal. In receiver operation, crystals are chosen so the third injection signal to produce the 500-kHz third if signal. In receiver operation, crystals are chosen so the third injection signal is either 500 kHz above or 500 kHz below the second if signal frequency. The resulting 500-kHz third if signal is applied to 3rd if and squelch module 1A3 where it is filtered, amplified, demodulated, and noise-limited. Normal audio from 3rd if and squelch module 1A3 is fed through volume control 1A9R4 to audio amplifiers of audio module 1A4 and to the H-33E/PT handset. An output of the audio detector of 3rd if and squelch module 1A3 is fed through the main receiver buffer amplifier stages and the squelch circuits of 3rd if and squelch module 1A3. Avc is also fed to an if amplifier stage in 1st and 2nd if amplifier module 1A2 and to four rf amplifier stages in rf and power amplifier module 1A8. a. DESCRIPTION. (Refer to figure 4-4.) - The main receiver functional section is composed of the circuits of the rf and power amplifier module 1A8.

b. TEST DATA. - In order to determine whether or not the main receiver functional section is faulty, perform the test procedure in step (3) of this paragraph. If the results of the performance test prove the main receiver functional section procedure in step (3) of this paragraph. If the results of the performance test prove the main receiver functional section to be operative, proceed to the next functional section performance test to locate the faulty functional section or sections. Steps (1) through (3) of this paragraph provide the necessary test data to aid in pinpointing or eliminating the functional section under consideration. Refer to the main receiver functional section of the servicing block diagram (figure 4-19) for location of test points, signal flow, and auxiliary circuits such as power distribution, control, and frequency generation. To determine whether or not the main receiver functional section is operative, connect RT-695A/ PRC-41 into the test setup described in step (2) of this paragraph and subject the equipment to the main receiver functional section performance test

(1) TEST EQUIPMENT REQUIRED. (a) Signal Generator AN/USM-44A.

(b) Signal Generator Pad CN-315/

**URM-26** 

(c) Fuse MX-1730/U.

(C) FUSE MX-1730/U.
(d) Output Power Meter ME-2/U.
(2) TEST SETUP. - Set Radio Receiver-Transmitter RT-695A/PRC-41 on the test bench, and connect Power Supply PP-3700/PRC-41 to provide the required primary power. Connect Signal Generator AN/USM-44A in series with Signal Generator Pad CN-315/URM-26, Fuse MX-1730/U, and a length of RG-213/U coaxial cable to antenna connector 1A9J28 on the RT-695A/PRC-41. Connect junction box (figure 5-1 and 5-2) to the H33E/ PT connector on the front panel of RT-695A/PRC-41. Connect Output Power Meter ME-2/U to NORMAL AUDIO OUTPUT connector on the junction box. Rotate the VOL control on RT-695A/PRC-41 to maximum clockwise position and the SQUELCH control to maximum counterclockwise position.

counterclockwise position (3) FUNCTIONAL SECTION PERFORMANCE TEST.

#### CAUTION

Do not key RT-695A/PRC-41 with AN/USM44A connected to antenna connector 1A9J28. MX-1730/U provides protection for the signal generator attenuator in the event the RT-695A/ PRC-41 is accidentally keyed.

(a) Set AN/USM-44A to 225.0 MHz modulated 30 percent with 1000 Hz.
(b) Set OFF-T/R-T/R/G-DIAL LIGHT switch to T/R. Turn on AN/USM-44A, and allow equipment time to warm up. Carefully adjust AN/USM44A signal output around the 225.0-MHz point to obtain maximum indication on ME-2/U.
(c) Audio power output should be 50 milliwatts minimum. Repeat test on 304.7 and 399 9 MHz. If correct results are not obtained, main receiver functional section is defective.
c. MAIN RECEIVER CIRCUIT DESCRIPTION. - As a further aid in explanation of the main receiver functional section, each circuit making up the functional section is discussed in order of signal flow. Only the circuits that are uncommon (those not covered in the Handbook of Electronic Circuits, NavShips 900, 000.102) will be discussed. See schematic diagrams of section 5 and functional section or servicing block diagrams of this section. Test data is provided to aid in isolation of trouble to a circuit or group of circuits and consequently to a module. Refer to applicable tuning and adjustment procedures of paragraph 5-3. Perform these procedures only when absolutely necessary or following repair as required. as required.

(1) FUNCTIONAL DESCRIPTION. - The main receiver functional section is composed of the circuits of rf and power amplifier, 1st and 2nd if

amplifier, 3rd if and squelch, and audio modules along with switching and interwiring of the main chassis. Discussion of the circuits of main receiver functional section is presented by module in the order of signal flow. See main receiver functional section block diagram, figure 4-4, as an aid in understanding the theory of operation of the main receiver functional section. A signal in the 225.0- to 399.9-MHz range appearing on either Antenna AS-1404/PRC-41 or Antenna AS-1405/PRC-41 is fed through antenna connector 1A9J28 to contact 2 of receiver-transmit switch 1A9S1A and fed to the input of the rf and power amplifier module.

(a) RF AND POWER AMPLIFIER MODULE 1A8. - With exception of the avc amplifier, the same stages of rf and power amplifier module are used in both receive and transmit operations. In receive operation, signals in the 225.0- to 399.9 MHz range are fed through switch contacts of receive transmit switch 1A9S1A to rf and power amplifier module. Four stages of rf amplification, a power amplifier, and an avc amplifier of the rf and power amplifier module are used in receive operation. See figure 5-92.

<u>1.</u> RF AMPLIFIERS V1 THROUGH V4. - Within the rf and power amplifier module, the signal is applied to the cathode of grounded grid rf amplifier V1. The output of first rf amplifier V1 is coupled through grounded grid amplifiers V2 and V3 to rf driver V4. The output from rf driver V4 is capacitivety coupled to the cathode of power amplifier V5.

2. POWER AMPLIFIER V5. - Power amplifier stage V5 is a triode tube. In receive operation, the stage acts as a class A amplifier. The output is taken from the plate of power amplifier V5 and fed through contacts of switch 1A9S1 B to diode mixer 1A9CR1. Tuned circuits Z1 through Z5 of the rf and power amplifier module are adjusted to the desired signal in the range of 225.0 to 399.9 MHz. These tuned circuits are mechanically ganged to the frequency selector knobs on the front panel of RT-695A/PRC -41.

<u>3</u>. AVC AMPLIFIER Q1. - Avc voltage from 3rd if and squelch module is applied to the base of avc transistor amplifier Q1. An increase to input signal results in a decrease of the base current and thus increases the collector voltage of Q1. The amplifier avc voltage is applied to the cathode circuits of rf amplifiers V1, V2, V3, and V4. In transmit operation, there is no output from avc amplifier Q1 because the collector of Q1 is grounded by relay 1A9K1.

4. OUTPUT CONTROL RELAY K1. During receive operation, the plate voltage applied to tubes V4 and V5 is lowered by means of dropping resistors R25 and R26. This reduces the amplifier gain to prevent overloading first mixer diode 1A9CR1 when high-level, off-channel signals are applied to the receiver. During transmit, the power amplifier output is applied to the antenna instead of 1A9CR1. Relay K1 is energized, applying full plate voltage to V4 and V5 through relay contacts which provide maximum amplifier gain.

(b) 1ST AND 2ND IF AMPLIFIER MODULE 1A2. - Some stages of the 1st and 2nd if amplifier module are used in both receive and transmit operation. Two if amplifier stages, second and third receiver mixer stages, and high- and low-frequency oscillator stages of the 1st and 2nd if amplifier module are used in receive operation (figure 5-86).

1. IF AMPLIFIERS Q1 AND Q2. The 200-to 370-MHz output from the spectrum generator module is mixed with received signal in the 225- to 399.9-MHz range in diode mixer 1A9CR1. Refer to paragraph 4-6e for a discussion of the frequency generation circuits. The difference frequency is selected to produce the 20- to 29.9-MHz intermediate frequency) from diode mixer 1A9CR1 is fed to two 20- to 29.9-MHz if signal (first intermediate frequency) from diode mixer 1A9CR1 is fed to two 20- to 29.9-MHz if amplifiers. These amplifiers consist of three sets of capacitive coupled parallel-tuned circuits and transistors 1A2Q1 and 1A2Q2. The 20- to 29.9-MHz signal is first applied to a set of parallel-tuned circuits and is then applied through coupling capacitor C4 to base number 1 of tetrode if amplifier transistor Q1. The output from the collector of Q1 is applied to a set of parallel-tuned circuits and receiver mixer Q3, or, in transmit operation, through capacitor C21 to the contacts of receive-transmit switch 1A9S1B. The gain of first if amplifier Q1 is controlled by a positive increasing avc voltage applied to the base number 2 of Q1. Selectivity is determined by the three sets of parallel-tuned circuits: one at the output of Q1 and the input of Q2, and one at the output of Q2. The inductances of these parallel-tuned circuits are mechanically ganged and are linearly positioned in 0.1-MHz steps at 3.0 degrees per increment.

2. SECOND RECEIVER MIXER Q3. -During receive operation, high frequency oscillator Q5 provides a 17.1- to 26.1-MHz output to the emitter of second receiver mixer Q3 through resistor R28 and contacts of relay K2. The first intermediate frequency (20 to 29.9 MHz) is applied to base number 1 of tetrode transistor Q3. The difference frequency (2.9 to 3.8 MHz) of the first intermediate frequency and high-frequency oscillator Q5 is selected by tunable bandpass filters,

consisting of inductors L19 through L24 and capacitors C24 through C27, C52, and C53. The difference frequency (2.9 to 3.8 MHz) is the second intermediate frequency in receive operation. Inductors L19, L21, and L23 are ganged and are linearly positioned in 0.1-MHz increments at 30 degrees per step. Inductors L20, L22, and L24 are inductive trimmers.

3. THIRD RECEIVER MIXER AND TRANSMITTER BUFFER Q6. - Third receiver mixer



transistor Q6 is similar in operation to Q3 in receive operation. The 2.9- to 3.8-MHz second intermediate frequency is coupled to the emitter of third receiver mixer Q6 from the tunable bandpass filter. The 2.9 - to 3.8-MHz low-frequency oscillator Q7 third injection frequency is applied to the base of receiver mixer Q6. The third injection frequency (2.9 to 3.8 MHz) differs from the second intermediate frequency either plus or minus 500 kHz. Second intermediate frequencies of 2.9 through 3.3 MHz have a third injection frequency of 3.4 through 3.8 MHz respectively. Second intermediate frequencies of 3.4 through 3.8 MHz have a third injection frequency of 2.9 through 3.3 MHz respectively. The collector output of Q6 (third intermediate frequency) is capacitively coupled to 500-kHz filter 1A3FL1 of 3rd if and squelch module. Refer to figure 5-87.

<u>4.</u> RELAY SWITCHING. - Relays K1, K2, and K3 of 1st and 2nd if amplifier module are energized during transmit operation. When relay K1 operates, voltage is applied through L1 to collector of Q4, capacitor C1 is removed from the input tuned circuit, and C33 is switched across the tuned circuit consisting of C2 and L1. Relay K2 switches the high-frequency oscillator output from 2nd receiver mixer Q3 to transmitter mixer Q4. By this operation, the emitter circuit of Q3 is opened and its base is grounded, biasing off the transistor. Relay K3 switches the rotors of S3. With K3 energized, the low-frequency oscillator frequency and center frequency of the second intermediate frequency are the same. In order to reduce low-frequency oscillator harmonics, this Q6 output is fed back through the tunable bandpass filter to the base of transmitter mixer Q4. During transmit operation, the low frequency oscillator Q7 signal is applied to the base of Q6, the output is taken from the emitter, and Q6 operates as a common collector amplifier.

(c) 3RD IF AND SQUELCH MODULE 1A3. - 3rd if and squelch module 1A3 consists of three 500-kHz if amplifier stages, two detectors, two direct-coupled avc amplifiers, and a 2-transistor carrier-operated squelch circuit. 3rd if and squelch module 1A3 receives a 500-kHz if signal from third receiver mixer Q6 in 1st and 2nd if amplifier module 1A2 and supplies a detected audio signal that is properly noise limited and controlled by a carrier-operated squelch system to impedance matching amplifier Q1 of audio module 1A4. An output of audio detector Q4 from 3rd if and squelch module 1A3 is supplied to the main receiver buffer amplifier where it is amplified and appears at J14 as an auxiliary audio signal. In addition, 3rd if and squelch module 1A3 supplies avc voltage for avc amplifier Q1 of rf and power amplifier module 1A8, supplies a positive avc voltage to amplifiers Q1 and Q2 of 3rd if and squelch module 1A3, and supplies a positive avc voltage to first if amplifier Q1 of 1st and 2nd if amplifier module 1A2. Refer to figure 5-87.

<u>1</u>. FILTER 1A3FL1 AND IF AMPLIFIERS Q1, Q2, AND Q3. - The 500-kHz if amplifier consists of filter FL1 and broadband amplifiers Q1, Q2, and Q3. An amplitude modulated if signal is fed to filter FL1 from third receiver mixer Q6 of 1st and 2nd if amplifier module 1A2. The bandpass of filter FL1 is approximately 80 kHz wide at the 6-dB points and provides the necessary selectivity for the if amplifier. The output of FL1 is amplified by three broadband amplifiers; two identical npn tetrode stages, Q1 and Q2; and npn triode stage Q3. The three amplifier stages are tuned by self-resonant, low-Q inductor collector loads L1, L2, and L3. The overall amplifier bandwidth is approximately 150 kHz. A positive-going voltage is applied to the number 2 bases of tetrode stages Q1 and Q2 for ave.

<u>2</u>. AUDIO DETECTOR Q4. - The output of third if amplifier Q3 is applied to audio detector Q4 (an npn silicon transistor). The collector load of this stage develops the audio for a diode noise limiter that clips the positive peak of any signal modulated more than the limits set by clipping adjustment R26. The limited output of audio detector Q4 is the audio output of 3rd if and squelch module 1A3. This output signal is applied through the volume control to the audio amplifier module and to J13 and J14 as the normal audio signal. This audio detector signal is also applied through the main receiver buffer amplifier to J14 as an auxiliary audio output.

<u>3</u>. AVC DETECTOR Q6. - The output of third if amplifier Q3 is also applied to the base of avc detector Q6. The collector load is a dc load that supplies the ave. Detector Q6 is a silicon pnp type transistor.

<u>4</u>. AVC AMPLIFIERS Q5 AND Q7. - The avc amplifier circuits consist of two transistor dc amplifiers, Q5 and Q7. Positive avc amplifier Q5 is an npn silicon transistor in a dc emitter follower circuit. The base signal for Q5 is obtained from avc detector dc load R21 and is a positive dc voltage. This voltage increases as the if signal to the detector increases, causing a positive-going avc voltage to be developed at the emitter of Q5. This positive-going avc voltage is fed to number 2 bases of Q1 and Q2 of 3rd if and squelch module 1A3 through diode gating circuit CR4 and to number 2 base of Q1 of 1st and 2nd if amplifier module 1A2. A portion of the positive avc voltage is also applied to the base of negative avc amplifier Q7 through gating diode CR5. The negative avc amplifier is a pnp silicon transistor in a common-emitter dc amplifier circuit with a negative collector supply voltage. As the positive base voltage increases, Q7 approaches cutoff, and the collector voltage approaches zero. This negative-going avc voltage is fed through gating diode CR2 to avc amplifier Q1 of rf and power amplifier module 1A8.

<u>5</u>. DC AMPLIFIER Q10 AND SQUELCH RELAY AMPLIFIER Q8. - The squelch circuit consists of dc amplifier Q10, squelch relay amplifier Q8, and squelch relay K1. Dc amplifier Q10 is a pnp silicon transistor, and squelch relay

#### NAVSHIPS 0967-872-5020

AN/PRC-41A

4-3c(1)(c)5

# TM-03816B-35/2

TROUBLESHOOTING

# TM11 -5820-510-35-1

amplifier Q8 is an npn silicon transistor. A portion of the positive avc emitter voltage, developed by positive avc amplifier Q5, is applied to the emitter of dc amplifier Q10. The collector of Q10 is direct coupled to the base of squelch relay amplifier Q8 that operates relay K1. The contacts of energized relay K1 remove the ground from the audio output 3rd if and squelch module 1A3. When SQUELCH control 1A9R5 is adjusted for maximum sensitivity (full cow), the avc voltage resulting from receiver noise will cause the squelch circuit to energize relay K1. As SQUELCH control 1A9R5 is progressively adjusted in the clockwise direction, the bias voltage developed at the cathode of gating diode CR8 increases, requiring an increase in input signal to develop adequate avc voltage to forward bias dc amplifier Q10 and consequently turn on squelch relay amplifier Q8, energizing squelch relay K1.

(d) AUDIO MODULE 1A4. - The audio output from 3rd if and squelch module 1A3 and guard receiver module 1A7 are applied across the main and guard input level control potentio meters 1A4R19 and 1A4R20, through volume control 1A9R4 on the front panel of Radio Receiver-Transmitter RT-695A/ PRC-41 to the base of audio amplifier 1A4Q1. The output of Q1 is fed through a low-pass filter, consisting of L2 and capacitor C4, to audio driver 1A4Q2 which is transformer coupled by audio transformer T1 to push-pull class B common-collector audio output transistors 1A4Q3 and 1A4Q4. The audio output from transistors Q3 and Q4 is fed through output transformer T2 to Handset H-33E/PT and is present at J14-J. Refer to figure 5-88.

(2) TEST DATA. - If the main receiver functional section has been determined defective, it is necessary to trace the signal path through the main receiver section. Perform the steps of procedure of table 4-4 to isolate the trouble to a defective circuit or circuit group. Refer to figure 4-19, servicing block diagram, and figure 4-1 for location of test points and signal flow of the main receiver functional section.

(a) TEST EQUIPMENT REQUIRED.

- 1. Oscilloscope AN/USM-105A.
- 2. Audio Oscillator TS-382B/U.
- 3. Signal Generator AN/USM-44A.
- Signal Generator AN/URM-25D. 4
- 5. Electronic Multimeter TS-505/U.
- 6. Output Power Meter ME-2/U.
- Signal Generator Pad CN-315/

URM-26.

Fuse MX-1730/U. 8.

(b) TEST SETUP. - Set RT-695A/PRC-41 on the test bench with CY-3884/PRC-41 removed, and connect Power Supply PP-3700/PRC-41 to provide required primary power. Refer to figure 5-21. Connect junction box (figures 5-1 and 5-2), fabricated from bulk supplies, to KY-38 connector 1A9J14 on front panel of RT-695A/PRC-41.

(c) MAIN RECEIVER FUNCTIONAL SECTION CIRCUIT TROUBLE ISOLATION. - To determine which circuit or circuit group is defective, perform the steps of table 4-4 in sequence. After each preliminary action, compare indications obtained with the expected result in NORMAL INDICATION column. If indications are normal, proceed to the next step. If abnormal indications are observed, follow the procedures outlined in NEXT STEP column. Before performing the steps of table 4-4, set the VOL control on RT-695A/PRC-41 to maximum clockwise position and the SQUELCH control to maximum counterclockwise position. Set OFF-T/R-T/R/G-DIAL LIGHT switch to T/R and allow a 5-minute warmup period.

TABLE 4-4. MAIN RECEIVER FUNCTIONAL SECTION, TROUBLE ISOLATION

Step	TEST POINT	PRELIMINARY ACTION	NORMAL INDICATION	NEXT STEP
1	<ul> <li>① KY-38 connec-</li> <li>to 1A9J14 on front</li> <li>panel</li> <li>②</li> <li>Test jack 1A4J2 on</li> <li>audio module</li> </ul>	Connect Output power Meter ME- 2/U to NORMAL AUDIO OUTPUT on junction box. Connect Audio Oscillator TS-382B/U to test jack 1A4J1 on audio amplifier module set at 0.03-volt output at 1000 Hz. Connect Oscilloscope AN/USM-105 to test jack 1A4J2.	ME-2/U should indicate approx 50 milliwatts and a sine-wave signal should be observed on AN/USM- 105A	If normal indications are not obtained, make voltage measurements on a audio module 1A4, (table 5-5). Remove audio module from main chassis and use extension cable W5 (figure 1-12 and table 1-4).
2 (Cont.)	③ KY-38 connec- on front panel	Disconnect wire from pin 1 of 3 <sup>rd</sup> if and squelch module connector 1A9J7. Connect coaxial extension	ME-2/U should indicate approximately 10 milliwatts.	If normal indications are not obtained, make voltage measurements on 3 <sup>rd</sup> if and squelch

AN/PRC-41A

# TROUBLESHOOTING

### NAVSHIPS 0967-872-5020

Table 1-4

#### TM-03816B-35/2 TM11 -5820-510-35-1

# TABLE 4-4. (Continued)

Step	TEST POINT	PRELIMINARY ACTION	NORMAL INDICATION	NEXT STEP
2		cable W7 and coaxial connector adapter A1 (figure 1-12 and table 1-4) to pin 1 of connector 1A9J7 and chassis ground. Connect - Signal Generator AN URM-25D, to adapter A1 set to 500 kHz, modulated 30% with 1000 Hz, with a signal output of 50 microvolts Connect TS- 505/U to test Jack 1A3J1 A maximum indication on TS- 505/U will indicate correct signal generator frequency in this and succeeding steps.		module 1A3 (table 5-4). Remove 3rd if and squelch module from main chassis and use extension cable W4(figure 1-12 and table, 1-4).
3	Same as step 2	Disconnect extension cable W7 from pin 1 of 1A9J7 and reconnect wire removed this pin in step 2 Remove plug 1A9P2 from jack 1A8J17 on transmit- receive coaxial switch in main chassis. Connect Signal Generator AN/USM-44A to plug 1A9P2 using coaxial adapter A1 Set AN/USM-44A to 29.9 MHz, modulated 30% with 1000 Hz, with a signal output of 50 microvolts. Set frequency control knobs on front panel of RT-695A/ PRC-41 to 399 9 MHz Progressively change frequency of AN/USM-44A to 28.8, 27 7, 20 0 MHz and RT- 695A/PRC-41 frequency control knobs to 398.8, 397 7 . 390 0 MHz. Progressively change frequency	ME-2/U should indicate approx 20 milliwatts. If normal indication is obtained on all combinations of test frequencies, proceed to step 7.	If normal indication is not obtained on all test frequencies, proceed to step 4.
4	③ Test jack 1A2J2 on 1 <sup>st</sup> and 2 <sup>nd</sup> if amplifier module	of AN/USM-44A to 28.8; 27.7; Connect TS-505/U to test jack 1A2J2. Rotate 10-MHz frequency control knob on front panel of RT-695A/ PRC-41 to all 10 positions (9, 8, 0).	390.0 MHz. Measure -2 5 to -4 0 volts dc on each position of 10-MHz knob.	If normal indication is not obtained measure voltages of transistor 1A2Q7 given in table 5-3. Transistor 1A2Q7 is accessible through opening on top side of 1st and 2nd if amplifier module. Remove module side cover and visually check switch 1A2S3 for bent contact arms.

# NAVSHIPS 0967-872-5020

AN/PRC-41A

4-4

#### TM-03816B-35/2 TM11 -5820-510-35-1

TROUBLESHOOTING

# TABLE 4-4. (Continued)

Step	TEST POINT	PRELIMINARY ACTION	NORMAL	NEXT STEP
			INDICATION	
5		Connect TS-505/U to test jack 1A2J1. Rotate 1-MHz frequency control knob on front panel of RT-695A/ PRC-41 to all 10 positions (9, 8, ., . 0).	Measure -0.5 to -1 5 volts dc on each position of 1-MHz knob.	If normal indication is not obtained. remove module side cover, and measure transistor 1A2Q5 given in table 5-3 Check setting of inductors 1A2L9 through 1A2L18 (17.1- to 26 1-MHz oscillator tuning slugs) according to procedure of paragraph 5-3b.
6	None	Remove 1st and 2nd if amplifier module 1A2 from main chassis, remove side covers and end cover. and connect module to main chassis using extender cable W2	Measure voltages given in table 5-3.	If indications of table 5-3 are not obtained, isolate defective detail part by indication of voltage measurements.
7	(2) Filament circuit test jack 1A9J25	Connect TS-505/U to test jack 1A9J25 on rear plate of main chassis.	Measure +6.3 volts dc.	If correct indication is not obtained. check individual tube filaments in rf and power amplifier module 1A8 and spectrum generator module 1A5.
8	⑤ Test jack 1A5J2 on spectrum generator module 1A5	Connect TS-505/U to test jack 1A5J2 Rotate tens hundreds-MHz frequency control knob on front panel of RT-695A/PRC-41 to all 18 positions, (39, 38,22).	Measure -1.5 to -5 volts dc If normal indication is obtained on all frequencies, proceed to step 10.	If normal indication is not obtained, proceed to step 9.
9	(19 Test jack 1A5J1 on spectrum generator module 1A5	Connect TS-505/U to test jack 1A5J1 Switch tens- hundreds-MHz frequency control knob on front panel of RT-695A/PRC-41 to all 18 positions, (39, 38, . 22).	Measure -5 to -20 volts dc. If normal indication on all frequencies, is obtained, proceed to step 10.	If normal indication is not obtained, proceed to step 9.
10	None	Remove spectrum generator module 1A5 from main chassis, and remove four side covers. Connect module to main chassis using extension cable W3	Measure voltages given in table 5-6.	If indications of table 5-6 are not obtained, isolate defective detail part by indication of voltage measurements given in table 5-7.

AN/PRC-41A

# TROUBLESHOOTING

#### NAVSHIPS 0967-872-5020

#### TM-03816B-35/2 TM11 -5820-510-35-1

# TABLE 4-4. (Continued)

Step	TEST POINT	PRELIMINARY ACTION	NORMAL INDICATION	NEXT STEP
11	None	Close switch S1 on junction box while observing action of transmit relay 1A9L9 on switch Open switch S1 on junction box while observing action of receive relay 1A9L7 on transmit-receive coaxial switch.	Transmit relay 1A9L6 shall rotate 45 degrees and return to original position very rapidly when switch S1 is closed Rotate armature of 1A9L9 manually 45 degrees in clockwise direction and release This shall cause no effect Armature shall rotate easily with no load except return spring. Receive relay 1A9L7 shall function same as 1A9L9 when switch S1 on junction box is opened.	If normal Indication is not obtained, check 26.5volts dc to 1A9L9 and 1A9L7 with switch S1 on junction box closed and opened respectively. Check switches S1A and SIB for continuity and first mixer for defective components (figure 4-6).
12	Same as step 2	Replace plug 1A9P2 on jack 1A9J17 Disconnect plug 1A9P7 from Jack 1A9J19 on transmit-receive coaxial switch Connect AN/ adapter A2, extension coaxial cable W6, and coaxial connector adapter A1. Set AN/USM-44A to 399 9 MHz, modulated 30% with 1000 Hz, with an output of 5 microvolts. Set frequency control knobs on front panel of RT- 695A/PRC-41 to 399 9-MHz	ME-2/U should indicate approximately 30 milliwatts If normal indication is obtained, proceed to step 14	If normal Indication is not obtained, proceed to step 13
13	None	Remove rf and power amplifier- module 1A8 from main chassis, remove four covers, and connect module to main chassis using extension cable W1 and coaxial extension cable W8.	Measure voltages given in table 5-10.	If Indication is not obtained, isolate detail defective part by indication of voltage measurements and use of resistance measurements (table 5- 11).
14	Same as step 2	Replace plug 1A9P2 on jack 1A9J17. Remove plug 1A9P9 from jack 1A9J22 on bandpass filter 1A9FL8 modulated 30% with 1000 Connect AN/USM-44A to plug 1A9P9 using coaxial connector adapter A1. Set AN/USM-44A to 399.9 MHz modulated 100 Hz with an output of 5 microvolts.	ME-2/U should indicate approximately 20 milliwatts.	If normal Indication is not obtained, check transmit-receive switch S1A for continuity from 1A9J21 to 1A9J19

Table

4-4

4-4

#### NAVSHIPS 0967-872-5020

AN/PRC-41A

TM-03816B-35/2 TM11 -5820-510-35-1 TROUBLESHOOTING

# TABLE 4-4. (Continued)

Step	TEST POINT	PRELIMINARY ACTION	NORMAL INDICATION	NEXT STEP
15	Same as step 2	Replace plug 1A9P9 on jack 1A9J22. Connect AN USM-44A to antenna connector 1A9J28 Set AN/ USM-44A to 399.9 MHz modulated 30% with 1000 Hz.	ME-2/U should indicate approximately 20 milliwatts	If normal indication is not obtained check antenna connector, cables, and bandpass filter 1A9FL8 for continuity or shorts. Filter 1A9FL8 does not have dc continuity from input to output.

# 4-4. OVERALL GUARD RECEIVER FUNCTIONAL SECTION.

a. DESCRIPTION. (Refer to figure 4-19.) -The guard receiver functional section is composed of circuits of guard receiver module 1A7 along with switching and interwiring of the main chassis of Radio Receiver-Transmitter RT-695A/PRC-41. The guard receiver module uses the same antenna and final audio amplifier stages as the main receiver functional section but has its own squelch, detector, first audio amplifier, and avc stages. A strong signal on the main channel will not affect the sensitivity of the guard receiver, providing the main receiver is not set to 243.0 MHz. The guard receiver frequency can be changed (in the range of 238 to 248 MHz) by changing the guard injection oscillator crystal and returning the rf tuned circuits. The guard receiver module is a single-conversion superheterodyne receiver normally fixed tuned to 243.0 MHz. The audio output is applied to the guard input level control potentiometer in audio module 1A4, amplified by stages of the audio module, and applied to Handset H-33E/PT. The guard receiver audio out put is also applied to guard audio buffer amplifier 1A9A2 in the main chassis where it is amplified and appears at KY-38 connector 1A9J14-S as a low level auxiliary guard output. The rf input signal from contacts of receiver-transmit switch 1A9S1 is fed to two stages of rf amplification. The output URM-26. Of the rf amplifier stages is fed to transistor mixer Q3 where the 243.0-MHz rf signal is mixed with an injection signal of 22.45 MHz from diode ČR7 and tuned circuit L7, C22, and C23, which act as a frequency doubler for oscillator Q4. The difference between these two frequencies is selected to provide the 20.55-MHz intermediate signal. The output from transistor mixer Q3 is coupled through coupling capacitor C21 to first if amplifier Q5. The output of Q5 is coupled to second if amplifier Q6 through crystal filter FL1. Q7, Q8, and Q9 are the third, fourth, and fifth if amplifiers. The output of these if amplifier stages is fed to detector Q10 which serves as both an audio detector and an avc detector. The audio output from the detector is applied to audio 4-20 amplifier Q15. The output from audio amplifier Q15 is fed to the guard input level control in the audio module and to the guard audio buffer amplifier. The output from detector Q10 is applied to avc amplifier Q11. The output from Q11 provides avc voltage to the first, second, and third if amplifier stages to rf avc amplifier Q12 and to squelch amplifier Q13.

b. TEST DATA. - In order to determine whether or not the guard receiver functional section is faulty, perform the test procedures of paragraph 4-4b(3). If the results of the performance test prove the guard receiver functional section to be operative, proceed to the next functional section performance test to locate the faulty functional section or sections. Steps (1) through (3) below provide the necessary test data to aid in pinpointing or eliminating the functional section under consideration. Refer to the guard receiver functional section of the servicing block diagram (figure 4-19) for location of test points, signal flow, and auxiliary circuits such as power distribution and control. To determine whether the guard receiver functional section is operative, connect RT-695A/PRC -41 into the test setup described in step (2) and subject the equipment to guard receiver functional section performance test.

(1) TEST EQUIPMENT REQUIRED.

- (a) Signal Generator AN/USM-44A.
- (b) Signal Generator Pad CN-315/

URM-26

(c) Fuse MX-1730/U.

(d) Output Power Meter ME-2/U.

(2) TEST SETUP. - Set RT-695A/PRC-41 on the test bench, and connect Power Supply PP-3700/ PRC-41 to provide the required primary power. Connect AN/USM-44A in series with Signal Generator Pad CN-315/URM-26, MX-1730/U, and a length of RG213/U coaxial cable to antenna connector 1A9J28 on the RT-695A/PRC-41. Connect the junction box (figures 5-1 and 5-2) to the KY-38 connectors 1A9J14 on the front panel of RT-695A/PRC-41. Connect ME2/U to the NORMAL AUDIO OUTPUT connector on the junction box. Set VOL and SQUELCH controls on the front panel of RT-695A/PRC-41 to maximum clockwise position.

4-4b(3)

TROUBLESHOOTING

#### TM-03816B-35/2 TM11 -5820-510-35-1

#### (3) FUNCTIONAL SECTION PERFORMANCE TEST.

#### CAUTION

Be careful not to key Radio Receiver-Transmitter RT-695A/PRC-41 with Signal Generator AN/USM-44A connected to antenna connector J28. Fuse MX-1730/U provides protection for the signal generator attenuator in the event the RT-695A/PRC41 is accidentally keyed.

 (a) Set AN/USM-44A to 243.0 MHz, modulated 30 percent with 1000 Hz, with an output level of 1000 microvolts.
 (b) Set OFF-T/R-T/R/G-DIAL LIGHT switch to T/R/G. Turn on AN/USM-44A, and allow equipment time to warm up. Carefully adjust AN/ USM-44A signal output about 243.0 MHz to obtain maximum indication on ME-2/U.

(c) Audio power output should be 50 milliwatts minimum. If this result is not obtained, the guard receiver functional section is defective.

c. GUARD RECEIVER CIRCUIT DESCRIPTION. As a further aid in explanation of the guard receiver functional section, each circuit making up a functional section is discussed in the order of signal flow. Only circuits which are uncommon (those not covered in the Handbook of Electronic Circuits, NavShips 900, 000.102) will be discussed. See the schematic diagram (figure 5-91) and the servicing block diagram (figure 4-19). Test data is provided to aid in isolation of trouble to a circuit or group of circuits. Refer to the applicable tuning and adjustment procedures of

 paragraph 5-3a. Perform these procedures only when necessary or following repair as required.
 (1) FUNCTIONAL DESCRIPTION. - The guard receiver functional section consists of the circuits of guard receiver module 1A7 and switching and interwiring of the main chassis of RT-695A/ PRC-41. Audio output of the guard receiver module is amplified by audio module 1A4 and applied to H33E/PT. Discussion of circuits of the guard receiver functional section is presented by circuit in order of signal flow. See the guard receiver functional section block diagram of the servicing block diagram (figure 4-19) as an aid in understanding the theory of operation of guard receiver functional section, and refer to figure 5-91 also.

(a) GUARD RECEIVER MODULE 1A7. A 243.0-MHz signal appearing on either Antenna AS-1404/PRC-41 or Antenna AS-1405/PRC-41 is fed through antenna connector 1A9J28 and through receive-transmit switch 1A9S1A to input of the guard receiver module at connector 1A9J9A1.

1. RF AMPLIFIERS Q1 AND Q2. The guard receiver module rf amplifiers consist of transistor stages Q1 and Q2. The guard receiver rf input signal is applied to the base of first rf amplifier Q1 through a parallel-tuned circuit, consisting of inductor L1 and capacitors C1 and C2, and coupling capacitor C4. The amplified output of Q1 is taken from its collector, applied to a tuned circuit, consisting of inductor L2 and capacitors C6 and C7, and coupled through capacitor C9 to the base of second rf amplifier Q2. The amplified rf output is taken from the collector of Q2, applied to a parallel-tuned circuit, coupled through capacitor C14 to another parallel-tuned circuit, and coupled through capacitor for through capacitor for the collector of Q2, applied to a parallel-tuned circuit, coupled through capacitor for the capacitor for the collector of Q2, applied to a parallel-tuned circuit, coupled through capacitor for for the collector of Q2, applied to a parallel-tuned circuit, coupled through capacitor for for the collector of Q2, applied to a parallel-tuned circuit, coupled through capacitor for for the collector of Q2, applied to a parallel-tuned circuit, coupled through capacitor for for the collector of Q2, applied to a parallel-tuned circuit, coupled through capacitor for for the collector of Q2, applied to a parallel-tuned circuit, and coupled through capacitor for for the collector for for the circuit for the collector for for the circuit for the collector for for for the circuit for the collector for for for the circuit for the collector for for for the circuit for t capacitor C17 to the base of mixer Q3. Collectors of rf amplifiers Q1 and Q2 are connected to the avc output taken from collector of rf avc amplifier Q12.

2. OSCILLATOR Q4. - The output of rf amplifiers Q1 and Q2 is fed to mixer Q3 where the 243.0-MHz rf signal is mixed with the injection signal of 222.45 MHz to obtain the 20.55-MHz intermediate frequency signal. The injection signal is developed by grounded base crystal oscillator Q4 using a fifth overtone crystal. This frequency is doubled by frequency doubler CR7 and tuned circuit L7, C22, and C23 and is applied to mixer Q3. The mixer is followed by a 5stage tuned if amplifier using tetrode transistors.

3. IF AMPLIFIERS Q5, Q6, Q7, Q8, AND Q9. - The 20.55-MHz if signal is taken from transistor mixer Q3 and applied through coupling capacitor C21 to the base of first if amplifier Q5; Resonating inductors L11 through L15, connected to the collectors of if amplifiers Q5 through Q9 respectively, are tuned to 20.55 MHz. Intermediate-frequency amplifiers Q5, Q6, and Q7 are controlled by the positive avc connected to their number 2 bases. The number 2 base of if amplifier Q8 is connected to squelch potentiometer R46. Adjustment of R46 changes the bias voltage on number 2 base of Q8 thus varying the squelch operating level. If amplifier Q9 output is capacitively coupled to the base of detector Q10.

4. DETECTOR Q10. - The amplified if output from fifth if amplifier Q9 is detected by class B operated detector Q10. The detected audio signal is taken from the collector of Q10 and applied to the base of audio amplifier Q15 and to the collector of squelch amplifier Q13 through noise limiter diode CR2. The collector output from detector Q10 is also applied to the base of avc amplifier Q11, providing a positive avc output.

5. AUDIO AMPLIFIER Q15 AND SQUELCH AMPLIFIER Q13. - The detected audio out-put from Q10 is amplified by audio amplifier Q15. The amplified output from the collector of audio amplifier Q15 is applied to audio amplifier 1A4Q1 of the audio module. Squelch amplifier Q13 biases off audio amplifier transistor Q15 when there Carrier signal being received on the guard channel. The emitter of audio amplifier Q15 has a fixed bias of approximately 17 volts. Under nosignal condition, the base bias of Q15 is approximately 18 volts. When a signal appears on the guard channel, the avc voltage from avc amplifier Q11 begins to rise. Due to the rise in avc voltage,

Paragraph

AN/PRC-41A

4-4C(1)(a)5

#### TM-03816B-35/2 TM11 -5820-510-35-1

TROUBLESHOOTING

the base bias voltage of squelch amplifier Q13 rises and overcomes the fixed bias provided the emitter of squelch Q13 by zener diode CRT. The squelch amplifier then conducts and draws collector current. This lowers the base bias on audio amplifier Q15 and causes it to conduct. Audio amplifier Q15 coll lector current then raises the base bias of squelch amplifier Q13 to hold the circuit on. When the carrier input is removed from the guard receiver module, the action is reversed. The avc voltage drops, squelch amplifier Q13 becomes biased off, and stops conducting. This action makes the base bias of audio amplifier Q15 rise and it also stops conducting. The absence of collector current of Q15 then lowers the base bias of squelch amplifier Q13 to hold the circuit off. 6. AVC AMPLIFIERS Q11 AND Q12.- Two avc voltages are used to control the gain of the guard receiver module. The first is a positive avc voltage applied to three if amplifier stages and to the base of squelch amplifier Q13 to squelch the audio output of Q15 when no signal is being received. The second is a decreasing positive avc voltage applied to the collectors of the first and second rf amplifiers and mixer. Test jack 1A7J2 provides a test point for measurement of avc amplifier Q11 avc voltage. Zener diode CR3 establishes a voltage reference level that must be exceeded by strength of input signal before avc amplifier Q12 is permitted to provide avc voltage to the rf amplifiers.

- (2) TEST DATA. If the guard receiver functional section has been determined defective, trace the signal path through the guard receiver module to locate the defective circuit or circuit group. To isolate trouble within the guard receiver functional section. perform the steps of table 4-5. Refer to figure 4-1 and the servicing block diagram figure 4-19 for location of test points and signal flow of guard receiver functional section.
- (a) TEST EQUIPMENT REQUIRED.
  - 1. Oscilloscope AN/USM-105A.
  - 2. Signal Generator AN/USM-44A.
  - 3. Signal Generator Pad CN-315/

URM-26.

- 4. Fuse MX-1730/U.
- 5. Electronic Multimeter TS

505/U.

(b) TEST SETUP. - Set RT-695A/PRC41 on the test bench with CY-3884/PRC-41 removed, and connect PP-3700/PRC-41 to provide required primary power. See figure 5-21. Connect junction box (figures 5-1 and 5-2) to KY-38 connector 1A9J14 on front panel of RT-695A/PRC-41.

(c) GUARD RECEIVER FUNCTIONAL SECTION CIRCUIT TROUBLE ISOLATION. - To determine which circuit or circuit group is defective, perform the steps of table 4-5 in the sequence listed. After each preliminary action. compare the indications obtained with the expected result in the NORMAL INDICATION column. If the indications are normal, proceed to the next step. If abnormal indications are observed, follow the procedures outlined in the NEXT STEP column. Before performing the steps of table 4-5, set the VOL and SQUELCH controls on front panel of RT-695A/PRC-41 to maximum clockwise. Set OFF-T/R-T/R/G-DIAL LIGHT switch to T/R/G.

# TABLE 4-5. GUARD RECEIVER FUNCTIONAL SECTION, TROUBLE ISOLATION

Step	TEST POINT	PRELIMINARY ACTION	NORMAL INDICATION	NEXT STEP
1	③KY-38 connector	Connect ME-2/U to NORMAL AUDIO	ME-2/U indicate approximately 20	If normal indication is not
	1A9J14 on front panel	OUTPUT jack on junction box.	milliwatts, and a sinewave signal	obtained, check operation of
			should be USM-105A.	audio module according to step 1
	iunction			functioning properly proceed to
	junction			step 2.
	<sup>②</sup> Test jack	Connect An/USM-105A to test jack 1A4J2.		
	-	Remove guard receiver module 1a7 from		
	1A4J2 on audio module	main chassis, remove side covers, and		
		connect the module to main chassis using		
		extension cable W3. Connect AN/USM-44A		
		of extension cable WZ and coavial adapter		
		A1 Set AN/USM-44A to 20 55 MHz		
		modulated 30% with 1000 Hz, with an output		
		level of 500 microvolts.		

AN/PRC-41A

# TROUBLESHOOTING

#### NAVSHIPS 0967-872-5020

# Paragraph

#### 4-4b(3)

#### TM-03816B-35/2 TM11 -5820-510-35-1

# TABLE 4-5. (Continued)

Step	TEST POINT	PRELIMINARY ACTION	NORMAL INDICATION	NEXT STEP
2	Same as step 1	Same as step 1	Measure voltages given in table 5-9 for guard receiver transistors Q5 through Q15	If indications of table 5-9 are not obtained, isolate defective detail part by indications of voltage measurements.
3	⑦ Test jack	Connect TS-505/U to test jack 1A7J1.	Measure -1 to -3 volts dc	If normal indication is not obtained, measure voltages of oscillator transistor Q4 given in table 5-9. Check setting of inductor L10 according to procedure of paragraph 5-3 b(3)(e).
	1A7J1 on guard receiver module			
4	NORMAL AUDIO OUTPUT jack on junction box	Remove AN/USM-44A and 10-pF capacitor from base number 1 of Q5 Disconnect plug 1A9P4 from jack 1A9J19 on transmit receive coaxial switch. Connect AN/USM-44A to plug 1A9P4 using coaxial connector adapter A1. Set AN/USM-44A to 243 0 MHz, modulated 30% with 1000- Hz, with an output level of 10 microvolts	ME-2/U should indicate approximately 20 milliwatts.	If normal indication is not obtained, measure voltages on rf amplifier transistors Q1 and Q2 and mixer transistor Q3 (table 5-9)
5	Same as step 4.	Disconnect AN/USM-44A from plug 1A9P4 Reconnect plug 1A9P4 to jack 1A9S1 receive coaxial switch. Connect AN/USM-44A to antenna connector 1A9J28. Set AN/USM- 44A to243 0 MHz, modulated 30% with 1000 Hz, with an out put level of 10 microvolts. Set frequency control knobs on front panel of RT- 695A/PRC-41 to any frequency other than 243 0 +5 MHz.	ME-2/U should indicate approximately 20 milliwatts.	If normal indication is not obtained, check continuity of transmit from jack 1A9J21 to Jack 1A9J20 Check antenna connector 1A9J28 and coaxial cables for continuity or shorts

# 4-5. OVERALL TRANSMITTER FUNCTIONAL SECTION.

a. DESCRIPTION. (Refer to figure 4-5.) - The transmitter functional section is composed of circuits of the modulator, rf and power amplifier, and 1st and 2nd if amplifier modules along with switching and interwiring of the main chassis of RT-695A/PRC-41.

The transmitting circuits use, with some switching differences in the 2.9- to 3.8-MHz crystal circuits, the same crystalcontrolled frequency generating circuits as the main receiver. Refer to figure 4-11. In transmit operation, the 2.9- to 3.8-MHz injection frequency is the same as the second intermediate frequency. The 2.9- to 3.8-MHz low-frequency injection oscillator frequencies of 1st and 2nd if amplifier

AN/PRC-41A

4-4C(1)(a)5

#### TM-03816B-35/2 TM11 -5820-510-35-1

TROUBLESHOOTING

module is mixed with high-frequency injection oscillator frequencies to produce rf signals between 20 and 29.9 MHz. The 20- to 29.9-MHz signals are amplified in the 20- to 29.9-MHz if amplifier stages of 1<sup>st</sup> and 2nd if amplifier module and applied through contacts of receiver-transmit switch 1A9S1B to diode mixer 1A9CR1. The 20- to 29.9-MHz signals are mixed with the 200- to 370-MHz signals from spectrum generator module in diode mixer 1A9CR1 to produce a carrier between 225.0 and 399.9 MHz. An audio input signal from H-33E/PT is applied to the input of modul lator module. Output from the modulator module is applied to the driver and power amplifier stages of rf and power amplifier module. The carrier is modulated in the driver and power amplifier stages of the rf and power amplifier module by audio from modulator module and applied to the antenna for radiation. All tunable circuits in the transmitter functional section are mechanically linked to three control knobs located on the front panel of RT-695A/PRC-41.

b. TEST ĎATA. - In order to determine whether the transmitter functional section is faulty, perform the test procedures of step (3) of this paragraph. If results of the performance test prove the transmitter functional section to be operative, proceed to the next functional section performance test to locate the faulty functional section or sections. Steps (1) through (3) below provide the necessary test data to aid in pinpointing or eliminating the functional section under consideration. Reference is made to the transmitter functional section of the servicing block diagram (figure 4-19) and to figure 4-1 for location of test points, signal flow, and auxiliary circuits such as control and frequency generation. To determine whether the transmitter functional section is operative, connect RT-695A/PRC-41 into test setup described in paragraph 4-5b(2), and subject the equipment to transmitter functional section performance test.

(1) TEST EQUIPMENT REQUIRED. - The required test equipment is RF Wattmeter AN/ URM-43C.

(2) TEST SETUP. switch to T/R. MHz.H- 33E/PT.- Set RT-695A/PRC-41 on the test bench with CY-3884/PRC-41 removed, and connect PP-3700/PRC-41 to provide required primary power. Refer to figure 5-21. Connect AN/ URM-43C to antenna connector 1A9J28 on RT-695A/ PRC-41, test point 5. Connect H-33E/PT to the H33E/PT connector on front panel of RT-695A/ PRC- 14.

- (3) FUNCTIONAL SECTION PERFORMANCE TEST.
- (a) Set OFF-T/R-T/R/G-DIAL LIGHT
- (b) Set frequency selector knobs to 225.0
- (c) Push press-to-talk switch on
- (d) Measure power output at 225.0 MHz. Reset RT-695A/PRC-41 to 304.7 and 399.9 MHz. Power output should be 3 watts average across range.

c. TRANSMITTER CIRCUIT DESCRIPTION. As a further aid in explanation of the transmitter functional section, each circuit making up the functional section is discussed in order of signal flows. Only circuits that are uncommon (those not covered in the Handbook of Electronic Circuits, NavShips 900,000.102) will be discussed. See schematic diagrams of section 5 and the transmitter functional section and servicing block diagrams of this section. Test data is provided to aid in isolation of trouble to a module and then to a circuit or group of circuits. Refer to the applicable tuning and adjustment procedures of paragraph 5-3. Perform these procedures only when necessary or following repair as required.

(1) FUNCTIONAL DESCRIPTION. - The transmitter functional section is composed of circuits of the modulator, rf and power amplifier, and 1st and 2nd if amplifier modules along with switching and interwiring of the main chassis. Discussion of circuits of the transmitter functional section is presented by module in order of signal flow. See transmitter functional section block diagram (figure 4-5) as an aid in understanding theory of operation of the transmitter functional section.

(a) 1ST AND 2ND IF AMPLIFIER MODULE 1A2. - The 1st and 2nd if amplifier module has some of its stages used in both receive and transmit operation. With RT-695A/PRC-41 in transmit operation, the third receiver mixer and transmitter buffer, transmitter mixer, and two if amplifier stages of 1st and 2nd if amplifier module are used. The low-and high-frequency oscillator stages provide required injection frequencies. Refer to paragraph 4-6f for a description of the frequency generation circuits.

<u>1</u>. THIRD ŘECEIVER MIXER AND TRANSMITTER BUFFER Q6. - In transmit operation, low-frequency oscillator Q7 is shifted 500 kHz in frequency from that of receive operation. Transistor Q6, used as the third receiver mixer in receive operation, acts as a common collector amplifier in transmit operation and applied 2.9- to 3.8-MHz low-frequency oscillator Q7 signal to a tunable bandpass filter. This filter consists of inductors L19 through L24 and capacitors C24 through C28, C52, and C53. The output from the bandpass filter is capacitively coupled to base number 1 of first transmitter mixer Q4. During transmit operation, relay 1A2K2 removes high-frequency oscillator Q5 injection from second receiver mixer Q3 and grounds the base of Q3 at the same time. This prevents feedback through Q3 from collector to base number 1.

2. TRANSMITTER MIXER Q4. During transmit operation, transmitter mixer Q4 has voltage applied to its collector, high-frequency oscillator Q5 signal output, 17.1 to 26.1 MHz, to its emitter, and low-frequency oscillator Q7 output, 2.9 to 3.9 MHz, to its base. The sum of these mixed frequencies, 20 to 29.9 MHz, is the first immediate frequency and is fed to the 20- to 29.9-MHz if amplifiers, consisting of transistors Q1 and Q2 and their associated parallel tuned circuits. Capacitor C33 tunes the output of Q4

AN/PRC-41A TROUBLESHOOTING

NAVSHIPS 0967-872-5020 TM-03816B-35/2 TM 11-5820-510-35-1



Figure 4-5. Radio Receiver-Transmitter RT-695A/PRC-41, Transmitter Functional Section Block Diagram

so the input circuit of Q1 is tuned in both receive and transmit operations (C1 is removed during transmit). 3. IF AMPLIFIERS Q1 AND Q2 AND SECOND TRANSMITTER MIXER 1A9CR1. With receive-transmit relay 1A2K1 in transmit position, the 20- to 29.9-MHz if signal from transmitter mixer Q4 is fed to if amplifiers Q1 and Q2. These amplifiers consist of three sets of capacitive coupled parallel-tuned circuits and transistors Q1 and Q2. The 20- to 29.9-MHz signal is applied to a parallel tuned circuit consisting of capacitor C2 and inductor L1 and is coupled by capacitor C3 to second tuned circuit. The second tuned circuit consists of capacitor C5 and inductor L2. The signal is then applied through coupling capacitor C4 to base number 1 of tetrode transistor if amplifier Q1. The output from collector of Q1 is applied to a parallel-tuned circuit consisting of capacitors C8 and C9 and inductor L3. The signal is then coupled through capacitor C10 to a parallel-tuned circuit consisting of capacitor C12 and inductor L4 and applied through coupling capacitor C56 to the base of if amplifier Q2. The ouput from collector of Q2 is applied to a parallel-tuned circuit consisting of capacitors C15 and C16 and inductor L5. The signal is then coupled through capacitor C17 to a paralleltuned circuit consisting of capacitor C20 and inductor L6, through capacitor C21 to the output of the 1st and 2nd if amplifier module. This output is fed through contacts of receive transmit switch 1A9S1B to diode mixer 1A9CR1 (the second transmitter mixer). In diode mixe r 1A9CR1, the 20 to 29.9-MHz first transmitter intermediate frequency is mixed with the 200- to 370-MHz output from spectrum generator module (refer to paragraph 4-6f) to produce the final output frequency in the 225.0- to 399.9-MHz range.

(b) RF AND POWER AMPLIFIER MODULE 1A8. - With exception of the avc amplifier, the same stages of rf and power amplifier module are used in both receive and transmit operations. In receive operation, 225.0- to 399.9-MHz signals are applied through contacts of receive-transmit switch 1A9S1B to the input of rf amplifier stages of rf power amplifier module. In transmit operation, output signals in the 225.0- to 399.9-MHz range are taken from diode mixer 1A9CR1 and applied to first rf amplifier V1. Signals in this range are amplified by V2, V3, and V4 and applied to power amplifier V5 in the same way as in receive operation. Refer to paragraph 4-3c(1)(a)1. Output from the modulator module is applied to the plates of rf driver V4 and power amplifier V5. Power amplifier V5 output is fed through contacts of receive-transmit switch 1A9S1A to the antenna. Refer to paragraph 4-3c(1)(a)2.

(c) MODULATOR MODULE 1A6. - When the press-to-talk button on Handset H-33E/PT is pressed, RT-695A/PRC-41 is switched to transmit operation, and the audio signal from the microphone is applied to the input of the modulator module. The audio signal is coupled to the tease of audio driver Q1. Amplified output from the collector of audio driver Q1 is transformer coupled through transformer T1 to the bases of push-pull modulator transistors Q2 and Q3. Resistor R9 sets the saturation level of transistors Q2 and Q3 so that both positive and negative peaks are clipped equally to establish the clipping level between 70- and 90-percent modulation. Output from Q2 and Q3 is taken from their emitters and transformer coupled through transformer T2 to the plates of driver amplifier V4 and power amplifier V5 in the rf and power amplifier module.

(2) TEST DATA. - If the transmitter functional section has been determined to be defective, it is necessary to trace the signal path through the transmitter section. Perform the steps of procedure of table 4-6 to isolate the trouble to a defective circuit or circuit group. Refer to the servicing block diagram (figure 4-19) and figure 4-1 for location of test points and signal flow of transmitter functional section.

(a) TEST EQUIPMENT REQUIRED.

1. RF Wattmeter AN/URM-43C.

2. Electronic Multimeter TS

505/U.

Audio Oscillator TS-382B/U.

(b) TEST SETUP. - Set RT-695A/ PRC-41 on test bench with the CY-3884/ARC-41 removed, and connect PP-3700/PRC-41 to provide required primary power (figure 5-21). Connect junction box (figures 5-1 and 5-2) to KY-38 connector 1A9J14 on front panel of RT-695A/PRC-41.

(c) TRANSMITTER FUNCTIONAL SECTION CIRCUIT TROUBLE ISOLATION. - To determine which circuit or circuit group is defective, perform the steps of procedure of table 4-6 in sequence. After each preliminary action, compare indications obtained with the expected result in the NORMAL INDICATION column. If the indications are normal, proceed to the next step. If abnormal indications are observed, follow the procedures outlined in the NEXT STEP column. Before performing the steps of table 4-6, set OFF-T/R-T/R/G-DIAL LIGHT switch to T/R, and allow 5-minute warmup period.

#### 4-6. OVERALL SUBORDINATE CIRCUITS.

a. DESCRIPTION. (Refer to figure 4-19.) - Subordinate circuits are those circuits which support the main functions (main receiver, guard receiver, and transmitter functional sections) of Radio Receiver-Transmitter RT-695A/PRC-41. These circuits are considered to be the buffer amplifier assembly, main chassis, primary power distribution, frequency generation, control, and antenna circuits. Because these circuits provide functions that are essentially separate, no attempt is made to balance one against the other or to provide an all-inclusive trouble isolation test for the overall subordinate circuits. In this paragraph, a description followed by test data for trouble isolation is provided for each subordinate circuit. Primary power required for operation may be

Table 4-6

### NAVSHIPS 0967-872-5020 TM-03816B-35/2 TM11 -5820-510-35-1

#### AN/PRC-41A TROUBLESHOOTING

# TABLE 4-6. TRANSMITTER FUNCTIONAL SECTION, TROUBLE ISOLATION

Step	TEST POINT	PRELIMINARY ACTION	NORMAL INDICATION	NEXT STEP
1	3 Test jack 1A2J2 on 1 <sup>st</sup> and 2 <sup>nd</sup> if amplifier module	Connect TS-505/U to test jack 1A2J2. Connect AN URM-43C to antenna connector 1A9J28 on front panel of RT-695A/PRC-41. Unless otherwise instructed, AN/URM-43C shall be connected to 1A9J28 for entire transmitter functional section tests. Set frequency control knobs on front panel of RT-695A/PRC-41 to 399.9-MHz. Close switch S1 on junction box. Rotate the tenth MHz frequency control knob to all ten positions, 9, 9, 0. Do not leave switch S1 closed for extended period of time. Open switch S1 after completion of each test.	Measure -2.5 to -4.0 volts dc on each position of 10-MHz knob.	If normal indication is not obtained, measure voltages of transistor 1A2Q7 given in table 5-3. Transistor 1A2Q7 is accessible through opening on top side of 1 <sup>st</sup> and 2 <sup>nd</sup> if amplifier modules. Remove module side cover, and visually check switch 1A3S3 for bent contact arms.
2	④ Test jack 1A2J1 on 1 <sup>st</sup> and 2 <sup>nd</sup> if amplifier module	Connect TS-505/U to test jack 1A2J1. Close switch S1 on junction box. Rotate 1 MHz frequency control knob to all ten positions 9,8, 0.	Measure -0.5 to -1.5 volts dc on each position of 1-MHz knob.	If normal indications is not obtained, remove module side cover, and measure voltages of transistor 1A2Q5 given in table 5-3. Check setting of inductors 1A2L9 through 1A2I18. (17.1 to 26.1-MHz oscillator tuning slugs) according to procedure of paragraph 5-3b.
3	3	Connect TS-505/U to test jack	Measure -0.5 to -2.0 volts dc. If	If normal indication is not obtained,
	Test jack 1A2J3 on 1 <sup>st</sup> and 2 <sup>nd</sup> if amplifier module	1A2J3. Close switch S1 on junction box.	to step 5.	proceed to step 4.
4	None	Remove 1 <sup>st</sup> and 2 <sup>nd</sup> if amplifier module 1A2 from main chassis, remove side covers, and connect the module to main chassis, remove side covers, and connect the module to main chassis using extension cable W2. Close switch S1 on junction	Measure voltage given in table 5-3 for transistors Q1, Q2, Q4, and Q6.	If indications of table 5-3 are not obtained, isolate defective detail part by indication of voltage measurements.

ORIGINAL

box.

# TABLE 4-6. (Continued)

STEP	TEST POINT	PRELIMINARY ACTION	NORMAL INDICATION	NEXT STEP
5	22 Test jack 1A9J27 on transmit-receive switch in main chassis	Replace 1st and 2nd if am- plifier module 1A2 in main chassis. Connect TS-505/U to test jack 1A9J27 located on side of transmit-receive coaxial switch in main chassis Close switch S1 on junction box.	Measure +0.5 to +2.0 volts dc.	If normal indication is not obtained, check spec- trum generator module 1A5 according to the procedure of steps 5, 6, and 7 of table 4-10.
6	9 Test point 1A8J7 on rf amplifier receiver and transmitter module	Connect TS-505/U to test jack 1A8J7 on rf and power amplifier module 1A8. Close switch S1 on junction box.	Measure +1.2 to +1.8 volts dc. If normal indication is obtained, proceed to step 8.	If normal indication is not obtained, proceed to step 7.
7	None	Remove rf and power amplifier module 1A8 from main chassis, remove covers, and connect module to main chassis using extension cable W1 and coaxial extension cable W8. Close switch on junction box.	Measure voltages given in table 5- 10	If indication of table 5-10 are not obtained, isolate defective detail part by indication of voltage measurements and use of resistance measurements given in table 5-11.
8	None	Replace rf and power amplifier module 1A8 in main chassis. Remove plug 1A9P9 from jack 1A9J22 on bandpass filter 1A9FL1 in main chassis. Connect AN/URM-43C to plug 1A9P9 using coaxial cable W6 and coaxial connector adapter A1. Close switch S1 on junction box.	Measure at least 3 watts average across the frequency range of 225.0 to 399.9 MHz.	If normal indication is not obtained, set OFF-T/R- T/R/G- DIAL LIGHT switch to OFF with switch S1 on junction box closed. Check continuity of transmit-receive switch 1A9S1A by measuring zero resistance from 1A9J16 to 1A9J21.
9	5	Connect AN/URM-43C to antenna connector 1A9J28 on front panel of RT- 695A/PRC-41. Replace plug 1A9P9 on jack 1A9J22 on bandpass filter 1A9FL1. Close switch S1 on junction box.	Measure at least 3 watts average across the frequency range of 225.0 to 399.9 MHz.	If normal indication is not obtained, check antenna connector 1A9J28, bandpass filter 1A9FL1, and coaxial cables for continuity or shorts. Bandpass filter 1A9FL1 does not have a dc path from input to output.
10	11 Test jack 1A6J2 on modulator module	Connect TS-382B/U to MIC INPUT jack on junction box. Connector TS-505/U to MIC INPUT OPEN CIRCUIT jack on junction box. Set TS-382B/U to 1000 Hz with an output level to indicate 0.7 volt ac on TS-505/U to test jack 1A6J2 on modulator module 1A6. Close switches S1 and S2 on junction box.	Measure 8 to 10 volts ac.	If normal indication is not obtained, proceed to step 11.

TABLE 4-6	. (Continued)
-----------	---------------

STEP	TEST POINT	PRELIMINARY ACTION	NORMAL INDICATION	NEXT STEP
11	None	Remove modulator module 1A6 from main chassis, remove cover, and connect module to main chassis using extension cable W5. Audio oscillator shall be set same as for step 10. Close switches S1 and S2 on junction box.	Measure voltages on audio driver transistor Q1 and modulator transistors Q2 and Q3 given in table 5-8.	If normal indications are not obtained, isolate defective detail part by indication of voltage measurements.

obtained from Storage Battery BB-451/U, Power Supply PP-3700/PRC -41, or from an aircraft or vehicular power source. Power distribution circuits are common to the three main functional sections. Frequency generation circuits are common to both main receiver and transmitter functional sections. Control circuits are inclusive of mechanical linkages required for frequency selection and frequency indication on the megahertz frequency selectors, volume adjustment, squelch adjustment, and function selection. These control circuits do not include fine adjustment trimmers or potentiometers used to obtain optimum performance.

b. AF AMPLIFIER ASSEMBLY (BUFFER). The buffer amplifier assembly provides the wide-band audio and lowlevel auxiliary guard outputs necessary for secure voice capability.

(1) FUNCTIONAL DESCRIPTION OF AF AMPLIFIER ASSEMBLY (BUFFER). - The af amplifier assembly (buffer) provides amplification of the auxiliary guard and main receiver auxiliary audio signals that are applied to the TSEC/KY-38 security equipment when in the appropriate mode. The detected audio signal from collector of 1A3Q4 is fed to the input of main receiver buffer amplifier 1A9A1. The output of this circuit is present at output connector 1A9J14-N. This output is referred to as the wide-band auxiliary audio output. When the guard receiver of RT- 695A/PRC -41 is receiving a guard signal, output of the guard receiver is applied to the guard receiver buffer amplifier 1A9A2. The guard signal is amplified in this circuit and is present at output connector 1A9J14-S as a low-level auxiliary guard output. Output of the guard receiver is also applied to audio module 1A4 as normal guard audio output.

(2) TEST DATA. - Testing the af amplifier assembly (buffer) involves checking each of the subordinate circuit outputs. The main receiver auxiliary audio output is on pin 1A9J14-N. The low-level auxiliary guard receiver output is on pin 1A9J14-S. Refer to the servicing block diagram (figure 4-I9) for signal flow of af amplifier assembly (buffer) section. To determine if the buffer amplifier is operating properly, perform the steps of procedure in table 4-7. The main receiver functional section must be operating properly.

(a) TEST EQUIPMENT REQUIRED.

- 1. Signal Generator AN/USM-44A
- 2. Signal Generator Pad CN-315/URM-26
- 3. Fuse MX- 1730/U
- 4. Electronic Multimeter TS-505/U

(b) TEST SETUP. - Set RT-695A/PRC-41 on the test bench with case CY-3884/PRC-41 removed, and connect PP-3700/PRC-41 to provide required primary power. Refer to figure 5-21. Connect junction box to KY-38 connector 1A9J14 on front panel of RT-695A/PRC-41.

(c) AF AMPLIFIER ASSEMBLY(BUFFER) FUNCTIONAL SECTION CIRCUIT, TROUBLE ISOLATION. -To determine which circuit is defective, perform the steps of procedure of table 4-7 in sequence. After each preliminary action, compare indications obtained with the expected result in the NORMAL INDICATION column. If the indications are normal, proceed to the next step. If abnormal indications are observed, follow the procedure in the NEXT STEP column. Before performing the steps of table 4-7, set the SQUELCH control on

#### TABLE 4-7. AF AMPLIFIER ASSEMBLY (BUFFER) CIRCUITS, TROUBLE ISOLATION

STEP	TEST POINT	PRELIMINARY ACTION	NORMAL INDICATION	NEXT STEP
1	Number 2 terminal on	Connect multimeter to terminal number	Measure 26.5 volts dc.	If normal indication is not
	buffer amplifier 1A9A1 and	2 of buffer amplifier modules 1A9A1		obtained, check power
	1A9A2 in main chassis	and 1A9A2.		distribution circuits.
2	MAIN RCVR AUX AUDIO	Connect AN/USM-44A to antenna	Measure at least 0.25 volt ac.	If normal indication in not
	output jack on junction box	connector 1A9J28. Set AN/USM-44A		obtained, replace buffer
		to 304.7 MHz, modulated 30% with		amplifier module 1A9A1.
		1000 Hz, with a signal output level of 3		
		microvolts. Set frequency control knobs		
		on front panel of RT-695A/PRC-41 to		
		304.7 MHz. Connect TS-505/U to		
		MAIN RCVR AUX AUDIO jack on		
		junction box.		
3	LOW LEVEL AUX	Set AN/USM-44A to 243.0 MHz,	Measure at least 0.15 volt ac.	If normal indication not
	GUARD AUDIO output	modulated 30% with 1000 Hz, with a		obtained, replace buffer
	jack on junction box	signal output level of 5 microvolts.		amplifier module 1A9A2.
		Connect TS-505/U to LOW LEVEL		
		AUX GUARD AUDIO jack on junction		
		box.		

RT-695A/PRC-41 to maximum counterclockwise position and OFF-T/R-T/R/G-DIAL LIGHT switch to T/R.

c. MAIN CHASSIS 1A9. - The main chassis of RT-695A/PRC-41 forms a mounting base for eight modular subassemblies and a control panel. The main chassis contains mechanical linkage between the frequency selector knobs on the control panel and tuned circuits of three of these modules. Refer to paragraph 4-6g(1)(a). The modules are secured to the main chassis by captive screws. Electrical interconnection between modules is provided by multipin connectors on the main chassis which mate with the applicable modules. Refer to figure 5-93. The main chassis also contains the receive-transmit switch, mixer circuit, control relays, fuses, cabling, af amplifier assembly (buffer), and filament regulator transistor 1A9Q1. (1) FUNCTIONAL DESCRIPTION.

(a) RECEIVE-TRANSMIT SWITCH. (Refer to figure 4-6.) - The receive-transmit switch is a rotary type coaxial switch having two rotor sections and is located in the main chassis. This switch may be actuated to one position in either direction by rotary relays 1A9L6 and 1A9L7. When the press-to-talk button on Handset H-33E/PT is pressed (equipment placed in transmit operation), relays 1A9K1, 1A9K3, and 1A9K4 become energized. Relay 1A9L6 then becomes energized, since it is grounded through contacts 8 and 10 of switch S1A and is provided 26.5 volts dc through contacts of relay 1A9K1. Relay 1A9L6 causes coaxial switches S1A and SIB to be rotated clockwise one position. At this time, the ground is removed from relay 1A9L6 and it is deenergized. The opposite condition exists when the press-to-talk button is released (equipment placed in receive operation). Releasing the press-to-talk button places 26.5 volts dc on relay 1A9L7 through contacts on relay 1A9K1. Relay 1A9K1. Relay 1A9L7 is energized, since it is grounded through contacts 3 and 6 of switch S1A. Relay 1A9L7 causes coaxial switches S1A and SIB to rotate counterclockwise one position. At this time, the ground is removed from relay 1A9K1. Relay 1A9L7 and deenergized. Since it is grounded through contacts 3 and 6 of switch S1A. Relay 1A9L7 causes coaxial switches S1A and SIB to rotate counterclockwise one position. At this time, the ground is removed from relay 1A9L7 and deenergized.

(2) TEST DATA. - Trouble isolation considerations for the main chassis of RT-695A/PRC-41 are concerned with main chassis cabling, detail parts of filtering and voltage regulation circuits, receive-transmit relays, and mechanical linkages. Refer to paragraph 4-6g for information pertaining to control circuits. Trouble isolation of the wiring of the main chassis cable and detail parts involves point-to-point continuity testing by use of figures 5-85 through 5-93. Refer to step 11 of table 4-4, step 5 of table 4-5, and step 8 of table 4-6 for testing procedures of the receive-transmit switch. Check filament regulator 1A9Q1 voltages by use of table 5-12. Mechanical troubles may be located by inspection.

d. PRIMARY POWER CIRCUITS.

(1) FUNCTIONAL DESCRIPTION.

(a) STORAGE BATTERY BB-451/U.

Storage Battery BB-451/U is used to provide



Figure 4-6. Receive-Transmit Switch, Simplified Schematic Diagram

required dc primary supply for RT-695A/PRC-41. The BB-451/U is made up of 16 series-connected cells. Each cell provides 1.86 volts open circuit when fully charged with a capacity of 25 ampere hours (nominal). The BB-451/U may be operated in any altitude without spilling out the electrolyte.

(b) POWER SUPPLY PP-3700/PRC-41. (Refer to figures 4-19 and 5-86.) - Power Supply PP-3700/PRC-41 is used to provide a required 26.5-volt dc primary supply for RT-695A/PRC-41 when a 115- (cables normally connected for 115 volts ac) or 230-volt, 50- to 400-Hz supply is available. The 115 or 230-volt, 50- to 400-Hz primary supply is stepped down to 45 volts by transformer T1, rectified, filtered, and made available to RT-695A/PRC-41 (figure 5-82). With the OFF-T/R-T/R/G-DIAL LIGHT switch on the front panel of RT-695A/PRC-41 in any position except OFF, power is applied through the filter to series regulator Q3. Dc amplifiers Q1 and Q2 supply required filtered base current for series regulator Q3. Diode CR7, in combination with a variable resistor, provides a voltage reference of 28 volts at the base of dc amplifier Q1. An overload protection circuit is provided by diode CR9, overload trip Q4, and voltage reference diode CR8. If the 26.5-volt dc output supply should become shorted and draw



Paragraph

4-6d(1)(b)

HOURS OF OPERATION Figure 4-7. Storage Battery BB-451/U, Typical Rate of Discharge

excessive current, voltage at the emitter of series regulator Q3 is caused to drop below the reference level of 15 volts dc set by zener diode CR8. This permits voltage reference diode CR9 and overload trip Q4 to conduct sufficiently to lower the reference voltage at the base of dc amplifier Q1, which prevents conduction of dc amplifiers Q1 and Q2 and series regulator Q3. The dc amplifiers and series regulator will remain turned off even though the higher load may be removed. The output circuit is held open until the OFF-T/R-T/R/G-DIAL LIGHT switch is set to OFF position for at least 5 seconds and then returned to one of the other positions of this control.

(2) STORAGE BATTERY BB-451/U, TEST DATA. - Trouble isolation considerations for Storage Battery BB-451/U are concerned with its ability to provide the required dc voltage for operation of RT-695A/PRC-41. Normal procedure for checking the level of charge of a battery is by measuring the terminal voltage developed across a very high load resistance. This procedure provides very little indication when applied to BB-451/U because of the characteristics of charge and discharge (figure 4-7). Instructions for battery use and a log for recording battery charges are printed on the inside of the battery cover.

(3) POWER SUPPLY PP-3700/PRC-41, TEST DATA. - Trouble isolation considerations for Power Supply PP-3700/PRC-41 are concerned with the ability of the power supply to provide the required 26.5 volts for operation of RT-695A/PRC-41. If PP-3700/PRC-41 does not provide the required 26.5-volt dc supply, connect it into the test setup described in paragraph 4-6d(3)(b), and perform the trouble isolation procedures of table 4-7.

(a) TEST EQUIPMENT REQUIRED. Multimeter AN/PSM-4C and a power resistor are required for trouble isolation procedures.

(b) TEST SETUP. - Set Power Supply PP-3700/PRC-41 on the test bench with the side which clamps to the rear of RT-695A/PRC-41 upward. Loosen two captive screws in top plate, and remove main chassis from the dust cover. Connect Power Electric Cable Assembly CX-8687/PRC-41 between ac power connector on PP-3700/PRC-41 dust cover and a 115-volt, 50- to 400-Hz primary power source. Refer to figure 4-8.

(c) POWER SUPPLY PP-3700/PRC-41, CIRCUIT TROUBLE ISOLATION. - To determine which circuit or circuit group is defective, perform the steps of procedure of table 4-8 in the sequence listed. After each preliminary action, compare the indications obtained with expected results in the NORMAL INDICATION column. If indications are normal, proceed to the next step. If abnormal indications are observed, follow procedures outlined in the NEXT STEP column.

# e. POWER DISTRIBUTION CIRCUITS.

AN/PRC-41A

(1) FUNCTIONAL DESCRIPTION. (Refer to figure 4-9.) - Power distribution circuits are the



Figure 4-8. Power Supply PP-3700/PRC-41, Test Setup



### TABLE 4-8. POWER SUPPLY PP-3700/PRC-41. TROUBLE ISOLATION

STEP	TEST POINT	PRELIMINARY ACTION	NORMAL INDICATION	NEXT STEP
1	12 Wiring side of plugs P1 and P3 (figure 4-8)	Connect a jumper between P3 and P2. Adjust power resistor for 8 ohms, and connect it between P1 and P3. Connect AN/PSM-4C to test point 12 to P1, and negative lead to P3.	Measure 26.5 volts dc.	If output voltage is not 26.5 volts dc but is less than 30 volts dc and greater than 24 volts dc, adjust variable resistor R6 for 26.5-volt dc output. If output cannot be adjusted to 26.5 volts dc, proceed to step 4. If output voltage is zero, proceed to step 2.
2	Visual	Remove ac input power to PP- 3700/PRC-41. Check fuses F1 and F2.	Fuses F1 and F2 check normal.	Proceed to step 3.
3	14 Terminals 5 and 6 of power transformer T1 (figure 4.8)	Connect AN/PSM-4C from terminal number 5 of transformer T1 to ground	Measure 45 <u>+</u> 3 volts dc .	If indication is abnormal, check transformer T1.
4	13 Refer to figure 4-8	Connect AN/PSM-4C between test point 13 and ground.	Measure 41 $\pm$ 3 volts dc.	If voltage is zero, check diodes CR1 through CR4 and capacitor C10
5	15 Red lead on inductor L1	Connect AN/PSM-4C between test point 15 and ground.	Measure 41 ±3 volts dc.	If normal indication is not obtained, remove jumper wire from P3 momentarily, reconnect it to P3, and listen for relay K1 to click. If relay K1 does not click, check relay K1. If relay K1 clicks, check inductor L1 and capacitor C1.
6	16 Refer to figure 4-8	Connect AN/PSM-4C between test point and 16 and ground.	Measure 27 <u>+</u> 3 volts dc	If normal indication is not obtained, measure voltages on transistors Q1, Q2, and Q4 given in table 5-13. Isolate defective component by indication of voltage measurements and checking associated circuits.
7	12 Plug P3 (figure 4-8)	Connect AN/PSM-4C between 12 test point ground.	Measure 27 <u>+</u> 2 volts	If normal indication is not obtained , measure voltage on transistor Q3 given in table 5-13. Transistor Q3 is mounted in heat sink in dust cover. Check cabling between chassis and transistor Q3.

#### TABLE 4-8. (Continued)

STEP	TEST POINT	PRELIMINARY ACTION	NORMAL INDICATION	NEXT STEP
8	12	Connect AN/PSM-4C to test point 12,	Measure 26.5 volts dc before	If normal indications are
		positive lead to P3, and negative lead to	connecting jumper, 0 volt dc after	not obtained, measure
	Refer to figure 4-8	P1. Connect jumper from P1 to P3	shorting P1 to P3.	voltages on transistors
		momentarily, then remove jumper.		Q1, Q2, Q3, and Q4 given
				in table 5-14. Isolate
				defective component by
				indication of voltage
				measurements and
				checking associated
				circuits. Voltage at test
				point 12 should return to
				26.5 volts dc by
				disconnecting jumper from
				P2 to P3, waiting 5
				seconds then reconnecting
				jumper from P2 to P3.

switching and circuits which provide required power for operation of the circuits and detail parts or Radio Receiver-Transmitter RT-695A/PRC-41. Primary 26.5 volts dc for operating RT-695A/PRC-41 is applied through electrical contacts of Receiver-Transmitter Case CY-3884/PRC-41 to the main chassis of RT695A/PRC-41. This primary power may be supplied by Storage Battery BB-451/U, Power Supply PP3700/PRC-41, or by an aircraft or vehicular power source. The 26.5-volt dc primary supply is used to provide power for the 180- and -6.8 volts dc supplies in the dc power supply module, to energize relays for receive-transmit switching, and for filament and transistor voltages.

(a) DC POWER SUPPLY MODULE 1A1. (See figure 5-85.) - When the OFF-T/R-T/R/G-DIAL LIGHT switch is placed in T/R position, the 26.5-volt dc primary supply is applied to terminal 4 of transformer 1A1T1 and the collectors of both switching transistors 1A1Q1 and 1A1Q2 of the dc power supply module. Since no two transistors are exactly alike, either Q1 or Q2 will begin to conduct because of positive biasing of the bases of Q1 and Q2. Assuming Q1 begins to conduct first, current flows through that section of the primary winding of transformer T1 between terminals 3 and 4. As the field about winding 3-4 expands, an induced voltage is developed across the secondary 8-9 and the base feedback windings 1-2 and 6-7. Transistor Q1 continues to conduct until the core of transformer T1 becomes saturated. At saturation, the base voltage of Q1 goes to zero and Q1 stops conduction. The magnetic field about the primary winding of 1-2 and 3-4 begins to collapse, and Q2 is switched on and begins to conduct. Switching transistor Q2 continues to conduct until it has saturated the core of transformer T1, and then Q1 is switched on again. Transistors Q1 and Q2 provide an 800-Hz output to bridge rectifiers CR1 through CR4. Inductor L1 and capacitors C1, C2, and C3 form a low-pass filter. Resistor R5 is a bleeder resistor for the filtered output. Zener diode CR5 is connected between ground and the B-supply and provides the -6.8-volt dc output. The B+ supply is 180 volts dc and is made available to modulator, spectrum generator, and rf and power amplifier modules.

(b) FILAMENT CIRCUITS. - The vacuum tubes requiring filament voltage are contained in spectrum generator and rf and power amplifier modules. Refer to figure 4-10. These filaments are connected in series parallel. All tubes require 6.3 volts for filament operation. Resistors 1A9R2, 1A9R3, 1A9R6, and 1A9R8 are voltage equalizing resistors. Transistor 1A9Q1 is a filament voltage regulator.

# AN/PRC-41A TROUBLESHOOTING



Figure 4-9. Radio Receiver-Transmitter RT-695A/PRC-41, Power Distribution, Simplified Schematic Diagram ORIGINAL **4-37, 4-38** 

# NAVSHIPS 0967-982-5020 TM-03816B-35/2 TM11-5820-510-35-1

Figure 4-9



Figure 4-10. Radio Receiver-Transmitter RT-695A/PRC-41, Filament Circuits

(c) RECEIVE-TRANSMIT RELAYS, DIAL LIGHT, AND GUARD RECEIVE MODULE. The 26.5-volt dc primary supply is available to receive-transmit relays, dial lights, and transistor stages of the guard receiver module. Refer to paragraphs 4-6g(1)(b) and 4-6g(1)(c) for a description of power distribution and operation of receiver-transmit relays. Dial lights are lit when OFF-T/R-T/R/G DIAL LIGHT switch is in the DIAL LIGHT position for lighting the front panel frequency indicator of RT-695A/PRC-41. When OFF-T/R-T/R/G-DIAL LIGHT switch is placed in T/R/G position, 26.5-volt dc primary supply is applied to the collector and base of guard receiver voltage regulator 1A7Q14. The emitter of voltage regulator Q14 provides a regulated supply

for transistor stages of the guard receiver module.

(2) TEST DATA. - If the power distribution circuits are suspected or known to be defective, perform the steps of procedure of table 4-9 to isolate the trouble to a defective circuit or circuit group. Refer to the servicing block diagram, figure 4-19, for location of test points.

(a) TEST EQUIPMENT REQUIRED. Test equipment required consists of Electronic Multimeter TS-505/U.
 (b) TEST SETUP. - Set RT-695A/ PRC-41 on the test bench with CY-3884/PRC-41 removed. To provide required power, connect PP3700/PRC-41 to the RT-695AJPRC-41 (figure 5-21). Set OFF-T/R-T/R/G-DIAL LIGHT switch to T/R position. Connect Handset H-33E/PT to connector 1A9J13 on front panel of RT-695A/PRC-41.

#### AN/PRC-41A TROUBLESHOOTING

STEP	TEST POINT	PRELIMINARY ACTION	NORMAL INDICATION	NEXT STEP
1	Fuse 1A9F1 on rear of RT- 695A/PRC-41 main chassis	Connect Electronic Multimeter TS- 505/U from fuse test jack to ground.	Measure 26.5 volts dc.	If normal indication is not obtained, check PP- 3700/PRC-41 and fuse 1A9F1.
2	Collector of 1A9Q1	Connect TS-505/U from collector of 1A9Q1 to ground.	Measure 26.5 volts dc.	If normal indication is not obtained, check fuse 1A9F1 and OFF-T/R- T/R/G/ DIAL LIGHT switch 1A9S3.
3	Emitter of 1A9Q1	Connect TS-505/U from emitter of 1A9Q1 to ground.	Measure 18.9 volts dc.	If normal indication is not obtained, check voltage at base of transistor 1A9Q1 (measure 20 volts dc).
4	21 See test jack 1A9J25.	Connect TS-505/U from 1A9J25 to ground	Measure 6.3 volts dc.	If normal indication is not obtained, measure tube filaments in spectrum generator and rf and power amplifier module for continuity (measure approximately 4 ohms).
5	Test jack 1A1J2	Connect TS-505/U from 1A1J2 to ground.	Measure 26.5 volts dc.	If normal indication is not obtained, remove dc power supply module, and check continuity between pin 3 of 1A9J5 and fuse test jack 1A9F1.
6	Test jack 1A1J3	Set TS-505/U to 250-volt dc range, and connect it from 1A1J3 to ground.	Measure 180 volts dc.	If normal indication is not obtained, check transistors Q1 and Q2, bridge rectifiers CR1 through CR4, and associated circuits of dc power supply module.
7	Test jack 1A1J1	Set TS-505/U to 50-volt dc range, and connect it from 1A1J1 to ground.	Measure -6.8 volts dc.	If normal indication is not obtained, check zener diode 1A1CR5.
8	18 Refer to figure 4-1	Connect TS-505/U from junction of capacitor 1A9C2 and inductor 1A9L2	Measure 26.5 volts dc.	If normal indication is not obtained, check continuity of inductor 1A9L2.
9	Pin 9 of connector 1A9J8	Remove audio module. Connect TS- 505/U from pin 9 of 1A9J8 to ground.	Measure 24.5 volts dc.	If normal indication is not obtained, measure continuity between pin 9 of 1A9J8 and 1A9J2.

#### TABLE 4-9. POWER DISTRIBUTION CIRCUITS, TROUBLE ISOLATION

ORIGINAL

4-40

# AN/PRC-41A TROUBLESHOOTING

# NAVSHIPS 0967-872-5020 TM-03816B-35/2 TM11-5820-510-35-1

# TABLE 4-9. (Continued)

STEP	TEST POINT	PRELIMINARY ACTION	NORMAL INDICATION	NEXT STEP
10	Pins 5 and 11 of connector 1A9J7	Replace audio module, and remove 3 <sup>rd</sup> if and squelch module. Connect TS- 505/U in turn from 1A9J7 to pin 5 to ground and pin 11 to ground.	Measure 24.5 volts dc.	If normal indication is not obtained, measure continuity from pin 5 1A9J7 to 1A9L2. Check relay 1A9K3.
11	Pin 3 connector 1A9J6	Replace 3 <sup>rd</sup> if and squelch module. Set frequency selector knobs to 299.9 MHz. Remove 1 <sup>st</sup> and 2 <sup>nd</sup> if amplifier module. Press press-to-talk button on H- 33E/PT. Connect TS-505/U from pin 3 of 1A9.16 to ground.	Measure 26.5 volts dc.	If normal indication is not obtained, check relay 1A9K3.
12	Pin 6 of connector 1A9J6	Unkey press-to-talk button on H- 33E/PT. Connect TS-505/U from pin 6 of 1A9J6 to ground.	Measure 24.5 volts dc.	If normal indication is not obtained, measure continuity between pin 6 of 1A9J6 and 1A9L2.
13	Pin 3 of connector 1A9J9	Replace 1 <sup>st</sup> and 2 <sup>nd</sup> if amplifier module, and remove guard receiver module. Connect TS-505/U from pin 3 of 1A9J9 to ground. Set OFF-T/R-T/R/G-DIAL LIGHT switch to T/R/G.	Measure 26.5 volts dc.	If normal indication is not obtained, check function switch 1A9S3 and relay 1A9K4.
14	Pin 3 of connector 1A9J10	Replace guard receiver module, and remove modulator module. Connect TS-505/U from pin 3 of 1A9J10 to ground. Press press-to-talk button on H-33E/PT.	Measure 26.5 volts dc.	If normal indication is not obtained, measure continuity between pin 3 of 1A9J10 to 1A9L2. Check relay 1A9K3.
15	Pin 8 of connector 1A9J10	Connect TS-505/U from pin 8 of 1A9J10 to ground.	Measure 24.5 volts dc.	If normal indication is not obtained, measure continuity between pin 8 of 1A9J10 to 1A9J10. Check relav 1A9K3.
16	Pin 20 of connector 1A9J12	Unkey press-to-talk button on H- 33E/PT. Replace modulator module. Remove rf and power amplifier module. Connect TS-505/U from pin 20 of 1A9J12 to ground.	Measure 24.5 volts dc.	If normal indication is not obtained, measure continuity between pin 20 of 1A9J12 and 1A9L2.

Paragraph 4-6e(2)(b)

#### NAVSHIPS 0967-872-5020 TM-03816B-35/2 TM11-5820-510-35-1

(c) POWER DISTRIBUTION CIRCUIT, TROUBLE ISOLATION. - To determine which circuit or circuit group of the power distribution circuits is defective, perform the steps of procedure of table 4-9 in the sequence listed. After each preliminary action, compare indications obtained with expected results in the NORMAL INDICATION column. If indications are normal, proceed to the next step. If abnormal indications are observed, follow procedures outlined in the NEXT STEP column.

f. FREQUENCY GENERATOR CIRCUITS. - The frequency generator circuits are composed of circuits of the spectrum generator module and high- and low- frequency oscillator stages in the 1st and 2nd if amplifier module. These circuits provide required frequency injection for both the main receiver and transmitter functional sections. The following frequency generation discussion is applicable to that required for frequency conversion and carrier development of the main receiver and transmitter functional sections of Radio Receiver-Transmitter RT-695A/PRC-41. Refer to paragraph 4-4a for a discussion of frequency generation for the guard receiver functional section.

(1) FUNCTIONAL DESCRIPTION, RECEIVE OPERATION. - Three injection frequency signals for RT-695A/PRC-41 are obtained from three crystal controlled oscillators. The first injection signal, 200 to 370 MHz, is obtained from the spectrum generator module, while the second, 17.1- to 26.1-MHz, and third, 2.9- to 3.8-MHz, injection signals are obtained from high- and low-frequency oscillator stages in the 1st and 2nd if amplifier module. Figure 4-11 lists the three injection frequencies for each frequency in the operating range of RT-695A/PRC-41. In receive operation, signals in the 225.0 to 399.9-MHz range are mixed with appropriate spectrum generator module output frequency in the range of 200 to 370 MHz to produce the 20- to 29.9 MHz first if frequency. The first intermediate frequency is then mixed with the appropriate high- frequency oscillator output in the range of 17.1 to 26.1 MHz to provide the second intermediate frequency in the 2.9- to 3.8-MHz range. The second intermediate frequencies are then mixed with the appropriate low-frequency oscillator output in the range of 2.9 to 3.8 MHz to provide the third intermediate frequency of 500 kHz.

(2) FUNCTIONAL DESCRIPTION, TRANSMIT OPERATION. - In transmit operation, the appropriate low- and high- frequency oscillator crystal frequencies are mixed in transmitter mixer 1A2Q4 of 1st and 2nd if amplifier module to produce the intermediate frequencies in the range of 20.0 to 29.9 MHz. This intermediate frequency is then mixed in diode mixer 1A9CR1 with the appropriate spectrum generator module output frequency to produce an output signal in the 225.0- to 399.9-MHz range.

(3) IF FREQUENCY CALCULATION. - The following procedure may be used to find the three injection frequencies from the final transmit or receive operation frequency. Refer to figure 4-11. Example: 294.6 MHz Subtract 20.0 MHz from the operating frequency. For this example, 20 from 294.6 gives

Example: 294.6 MHz <u>-20.0 MHz</u> <u>27</u> 4 6 MHz 270.0 MHz	Subtract 20.0 MHz from the operating frequency. For this example. 20 from 294.6 gives 274.6. The first two digits of this number (10-MHz digits) with an added 0 gives 270MHz, which is the spectrum generator module injection frequency.
29 <u>4</u> .6 MHz 4.0 <u>+17.0</u> 21.1 MHz	To determine the 17.1- to 26.1-MHz oscillator frequency, add 17.1 to the unit digit. In this case $4 + 17.1 = 21.1$ MHz, which is the high-frequency oscillator injection .frequency.
294.6 MHz	In transmit operation, the low-frequency oscillator frequency can be determined by adding 2.9 MHz to the 10th megahertz digit. In this case, $0.6 + 2.9 = 3.5$ MHz which is the low-frequency oscillator injection frequency.
0.6 MHz	
<u>+2.9</u>	
3.5 MHz	
In receive operation, the the oscillator frequence frequency of the second seco	the low-frequency oscillator is shifted +500 kHz from the predetermined transmit frequency. Since by varies from 2.9 to 3.6 MHz, this frequency can be determined from the transmit oscillator
Example: 294.6 MHz	294.2 MHz
<u>^ </u>	

Example: 294.6 MHz	294.2 MHz
0.6	0.3
+2.9	+2.9
3.5 MHz Transmit	3.1 MHz
frequency	
<u>-500 kHz</u>	+.500 kHz
3.0 MHz Receive	3.6 MHz
frequency	

(a) SPECTRUM GENERATOR MODULE. (Refer to figure 5-89.) - The spectrum generator module supplies an injection signal of 200 to 370 MHz in 10-megahertz steps to the receiver first mixer or the transmitter second

mixer (diode mixer 1A9CR1). The spectrum generator module consists of a crystal oscillator followed by a multiplier and a 2-stage rf amplifier. The crystal oscillator is a series mode crystal oscillator with 1 of 18 crystals in the cathode circuit of V1. The crystals are opened on the fifth mode. Both the crystal and the plate tank circuit of V1, which is tuned to the crystal frequency, are switched to obtain the desired frequency. Capacitor C6 is the capacity portion of the plate tank.



Figure 4-11. Main Receiver and Transmitter, Functional Section, Frequency Scheme

Choke coil L19 tunes out crystal reactance as different crystals are selected. Tuned circuit Z1 and capacitor C10, in plate circuit of V2, is tuned to triple or quadruple the oscillator frequency for output frequencies from 200 to 370 MHz. This signal is fed to two grounded-grid amplifiers, V3 and V4. The plate circuits of V3 and V4 are tuned with plate tank circuits identical to the plate tank circuit of V2. When a new frequency is selected, both the inductance and capacitance of tuned circuits Z1, Z2, and Z3 are varied to obtain the proper resonant frequency. The output is fed to mixer 1A9CR1 in the main chassis.

(b) HIGH-FREQUENCY OSCILLATOR 1A2Q5. - High-frequency oscillator 1A2Q5 is located in the 1st and 2nd if amplifier module. This 17.1-to 26.1-MHz oscillator is a feedback type oscillator employing series-mode crystals. The oscillator output is switched from second receiver mixer 1A2Q3 to transmitter mixer 1A2Q4 by energizing relay 1A2K2. The undesired harmonics of high-frequency oscillator Q5 are attenuated by a low-pass filter composed of R28 inductors L26 through L29 and the input capacitance of the respective mixers.

(c) LOW-FREQUENCY OSCILLATOR 1A2Q7. -Low-frequency oscillator 1A2Q7 is located in the 1st and 2nd if amplifier module and is similar to a Colpitts vacuum tube oscillator. Switch S3 switches 2.9- to 3.8-MHz crystals in transmit operation, and with the aid of relay 1A2K3, changes crystal frequency by ±500 kHz for receive operation. With 1A2K3 in the receive position, low-frequency oscillator

1A2Q7 frequency is +500 kHz from the second intermediate center frequency. With 1A2K3 in the transmit position (energized), the low-frequency oscillator and second intermediate frequencies are identical. Relay 1A2K3 also grounds the low-frequency oscillator receive crystal in this position. The low frequency oscillator signal is fed through transistor 1A2Q6, which acts as a buffer amplifier for the low- frequency oscillator in transmit operation, to transmitter mixer 1A2Q4.

(4) TEST DATA. - Trouble isolation considerations for the frequency generation circuits are concerned with their ability to provide the required frequency injection for receive and transmit operations of RT-695A/PRC-41. If trouble is known or suspected as being caused by a defective frequency generation circuit, connect RT-695A/PRC-41 into the test setup described in paragraph 4-6f(4)(b), and perform the trouble isolation procedures of table 4-10. Refer to figure 4-1 and the servicing block diagram (figure 4-19) for location of test points and signal flow of the frequency generation circuits.

505/U.

(a) TEST EQUIPMENT REQUIRED. - Test equipment required consists of Electronic Multimeter TS-

(b) TEST SETUP. - Set RT-695A/PRC-41 on the test bench with CY-3884/PRC-41 removed, and connect PP-3700/PRC-41 to provide required primary power. Refer to figure 5-21. Set frequency selector knobs on front panel of RT-695A/PRC-41 to 399.9 MHz. Set OFF-T/R-T/R/G-DIAL LIGHT switch to T/R.

(c) FREQUENCY GENERATION CIRCUIT, TROUBLE ISOLATION. - To determine which frequency generation circuit and which detail part of the generation circuit is defective, perform the steps of procedure of table 4-10 in the sequence listed. After each preliminary action, compare indications obtained with expected result in the NORMAL INDICATION column. If indications are normal, proceed to the next step. If abnormal indications are observed, follow procedures outlined in the NEXT STEP column. Frequency generation circuits are located in the spectrum generator module (200- to 370-MHz output), and the 1st and 2nd if amplifier module (high-frequency oscillator 1A2Q5, 17.1- to 26.1-MHz output, and low-frequency oscillator 1A2Q5, 2.9- to 3.8-MHz output).

g. CONTROL CIRCUITS.

(1) FUNCTIONAL DESCRIPTION. - The control circuits are composed of the controls on the front panel of Radio Receiver-Transmitter RT-695A/PRC-41, and circuits and mechanical linkages between front panel and the controlled module or circuit. Table 3-1 of Operator's Section of Installation and Operation Technical Manual for Radio Set AN/PRC-41A, NavShips 0967-872-5010, lists and describes the controls of RT-695A/PRC-41. The control circuits include application of primary power; selection of either transmit/receive, transmit/receive/guard, or transmit/receive/guard and dial light operation; control of receiver output level; setting of main receiver squelch level; and frequency selection. The three frequency selector knobs on the front panel of RT-695A/PRC-41 are mechanically linked to the tuned circuits of rf and power amplifier, 1st and 2nd if amplifier, and spectrum generator modules. Rotation of any of the three frequency indicator window on the front panel of RT-695A/PRC-41. The control (VOL) volume permits adjustment of the audio output level of both the main and guard receivers. The SQUELCH control provides an adjustment which will permit main receiver audio output only when a signal is present at the input of the receiver.

(a) MECHANICAL FREQUENCY SELECTION. (Refer to figure 4-12.) - The selection of a desired frequency within the range of 225.0 to 399.9 MHz is provided by three control knobs on the control panel and is indicated by the frequency dial reading in the MHz frequency indicator window. Assume the MHz frequency indicator window shows 225.0 MHz. If the 10-MHz knob (located on the left of the control panel) is rotated one detent position clockwise, the MHz frequency indicator window would show 235.0 MHz. If the 1-MHz knob (located in the center of the control panel) is rotated one detent position clockwise, the MHz frequency indicator window would show 236.0 MHz. If the 0.1-MHz knob (located on the right side of the control panel) is rotated one detent position clockwise, the MHz frequency window would show 236.1 MHz. A change of one detent position of the 10-MHz knob causes the coupler that drives the rf and power amplifier module to be rotated through 10 degrees and the coupler that drives the spectrum generator module to be rotated through 20 degrees. A change of one detent position of the 1-MHz knob causes the coupler that drives the rf and power amplifier module to be rotated through 1 degree; the coupler that selects crystals and oscillator coils for the high-frequency oscillator of 1st and 2nd if amplifier module to be rotated through 30 degrees; and the coupler that adjusts the high-frequency slug rack on 1st and 2nd if amplifier module to be rotated through 30 degrees. A change of one detent position of the 0.1-MHz knob causes the coupler that drives the low-frequency slug rack and selects crystal for low-frequency oscillator of the 1st and 2nd if amplifier module to be rotated through 30 degrees; the coupler that drives the high-frequency slug rack to be rotated through 3 degrees; and the coupler that drives the rf and power amplifier module to be rotated through 0.1 degree. In this way, the electrical circuits of the RT-695A/PRC-41 may be adjusted to select any desired frequency in the range of 225.0 to 399.9 MHz.

(b) RECEIVE OPERATIONAL SWITCHING. -Figure 4-13 shows Radio Receiver-Transmitter RT-695A/PRC-41 in receive operation. With the OFF-T/R-T/R/G-DIAL LIGHT switch changed from OFF to any of the other positions, the 26.5-volt dc primary power is applied to the circuits of

#### Table 4-10

#### TABLE 4-10. FREQUENCY GENERATION CIRCUIT TROUBLE ISOLATION

STEP	TEST POINT	PRELIMINARY ACTION	NORMAL INDICATION	NEXT STEP
1	3 Test jack 1A2J2 on 1 <sup>st</sup> and 2 <sup>nd</sup> if amplifier module	Connect Electronic Multimeter TS- 505/U between test jack 1A2J2 and	Measure -2.5 to -4.0 volts dc.	If indication is not obtained, proceed to step
2	Same as step 1	Same as step 1 except rotate the 10 <sup>th</sup> MHz frequency selector knob from .9 through .0.	Same as step 1.	If indication is not obtained, check those low- frequency crystal circuits associated with selector knob setting, and measure voltages of transistor 1A2Q7 (table 5-3).
3	4 Test jack 1A2J1 on 1 <sup>st</sup> and 2 <sup>nd</sup> if amplifier module	Connect TS-505/U between test jack 1A2J1 and ground.	Measure -0.5 to -1.5 volt dc.	If indication is not obtained, proceed to step 4.
4	Same as step 3	Same as step 3 except rotate 1-MHz frequency selector knob from 9. to 0	Same as step 3.	If normal indication is not obtained, check those high-frequency crystal circuits associated with frequency selector knob setting, and measure voltages of transistor 1A2Q5 (table 5-3).
5	19 Test jack 1A5J1 on spectrum generator module.	Connect TS-505/U between test jack 1A5J1 and ground.	Measure -5 to -20 volts dc.	If normal indication is not obtained, proceed to step 6.
6	Same as step 5	Same as step 5, but rotate hundreds- tens frequency selector knob from 39 to 22.	Same as step 5.	If normal indication is not obtained, check those spectrum crystal circuits as associated with frequency selector knob setting, and measure voltages of oscillator 1A5V1 (table 5-6).
7	5 Test jack 1A5J2 on spectrum generator module	Connect TS-505/U between test jack 1A5J2 and ground. Set function switch to T/R.	Measure -1.5 to -5 volts dc.	If normal indication is not obtained, make voltage and resistance checks of multiplier 1A5V2, and of amplifiers 1A5V3 and 1A5V4 (tables 5-6 and 5-7) to aid in location of defective detail.


Figure 4-12. Frequency Selection, Functional Diagram

RT-695A/PRC-41. In receive operation (press-to-talk button on H-33E/PT not depressed), signals in the 225.0- to 299.9-MHz range are received at the antenna, coupled through switch contacts 2 and 5 of receive-transmit switch 1A9S1A, and applied to the input of the rf and power amplifier module. The output signal from rf and power amplifier module is coupled through contacts 2 and 4 of receive-transmit switch 1A9S1B to diode mixer 1A9CR1 where it is mixed with the spectrum generator module output signal. The frequency difference between the spectrum generator signal and the received signal is fed through contacts 11 and 12 of receive-transmit switch 1A9S1B to contacts of deenergized relay 1A2K1, and to the input of 20- to 29.9-MHz if amplifiers of 1st and 2nd if amplifier module. In the 1st and 2nd if amplifier module, the resulting signal is further mixed with the output of the low and high-frequency oscillators to produce the final intermediate frequency. The output of 1st and 2nd if amplifier module is coupled to the input of 3rd if and squelch module, where it is detected and fed through contacts of deenergized relay 1A9K4 to the audio module input. The output of the audio module is coupled to H-33E/PT and to pin J of K4-38 connector 1A9J14. When squelch relay 1A3K1 is energized, pin K of 1A9J13 is grounded for automatic relay operation.

(c) TRANSMIT OPERATIONAL SWITCHING. - Figure 4-14 shows Radio Receiver-Transmitter RT-695A/PRC-41 in transmit operation.

Depressing press-to-talk button on the H-33E/PT provides a ground for relays in the main chassis (relays 1A9K1, 1A9K3, and 1A9K4). Energizing relay 1A9K1 causes relay 1A9L6 of the receive-transmit switch to be energized and rotates the receive-transmit coaxial switch to the transmit position. Contacts of relay 1A9K1 ground the collector of transistor 1A9K1 in rf and power amplifier module. Contacts of relay 1A9K3 remove the filtered 24.5-volt dc supply from the receive circuits, apply it to the transmit circuits, and apply 26.5 volts dc to the modulator module and to the relays of 1st and 2nd if amplifier module. Contacts of relay 1A9K4 break the audio output connection from the 3rd if and squelch module to the audio module, apply side-tone from modulator module to the audio module, and remove 26.5-volt dc supply from the guard receiver module. Contacts of energized relay K3 in the 1st and 2nd if amplifier module connect the transmit rotor of crystal switch S3 to the low- frequency oscillator and ground the receive crystal. Contacts of energized relay K2 of 1st and 2nd if amplifier module inject the signal from high- frequency oscillator into the transmitter mixer and ground the base of Q3. The sum of these two oscillator output frequencies, in the 20- to 29.9-MHz range, is fed through contacts of energized relay K1 of 1st and 2<sup>nd</sup> if amplifier module to the 20- to 29.9-MHz if amplifiers. The output of the 20- to 29.9-MHz if amplifiers is fed through contacts 10 and 11 of receive-transmit switch 1A9S1B to diode mixer 1A9CR1. The 20to 29.9-MHz if signals are mixed with spectrum generator module output signals in the diode mixer to produce output signals in the 225- to 399.9 MHz range. The output from diode mixer 1A9CR1 is fed through contacts 4 and 8 of receivetransmit switch 1A9S1B to the input of rf and power amplifier module. Output from the rf and power amplifier module is fed through contacts 9 and 2 of receive-transmit switch 1A9S1 to the antenna for radiation.

(2) CONTROL CIRCUITS TROUBLE ISOLATION. - Troubleshooting considerations for the control circuits of Radio Receiver-Transmitter RT-695A/PRC-41 are concerned with the ability of the operating controls to control the functional sections and circuits of RT-695A/PRC-41. Since control is very closely related to the particular functional section which is in operation, no special test is provided. Tests for the control circuits are covered as part of the test data of the respective functional sections or circuits. The inability of any one of the control circuits to function properly is an indication of control malfunction. Further troubleshooting of these circuits may be effected through visual inspection and continuity testing.

h. OMNIDIRECTIONAL AND DIRECTIONAL ANTENNAS. - Radio Receiver-Transmitter RT-695A/ PRC-41 may utilize either Antenna AS-1405/PRC-41, directional, or Antenna AS-1404/PRC-41, omnidirectional, (AS-1404/PRC-41 and AS-1405/PRC-41 should each indicate an open circuit).

(1) ANTENNA AS-1405/PRC-41. - Antenna AS-1405/PRC-41 is used with RT-695A/PRC-41 for reception and transmission of signals in the 225.0- to 399.9-MHz range without electrical or physical adjustments, other than extending the elements which are normally collapsed for storage. This antenna is a planar log periodic dipole array which provides unidirectivity characteristics. The nominal input impedance is 50 ohms. The azimuthal plane voltage patterns are shown in figure 4-15 for several frequencies in the range. The voltage standing-wave ratio for frequencies between 220 and 400 MHz is shown in figure 4-16. AS-1405/PRC-41 is operated on Mast AB-777/PRC-41.

(2) ANTENNA AS- 1404/PRC -41. - Antenna AS-1404/PRC-41 is used with RT-695A/PRC-41 for reception and transmission of signals in the 225.0- to 399.9-MHz range without electrical or physical adjustments. The AS-1404/PRC-41 has a nominal input impedance of 50 ohms. The azimuthal plane voltage pat-terns are shown in figure 4-17 for several frequencies in the range. These patterns apply when AS-1404/PRC-41 is mounted and operated on RT-695A/PRC-41. The voltage standing-wave ratios for frequencies between 220 and 400 MHz are shown in figure 4-18 when the AS-1404/PRC-41 is mounted and operated on the RT-695A/PRC-41. When AS-1404/PRC-41 is operated on Mast AB-777/PRC-41, the voltage standing-wave ratio properties and the voltage patterns are approximately the same as when operated on RT-695A/PRC -41.

4-7. HISTORY OF MODIFICATIONS.

Effective the date of this manual, there have been no modifications to Radio Set AN/PRC-41A.

# ORIGINAL 4-47, 4-48



Figure 4-13. Radio Receiver-Transmitter TR-695A/PRC-41, Receiver, Functional Switching Diagram

Figure 4-13

TPI-5043-016



Figure 4-14. Radio Receiver-Transmitter RT-695A/PRC-41, Transmitter, Functional Switching Diagram

ORIGINAL

4-51, 4-52

Figure 4-15



Figure 4-15. Antenna AS-1405/PRC-41, Azimuthal Plane Voltage Pattern

Figure 4-16

AN/PRC-41A TROUBLESHOOTING



Figure 4-16. Antenna AS-1405/PRC-41, Voltage Standing Wave Ratio



340 MHZ

Figure 4-17. Antenna AS-1404/PRC-41, Azimuthal Voltage Pattern

400MHZ

ORIIGINAL 4-55





Figure 4-18. Antenna AS-1404/PRC-41, Voltage Standing-Wave Radio



Figure 4-19. Radio Set AN/PRC-41A, Functional Section Servicing Block Diagram (Sheet 1 of 2)

Original

4-57, 4-58

# Figure 4-19

# AN/PRC-41A TROUBLESHOOTING



Figure 4-19. Radio Set AN/PRC-41A, Functional Section Servicing block diagram (Sheet 2 of 2)

Original

4-59, 4-60

NOT PART OF CHASSIS 149

# SECTION 5 MANTENANCE

## 5-1. FAILURE, AND PERFORMANCE AND OPERATIONAL REPORTS.

#### Note

The Naval Electronics Systems Command no longer requires the submission of failure reports for all equipments. Failure reports and performance and operational reports are to be accomplished for designated equipments (refer to Electronics Installation and Maintenance Book, NavShips 900.000) only to the extent required by existing directives. All failures shall be reported for those equipments requiring the use of failure reports.

## **5-2. PREVENTIVE MAINTENANCE.**

a. MAINTENANCE STANDARDS.

(1) GENERAL. - This section of the handbook provides complete and comprehensive preventive maintenance information so that proper equipment operation can be maintained.

(2) TEST EQUIPMENT AND SPECIAL TOOLS. -The following test equipment is required for performance of the preventive maintenance procedures of this section. No special tools are required.

- (a) Signal Generator AN/USM-44A.
- (b) Signal Generator AN/URM-25D.
- (c) RF Wattmeter AN/URM-43C
- (d) Oscilloscope AN/USM-105A.
- (e) Output Power Meter ME-2/U.
- (f) Audio Oscillator TS-382B/U.
- (g) Fuse MX-1730/U.
- (h) Electronic Multimeter TS-505/U.
- (i) Frequency Counter AN/USM-122.
- (j) Electronic Frequency Converter CV-394/USA-5.
- (k) Transfer Oscillator CM-102/USM-73.
- (I) Adjustable Attenuator CN-318/G.
- (m) Signal Generator Pad CN-315/URM-26.
- (n) Coaxial Crystal Detector HP-420A.
- (o) Power Resistor 204C.
- (p) Multimeter AN/PSM-4C.
- (q) Output Meter ME-6C/U.

(3) SPECIAL PROCEDURES. - Except for daily operational tests, all preventive maintenance tests are to be performed on a test bench having available the test equipment listed under paragraph 5-2a(2). Tests are to be performed with PP-3700/PRC-41 supplying required primary power. See figure 5-21. Initially, the OFF-T/R-T/ R/G-DIAL LIGHT switch should be in the OFF

position for making required test setups.

(a) JUNCTION BOX. - Figure 5-1 illustrates schematically a j unction box which is to be fabricated from bulk supplies. Preventive maintenance procedures make reference to and illustrate the use of this junction box. The junction box takes the place of H-33E/PT. It provides a means of keying the transmitter, a connection for audio outputs, and a dummy microphone load for signal input to RT-695A/PRC-41. Figure 5-2 provides a suggested layout for connectors and switches, identifying jacks and switch markings, and part numbers of the switch and connectors which must be used to be compatible with RT-695AJPRC-41 and the test equipment listed m paragraph 5-2a(2).

(b) Special Purpose Cable Assembly CX10831/PRC-41A Test. Refer to new figure 5-85.1 and 5-85.2 and perform a point-to-point continuity test (NMT 1.0 ohms). Use whichever figure is applicable.

(4) REFERENCE STANDARD PROCEDURES

#### NOTE

The procedures listed below consist of the minimum number of reference standards which will indicate, when completed, the relative performance of the equipment. Each group of tests represents a functional section of the equipment. The procedures are listed in the suggested sequence of performance; however, deviation from the listed order will m no way affect the results unless otherwise noted.

(5) PREVENTIVE MAINTENANCE PROCEDURES. (Refer to figures 5-4 through 5-20.) The following preventive maintenance procedures are arranged numerically within each maintenance period. Because of the nature of this equipment, all tests, except the dally operational tests, are to be performed on the test bench at a facility having the necessary test equipment. Bench performance tests are to be performed on a monthly and semiannual basis.

(6) PERIODIC SCHEDULE CHARTS. - The operator or faculty IS to prepare periodic schedule charts for recording and checking the results obtained from the preventive maintenance procedures listed In reference standards procedures, table 5-1. Figure 5-3 provides examples of the sort of periodic schedule charts which are to be prepared. These charts are to run for a period of 2 years for each of the reference standard daily, monthly, or semiannual tests.



Figure 5-1. Junction Box, Schematic Diagram



TP1-5046-013

Figure 5-2. Junction Box, Suggested Layout for Fabrication **5-2** 

## **TABLE 5-1**. REFERENCE STANDARDS PROCEDURES

SECTION	ACTION REQUIRED	REFERENCE
Main receiver	Make operational check.	D1
	Visually inspect gear train.	M1
	Record sensitivity.	M4
	Remove module covers to visually inspect detail parts.	SA1
	Record selectivity.	SA2
	Record avc characteristics.	SA4
	Record audio power output.	SA5
	Make audio capability check in addition to distortion check.	SA8
Guard receiver	Make operational check.	W1
	Record signal input level required to produce audio output.	M2
	Record sensitivity.	M3
	Record selectivity.	SA3
	Record avc characteristics.	SA6
	Record audio power output.	SA7
	Make audio capability check in addition to distortion check.	SA9
Transmitter	Make operational check.	D2
	Record power output.	M5
	Record percentage of modulation.	M6
	Record frequency stability.	SA10
	Record modulation fidelity and distortion.	SA11
Power Supply		SA12
PP-3700/PRC-41	Record 26.5-volt output supply voltage and ac ripple.	

Enter the name of the month in which the maintenance step is begun in the first empty block of the top row. Fill in the names of the months consecutively thereafter for a period of two years. Log the result in appropriate space after performing step 1.



Time Schedule: Check (  $\checkmark$  ) if communications was established, and initial 1st Year of Operation.

Day	19	19	19	19	19	19	19	19	19	19	19	19
1				-								
2												
3												
4												
5												
6												
7												
8												
9												
10												
11												

Time Schedule: Record and Initial

	1st year of Operation											
Month	19	19	19	19	19	19	19	19	19	19	19	19
Step												
Step												
Step												

					2nd ye	ar of Ope	ration					
Month	19	19	19	19	19	19	19	19	19	19	19	19
Step												
Step												
Step												

Figure 5-3. Examples of Periodic Schedule Charts To Be Prepared for Recording Preventive Maintenance Results

			STEPS D1 and D2
			5
OPERATING CON	DITIONS AND CONTROL SETTI	NGS:	
OFF-T/R-T/R/G-DI SQUELCH: Fully o VOL: Fully clockwi	AL LIGHT: OFF. counterclockwise. se, adjust for satisfactory receptio	n.	
STEP NO	ACTION REQUIRED	READ INDICATION ON	REFERENCE STANDARD
D1 and D2	Establish communications with another Radio Set AN/PRC- 41A or other uhf transceiver.	Listen for signal in Handset H-33E/PT.	Ability to receive on main receiver and to transmit.
	PROCEDURE: Set OFF-T/R-T/F warmup. Establish communicat Press press-to-talk button on H-	R/G-DIAL LIGHT switch to T/R, and ion with a netting AN/PRC-41A equ 33E/PT for transmissions, release f	allow equipment 5 minutes ipment or other uhf transceiver. or reception.

Figure 5-4. Preventive Maintenance Test, D1 and D2



Figure 5-5. Preventive Maintenance Test, W1

			STEP M1
OPERATI	NG CONDITIONS AND CONTROL SE	ETTINGS.	
Remove p RT-695A/I	rimary power from Radio Receiver-Tra PRC-41 and remove CY-3884/PRC-41	nsmitter case.	
STEP NO	ACTION REQUIRED	READ INDICATION ON	REFERENCE STANDARD
M1	Visually inspect main chassis gear train.	Visual.	Mechanically tuned modules can be adjusted mechanically by frequency adjustment of front panel frequency selector knobs. There shall be no evidence of damage.
	gear tram for evidence of wear and d throughout their range.	eterioration and excessive backlas	sh. Rotate frequency selector knobs

Figure 5-6. Preventive Maintenance Test, M1

STEP M	2		
OPERAT OFF-T/R SQUELC VOL: Ma	SIGNAL GEN AN/USM- SIGNAL GENERATOR CN-315/URM-20 CN-315/U	REATOR 44A J28 RT-695A/PRC-41 ANY FREQUENCY EXCEPT 243.0-MHZ W KY-38 CONNECTOR JUNCTION E	TR
STEP NO	ACTION REQUIRED	READ INDICATION ON	REFERENCE STANDARD
M2 (Cont)	Record signal input level required to produce audio output from guard receiver. PROCEDURE Connect AN/USM-44 coaxial cable to antenna connector 1 Connect junction box (figures 5-1 and Connect ME-2/U to NORMAL AUDIO minutes warmup. Adjust AN/USM-4 T/R/G-DIAL LIGHT switch to T/R/G	Signal output level indicator on Signal Generator AN/ USM-44A and Output Power Meter ME-2/U. A in series with a 6-dB pad, Fuse MX- A9J28 on front panel of Radio Received 5-2) to KY-38 connector 1A9J14 on f O OUTPUT Jack on junction box. Turn 4A to 243.0 MHz, modulated 30 perceived	Guard receiver should squelch on between2 and 5 microvolts. 1730/U, and a length of RG-213/U er-Transmitter RT-695A/PRC-41. front panel of RT-695A/PRC-41. on test equipment, and allow 10 nt with 1000 Hz. Set OFF-T/R-
	Figure 5-7. Prev	ventive Maintenance Test, M2 (Sheet ?	1 of 2)

				STEP M2
STEP NO	ACTION REQUIRED	READ INDICATION ON	REFERENCE STA	NDARD
M2 (Cont)	Be careful not to key RT-695A/ MX-1730/U provides protection transmitter output power to sign Set the frequency selector knobs on MHz. Beginning with zero output fro 2/U.	CAUTION PRC-41 with AN/USM-44A connected in event RT-695A/PRC-41 is accid al generator output will result in attend front panel of RT-695A/PRC-41 to an m AN/USM-44A, increase output until	d to antenna connect dentally keyed. App uator damage. y frequency other that an indication is obta	tor 1A9J28. blication of an 243.0 ±5 ained on ME-





Г

STEP M	3						
OPERAT	PERATING CONDITIONS AND CONTROL SETTINGS:						
OFF-T/R SQUELC VOL: Ma	-T/R/G-DIAL LIGHT: OFF. H. Maximum clockwise. Iximum clockwise.						
STEP NO	ACTION REQUIRED	READ INDICATION ON	REFERENCE STANDARD				
M3	Record guard receiver sensitivity.	Output Power Meter ME-2/U, Output Meter ME-6C/U.	NORMAL AUDIO SENSITIVITY10 dB or more; LOW LEVEL AUX GUARD AUDIO sensitivity 6 dB or greater.				
	PROCEDURE: Connect output of Signal Generator AN/USM-44A m series with a 6-dB pad through a length of RG-213/U coaxial cable to antenna connector 1A9J28 on front panel of Radio Receiver-Transmitter RT- 695A/PRC-41. Connect junction box (figures 5-1 and 5-2) to KY-38 connector 1A9J14 on front panel of RT- 695A/PRC-41 with switches in open position Connect Meter ME-2/U to NORMAL AUDIO OUTPUT jack on junction box. Connect ME-6C/U to LOW LEVEL AUX GUARD AUDIO jack on junction box. Connect Electronic Multimeter TS-505/U to avc Jack 1A7J2 on guard receiver module.						
		Note					
	Main receiver module must be	squelched off during guard receiver s	ensitivity test.				
	Adjust AN/USM-44A to produce an output of 5 microvolts modulated 30 percent with 1000 Hz. Set OFF-T/R-T/R/G-DIAL LIGHT switch to T/R/G, and allow 5 minutes warmup. Adjust AN/ USM-44A to 243.0 <i>MHz</i> as indicated by a maximum avc voltage on TS-505/U. Record normal audio output LOW LEVEL AUX GUARD OUTPUT m decibels as measured on the ME-2/U and ME6C/U Remove modulation, and again record ME-2/U and ME-6C/U indication Signal-plus-noise to noise ratio is difference of recorded values with and without modulation.						

Figure 5-8. Preventive Maintenance (Sheet 2 of 2)

				STEP M4				
	OPERATING CONDITIONS AND CONTROL SETTINGS:							
OPERATI	NG CONDITIONS AND CONTROL S	ETTINGS:						
OFF-T/R- SQUELCH VOL: Max S1 and S2	T/R-G-DIAL LIGHT: OFF. I: Maximum counterclockwise. kimum clockwise. Switches (junction box): Open positic	on.						
STEP NO	ACTION REQUIRED	READ INDICATION	REFERENCE STA	NDARD				
M4	Record main receiver sensitivity.	Output Power Meter ME-2/U Electronic Multimeter TS-505/U.	NORMAL AUDIO S 10 dB or more, MA AUDIO sensitivity 6	SENSITIVITY IN RCVR AUX 6 dB or greater.				
	PROCEDURE: Connect Signal Generator AN/USM-44A in series with a 6-dB pad and a length of RG-213/U coaxial cable to antenna connector 1A9J28 on front panel of Radio Receiver-Transmitter RT-695A/PRC-41, Connect junction box (figures 5-1 and 5-2) to KY-38 connector 1A9J14 on front panel of RT-695A/PRC-41. Connect ME-2/U to MAIN RCVR AUX AUDIO OUTPUT jack on junction box. Connect TS-505/U to avc jack 1A3J1 on 3rd if NORMAL AUDIO OUTPUT jack on junction box. Connect ME-6C/U to avc jack 1A3J1 on 3 <sup>rd</sup> if and squelch module.							
	Note							
	Guard receiver module m	ust be squelched off during main rece	eiver sensitivity test.					
	Adjust AN/USM-44A to produce an of T/R/G-DLAL LIGHT switch to T/R, at RT-695A/PRC-41 to 225.0 MHz. Se TS- 505/U. Record normal audio ou ME-2/U and ME-6C/U. Remove more noise ratio is difference of recorded of MHz.	butput of 3.mlcrovolts modulated 30 per nd allow 5 minutes for warmup. Set fr et AN/USM-44A to 225.0 MHz as indic tput and MAIN RCVR AUX AUDIO Of dulation, and again record indications value with and without modulation. Re	ercent with 1000 Hz. equency controls on ated by a maximum a JTPUT in decibels a in decibels. Signal-p epeat procedure at 3	Set OFF-T/R- front panel of avc voltage on s measured on olus-noise to 04.7 and 399.9				

STEP M5							
RF WATTMETER AN/USM-43C 2 FT OF RG-213/U RECEIVER-TRANSMITTER RT-695A/PRC-41 KY-38 CONNECTOR							
OPERATING OFF-T/R-T/R/	CONDITIONS AND CONTROL SET G-DIAL LIGHT: OFF	TINGS:					
STEP NO	ACTION REQUIRED	READ INDICATION ON	REFERENCE STANDARD				
M5	Record transmitter output.	RF Wattmeter AN/USM-43C.	2.25 watts minimum below 258 MHz. 3-watts power (minimum) of the channels at 258 MHz and above.				
	PROCEDURE: Connect a length o Radio Receiver-Transmitter RT-69 5-2) to KY-38 connector 1A9J14 or OFF-T/R-T/R/G-DIAL LIGHT switcl and measure power output for enor example, 225.0, 236.1, 247.2, etc.	f RG-213/U coaxial cable between a 5A/PRC-41 and AN/USM-43C. Conr n front panel of RT-695A/PRC-41 wit h to T/R, and allow 5 minutes warmu ugh frequencies to check all position	ntenna connector 1A9J28 on nect junction box (figures 5-1 and th switch in open position. Set up. Close switch on junction box, is of frequency selector knobs, for				



Figure 5-11. Preventive Maintenance Test, M 6 (Sheet 1 of 2)

STEP M6			
STEP NO	ACTION REQUIRED	READ INDICATION ON	REFERENCE STANDARD
M6 (Cont)	PROCEDURE: Connect Adjustable 695A/PRC-41 and RF Wattmeter AN output of CN-318/G to input of Trans 73 to Oscilloscope AN/USM-105A. C RT-695A/PRC-41. Connect Audio C Electronic Multimeter TS-505/U to M DIAL LIGHT switch to T/R, and adjust test equipment, and allow 10 minute 102/USM-73 to produce a difference input level of TS-382B/U for 0.7 volt from the MIC INPUT (OPEN CIRCU obtain a modulation envelope pattern Percent modulation = (P to P) - (V to (P to P) + (V to Check percent of modulation at 304.	Attenuator CN-318/G between antenn V/URM-43C using a 2-ft length of RG-2 ofer Oscillator CM-102/USM-73. Conn Connect Audio 5-1 and 5-2) to KY-38 of Dscillator TS-382B/U to MIC INPUT jac IIC INPUT (OPEN CIRCUIT) jack on just frequency control knobs for a frequency s warmup. Close switch S1 on junction frequency of approximately 200 kHz of (open circuit) at 1000 Hz as measured (open circuit) at 1000 Hz as measured iT) jack. Close switch S2 on junction n. Calculate percentage of modulation $\frac{0.V \times 100.}{100}$ to V) 7 and 399.9 MHz.	a connector 1A9J28 onRT- 213/U coaxial cable. Connect ect VIDEO output of CM-102/USM- connector 1A9J14 on front panel of ck on junction box. Connect unction box. Set OFF-T/R-T/G- ency of 225.0 MHz. Apply power to on box, and adjust frequency of CM- on AN/USM-105A. Adjust signal d on TS-505/U. Remove TS-505/U box, and adjust AN/USM-105A to n according to following formula:

Figure 5-11	Preventive	Maintenance	Test	M 6 (	Sheet 2 of 2	2)
i igule J-i i.	Flevenuve	Maintenance	i esi,	101 0 (		<u>-</u> )

STEP SA1			
OPERATIN	IG CONDITIONS AND CONTR	OL SETTINGS	
Remove pr	imary power from Radio Receiv	er-Transmitter RT-695A/PRC-4	1, and remove CY-3884/PRC-41 case.
STEP NO	ACTION REQUIRED	READ INDICATION ON	REFERENCE STANDARD
SA1	Visually inspect detail parts	Visual.	Detail parts and wiring of modules of RT- 695A/PRC-41 shall show no Visual evidence of damage or deterioration.
PROCEDURE: Remove RT-695A/PRC-41 modules from main chassis Remove module side covers, and visually inspect each module for evidence of charred or discolored wiring and detail parts, or other indications of damage.			

Figure 5-12. Preventive Maintenance Test, SA1

				STEP SA2
	ELECTRONIC FREQUENCY CONVERTER CU-394/USA-5	FREQUENCY COUNTER AN/USM-122	TRANSFER OSCIL CM-IO2/USM	LLATON -73
OPERAT	SIGNAL GENERATOR PADO CN-315/URM-26 HX-1730/U			
OFF-T/R SQUELC VOL: Ma	-T/R/G-DIAL LIGHT: OFF. CH: Maximum counterclockwise. aximum clockwise.			
STEP NO SA2	ACTION REQUIRED Record main receiver selectivity.	READ INDICATION ON Electronic Multimeter TS-505/U and Frequency Counter AN/USM- 122.	REFERENCE ST 80 kHz minimum at 6 and 150 kHz maximu level.	TANDARD -dB level, m at 60-dB
PROCEDURE: Connect Signal Generator AN/USM-44A in series with a 6-dB pad through a length of RG- 213/U coaxial cable to antenna connector 1A9J28 on front panel of Radio Receiver-Transmitter RT- 695A/PRC-41. Connect junction box (figures 5-1 and 5-2) to KY-38 connector 1A9J14 on front panel of RT- 695A/PRC-41. Connect audio Output Power Meter ME-2/U to NORMAL AUDIO OUTPUT jack on junction box. Connect TS-505/U to avc jack 1A3J1 on 3rd if and squelch module. Set Electronic Frequency Converter CV-394/USA-5 in place in AN/USM-122. Connect CV-394/USA-5 to Transfer Oscillator CM-102/USM-73. Turn on AN/USM-122 and CM-102/USM-73, and allow 15 minutes warmup. Set RT-695A/PRC-41 and AN/USM-44A (Cont) to 304.7 MHz unmodulated.				

Figure 5-13. Preventive Maintenance Test, SA2 (Sheet 1 of 2)

STEP SA2			
STEP NO	ACTION REQUIRED	READ INDICATION ON	REFERENCE STANDARD
SA2 (Cont)		Note	
(0011)	AN/USM-44A must be a bandwidth.	accurately tuned to 304.7 MHz to	give correct indications of
	Set OFF-T/R-T/R/G-DIAL LIGH warmup. Increase signal output approximately +7 volts is indica output frequency of AN/USM-44 44A output connection from 1At 102/USM-73, and record freque 695A/PRC-41, decrease output frequency at this point. The diffe Repeat this procedure at 60-dB	T switch to T/R, and apply power from AN/USM-44A until a main re ted on TS-505/U. Increase AN/US 4A until TS-505/U again indicates 9J28 on front panel of RT-695A/P ency at this point. Reconnect output frequency until avc reference level erence between two measured fre level.	to AN/USM-44A. Allow 5 minutes eceiver avc reference level of SM-44A signal output 6 dB. Increase avc reference level. Remove AN/USM- RC-41, connect it to input of CM- ut of AN/USM-44A to 1A9J28 on RT- el is again obtained, and record quencies is bandwidth at 6-dB point.

Figure 5-13. Preventive Maintenance Test, SA2 (Sheet 2 of 2)



NAVSHIPS 0967-872-5020

TM-03816B-35/2

AN/PRC-41A

MAINTENANCE

Figure 5-14. Preventive Maintenance Test, SA3 (Sheet 1 of 2)

SA3			
STEP NO	ACTION REQUIRED	READ INDICATION ON	REFERENCE STANDARD
SA3 (Cont)	to 243.0 MHz unmodulated. Increase level of approximately +3 volts is indi output frequency of AN/USM-44A un output connection from 1A9J28 on fr and record frequency at this point. Re 695A/PRC-41, decrease output frequ this point. Repeat this procedure at 6	e signal output from AN/USM-44A until cated on TS-505/U. Increase AN/USM til TS-505/U again indicates avc refere ont panel of RT-695A/ PRC-41, conne econnect output of AN/USM-44A to an uency until avc reference level is again 60-dB level.	a guard receiver avc reference M-44A signal output 6 dB. Increase ence level. Remove AN/USM-44A ct it to input of CM-102/USM-73, tenna connector 1A9J28 on RT- obtained, and record frequency at
		CAUTION	
	Be careful not to key RT-695A/F Fuse MX-1730/U provides prote transmitter output power to sign	PRC-41 with AN/USM-44A connected ction in event RT-695A/PRC-41 is accurate al generator output will result in attenu	to antenna connector 1A9J28. sidentally keyed. Application of ator damage.



PROCEDURE: Connect Signal Generator AN/USM-44A through a 6-dB pad and a length of RG-213/U coaxial cable to antenna connector 1A9J28 on RT-695A/PRC-41. Connect junction box (figures 5-1 and 5-2) to KY-38 connector 1A9J14 on RT-695A/PRC-41. Connect ME-2/U to NORMAL AUDIO OUTPUT jack on junction box. Connect ME-6C/U to MAIN RCVR AUX AUDIO jack on junction box. Set OFF-T/R-T/R/G-DIAL LIGHT switch to T/R, and apply power to test equipment. Allow 10 minutes warmup. Set AN/USM-44A to 231.2 MHz modulated 30 percent with 1000 Hz at 1000 microvolts. Note normal audio power output indication on ME-2/U and MAIN RCVR AUX AUDIO output indication on ME-6C/U with RT-695A/PRC-41 tuned to 231.2 MHz. Vary signal input level to RT-695A/PRC-41 from 10 to 100,000 microvolts, and record the normal audio power output and MAIN RCVR AUX AUDIO output changes in decibels from 1000-microvolt reference. Increase signal input to 0.5 volt, and check to see that blocking does not occur. Repeat these measurements at 304.7 and 399.9 MHz.

Figure 5-15. Preventive Maintenance Test, SA4 and SA5 (Sheet 1 of 2) **ORIGINAL** 5-19

STEPS S	SA4 AND SA5			
STEP NO	ACTION REQUIRED	READ INDICATION ON	REFERENCE STANDARD	
SA5	Record main receiver audio power output and auxiliary audio output.	Output Power Meter ME-2/U, Output Meter ME-6C/U.	Normal audio output shall be at least 50 milliwatts, MAIN RCVR AUX audio output shall be at least 0.25 volt.	
0.25 volt. PROCEDURE: Connect Signal Generator AN/USM-44A through a 6-dB pad and length of RG-213/U coaxial cable to antenna connector 1A9J28 on RT-695A/PRC-41. Connect junction box (figures 5-1 and 5-2) to KY-38 connector 1A9J14 on RT-695A/PRC-41. Connect ME-2/U to NORMAL AUDIO AUDIO OUTPUT jack on junction box. Connect ME-6C/U to MAIN RCVR AUX AUDIO output jack on junction box. Set OFF-T/R-T/R/G- DIAL LIGHT switch to T/R, and apply power to test equipment. Allow 10 minutes warmup. Set AN/USM-44A to 304.7 MHz modulated 30 percent with 1000 Hz at 1000 microvolts. Record normal audio power output and MAIN RCVR AUX AUDIO output.				

Figure 5-15. Preventive Maintenance Test, SA4 and SA5 (Sheet 2 of 2)



Figure 5-16. Preventive Maintenance Test, SA6 and SA7 (Sheet 1 of 2)

STEPS S	SA6 AND SA7				
OPERAT	OPERATING CONDITIONS AND CONTROL SETTINGS:				
OFF-T/R SQUELC VOL: Ma S1 and S	-T/R/G-DIAL LIGI H: Maximum clo aximum clockwise 2 switches (junct	HT: OFF. ckwise. ion box): Open positi	ion.		
STEP NO	ACTION REQU	IRED	READ INDICATION ON	REFERENCE STANDARD	
SA6	Record guard re level auxiliary gu characteristics.	eceiver and low- uard audio avc	Output Power Meter ME-2/U, Output Meter ME-6C/U.	Normal audio power output and LOW LEVEL AUX GUARD AUDIO shall not vary more than ±3 dB from 1000-microvolt reference over 10- to 100,000-microvolt range and shall not block signals less than 0.5-volt input for all frequencies.	
	PROCEDURE: Connect Signal Generator AN/USM-44A through a 6-dB pad and a length of RG-213/U coaxia cable to antenna connector 1A9J28 on RT-695A/PRC-41. Connect junction box (figures 5-1 and 5-2) to KY-38 connector 1A9J14 on RT-695A/PRC-41. Connect ME-2/U to NORMAL AUDIO OUTPUT jack on junction box. Connect ME-6C/U to LOW LEVEL AUX GUARD OUTPUT jack on junction box. Set OFF-T/R-T/R/G-DIAL LIGHT switch to T/R/G. Apply power to test equipment, and allow 10 minutes warmup. Adjust AN/USM-44A to 243.0 MHz modulated 30 percent with 1000 Hz at 1000 microvolts. Note audio power output indication on ME-2/U and LOW LEVEL AUX GUARD OUTPUT indication on ME-6C/U with RT-695A/PRC-41 tuned to 231.2 MHz. Vary signal input level to RT-695A/PRC-41 from 10 to 100,000 microvolts, and record audio power output and LOW LEVEL AUX GUARD OUTPUT changes in decibels from 1000-microvolt reference. Increase signal input to 0.5 volt, and check to see that blocking does not occur.				
SA7	Record guard re output, and low- guard audio out	eceiver audio power level auxiliary put.	Output Power Meter ME-2/U, Output Meter ME-6C/U.	Normal audio output shall be at least 50 milliwatts. LOW LEVEL AUX GUARD AUDIO OUTPUT shall be at least 0.15 volt.	
	PROCEDURE: connector 1A9J 41 Connect ME- AUX GUARD O test equipment, 1000 Hz at 1000	Connect AN/USM-44 28 on RT-695A/PRC- 2/U to NORMAL AUI UTPUT jack on junct and allow 10 minutes microvolts. Record	4A through a 6-dB pad and length of R -41. Connect junction box to KY-38 co DIO OUTPUT jack on junction box. Co ion box. Set OFF-T/R-T/R/G-DIAL LIG s warmup. Set AN/USM-44A to 243.0 NORMAL AUDIO OUTPUT and LOW	G-213/U coaxial cable to antenna nnector 1A9J14 on RT-695A/PRC- onnect ME-6C/U to LOW LEVEL GHT switch to T/R/G. Apply power to MHz, modulated 30 percent with LEVEL AUX GUARD OUTPUT.	



Figure 5-17. Preventive Maintenance Test, SA8 and SA9 (Sheet 1 of 2)

				STEPS SA8 AND SA9
			1	
STEP NO	ACTION REQUIRED	READ INDICATION ON	RE	FERENCE STANDARD
SA8 (Cont)	PROCEDURE: Connect Signal Ger coaxial cable to antenna connector 1 external modulation input jack on AN connector 1A9J14 on RT-695A/PRC Connect ME-6C/U to MAIN RCVR A NORMAL AUDIO OUTPUT jack on j power to test equipment. Allow 10 m 44A to 225.0 MHz externally modula audio output waveform on AN/USM- AUX AUDIO output jack on junction AN/USM-105A for evidence of distor auxiliary audio output indication on N 30-percent modulation of AN/USM-4 from 1000-Hz reference as indicated maintaining 30-percent modulation of output in decibels from 1000-Hz reference MHz.	herator AN/USM-44A through a 6-dB p A9J28 on RT-695A/PRC-41. Connec I/USM-44A. Connect junction box (figu -41. Connect ME-2/U to NORMAL AL UX AUDIO output jack on junction box junction box. Set OFF-T/R-T/R/G-DIAI inutes warmup. Tune RT-695A/PRC-4 ted 30 percent with 1000 Hz at 1000 r 105A for evidence of distortion. Connec box. Observe main receiver auxiliary a tion. Note normal audio output indicat ME-6C/U. Vary frequency of TS-832B/U 4A, and measure the change of main I on ME-2/U. Vary the frequency of TS of AN/USM-44A, and measure the cha rence as indicated on ME-6C/U. Rep	vad and a t Audio C ures 5-1 a IDIO OU x. Conne L LIGHT 41 to 225 nicrovolts ect AN/U: audio out ion on M U from 30 receiver -382B/U nge of m eat proce	a length of RG-213/U Dscillator TS-382B/U to and 5-2) to KY-38 TPUT jack on junction box. ct AN/USM-105A to switch to T/R, and apply 5.0 MHz, and set AN/USM- s. Observe main receiver SM-105A to MAIN RCVR put waveform on E-2/U and main receiver 20 to 3500 Hz maintaining audio output in decibels from 300 Hz to 20 kHz ain receiver auxiliary audio edure for 304.7 and 399.9
SA9	Observe waveform, and record audio capability of guard audio output and low-level auxiliary guard output.	Oscilloscope AN/USM-105A, Output Power Meter ME2/U, Output Meter ME-6C/U.	Observe wave in should output a guard o than +1 reference	e waveform. For a sine- put, a sine-wave output be obtained. Normal audio and low-level auxiliary putput should not vary more , -3 dB from 1000-Hz ce.
	PROCEDURE: Connect Signal Ger coaxial cable to antenna connector 1 external modulation input jack on AN connector 1A9J14 on RT-695A/PRC Connect ME-6C/U to LOW-LEVEL A NORMAL AUDIO OUTPUT jack on j clockwise, and set OFF-T/R-T/R/G-D minutes warmup. Set RT-695A/PRC +5 MHz. Set AN/USM-44A to 243.0 Observe guard receiver audio output AN/USM-105A to LOW-LEVEL AUX audio output waveform on AN/USM- ME-2/U and low-level auxiliary guard to 3500 Hz maintaining 30-percent n output in decibels from 1000-Hz refe auxiliary guard audio output in decibel	herator AN/USM-44A through a 6-db p A9J28 on RT-695A/PRC-41. Connec I/USM-44A. Connect junction box (figu -41. Connect ME-2/U to NORMAL AU UX GUARD output jack on junction b junction box. Set SQUELCH control of DIAL LIGHT switch to T/R/G. Apply po -41 main receiver frequency controls to MHz, externally modulated 30 percent t waveform on AN/USM-105A for evid GUARD output jack on junction box. 105A for evidence of distortion. Note r d output indication on ME-6C/U. Vary f nodulation of AN/ USM-44A. Measure rence as indicated on ME-2/U, and me els from 1000-Hz reference as indicated	ad and a t Audio C ures 5-1 a IDIO OU ox. Conn n RT-695 wer to te to any fre with 100 ence of d Observe normal au requency the chan easure th ed on ME	length of RG-213/U Descillator TS-382B/U to and 5-2) to KY-38 TPUT jack on junction box. ect AN/USM-105A to 5A/PRC-41 maximum st equipment and allow 10 equency other than 243.0 00 Hz at 1000 microvolts. listortion. Connect low-level auxiliary guard udio output indication on of TS-382B/U from 300 ige of guard receiver audio he change of low-level E-6C/U.



Figure 5-18. Preventive Maintenance Test, SA10

AN/PRC-41A	NAVSHIPS 0967-872-5020	Figure
MAINTENANCE	TM-03816B-35/2	5-19
	I M11-5820-510-35-1	



STEP SA1	1		
STEP NO	ACTION REQUIRED	READ INDICATION ON	REFERENCE STANDARD
SA11 (Cont)	connect Electronic Multimeter T3 695A/PRC-41 to 304.7 MHz, and equipment, and allow 10 minute volt on TS-505/U. Close switche on AN/USM-105A for evidence of reference voltage of detected tra voltage as indicated on TS-505/ reference mic input closed circuit transmitter output in decibels fro	S-505/U to MIC INPUT (OPEN CIRC d set OFF-T/R-T/R/G-DIAL LIGHT sv s warmup. Set TS-382B/U for 1000 H s S1 and S2 on junction box. Observ of distortion. Connect output of HP-42 unsmitter output as indicated on ME-6 U. Vary frequency of TS-382B/U from it voltage as indicated on TS- 505/U. m 1000-Hz reference as indicated or	UIT) jack on junction box. Tune RT- vitch to T/R. Apply power to test Hz at a signal output level to indicate 0.5 re detected transmitter output waveform 20B to ME-6C/U. Note audio output 6C/U and mic input closed circuit in 300 Hz to 20 kHz maintaining Measure the change of detected in ME-6C/U.





Figure 5-20. Preventive Maintenance Test, SA12 (Sheet 1 of 2)

Change 1 5-26
				SA12			
OPERATIN	OPERATING CONDITIONS:						
Connect Po	wer Supply PP-3700/PRC-41 into tes	t setup illustrated above.					
STEP NO	ACTION REQUIRED	READ INDICATION ON	REFERENCE STAN	IDARD			
SA12	Measure 26.5-volt dc primary output under load conditions and ac ripple.	Multimeter AN/PSM-4A.	Power Supply PP-3700/ primary dc output voltag 26.5 volts dc across an load. Ac ripple shall not 10 millivolts.	PRC-41 e shall be 8-ohm exceed			
	PROCEDURE: Set Power Supply I Receiver-Transmitter RT-695A/PRC chassis from dust cover. Connect F connector J1 and 115-volt, 50- to 40 connect between P1 and P3. Conn to P3. Connect output meter ME-60 If measured output voltage as indica greater than 24 volts dc, adjust varia	PP-3700/PRC-41 on test bench with s C-41 upward. Loosen two captive scre Power Electric Cable Assembly CX-86 00-Hz primary power source. Adjust p ect AN/PSM-4A to PP-3700/PRC-41 C/U across power resistor 240C. Conr ated on AN/PSM-4C is not 26.5 volts able resistor R6 on PP-3700/PRC-41	side which clamps to rear ews in top plate, and remo 687/PRC-41 between dus ower resistor 240C for 8 c , positive lead to P1, nega nect a jumper between P3 dc but is less than 30 volt to 26.5-volt dc output.	of Radio ive main t cover ohms, and tive lead and P2. s dc and			

Figure 5-20. Preventive Maintenance Test, SA12 (Sheet 2 of 2)

#### 5-3. TUNING AND ADJUSTMENT

a. GENERAL. - The following paragraphs contain alignment and adjustment procedures necessary to maintain optimum equipment performance. When performing overall alignment of the equipment, procedures must be performed in the order in which they occur in the following paragraphs. Before attempting alignment, read the complete alignment procedure carefully to become familiar with the steps involved. Do not attempt alignment of equipment as a substitute for trouble isolation.

#### CAUTION

Retracking should be performed only when it has been established that misalignment (mistracking) is the cause of abnormal operation.

(1) TEST EQUIPMENT. - Each align mentor adjustment procedure lists the test equipment required to perform that procedure.

(2) SPECIAL TOOLS AND CABLES. - The special tools supplied with Radio Set Accessory Kit MK-706/PRC-41 are listed in table 1-5 and illustrated in figure 1-13. The special cables supplied with MK-706/PRC-41 are listed in table 1-4 and illustrated in figure 1-12. These special tools and cables are required for the performance of alignment and adjustment procedures.

(3) PRELIMINARY TEST SETUP. - Refer to figure 5-21. Remove Receiver-Transmitter Case CY3884/PRC-41 from Radio Receiver-Transmitter RT-695A/PRC-41, and connect the dc maintenance adapter to the rear of RT-694A/PRC-41. Connect Power Electric Cable Assembly CX-8687/PRC-41 between Power Supply PP-3700/PRC-41 and 115-volt 50- to 400-Hz primary source. Connect Power Electric Cable Assembly CX-8686/PRC-41 between PP-3700/PRC-41 and the dc maintenance adapter. Connect junction box (figures 5-1 and 5-2) to KY-38 connector 1A9J14 on RT-695A/PRC-41. Connect Output Power Meter ME-2/U to normal audio output jack on junction box to provide required load for audio output circuits. Before performing alignment procedures, turn on equipment, and adjust frequency selector knobs for 225.0 MHz. (For removal of the 1st and 2nd if amplifier module, set the frequency selector knobs to 229.9 MHz.) Set OFF-T/R-T/R/G-DIAL LIGHT switch to OFF, and remove spectrum generator, 1st and 2nd if amplifier, and rf and power amplifier modules according to procedures of paragraph 5-4b. Check position of the couplers on main chassis to see that they are in the relative positions shown in figure 5-22. If the main chassis couplers are properly



Figure 5-21. Radio Receiver-Transmitter RT-695A/PRC-41, Standard Test Bench Setup positioned, replace these modules according to module replacement procedures of paragraph 5-4g.

- b. ALIGNMENT OF HIGH- AND LOW-FREQUENCY OSCILLATORS OF 1ST AND 2ND IF AMPLIFIER MODULE.
  - (1) TEST EQUIPMENT REQUIRED.
    - (a) Electronic Multimeter TS-505/U.
    - (b) Output Power Meter ME-2/U.
  - (2) ALIGNMENT SETUP.
    - (a) Perform preliminary test setup procedure of paragraph 5-3a(3).
    - (b) Set frequency selector knobs on RT-695A/PRC-41 to 329.0 MHz.
    - (c) Connect TS- 505/U to test jack 1A2J1.
    - (d) Set OFF-T/R-T/R/G-DIAL LIGHT switch to T/R, and allow 10 minutes warmup.
  - (3) PROCEDURE.

(a) Adjust the 26.1-high-frequency oscillator slug (located on top of 1 stand 2nd if amplifier module) for maximum indication on TS- 505/U approximately -0.7 volt dc).

(b) Change frequency selector knobs on RT-695A/PRC-41 to 328.0 MHz, and adjust 25.1-high-frequency oscillator slug as described for adjustment of 26.1-high-frequency oscillator slug in (a).

- (c) Adjust 24.1-high-frequency oscillator slug with frequency selector knobs set to 327.0 MHz as in (a).
- (d) Adjust 23.1-high-frequency oscillator slug with frequency selector knobs set to 326.0 MHz as in (a).
- (e) Adjust the 22.1-high-frequency oscillator slug with frequency selector knobs set to 325.0 MHz as in
- (a).
- (f) Adjust 21.1-high-frequency oscillator slug with frequency selector knobs set to 324.0 MHz as in (a).
- (g) Adjust 20.1-high-frequency oscillator slug with frequency selector knobs set to 323.0 MHz as in (a).
- (h) Adjust 19.1-high-frequency oscillator slug with frequency selector knobs set to 322.0 MHz as in (a).
- (i) Adjust 18.1-high-frequency oscillator slug with frequency selector knobs set to 321.0 MHz as in (a).
- (j) Adjust 17.1-high-frequency oscillator slug with frequency selector knobs set to 320.0 MHz as in (a).

(k) Switch back through range of 1-MHz frequency selector knob to 329.0 MHz, and observe that T S-505/U indicates approximately - 0.25 to - 1.5 volts on each 1-MHz position between 329.0 and 320.0 MHz.

(I) Connect TS- 505/U to test jack 1A2J2, and measure approximately 2.5 volts for each setting of 10th-MHz frequency selector knob. This indicates that low-frequency oscillator is functioning properly. If indication is not obtained, refer to applicable troubleshooting procedures of section 4.

c. ALIGNMENT OF 1ST AND 2ND IF AMPLIFIER STAGES OF 1ST AND 2ND IF AMPLIFIER MODULE.

- (1) TEST EQUIPMENT REQUIRED.
  - (a) Electronic Multimeter TS- 505/U.
  - (b) Signal Generator AN/USM-44A.
  - (c) Oscilloscope AN/USM-105A.
  - (d) Output Power Meter ME-2/U.
- (2) ALIGNMENT SETUP.
  - (a) Perform preliminary test setup procedure of paragraph 5-3a(3).

(b) Connect AN/USM-105A to normal audio output jack on junction box as an aid in tuning signal generator to desired frequency and as monitor during alignment process.

- (c) Set frequency selector knobs on Radio Receiver-Transmitter RT-695A/PRC-41 to 329.0 MHz.
- (d) Connect TS-505/U to avc test jack 1A3J1.



Figure 5-22. Main Chassis Couplers in the 225-MHz Position

(3) PROCEDURE.

(a) Connect AN/USM-44A to input of if module by removing plug 1A9P2 from jack 1A9J17, connecting BNC-to-Conhex adapter A1 to 1A9P2, and connecting AN/USM-44A to BNC-to-Conhexadap-ter. Set AN/USM-44A for 29.0 MHz modulated 30 percent at 1000 Hz.

(b) Increase output level of AN/USM-44A until signal can be tuned in by varying AN/USM-44A frequency slightly around 29.0 MHz. When AN/ USM-44A has been tuned to exactly 29.0 MHz, as indicated by maximum indication on TS505/U, maintain AN/USM-44A output at a minimum level to provide a usable signal on Oscilloscope AN/USM-105A.

- (c) Set OFF-T/R-T/R/G-DIAL LIGHT switch to T/R. Allow 10 minutes warmup.
- (d) Adjust trimmer capacitors C2, C5, C8, C12, C15, and C20 for maximum indication on TS- 505/U.

(e) Set frequency selector knobs to 321.0 MHz. Repeat step (b), except set AN/USM-44A to 21.0 MHz, and adjust inductor slugs L1, L2, L3, L4, L5, and L6 for maximum indication on the TS-505/U.

- (f) Repeat steps (d) and (e) until no further improvement is obtained.
- (g) Set frequency selector knobs on RT-695A/PRC-51 to 329.9 MHz.
- (h) Set AN/USM-44A for 29.9 MHz modulated 30 percent with 1000 Hz.
- (i) Repeat step (b), except vary AN/ USM-44A frequency slightly about 29.9 MHz.
- (j) Adjust inductors L20, L22, and L24 for maximum indication on TS- 505/U.
- (k) Set frequency selector knobs on RT-695A/PRC-51 to 329.0 MHz.
- (I) Set AN/USM-44A for 29.0 MHz mod-ulated 30 percent with 1000 Hz.
- (m) Repeat step (b).
- (n) Adjust inductors L19, L21, and L23 for maximum indication on TS-505/U.

(o) Repeat steps (g) through (n) above until no further improvement is observed. A good signal should be received with 10-microvolt or less output from AN/USM-44A.

(p) Remove AN/USM-44A, and reconnect plug 1A9P2, to jack 1A9J17.

(q) Remove modulator module from main chassis according to removal procedure, paragraph 5-4b, in order to remove B+ voltage from final stages. This prevents damage to final stages if they are not properly tuned.

# Paragraph 5-3c(3)(r)

### NAVSHIPS 0967-872-5020 TM-03816B-35/2 TM11-5820-510-35-1

# MHz.

(r) Connect TS-505/U to test jack 1A2J3. Set frequency selector knobs on RT-695A/PRC-51 to 329.9

(s) Close switch S1 on junction box and adjust capacitors 1A2C33 and 1A2C20 for maximum negative indication on TS- 505/U.

(t) Open switch S1 on junction box and replace modulator module according to replacement procedure, paragraph 5-4g.

d. ALIGNMENT OF SPECTRUM GENERATOR MODULE.

- (1) TEST EQUIPMENT REQUIRED.
  - (a) Electronic Multimeter TS-505/U.
  - (b) Frequency Counter AN/USM-122.
  - (c) Electronic Frequency Converter CV-394/USA- 5.
  - (d) Transfer Oscillator MC- 102/USM-73.
  - (e) Output Power Meter ME-2/U. (2) ALIGNMENT SETUP.
- (2) ALÍGNMENT SETUP.
  - (a) Perform preliminary test setup procedure of paragraph 5-3a(3).
  - (b) Set frequency selector knobs on Radio Receiver-Transmitter RT-695A/PRC-41 to 399.9 MHz.

(c) Remove red wire from collector lead of transistor 2A9Q1 on rear plate of main chassis, remove 12 cross-recessed screws attaching rear main chassis plate, and dress this plate clear of rear of main chassis.

(d) Loosen four captive screws on spectrum generator module, and remove module from main chassis. Be careful not to turn drive coupler on module. Remove module end cover having access holes for trimmer capacitors C10, C15, and C20 by removing eight cover screws. Replace spectrum generator module on main chassis according to replacement procedure, paragraph 5-4g.

(e) Remove plug 1A9P3 from jack 1A9J15 on transmit-receive coaxial switch. Connect CM-102/USM-73 to plug 1A9P3 using Conhex-to-BNC coaxial connector adapter A1.

(f) Set CV-394/USA- 5 in AN/USM- 122. Connect frequency meter output of CM-102/USM-73 to input jacks of CV-394/USA-5.

(g) Connect TS-505/U to test jack 1A5J1 on spectrum generator module.

(h) Set OFF-T/R-T/R/G-DIAL LIGHT switch to T/R, and allow 10 minutes warmup.

(3) PROCEDURE

(a) Adjust oscillator tuning slug for maximum indication on TS-505/U. (Oscillator tuning slug is accessible through top of spectrum generator module.) Check spectrum generator output frequency with AN/USM-122. Readjust oscillator tuning slug to produce an output frequency 2.5 to 5.0 kHz lower than nominal spectrum generator output frequency as listed in figure 4-11.

(b) Repeat step (a) for each 10-MHz increment through 229.9 MHz.

(c) Connect multimeter to test jack 1A5J2 on spectrum generator module, and rotate 10-MHz frequency selector knob throughout its range. Observe indication on TS-505/U to determine that a negative indication is obtained at all positions of frequency selector knob.

(d) Disconnect plug 1A9P3 from CM-102/USM-73, and reconnect it to jack 1A9J15.

(e) Adjust trimmer capacitors C 10, C15, and C20 for maximum negative indication on TS- 505/U.

(f) Set frequency selector knobs to 389.9 MHz. Refer to figure 5-23. Bend rotor blade segment which has just come into reach with stator on tuned circuits Z 1, Z2, and Z3 to obtain maximum indication on TS-505/U.

Note

All adjustment tabs are coded with a dot of paint as specified in the applicable tuning capacitor rotor tab illustration. The tab to be adjusted is identified further as the one partially meshed.





BOTTOM ROTOR BLADE

Figure 5-23. Spectrum Generator Module, Tuning Capacitor Rotor Tab Z1, Z2, Z3, Adjustment Diagram



Figure 5-24. Spectrum Generator Module, Rotor-Stator Coincidence at 225 MHz CAUTION

Never bend tabs out beyond a 30-degree angle. Excessive inward bending will cause tab to short against the stator. If a tuned circuit is far from resonance, it may be caused by circuit discontinuity. If this occurs, check inductance arm contacts and inductance rings for proper contact.

(g) Set frequency selector knobs to 379.9 MHz. Bend rotor segment which has come into mesh with stator on tuned circuits Z1, Z2, and Z3 to obtain maximum indication on TS-505/U.

- (h) Repeat step (g) above for each 10-MHz increment to 229.9 MHz inclusive. Refer to figure 5-24.
- (i) Repeat steps (e) through (h) until no further improvement is obtained.
- (j) Repeat steps (a) and (b).

Note

The configuration of the tabs after adjustment should be a smooth contour with no sharp discontinuities.

Rechannel the equipment after each tab is bent to remove any backlash of the gears.

(k) Remove spectrum generator module, replace end cover, and reinstall module on main chassis according to replacement procedure, paragraph 5-4g. Position rear plate in place on main chassis, and secure with 12 cross-recessed screws. Resolder red wire to transistor 1A9Q2 collector lead.

e. ALIGNMENT OF RF AND POWER AMPLIFIER MODULE.

- (1) TEST EQUIPMENT REQUIRED.
  - (a) Electronic Multimeter TS-505/U.
  - (b) RF Wattmeter AN/URM-43A.
  - (c) Output Power Meter ME-2/U. (2) ALIGNMENT SETUP.
  - (a) Perform preliminary test setup procedure of paragraph 5-3a(3).
- (b) Connect AN/URM-43A to antenna connector 1A9J28 on front panel of Radio Receiver-Transmitter RT-695A/PRC-41.
  - (c) Connect TS-505/U to V5 GRID test jack 1A8J8 on rf and power amplifier module 1A8.
- (d) Connect junction box (figures 5-1 and 5-2) to K4-38 connector 1A9J14 on front panel of RT-695A/PRC-51.
  - (e) Loosen two captive screws on guard receiver module 1A7, and remove module from main chassis.

(f) Loosen four captive screws on dc power supply module 1A1, and remove module from main chassis. Connect this module to main chassis using extension cable W5.





(g) Set OFF-T/R-T/R/G-DIAL LIGHT switch to T/R, and allow 10 minutes warmup. Set frequency selector knobs for 399.9 MHz.

(3) PROCEDURE.

- (a) Close switch S1 on junction box.
- (b) Adjust trimmer capacitors C7, C12 and C18 for a maximum negative indication on TS- 505/U.
- (c) Adjust trimmer capacitors C21 and C27 for a maximum indication on AN/URM-43C.
- (d) Set frequency selector knobs for 395.0 MHz.

(e) Bend rotor blade segment which has just come into mesh with stator on tuned circuits Z1, Z2, and Z3 for a maximum negative indication on TS-505/U (figure 5-25).

(f) Bend rotor blade segment which has just come into mesh with stator on tuned circuits Z4 and Z5 for a maximum indication on AN/URM-43C (figures 5-26 and 5-27).

(g) Repeat rotor tab bending on Z1 through Z5 for each 10-MHz increment of frequency selector knobs to 225.0 MHz.

Note

All adjustment tabs are coded with a dot of paint as specified in the applicable tuning capacitor rotor tab illustration. The tab to be adjusted is identified further as the one partially meshed, and the sequence will alternate from one side of stator to the other.



Figure 5-26. RF and Power Amplifier Module, Tuning Capacitor Rotor Tab Z1, Z4, and Z5, Adjustment Diagram



Figure 5-27. RF and Power Amplifier Module, Rotor-Stator Coincidence at 225 MHz CAUTION

An insulated tool must be used for bending rotor tabs. Tuned circuits Z4 and Z5 are at 180 volts dc. Never bend tabs out beyond a 30-degree angle. Excessive inward bending will cause tab to short against the stator. If a tuned circuit is far from resonance, it may be caused by a circuit discontinuity. If this occurs, check inductance arm contacts and inductance rings for proper contact. Occasionally open switch S1 on junction box during procedure to prevent exceeding normal operating temperature of RT-695A/PRC-41.

(h) Repeat steps (b) through (g) until no further improvement is obtained.

Note

The configuration of the tabs after adjustment should be a smooth contour with no sharp discontinuities. Rechannel the equipment after each tab is bent to remove any backlash of the gears.

(i) Rotate 10-MHz frequency control knob throughout its range. Observe transmitter rf power output indication on AN/URM-43C. If satis-factory rf power output curve is obtained, replace guard receiver module 1A7 and dc power supply module 1A1 on main chassis. If satisfactory rf power output curve is not obtained, proceed to step (j).

(j) Remove rf and power amplifier module end cover having access holes for capacitors C7, C12, C18, C21, and C27 by removing eight cover screws.

(k) Adjust capacitors C27 and C30 alternately until optimum setting of each is determined to obtain maximum indication on AN/URM-43C over frequency range of 225.0 to 399.9 MHz.

(I) Replace end cover on rf and power amplifier module.

(m) Replace guard receiver module 1A7 and dc power supply module 1A1 on main chassis.

f. ALIGNMENT OF GUARD RECEIVER MODULE.

(1) TEST EQUIPMENT REQUIRED.

(a) Electronic Multimeter TS-505/U.

(b) Signal Generator AN/USM-44A.

(c) Oscilloscope AN/USM-105A.

(d) Output Power Meter ME-2/U.

(2) ALIGNMENT SETUP.

(a) Perform preliminary test setup procedure of paragraph 5-3a(3).

Paragraph 5-3f(2)(b)

# NAVSHIPS 0967-872-5020 TM-03816B-35/2

# TM11-5820-510-35-1

(b) Connect AN/USM-105A to NORMAL AUDIO OUTPUT jack on junction box.

(c) Loosen two captive screws on guard receiver module 1A7, and remove module from main chassis. Remove side cover from guard receiver module exposing transistors Q1 through Q5 and associated circuits.

(d) Connect AN/USM-44A through a 10-pF capacitor to junction point of base number 1 of transistor 1A7Q5, resistors R16 and R17, and capacitor C67 all located on TB2. Refer to figures 5-46, 5-47, and 5-48. Set AN/USM-44A to 20.55 MHz, modulated 30 percent with 1000 Hz.

(e) Connect guard receiver module to main chassis using extension cable W3.

(f) Connect TS-505/U to test jack 1A7J2 on guard receiver module.

(g) Set OFF-T/R-T/R/G-DIAL LIGHT switch to T/R/G, and allow 10 minutes warmup. Set SQUELCH control on front panel full clockwise.

(h) Turn SQUELCH resistor 1A7R46 in guard receiver full clockwise.

(3) PROCEDURE.

(a) Increase output of AN/USM-44A un-til signal appears on AN/USM-105A.

(b) Adjust inductors L15, L14, L13, L12, L16, and L11 for maximum indication on TS-505/U. Repeat sequence until no further improvement is obtained.

Note

Maintain output of AN/USM-44A at minimum level which gives a usable indication on Oscilloscope AN/USM- 105A throughout the tuning process.

(c) Adjust AN/USM-44A for 20.57 MHz, and readjust L16 and L17 for maximum indication on TS-505/U.

(d) Disconnect AN/USM-44A and 10-pF capacitor from base number 1 of transistor 1A7Q5. Replace side cover on guard receiver module, and replace module on main chassis.

(e) Remove TS-505/U from test jack 1A7J2, and connect to test jack 1A7J1.

(f) Adjust inductor L10 for maximum negative indication on TS-505/U (approximately -2.0 volts). With inductor L10 not tuned, indication will be zero.

(g) Connect AN/USM-44A to antenna connector 1A9J28 on front panel of RT-695A/PRC-41.

(h) Connect TS-505/U to avc test jack 1A7J2 on guard receiver module.

(i) Tune AN/USM-44A to 243.0 MHz, modulated 30 percent with 1000 Hz by adjusting output frequency about 243.0 MHz until maximum indication is obtained on TS-505/U.

Adjust capacitor C22 and inductor L6 for maximum indication on TS- 505/U.

(k) Adjust capacitors C16, C12, C6, and C1 for maximum indication on TS-505/U.

(I) Repeat steps (i) and (k) until no further improvement is obtained.

(m) Remove modulation from AN/USM44A, and readjust capacitor C1 for minimum indication on ME-2/U.

g. FINAL ADJUSTMENTS.

(1) GENERAL. - After the foregoing alignment procedures have been completed, it is necessary to make the following final adjustments.

(a) 3RD IF AND SQUELCH MODULE. Make final adjustments to the 3rd if and squelch module as follows:

1. Set frequency selector knobs on RT-695A/PRC-41 to 304.7 MHz.

2. Connect Electronic Multimeter TS 505/U to 1A3J1 on 3rd IF and squelch module 1A3.

3. Connect Output Power Meter ME 2/U and Oscilloscope AN/USM- 105A to NORMAL AUDIO OUTPUT jack on junction box. Set ME-2/U to provide 300-ohm load for audio module.

4. Connect Signal Generator AN/ USM-44A to antenna connector 1A9J28 on front panel of RT-695A/PRC-41, and tune AN/USM-44A to 304.7 MHz by varying output frequency about 304.7 MHz until maximum indication is observed on TS-505/U. Increase AN/USM-44A output to 30 microvolts modulated 30 percent with 1000 Hz.

5. Disconnect TS- 505/U from 1A3J1, and connect to 1A8J11 on rf and power amplifier module 1A8. Adjust potentiometer 1A3R1 until voltage begins to rise (approximately 0.1 volt dc) as indicated on TS-505/U (low dc voltage scale).

6. Increase AN/USM-44A output level to 1000 microvolts, and adjust modulation to 55 percent at 1000 Hz.

7. Adjust potentiometer 1A3R26 until signal begins to clip as indicated on AN/USM105A.

8. Decrease AN/USM-44A modulation to 30 percent at 1000 Hz.

9. Adjust potentiometer 1A4R20 for 50 milliwatts as indicated on ME-2/U.

10. Decrease AN/USM-44A output level to 20 microvolts. Adjust SQUELCH control 1A9R5 fully clockwise. Adjust 1A3R46 until squelch breaks (audio signal appears on AN/USM- 105A).

(b) MODULATOR MODULE. - Make final adjustments to the modulator module as follows:

1. Connect AN/URM-43C to antenna connector 1A9J28 on front panel of RT-695A/ PRC-41.

2. Connect AN/USM-105A to test jack 1A6J2 on modulator module.

3. Connect TS-382B/U to MIC INPUT jack on junction box. Set TS-382B/U for 1000 Hz.

4. Connect TS-505/U to MIC INPUT (open circuit) jack on junction box. Adjust output of TS-382B/U to indicate 0.7 volt on TS-505/U.

5. Close switches S1 and S2 on junction box.

<u>6</u>. Adjust potentiometer 1A6R2 until signal on AN/USM-105A shows evidence of clipping.

7. Open switches S1 and S2 on junction box.

(c) GUARD RECEIVER MODULE. - Make final adjustments to the guard receiver module as follows:

<u>1</u>. Connect TS-505/U to test jack 1A7J2 on guard receiver module. Connect AN/USM-105A to normal audio output jack on junction box.

<u>2</u>. Connect AN/USM-44A to antenna connector 1A9J28 on front panel of RT-695A/PRC-41. Set OFF-T/R-T/R/G-DIAL LIGHT switch to T/R/G.

<u>3</u>. Tune AN/USM-44A to 243.0 MHz by varying output frequency about 243.0-MHz until maximum indication is observed on TS-505/U.

<u>4</u>. Adjust squelch resistor 1A7R46 in guard receiver module counterclockwise until guard receiver module is squelched off, and signal disappears on AN/USM-105A with AN/USM-44A set at 2-uV input.

<u>5</u>. Set AN/USM-44A to 243.0 MHz, 1000 microvolts, modulated 30 percent, and adjust 1A4R19 on audio amplifier module to indicate 50 milliwatts on ME-2/U.

<u>6</u>. Set frequency controls main receiver to 304.7 MHz, and set AN/USM-44A to 304.7 MHz, modulated 30 percent, with 1000-microvolt output, and readjust 1A4R20 on audio amplifier module to indicate 50 milliwatts on ME-2/U.

7. Repeat 5-3g(1)(c)5 and 5-3g(1) (c)6 until 50 milliwatts is obtained on both guard receiver and main

#### receiver. 5-4. REPAIR.

a. REMOVAL, REPAIR, AND REPLACEMENT OF PARTS, MODULES, AND UNITS. - This subsection contains removal, disassembly, repair, reassembly, and replacement information for modules and assemblies of Radio Receiver-Transmitter RT-695A/PRC-41. Remove Receiver-Transmitter Case CY-3884/PRC-41 from RT-695A/PRC-41 by loosening four captive screws at the rear of CY-3884/PRC-41. Removal of the CY-3884/PRC-41 case permits access to the modules of RT-695A/PRC-41. Repair procedures involve the isolation of a defective detail part by use of troubleshooting procedures of section 4, and removal and replacement of the defective detail part. Refer to paragraph 5-4c for disassembly procedures of the mechanical assemblies. Refer to paragraph 5-4e for reassembly procedures. All adjustments of this equipment are covered under paragraph 5-3.

b. MODULE REMOVAL. - All modules of RT-695A/PRC-41 may be removed by loosening the captive screws and lifting the module upward. Set frequency selector knobs on front panel of RT-695A/PRC-41 to 229.9 MHz before removing the 1st and 2nd if amplifier module. Set frequency selector knobs to 225.0 MHz before removing either spectrum generator or rf and power amplifier modules.

c. DISASSEMBLY. - A mechanical assembly must not be disassembled unless it has been determined to be necessary to correct a malfunction. Disassembly procedures must not be performed as a matter of exploratory trouble isolation. In most cases, trouble can be corrected by adjustment procedures and application of operational theory. If trouble is known to exist in a part or a small assembly of parts not readily accessible, use the disassembly instructions to provide a guide for complete disassembly. These instructions are not to infer that a module should be regularly disassembled, since disassembly will disturb factory alignment. Use disassembly procedures only to extent necessary to effect required overhaul. Disassembly is limited to removal of mechanical items such as gears, bearings, shafts, etc. Internal wiring, resistors, capacitors, or other electrical items are not removed unless such action is necessary for access to a part requiring overhaul. The module disassembly procedures make reference to exploded views which use reference designations for identifying detail parts and assemblies. While performing disassembly procedures, be careful to avoid excessive strain on internal connections and to prevent distortion of shafts and rotor assemblies. A wiring diagram should be drawn as an aid to reassembly before removal of any part requiring unsoldering of several wires. This diagram should show any color coding or markings of wires, and approximate location and identity of terminals to which they are connected. In cases where lacing is to be removed or where lead dress is important, this must also be shown.

#### CAUTION

Before removing matching or meshing parts, use a scribe or otherwise mark the relative positions of such parts to ensure correct reassembly.

(1) 1ST AND 2ND IF AMPLIFIER MODULE.-Remove 1st and 2nd if amplifier module from main chassis of RT-695A/PRC-41 according to procedures of paragraph 5-4b. To disassemble 1st and 2nd if amplifier module (figures 5-28 and 5-75), perform the following:

(a) Z1, Z2, AND Z3 SLUG RACK. - To disassemble the Z1, Z2, and Z3 slug rack and gain access to the amplifier subassembly, refer to figure 5-75, and perform the following:

<u>1</u>. Remove thirteen 2-56x3/16-inch self-locking screws, and remove amplifier covers MP16 and MP17 from electrical equipment chassis MP39.

<u>2</u>. From bottom of right front corner of 1st and 2nd if amplifier module, remove a 2-56 x 1/8-inch screw and 2-56 nut, and remove captive screw H2.

3. Remove spring pin from cams MP12 and MP13. Remove shouldered shaft MP14,

#### AN/PRC-41A MAINTENANCE



Figure 5-28. 1st and 2nd IF Amplifier Module, Switch and Cam Positioning Diagram

cams MP12 and MP13, and two flat washers H3. Remove Z1, Z2, and Z3 slug rack MP22.

<u>4</u>. Remove three 4-40x 1/8-inch and one 4-40 x 1/4-inch screws from top of electrical equipment chassis MP39. Unsolder electrical connections at points A (figure 5-40) on wired bracket, coaxial lead and ground connection on amplifier subassembly, and remove two 2-56 x 1/4-inch screws to free relay K3. For further disassembly of amplifier subassembly, see figure 5-78.

(b) Z4, Z5, AND Z6 SLUG RACK. - To disassemble Z4, Z5, and Z6 slug rack, see figure 5-75 and perform the following:

<u>1</u>. Remove thirteen 2-56x3/16-inch self-locking screws, and remove amplifier covers MP16 and MP17 from electrical equipment chassis MP39.

2. Remove spring pin from cams MP8 and MP9.

 $\overline{3}$ . Remove retaining ring from should red shaft MP11, two cams MP8 and MP9, and two flat washers H3. Remove Z4, Z5, and Z6 slug rack upward.

(c) OSCILLATOR AND SWITCHBOARD, AND RF COIL ASSEMBLIES. - To gain access to detail parts of either oscillator and switchboard assembly or rf coil assembly, remove amplifier cover MP16, remove retaining ring from shouldered shaft MP15, and remove shouldered shaft MP15 outward from electrical equipment chassis MP39. For further disassembly procedures, refer to figures 5-76 and 5-77. Refer to figure 5-75 for an exploded view of amplifier subassembly board number 1.

(2) RF AND POWER AMPLIFIER MODULE. - Remove rf and power amplifier module from main chassis of Radio Receiver-Transmitter RT-695A/PRC-41 according to paragraph 5-4b. To disassemble rf and power amplifier module, refer to figure 5-80, and perform the steps of either paragraph 5-4c(2)(a) or 5-4c(2)(b) as applicable.

(a) Z 1, Z2, AND Z3 TUNED CIRCUITS. - To disassemble Z1, Z2, and Z3 tuned circuits to gain access to a particular detail part, refer to figure 5-80, and perform the following:

<u>1</u>. Remove eight 2-56 x 1/8 inch cross-recessed screws, and remove side cover MP19. Remove plastic retaining ring MP1 and coupler MP5. Remove spring pin from shaft coupler MP4, and remove MP4 from gear and shaft MP22.

MP28.

2. Remove two 6-32 x 1/2-inch cross-recessed screws from gearplate MP28, and remove gearplate

<u>3</u>. Gears MP41, -MP21, MP22, and MP23 may be removed as required from bottom plate MP33. (Scribe gear and shaft MP22 and gears MP23 and MP41 for reassembly reference.)

<u>4</u>. Loosen two setscrews in shaft collar MP6(1), and slide MP6(1) and ground spring MP42(1) clear of retaining ring next to bottom plate bearing. Remove retaining ring.

<u>5</u>. Loosen two setscrews in shaft collars MP6(2), and MP6(3), and MP6(4). Position rotor of tuned circuit Z3 so that it is completely unmeshed with its stator, and slide straight shaft MP14 outward from bottom plate MP33 far enough to gain access to retaining ring which normally rides against bearing MP2(2) in top plate. Remove retaining ring. Remove straight shaft MP14, and remove rotors and shaft collars from tuned circuits Z1 through Z3.

(b) Z4 AND Z5 TUNED CIRCUITS. - To disassemble Z4 and Z5 tuned circuits to gain access to a particular detail part, refer to figure 5-80, and perform the following:

<u>1</u>. Remove eight 2-56 x 1/8-inch cross-recessed screws, and remove side cover MP16. Remove retaining ring MP1 and coupler MP5. Remove spring pin from shaft coupler MP4 and remove MP4 from gear and shaft MP22..

- MP28.
- 2. Remove two 6-32 x 1/2-inch cross-recessed screws from gearplate MP28, and remove gearplate

<u>3</u>. Remove spring pin from gear MP23, and remove gear MP23 from shaft MP32. (Scribe gear and shaft MP22 and gears MP23 and MP41 for assembly reference.)

<u>4</u>. Remove spring pin from gear MP41, and remove gear MP41. Remove gears MP21 and MP22.

 $\overline{5}$ . Remove four 2-57 x 3/8-inch self-locking screws at corners of bottom plate MP33 and one at location A. Remove six 2- 56 x 5/16 inch screws, nuts, and lockwashers from location B. Remove two 2- 56 x 3/8-inch screws securing plug P2; free plug P2. Remove two 2-56 x 3/16-inch screws, nuts, and lockwashers from connector P 1 and free connector P 1. Remove bottom plate MP33.

<u>6</u>. Loosen two setscrews in rotors MP29(1) and MP29(2). Loosen two setscrews in shaft collar MP6(5), and remove retaining ring (6). Remove capacitor shaft MP32.

(3) SPECTRUM GENERATOR MODULE. Remove spectrum generator module from main chassis of RT-695A/PRC-41 according to paragraph 5-4b. Disassemble spectrum generator module as required according to applicable paragraphs 5-4c(3)(a) through 5-4c(3)(c).

(a) CRYSTAL AND SWITCH SECTION S2. -To remove or gain access to a detail part in the crystal and switch S2 section of spectrum generator module, refer to figure 5-79, and perform the following:

<u>1</u>. Remove twenty-six 2-56 x 3/16 inch screws from side covers MP17 and MP18. Remove six 2-56 x 1/8-inch and five 2-56 x 3/16-inch screws from front cover MP19, and remove front cover MP19.

<u>2</u>. Remove four 4-40 x 1/4-inch and four 2-56 x 1/4-inch screws from crystal cover MP20. Unsolder test jack 1A5J1 from 100-k $\Omega$  resistor R3, and remove crystal cover MP20.

<u>3</u>. Remove thermal insulation MP35 from switch S2, and remove switch S2.

4. Remove 2-56 x 3/16-inch screw. lockwasher, and flat washer from end of straight shaft MP25. Remove switch rotor E5.

(b) COIL TURRET SECTION. - To remove or gain access to a detail part in the coil turret section of spectrum generator module, refer to figure 5-79, and perform the following:

1. Remove twenty-six 2-56 x 3/16 inch screws from side covers MP17 and MP18. Remove six 2-56 x 1/8-inch and five 2-56 x 3/16 inch screws from front cover MP19, and remove front cover MP19.

2. Remove plastic retaining ring MP1 and coupling MP9. Remove spring pin from shaft coupling MP6. and remove shaft coupling MP6. Remove retaining ring MP4.

3. Remove six 4-40 x 3/8-inch screws from mounting plate MP21. Remove two 2-56 x 3/8-inch screws, nuts, and lockwashers, securing connector P1, and free mounting plate MP21 from connector P1. Remove mounting plate MP21.

4. Loosen setscrews in shaft collar MP7, and remove gear MP13 and shaft collar MP7. Remove three 2-56 x 1/4-inch screws and coil assembly E13.

5. If it becomes necessary to replace straight shaft MP25, perform the disassembly procedures of paragraph 5-4c( $\overline{3}$ )(a) and preceding steps of this paragraph, remove retaining ring MP4, and remove straight shaft MP25. (c) TUNING CIRCUIT SECTION. - To remove or gain access to a detail part in the tuning circuit section of spectrum generator module, refer to figure 5-79, and perform the following:

1. Remove twenty-six 2-56 x 3/16 inch screws from side covers MP17 and MP18, and remove side covers MP17 and MP18. Remove six 2- 56 x 1/4-inch and two 2- 56 x 1/8-inch screws from rear cover MP16, and remove rear cover MP16.

2. Remove plastic retaining ring MP1 and coupling MP9. Remove spring pin from shaft coupling MP6, and remove shaft coupling MP6. Remove retaining ring MP4.

3. Remove six 4-40 x 3/8-inch screws from mounting plate MP21. Remove two 2-56 x 3/8-inch screws, nuts, and lockwashers, securing connector P1 and free mounting plate MP21 from connector P1. Remove mounting plate MP21.

4. Remove hexnut H3 and bearing housing MP10 from top of spectrum generator module. Remove annular bearing  $\overline{MP2}$  and retaining ring MP5.

5. Loosen two setscrews on shaft collars MP11 in tuned circuit Z1 and MP7 in tuned circuits Z2 and Z3.

6. Pull straight shaft MP29 outward toward coupler side of spectrum generator module to clear tuned circuits Z 1 through Z3 as required.

Remove three capacitor rotors E10 and grounding springs E3 and E4 as required.

(4) RADIO RECEIVER-TRANSMITTER RT-695A/PRC-41 MAIN CHASSIS. - Remove RT-695A/ PRC-41 modules from main chassis according to paragraph 5-4b. Disassemble main chassis of RT695A/PRC-41 as required according to the following procedures (figure 5-81). (a) Remove five 6-32 x 5/16-inch screws at location B. and loosen one screw at location A on rear plate MP1. (b) Remove five 6-32 x 5/16-inch screws at location D, and loosen one screw at location C on rear plate MP1.

(c) Remove four 6-32 x 1/4-inch screws at location E on frame MP4. Remove three 4-40 x 1/4-inch screws at location F on frame MP4.

(d) Remove four 6-32 x 1/4-inch screws at location J on frame MP3. Remove four 4-40 x 1/4-inch screws at location H on frame MP3.

Loosen setscrews in control knobs O1 through 06 on front panel MP52 and remove control knobs O1 (e) through 06

Remove locking nuts on connectors 1A9J14, 1A9J13, and 1A9J28 on front panel MP52 and free (f) connectors 1A9J14, 1A9J13, and 1A9J28.

(g) Loosen locking nuts on controls, and free control shafts from front panel MP52.

(h) Carefully remove side frames MP3 and MP4 and rear plate MP1 from gear frame MP53 and front panel MP52.

(i) Remove four screws and lockwashers from gear frame MP53 at location K, and separate front panel MP52 from gear frame MP53.

(j) Remove four 2-56 x 3/16-inch screws from detent MP39, remove screw attaching spring MP77 and sleeve MP45, and remove detent MP39.

(k) Loosen screws in four coupler clamps MP48, and remove couplers MP49 and MP50, and two couplers MP51.

(I) Remove four spacer nuts MP74 at location M on gearplate MP30. Remove gearplate MP30 from gear frame MP53. At this point, access to remaining detail parts may be obtained as required.

d. REPAIR. - Repair procedures involve isolation of defective electrical parts by use of the troubleshooting procedures of section 4 and by inspection of mechanical parts and assemblies, removal and replacement of defective detail part, and the necessary synchronization and adjustment. Reassemble the disassembled mechanical assembly according to the applicable procedures of paragraph 5-4e. Synchronize the assembly according to paragraph 5-4f and make the necessary alignment according to applicable procedures of paragraph 5-3. Lubricate equipment as required according to paragraph 5-6.

e. REASSEMBLY.

(1) 1ST AND 2ND IF AMPLIFIER MODULE. To reassemble disassembled 1st and 2nd if amplifier module, refer to figure 5-75, and perform the steps of paragraphs 5-4f(1)(a) through 5-4f(1)(c) as required.

(a) Z1, Z2, AND Z3 SLUG RACK. - To reassemble the Z1, Z2, and Z3 slug rack and replace the amplifier subassembly, refer to figure 5-75, and perform the following:

1. Replace amplifier subassembly in place in electrical equipment chassis MP39 and secure by use of three 4-40 x 1/8-inch and one 4-40 x 1/4-inch screws. Solder electrical connections at points A (figure 5-40) on wired bracket, coaxial lead and ground connections on amplifier subassembly, and secure relay K3 with two 2-56 x 1/4-inch screws.

2. Set Z1, Z2, and Z3 slug rack MP22 in place on electrical equipment chassis MP39. Replace shaft end of shouldered shaft MP14 into bearing of electrical equipment chassis MP39, through cam MP12, flat washer H3, slug rack MP22, second flat washer H3, second cam MP13, through switches of amplifier subassembly, and into bearing of electrical equipment chassis MP39.

3. Set 1st and 2nd if amplifier module on maintenance bench so it is resting on wired bracket assembly and slug racks are upward. Rotate coupling MP10 so the guide pin hole is located over silk-screened circle on coupler side of electrical equipment chassis MP39. Rotor of rotary switch on amplifier subassembly should be making contact with its stator at point B (figure 5-28) and high point on cams MP12 and MP13 should be positioned downward. Replace spring pins in cams MP12 and MP13, and connect two helical springs MP6. 4. Replace captive screw H2 in top front position, and replace 2-56 x 1/4-inch screw and nut in bottom of

right front corner of 1<sup>st</sup> and 2<sup>nd</sup> if amplifier module.

5. Replace amplifier side covers MP16 and MP17 and secure in place with thirteen 2-56 x 3/16-inch selflocking screws.

(b) Z4, Z5, AND Z6 SLUG RACK. - To reassemble Z4, Z5, and Z6 slug rack, refer to figure 5-75, and perform the following:

1. Set slug rack MP25 in place on electrical equipment chassis MP39.

Replace shaft end of shouldered shaft MP11 into bearing of electrical equipment chassis MP39, through cam MP8, flat washer H3, slug rack MP25, second flat washer H3, second cam MP9, and into bearing of electrical equipment chassis MP39.

Replace retaining ring in shouldered shaft MP11. Rotate coupling MP10 so the guide pin hole is located over silk-screened circle on coupler side of electrical equipment chassis MP39.

4. Rotate cams MP8 and MP9 so their flat sides are upward and to left, and the collar pinning holes match holes in shouldered shaft MP11. Replace spring pins in cams MP8 and MP9 and connect four helical springs MP6. Refer to figure 5-28.

5. Replace side covers MP16 and MP17, and secure in place with thirteen 2-56 x 3/16-inch screws.

(c) OSCILLATOR AND SWITCHBOARD, AND RF COIL ASSEMBLIES. - With the oscillator and switchboard, and rf coil assemblies in place on electrical equipment chassis MP39, replace the shaft of shouldered shaft MP15 into the bearing of chassis MP39. Replace the shaft of shouldered shaft MP15 through oscillator and switchboard assembly, rf coil assembly and into bearing of electrical equipment chassis MP39. With coupling MP 10 of shouldered shaft MP15 rotated so its guide pin hole is upward, electrical contacts of the rotary switches in oscillator and switchboard and rf coil assemblies should be upward and in line with the coupling MP10 guide pin hole (figure 5-28). Replace retaining ring on shouldered shaft MP15.

(2) RF AND POWER AMPLIFIER MODULE. - To reassemble the disassembled rf and power amplifier module, refer to figure 5-80, and perform the steps of paragraphs 5-4f(2)(a) through 5-4f(2)(b) as required.

(a) Z1, Z2, AND Z3 TUNED CIRCUITS. To reassemble Z1, Z2, and Z3 tuned circuits, refer to figure 5-80, and perform the following:

1. Pass straight shaft MP14 through bottom plate MP33 at location C. Within tuned circuit Z 1, replace grounding spring MP42(1), grounding spring contact MP8(1), shaft collars MP6(1) and MP6(2), and shaft collar and rotor MP10(1) over straight shaft MP14.

2. Continue replacing rotors, grounding springs, shaft collars, etc. on straight shaft MP14 as MP14 is passed through tuned circuits Z2 and Z3.

3. Replace retaining rings (1 and 2), and pass shaft MP14 into bearing MP2(2). Replace 5/8-inch spring pin in straight shaft MP14. Rotate straight shaft MP14 fully clockwise so that 5/8-inch spring pin is stopped by stop pin in bottom plate MP33.

4. Position rotors in tuned circuits Z1 through Z3 so that they are in mesh with their stators and the flat edges of stators and rotors are in coincidence at their back side. Refer to figure 5-27. Tighten all shaft collars.

5. Replace gear and shaft MP22 in position so scribes made during disassembly are matched. Replace gears MP21, MP41, and MP23. Matching scribes made during disassembly are matched. Replace gearplate MP28, and secure in place with two 6-32 x 1/2-inch cross-recessed screws.

6. Replace shaft coupler MP4, and secure in place with spring pin. Replace coupler MP5, and secure in place with retaining ring MP1.

(b) Z4 AND Z5 TUNED CIRCUITS. - To reassemble Z4 and Z5 tuned circuits, refer to figure 5-80, and perform the following:

1. Carefully pass capacitor shaft MP32 through MP13. Replace rotor MP29(2), shaft collar MP6(5), grounding spring contact MP8(2), grounding spring E1, and retaining ring (6) on capacitor shaft MP32. 2. Replace rotor MP29(1), shaft cam MP24 grounding spring MP11(3) and retaining ring MP11(3) and retaining mP11(3).

Replace rotor MP29(1), shaft cam MP24, grounding spring MP11(3), and retaining ring (5) on capacitor shaft MP32.

Replace bottom plate MP33 in position, and secure in place by replacing four

2-56 x 3/8-inch self-locking screws at corners and one at location A. Replace six 2-56 x 5/16-inch screws, nuts, and lockwashers at location B. Secure plug P2 in place with two 2-56 x 3/8-inch screws. Secure plug P1 in place by use of two 2-56 x 3/16-inch screws, nuts, and lockwashers. 4. Check to see that 5/8-inch spring pin in straight shaft MP14 is against its stop pin. Replace gear

MP23 on capacitor shaft MP32, and secure in place with its spring pin.

5. Replace gearplate MP28 and secure in place with two 6-32 x 1/2-inch cross-recessed screws. Replace shaft coupler MP4, and secure in place with spring pin. Replace coupler MP5, and secure in place with retaining ring MP1.

(3) SPECTRUM GENERATOR MODULE. - To reassemble the disassembled spectrum generator module, refer to figure 5-79, and perform the applicable procedures of paragraphs 5-4f(3)(a) through 5-4f(3)(c). (a) CRYSTAL AND SWITCH S2 SECTION. - To reassemble mechanical parts of crystal and switch S2

section, perform the following:

1. Replace switch rotor E5. Secure switch rotor E5 in place with 2-56 x3/16-inch screw, lock-washer, and flat washer.

2. Position crystal and switch S2 section in place over switch rotor E5. Fit resistor R3 through proper hole in printed circuit switch S2.

3. Replace thermal insulation MP35 over crystals on switch section S2.

4. Replace crystal cover MP20 and secure with four 4-40 x 1/4-inch and four 2-56 x 1/4-inch screws; connect test jack 1A5J1 to 100-k  $\Omega$  resistor R3.

5. Check synchronization according to paragraph 5-4f, and make required adjustments. Replace side covers MP17 and MP18, and secure in place with twenty-six 2-56 x 3/16-inch screws. Replace cover MP19 and secure in place with six 2-56 x 1/8-inch and five 2-56 x 3/16-inch screws.

(b) COIL TURRET SECTION. - To reassemble mechanical parts of coil turret section, refer to figure 5-79, and perform the following:

1. If straight shaft MP25 and shaft collar MP26 were removed, replace straight shaft in bearing of chassis MP28. Replace retaining ring MP4 on straight shaft MP25. Reassemble crystal and switch S2 section in accordance with paragraph 5-4c(3)(a).

2. Replace coil assembly E13 over straight shaft MP25, and secure to shaft collar MP26 with three 2-56 x 1/4-inch screws. Replace shaft collar MP7 and gear MP13 on straight shaft MP25, and tighten setscrews in shaft collar MP7.

3. Replace mounting plate MP21 in position on spectrum generator module, and carefully fit gears MP14 located on mounting plate MP21 to gears MP13 and MP33. Position keying tab of grounding spring E11 in keying hole on mounting plate MP21. Secure mounting plate MP21 in place with six 4-40 x 3/8-inch screws.

4. Secure connector P1 to mounting plate MP21 with two 2-56 x 3/8-inch screws, nuts, lockwashers, and solder lugs.

5. Replace retaining ring MP4 on straight shaft MP25. Replace shaft coupling MP6 on straight shaft MP25, and secure in place with spring pin. Replace coupling MP9, and secure in place with plastic retaining ring MP1.

6. Check synchronization according to paragraph 5-4f and make required adjustments. Replace side covers MP17 and MP18, and secure in place with twenty-six 2-56 x 3/16-inch screws. Replace cover MP19, and secure in place with six screws 2-56 x 1/8-inch and five 2-56 x 3/16-inch screws. (c) TUNING CIRCUIT SECTION. - To reassemble mechanical parts of tuning circuit section, refer to figure

5-79, and perform the following:

1. Replace straight shaft MP29 with gear MP33 into position in spectrum generator chassis MP30. In tuned circuits Z1, Z2, and/or Z3, replace capacitor rotors E10 and grounding springs E3 and E4 as required.

2. Replace retaining mP5 and bearing MP2 on retaining plate MP24 end of straight shaft MP29. Position keying tab of grounding spring E4 in keying hole on retaining plate MP24, and replace bearing housing MP10 over bearing MP2.

3. Replace hexnut MPH3 on bearing housing MP10 and tighten.

4. Replace grounding spring E11 on straight shaft MP29. Replace mounting plate MP21 in position on spectrum generator module, and carefully fit gear MP14 located on mounting plate MP21 to gears MP13 and MP33. Position keying tab of grounding spring E11 in keying hole on mounting plate MP21. Secure mounting plate MP21 in place with six 4-40 x 3/8-inch screws.

5. Secure connector P1 to mounting plate MP21 with two 2-56 x 3/8-inch screws, nuts, lockwashers, and solder lugs.

6. Replace retaining ring MP4 on straight shaft MP25. Replace shaft coupling MP6 on straight shaft MP25, and secure in place with spring pin. Replace coupling MP9, and secure in place with plastic retaining ring MP1.

7. Position and synchronize three capacitor rotors E10 according to paragraph 5-4f, and tighten setscrews in shaft collars MP7 and MP11. Replace side covers MP17 and MP18, and secure in place with twenty-six 2-56 x 3/16-inch screws. Replace rear cover MP16, and secure in place with six 2-56 x 1/4-inch and two 2-56 x 1/8-inch screws

(4) RADIO RECEIVER-TRANSMITTER RT-695A/PRC-41 MAIN CHASSIS. -To reassemble the disassembled main chassis of RT-695A/PRC-41, refer to figure 5-81, and perform the following:

(a) Replace detail parts as required on gear frame MP53. Replace gearplate MP30 in position on gear frame MP53, and secure in place with four spacer nuts at location M.

(b) Replace four coupler clamps MP48 and couplers MP51, MP50, and MP49 on gearshaft

MP42, shaft MP84, shaft of differential MP71 and gearshaft MP40. Tighten coupler clamp screws.

(c) Replace detent MP39, and secure in place with four 2-56 x 3/16 screws and lockwashers. Replace spring MP77 over sleeve MP45, and secure with screws.

(d) Fit front panel MP52 to gear frame MP53, and secure with four screws and lockwashers at position K.

(e) Carefully fit side frames MP3 and MP4 and rear plate MP1 over gear frame MP53.

Reconnect required electrical connections, and fit front panel MP52 to control shafts and connectors of side frames MP3 and MP4 and gear frame MP53.

(f) Secure all control shafts and connectors that project through front panel MP52 with their applicable locking nuts. Secure side frames MP3 and MP4 to front panel MP52 by replacing eight 6-32 x 1/4-inch screws and lockwashers at positions E and J.

(g) Replace five 6-32 x 5/16-inch screws at location B, and tighten one screw at location A on rear plate MP1.

(h) Replace five 6-32 x 5/16-inch screws at location D, and tighten one screw at location C on rear plate MP1.

(i) Replace three 4-40 x ¼-inch screws at location F on side frame MP4.

(j) Replace four 4-40 x ¼-inch screws at location H on side frame MP3.

(k) Replace all control knobs in applicable positions, and tighten setscrews.

f. SYNCHRONIZATION.

(1) 1ST AND 2ND IF AMPLIFIER MODULE. The 1st and 2nd if amplifier module is synchronized during reassembly. Loss of synchronization or improper angular positioning of switch rotors or couplings is an indication of a defective detail part which must be located and replaced. Refer to figure 5-28 for correct synchronization of switch rotors, cams, and couplers. Subject 1st and 2nd if amplifier module to the trouble isolation procedures described in applicable portion of section 4.

(2) RF AND POWER AMPLIFIER MODULE. - Synchronize rf and power amplifier module as follows:

(a) Rotate coupler on rf and power amplifier module so guide hole in coupler is directly over silk-screened circle on bottom plate of rf and power amplifier module and slots in coupling are perpendicular to a line drawn between module tapered guide pin.

(b) Check rotors and stators of tuned circuits of rf and power amplifier module to determine whether module is properly synchronized. Module is properly synchronized when tips of stator and rotor blades of tuned circuits coincide at bottom and are displaced at top as viewed from side of module. Check position of cam that drives variable output coupling capacitor C30. Cam is properly synchronized when cam follower is at low point of cam. Refer to figure 5-27.

(c) If synchronization is required, loosen applicable rotor shaft collar, and adjust rotors for coincidence with stator by lining up tips of stator and rotor blades at bottom (figure 5-27). Adjust rotors on shaft so stator is halfway between bases of rotor blades, tighten rotor shaft collar, and replace side covers.

(d) If synchronization is required, check that rf and power amplifier module is properly aligned according to alignment procedures of paragraph 5-3e.

(3) SPECTRUM GENERATOR MODULE. Synchronize tuned circuits and printed circuit switch S3 of spectrum generator module as follows:

(a) Rotate coupler on spectrum generator module so guide hole in coupler is directly over silk-screened circle on bottom plate and slots in coupling are perpendicular to a line drawn between module guide pins.
 (b) Visually check that inductor L1 is contacted by dual wiper of switch S3. Inductor L1 is color coded with

(b) Visually check that inductor L1 is contacted by dual wiper of switch S3. Inductor L1 is color coded with red and orange dots which may be seen through observation hole adjacent to coupler. Rotate coupler until red and orange dots can be observed, rotors of tuned circuits are fully meshed, and coupler indication is over silk-screened circle. A 2 to 1 gear ratio exists between tuned circuit shaft and turret shaft. Both tuned circuits and inductor turret must be properly positioned to avoid a 180-degree ambiguity.

(c) Check position of rotor blades of tuned circuits. They should be fully meshed as illustrated in figure 5-24. If all three tuned circuits are not properly meshed, loosen gearshaft collar which clamps drive gear to turret shaft. Rotate tuned circuit shaft until all three tuned circuits are properly meshed, and tighten setscrews in gearshaft collar. If an individual tuned circuit is out of mesh, loosen setscrews in collar, and rotate rotor to a fully meshed position as shown in figure 5-24. Do not disturb position of tuner shaft while making this adjustment. Adjust rotors on shaft so stator is halfway between bases of rotor blades, and tighten setscrews.

Note

Use care in tightening setscrews. Make sure setscrews are engaging a segment of rotor shaft assembly and that they are not riding on the slots.

(d) Adjust idler gears for minimum backlash and binding.

g. REPLÁCEMENT.

(1) SPECTRUM GENERATOR MODULE. -Set Radio Receiver-Transmitter RT-695A/PRC-41 frequency controls to 225.0 MHz, set coupler on spectrum generator module so guide hole in coupler is directly over silk-screened circle on bottom plate. Look through hole in side cover of module, and check

that rotor and stator blades are in coincidence. Insert spectrum generator module in place on main chassis, and tighten captive screws.

Note

It is possible to mesh the hole in spectrum generator module coupler with pin on main chassis coupler so that tuning capacitors are in mesh. Use a screwdriver on shaft at top of the module to effect final fitting before tightening captive screws.

(2) RF AND POWER AMPLIFIER MODULE. -With RT-695A/PRC-41 still set to 225.0 MHz, set the coupler on rf and power amplifier module so guide hole in coupler is directly over silk-screened circle on bottom plate. Insert rf and power amplifier module in place on main chassis, and tighten captive screws.

Note

Use a screwdriver on shaft at top of the module to effect final fitting before lightening captive screws.

(3) 1ST AND 2ND IF AMPLIFIER MODULE. - Set frequency selector knobs on RT-695A/ PRC-41 to 229.9 MHz. Set couplers on 1st and 2nd if amplifier module so they are in the same relative position as the main chassis couplers. Insert the 1st and 2nd if amplifier module in place on the main chassis. Observe that couplers are properly fitted and tighten the captive screws.

Note

Use a screwdriver on shafts at top of the module to effect final fitting before tightening captive screws.

#### 5-5. VOLTAGE AND RESISTANCE MEASUREMENTS.

Unless noted otherwise, all voltages are taken in receive operation with Radio Receiver-Transmitter RT-695/PRC-41 set to 304.7 MHz and a signal generator input signal of 304.7 MHz, 1000 Hz modulated 30 percent at 1000 microvolts. All resistance measurements must be taken with the equipment turned OFF. Resistance measurements are not provided for transistorized circuits, since many ohmmeters will burn out the transistor if not used properly. All module voltage and resistance measurements are to be taken with the module operated on the appropriate extender cable. Refer to table 1-4. Refer to tables 5-2 through 5-14 as required for voltage and resistance measurements.

#### 5-6. LUBRICATION

Lubrication instructions for the mechanical modules of Radio Receiver-Transmitter RT-695A/PRC-41 are contained in table 5-15. Lubrication points are designated in figures 5-29 through 5-34. These procedures should be performed only on those assemblies that obviously require lubrication. If the mechanical parts of these assemblies are suitably lubricated, do not apply lubricant, but if the lubricant is dry, caked, or dirty, clean the component with a suitable solvent, dry it with clean, oil-free compressed air, and apply the specified lubricant as instructed in table 5-15. When performing these procedures, check all setscrews in clamps and collars for secure fitting.

#### CAUTION

Overlubrication may cause serious damage to the equipment. Wipe excess lubrication from exposed parts immediately after lubrication. A thin protective film should be left after wiping. Take care not to saturate any electrical insulation or rubber with lubricant.

#### 5-7. PARTS LOCATION.

See figures 5-35 through 5-82 for location and identification of detail parts, adjustments, and test points. Refer to section 6 of this handbook for further information pertaining to any particular detail part or assembly.

Original 5-42

# **TABLE 5-2**. DC POWER SUPPLY MODULE, VOLTAGE MEASUREMENTS\*

TRANSISTOR	EMITTER	BASE	COLLECTOR		
Q1	0.2	-1.0	26.0		
Q1	0.2	-1.0	26.0		
*Connect dc power supply module to RT-695A/PRC-41 with extension cable W5					

# TABLE 5-3. 1ST AND 2ND IF AMPLIFIER MODULE, VOLTAGE MEASUREMENTS\*

TRANSISTOR	EMITTER	BASE 1	COLLECTOR	BASE 2			
Q1	2.7	3.3	25.0	1.0			
Q2	1.85	2.65	19.5				
Q3	4.1	4.5	25.5	2.4			
Q4	0.05	0.75	0.0	0.55			
Q5	3.8	3.9	23.0	2.5			
Q6	0.95	0.97	18.5				
Q7 6.0 6.2 14.0							
*Set RT-695A/PRC-41 to 3099 MHz, carefully remove 1st and 2nd If amplifier module, and connect module with extension cable W2							

# TABLE 5-4. 3RD IF AND SQUELCH MODULE, VOLTAGE MEASUREMENTS

TRANSISTOR	EMITTER	BASE 1	COLLECTOR	BASE 2			
Q1	6.0	4.5	12.5	6.8			
Q2	6.2	4.5	13.0	7.0			
Q3	5.6	6.2	13.0				
Q4	0.35	0.5	6.5				
Q5	9.3	10.0	13.5				
Q6	14.0	14.0	10.0				
Q7	9.5	8.7	-0.2				
Q8 0.0 0.8 0.2							
Q10 *1.3, **8.2 *0.6, **7.6 0.8							
Connect 3rd If and squelch module 1A3 to RT-695A/PRC-41 with extension cable W4.							
*Squelch control 1A9R5 adjusted fully counterclockwise.							
**Squelch control 1A	**Squelch control 1A9R5 adjusted fully clockwise.						

# **TABLE 5-5**. AUDIO MODULE, VOLTAGE MEASUREMENTS\*

TRANSISTOR	EMITTER	BASE	COLLECTOR			
Q1	10.0	10.5	17.5			
Q2	5.2	5.7	21.0			
Q3	0.2	0.8	23.0			
Q4	0.2	0.8	23.0			
*Connect audio module to RT-695A/PRC-41 with extension cable W5						

# TABLE 5-6. SPECTRUM GENERATOR MODULE, VOLTAGE MEASUREMENTS

TUBE	VOLTAGE	CATHODE	GRID	PLATE	HEATER	HEATER
V1		0.35	-1.5	125	0	6.3
V2		4.3	-4.4	205	0	6.3
V3		2.4	0.0	205	0	6.3
V4		4.4	0.0	205	0	6.3
*Connect spectrum generator module to RT-695A/PRC-41 with extension cable W3						

# **TABLE 5-7.** SPECTRUM GENERATOR MODULE, RESISTANCE MEASUREMENTS

TUBE RESISTANCE	CATHODE	GRID	PLATE
.V1	100	10 k	<b>450 k</b> Ω
V2	700	220 k	<b>450 k</b> Ω
V3	330	0	<b>450 k</b> Ω
V4	680	0	<b>450 k</b> Ω

# **TABLE 5-8**. MODULATOR MODULE, VOLTAGE MEASUREMENTS

TRANSISTOR	EMITTER	BASE	COLLECTOR		
Q1	1.6	2.15	25.0		
Q2	0.8	-0.45	22.5		
Q3	0.8	-0.45	22.5		
All measurements taken with a 1000-Hz, 0.7-volt (open circuit) input signal with RT-695A/PRC-41 set to					
304.7 MHz in transmit operation Connect modulator module to RT-695A/PRC-41 with extension cable W5					

TRANSISTOR	EMITTER	BASE 1	COLLECTOR	BASE 2		
Q1	7.8	8.6	8.2			
Q2	7.8	8.6	8.2			
Q3	7.0	7.5	8.2			
Q4	12.2	12.3	19.0			
Q5	3.6	4.0	18.0	3.8		
Q6	3.6	4.2	18.0	4.0		
Q7	3.6	4.0	18.0	4.0		
Q8	4.2	4.8	18.0	4.8		
Q9	3.0	3.8	18.0	2.6		
Q10	0.05	0.5	18.0			
Q11	17.0	16.5	9.5			
Q12	1.65	2.3	8.2			
Q13	4.5	5.2	4.6			
Q14	19.0	19.5	24.0			
Q15	Q15 14.5 14.0 7.8					
*Set RT-695A/PRC-41 to 304.7 MHz, operate module on extension cable W3 and set SQUELCH control						
maximum counterclock	wise. Adjust signal input	for 243 0, 1000 Hz, m	nodulated 30 percent at	1000 microvolts		

# **TABLE 5-9**. GUARD RECEIVER MODULE, VOLTAGE MEASUREMENTS\*

# TABLE 5-10. RF AND POWER AMPLIFIER MODULE, VOLTAGE MEASUREMENTS\*

TUBE VOLTAGE	CATHODE	CONTROL	GRID	PLATE	HEATER	HEATER
V1	2.45	0		195	6.3	12.6
V2	2.5	0		195	6.3	12.6
V3	2.7	0		195	6.3	12.6
V4	2.75	0		150	6.3	12.6
V5	0.2	-2.25	5	155	12.6	18.9
TRANSISTOR EMITTER		COLLI		ECTOR	BASE	
Q1	0	2.3			.45	

\*Measurements taken m transmit condition.

Connect rf and power amplifier module to RT-695A/PRC-41 with extension cables W1 and W8.

# TABLE 5-11. RF AND POWER AMPLIFIER MODULE, RESISTANCE MEASUREMENTS\*

TUBE RESISTANCE	CATHODE	CONTROL GRID	PLATE				
V1	<b>1.4k</b> Ω	0	<b>450k</b> Ω				
V2	<b>1.4k</b> Ω	0	<b>450k</b> Ω				
V3	<b>1.4k</b> Ω	470	<b>450k</b> Ω				
V4	<b>1.4k</b> Ω	470	<b>450k</b> Ω				
V5 <b>33 330</b> 450kΩ							
*All resistance values in ohms							

# TABLE 5-12. RADIO RECEIVER-TRANSMITTER RT-695A/PRC-41, MAIN

# CHASSIS VOLTAGE MEASUREMENTS

TRANSISTOR	EMITTER	COLLECTOR	BASE
Q1	18.9	26.5 V	19.5

# TABLE 5-13. POWER SUPPLY PP-3700/PRC-41, NORMAL VOLTAGE MEASUREMENTS\*

TRANSISTOR	EMITTER	COLLECTOR	BASE			
Q1	28.0	40.0	28.8			
Q2	27.3	40.0	28.0			
Q3	26.4	36.0	27.2			
Q4	14.0	28.8	14.5			
*Measurements taken in transmit condition (3 amperes) 115-volt input.						

# TABLE 5-14. POWER SUPPLY PP-3700/PRC-41, OVERLOAD TRIP VOLTAGE MEASUREMENTS\*

TRANSISTOR	EMITTER	COLLECTOR	BASE		
Q1	0.3	60	0.8		
Q2	0.11	60	0.3		
Q3	0.1	50	0.1		
Q4	0.7	0.8	1.4		
*Measurements taken m transmit condition (3 amperes) 115-volt input after overload trip has been actuated by shorting P1 to P3.					

# AN/PRC-41A MAINTENANCE

# NAVSHIPS 0967-872-5020 TM-03816B-35/2 TM 11-5820-510-35-1

# TABLE 5-15. LUBRICATION INSTRUCTIONS

MODULE	PART	REFERENCE	FIGURE	LUBRICANT	METHOD OF APPLICATION
Rf and power amplifier module	Contact arm and contact ring	1	5-29, 5-30	Note 1	Apply thin film with a pipe cleaner.
	Ground springs	2	5-29, 5-30	Note 1	Apply thin film with a pipe cleaner.
1st and 2nd if amplifier module	Switches	3	5-31	Note 1	Apply thin film with a pipe cleaner.
	Sleeve bearings	4	5-31	MIL-L-6085A (note 2)	Apply two drops with a pipe cleaner.
Spectrum generator module	Inductor arms and inductor rings	5	5-32	Note 1	Apply thin film with a pipe cleaner.
	Grounding springs	6	5-32	Note 1	Apply thin film with a pipe cleaner.
	Sleeve bearings	7	5-32	MIL-L-6085A (note 2)	Apply two drops with a pipe cleaner.
	Gear teeth and idler shafts	8	5-32	Texaco Uni-Temp Grease 500 (005-0662-00)	Brush
	Switch contacts	9	5-32	Note 1	Apply thin film with a pipe cleaner.
RT-695A/ PRC-41 main chassis	Switch contacts	10	5-33, 5-34	Note 1	Apply thin film with a pipe cleaner.
	Sleeve bearings	11	5-33, 5-34	MIL-L-6085A (Note 2)	Apply two drops with a pipe cleaner.
	Bevel gear teeth pivot post and detents	12	5-33, 5-34	Texaco Uni-Temp Grease 500 (005-0662-00)	Brush.
	Ball bearings	13	5-33, 5-34	MIL-L-6085A (Note 2)	Apply two drops with a dropper.
	Helical gears	14	5-33, 5-34	Note 1	Brush.
	Ends of shaft	15	5-33, 5-34	Note 1	Brush.
	Sleeve bearing	16	5-33, 5-34	MIL-L-6085A (note 2)	Apply two drops with a pipe cleaner
NOTES:		-	•	•	•

1. Lubricant consists of 10 parts by weight of MIL-G-3278 grease, 45 parts by weight of butyl alcohol,

and 45 parts by weight of xylene.

2. Federal stock number 9150-261-8298.

3. Federal stock number 9150-223-4129.

Figure 5-29-

#### NAVSHIPS 0967-872-5020 TM-03816B-35/2 TM11-5820-510-35-1



Figure 5-29. RF and Power Amplifier Module, Left Side View, Lubrication Points



Figure 5-30. RF And Power Amplifier Module, Right Side View, Lubrication Points



Figure 5-31. 1st and 2nd IF Amplifier Module, Lubrication Points



Figure 5-32. Spectrum Generator Module, Lubrication Points

Figure 5-33

#### NAVSHIPS 0967-872-5020 TM-03816B-35/2 TM11-5820-510-35-1

# AN/PRC-41A MAINTENANCE



# AN/PRC-41A MAINTENANCE

# NAVSHIPS 0967-872-5020 TM-03816B-35/2 TM11-5820-510-35-1





Figure 5-34. Radio Receiver-Transmitter RT-695A/PRC-41, Left Side View, Lubrication Points



Figure 5-36. Radio Set Accessory Kit MK-706/PRC-41, Parts List Unit Designation **ORIGINAL** 5-52



Figure 5-37. Radio Receiver-Transmitter RT-695A/PRC-41 and Receiver-Transmitter



Figure 5-38. DC Power Supply Module, Oblique View, Parts Location



Figure 5-39. DC Power Supply Module, Bottom View, Parts Location



Figure 5-40. 1st and 2nd IF Amplifier Module, Wired Bracket, Parts Location



Figure 5-41. 3rd IF and Squelch Module, Right Side View, Parts Location

ORIGINAL

5-56



Figure 5-42. 3rd IF and Squelch Module, Left Side View, Parts Location

ORIGINAL

5-57



Figure 5-43. Audio Amplifier Module, Rear View, Parts Location



Figure 5-44. Modulator Module, Front View, Parts Location



Figure 5-45. Guard Receiver Module, Right Side View, Parts Location



Figure 5-46. Guard Receiver Module, Right Side View, Parts Location



Figure 5-47. Guard Receiver Module, Left Side View, Parts Location



Figure 5-48. Guard Receiver Module, Left Side View, Parts Location

,



REAR VIEW TB4



Figure 5-49. Guard Receiver Module, Terminal Board, Parts Location

ORIGINAL

5-61



Figure 5-50. RF and Power Amplifier Module, Rear View, Parts Location

ORIGINAL

5-62


Figure 5-51. RF and power Amplifier Module, Left Side View, Parts Location



Figure 5-52. RF and Power Amplifier Module, Right Side View, Parts Location



Figure 5-53. Radio Receiver-Transmitter RT-695A/PRC-41, Left Side View, Parts Location



Figure 5-54. Radio Receiver-Transmitter RT-695A/PRC-41, Front Panel, Parts Location



Figure 5-56. Power Supply PP-3700/PRC-41, Dust Cover Displaced, Parts Location



Figure 5-57. Power Supply PP-3700/PRC-41, Bottom View, Dust Cover Removed, Parts Location



Figure 5-58. Power Supply PP-3700/PRC-41, Side View, Dust Cover Removed, Parts Location .



Figure 5-59. Power Supply PP-3700/PRC-41, End View, Dust Cover Removed, Parts Location



Figure 5-60. Power Supply PP-3700/PRC-41, Oblique View, Parts Location



Figure 5-61. Power Supply PP-3700/PRC-41, Side View, Dust Cover Removed, Parts Location



Figure 5-62. Antenna AS-1405/PRC-41, Parts Location

5-69

## AN/PRC-41A MAINTENANCE



Figure 5-63. Mounting MT-2976/PRC-41, Parts Location



Figure 5-64. Mounting MT-2976/PRC-41, Rear View, Parts Location

### AN/PRC-41A MAINTENANCE



Figure 5-65. DC Adapter Cable, Parts Location



Figure 5-66. Cable Assemblies, Part Location



Figure 5-67. Mast AB-777/PRC-41, Parts Location



Figure 5-68. Adjustment Antenna Mast Adapter, Parts Location

### AN/PRC-41A MAINTENANCE



Figure 5-69. Mounting MT-2977/PRC-41, Parts Location



Figure 5-70. Rucksack Frame and Harness



Figure 5-71. Antenna Mounting Bracket, Parts Location



Figure 5-72. Maintenance Cable Set, Parts Location



Figure 5-74. Directional Antenna Carrying Bracket, Parts Location



Figure 5- 75







Figure 5-76. 1st and 2nd IF Amplifier Module, Oscillator and Switchboard Assembly, Exploded View



Figure 5-77. 1st and 2nd IF Amplifier Module, RF Coil Assembly, Exploded View

5-79, 5-80



Figure 5-78. 1<sup>st</sup> and 2<sup>nd</sup> IF Amplifier Module, Amplifier Subassembly, Exploded View





Figure 5-79. Spectrum Generator Module, Exploded View



Figure 5-80. RF and Power Amplifier Module, Exploded View

ORIGINAL

5-83

 $\gamma_{j}^{2,0}$ 



Figure 5-81. Radio Receiver-Transmitter RT-695A/PRC-41, Main Chassis, Exploded View

5-84



Figure 5-82. AF Amplifier Assembly (Buffer)



Figure 5-83. Connector and Filter Assembly



Figure 5-84. Receive-Transmit Switch, Exploded View



Figure 5-85.Special Purpose Electrical Cable Assembly CX-10831/PRC-41A

### TM 11-5820-510-35-1/NAVELEX 0967-872-5020



### SCHMATIC DIAGRAM MODEL 1 CX-10831/PRC-41A 5995-00-135-0205



Figure No. 5-86.1. Schematic Diagrams, Model 1 and Model 2, CX-10831/PRC-41A

U.S. GOVERNMENT PRINTING OFFICE: 1980-665-119/1021

Change 1 5-86.1

AN/PRC-41A MAINTENANCE



Figure 5-86. Power Supply PP-3700/PRC-41, Schematic Diagram

**ORIGINAL** 5-87,5-88

Figure 5-86



Figure 5-87. DC Power Supply 1A1, Schematic Diagram

ORIGINAL 5-89, 5-90

AN/PRC-41A MAINTENANCE

### NAVSHIPS 0967-872-5020 TM-03816B-35/2 TM11-5820-510-35-1



NOTES:

- 1. UNLESS OTHERWISE INDICATED; ALL RESISTANCE VALUES ARE IN OHMS.
- ALL CAPACITANCE VALUES ARE IN PICOFARADS, AND 2. L26 THRU L29 ARE FERRITE BEADS.
- FRONT AND BACK ROTORS ARE ELECTRICALLY CONNECTED IN POSITION SHOWN, FIRST IF. IS TUNED TO 25 MHZ.
- 4. FRONT AND BACK ROTORS ARE SEPERATE. FRONT ROTOR FOR TRANSMIT; BACK ROTOR FOR RECEIVE. IN POSITION SHOWN, SECOND IF. IS TUNED FOR 2.9 MHZ.

Figure 5-88. 1<sup>st</sup> and 2<sup>nd</sup> IF Amplifier 1A2, Schematic Diagram

Figure 5-88



NOTE:

UNLESS OTHERWISE INDICATED, ALL RESISTANCE VALUES ARE IN OHMS, ALL CAPACITANCE VALUES ARE IN PICOFARADS, AND ALL INDUCTANCE VALUES ARE IN MICROHENRYS.

TPI-5003-014

Figure 5-89. 3<sup>rd</sup> IF and Squelch 1A3, Schematic Diagram

ORIGINAL 5-93, 5-94



TPI-5002-013

Figure 5-90. Audio 1A4, Schematic Diagram

ORIGINAL 5-95, 5-96

Figure 5-90 AN/PRC-41A MAINTENANCE



NOTES:

1. UNLESS OTHERWISE INDICATED; ALL RESISTANCE VALUES ARE IN OHMS, ALL CAPACITANCE

VALUES ARE IN PICOFARADS, AND ALL INDUCTANCE VALUES ARE IN MICROHENRYS. 2. L42 THRU L61 AND L71 THRU L74 ARE FERRITE BEADS.

Figure 5-91. Spectrum Generator 1A5, Schematic Diagram

ORIGINAL 5-97, 5-97





TPI-5001-013

Figure 5-92. Modulator 1A6, Schematic Diagram

ORIGINAL 5-99, 5-100

Figure 5-93



Figure 5-93. Guard 1A7, Schematic Diagram

ORIGINAL

5-101, 5-102

- NOTES: I. UNLESS OTHERWISE INDICATED, ALL RESISTANCE VALUES ARE IN OHMS, ALL CAPACITANCE VALUES ARE IN PICOFARADS, AND ALL INDUCTANCE VALUES ARE IN MICROHENRYS. 2. VALUES SELECTED TO PROVIDE SPECIFIED SQUELCH PERFORMANCE. 3. VALUE SELECTED TO PROVIDE SPECIFIED AUDIO CLIPPING LEVEL.

TPI-5007-016

Figure 5-94



ORGINAL

Figure 5-94. RF and Power Amplifier 1A8, Schematic Diagram

5-103

IOTE: UNLESS OTHERWISE INDICATED, ALL RESISTANCE VALUES ARE IN OHMS, ALL CAPACITANCE VALUES ARE IN PICOFARADS, AND ALL INDUCTANCE VALUES ARE IN MICROHENRYS.

TPI-5004-015

TM11-5820-510-35-1



Figure 5-95. Radio Set AN/PRC-41-A, Interconnection Diagram (sheet 1 of 2)

ORGINAL

5-104

## MAINTENANCE

# NAVSHIPS 0967-872-5020 MAINTENANCE

# Figure TM-03816B-35/2 TM11-5820-510-35-1



Figure 5-95. Radio Set AN/PRC-41A, Interconnection Diagram (sheet 2 of 2)



Figure 5-96. Radio Receiver-Transmitter RT-695A/PRC-41, Main Chassis, Wiring Diagram

5-106

SIZE OF WIRE		COVERING OF WIRE	COLOR CODE	
CODE	SIZE		CODE	TYPE
A B C D E F G H J K L M N P Q R T V W X Y Z	No. 22AWG No. 20 No18 No.16 No.14 No.12 No.10 No.8 No.6 No.4 No.2 No.1 No.0 No.000 No.000 No.000 No.000 No.28 No.26 No.24 No.19 No.30	S Shielded S J Shielded & Jacket	1 2 3 4 5 6 7 8 9 a b C d e f	Brown Red Orange Yellow Green Blue Violet Gray (Slate) White Clear Tan Pink Maroon Light Green Light Blue

Figure 5-97. Electrical Wire Code
AN/PRC-41A MAINTENANCE





Figure 5-98. Radio Set AN/PRC-41A, Schematic Diagram (sheet 1 of 3)

ORGINAL

5-109

Figure 5-98





ORGINAL

Figure 5-98

Figure 5-98. Radio Set AN/PRC-41A, Schematic Diagram (sheet 1 of 3) **5-110** 

# AN/PRC-41A MAINTENANCE

AN/PRC-41A MAINTENANCE

#### NAVSHIPS 0967-872-5020 TM-03816B-35/2 TM 11-5820-510-35-1



Figure 5-98. Radio Set AN/PRC-41A, Schematic Diagram (sheet 2 of 3)

Original

Figure 5-98 Figure 5-98





Figure 5-98. Radio Set AN/PRC-41A, Schematic Diagram (sheet 2 of 3)

ORGINAL

5-112

#### AN/PRC-41A MAINTENANCE

AN/PRC-41A MAINTENANCE NAVSHIPS 0967-872-5020 TM-03816B-35/2 TM 11-5820-510-35-1



Figure 5-98. Radio Set AN/PRC-41A, Schematic Diagram (sheet 3 of 3)

5-113, 5-114

ORGINAL

00.0 AUDIC M N P R S T U • • • • • • • • FLIO

Figure 5-98

#### NAVSHIPS 0967-872-5020 TM-03816B-35/2 TM11-5820-510-35-1

Figure 5-99





Figure 5-99. Mounting MT-2976/PRC-41, Wiring Diagram and Schematic Diagram

#### **SECTION 6**

#### PARTS LIST

#### 6-1. INTRODUCTION.

a. REFERENCE DESIGNATIONS - The unit numbering method of assigning reference designations has been used to identify units, assemblies, subassemblies, and parts. This method has been expanded as much as necessary to adequately cover the various degrees of subdivision of the equipment. Examples of this unit numbering method and typical expansions of the same are illustrated by the following:



Example 3:



Read as: First (1) resistor (R) of second (2) subassembly (A) of first (1) subassembly (A) of third (3) unit.

b. REF DESIG PREFIX. - Partial reference designations are used on the equipment and illustrations. The partial reference designations consist of the class letter(s) and the identifying item number. The complete reference designations may be obtained by placing the proper prefix before the partial reference designations. Prefixes are provided on illustrations following the notation REF DESIG PREFIX.

#### 6-2. LIST OF UNITS.

Table 6-1 is a listing of the units comprising the equipment. The units are listed by unit numbers in numerical order. Thus when the complete reference designation of a part is known, this table will furnish the identification of the unit in which the part is located, since the first number of a complete reference designation identifies the unit. Table 6-1 also provides the following information for each unit listed: (1) quantity per equipment, (2) official name, (3) designation, (4) colloquial name, and (5) location of the first page of its parts listing in table 6-2.

#### 6-3. MAINTENANCE PARTS LIST .

Table 6-2 lists all units and their maintenance parts. The units are listed in numerical sequence. Maintenance parts for each unit are listed alphabetically-numerically by class of part following the unit designation. Thus the parts for each unit are grouped together. Table 6-2 provides the following information: (1) the complete reference designation of each unit, assembly, subassembly, or part, (2) reference to explanatory notes in paragraph 6-6, (3) noun name and brief description, and (4) identification of the illustration which pictorially located the part.

Printed circuit boards, assembly boards, modules, etc. are listed first as individual items in the maintenance parts list. In addition, at the completion of a parts listing for each unit, the individual circuit board, assembly board, module, etc. is then broken down by components into separate parts listing. When there is a redundancy of such electronic assemblies in subsequent units, reference is made to the parts breakdown previously listed.

#### Note

Classified parts are designated by the following classification symbols placed in the NOTES column (in addition to any numerically identified notes) of the Maintenance Parts List: C, Confidential; CMH, Confidential- Modified Handling; S. Secret; TS, Top Secret. A brief description is given for all key parts (parts differing from any parts previously listed in this table) and sub-key parts (parts identical to a key part but appearing for the first time for a unit). The names and descriptions are omitted for other parts, but reference is made to the key or subkey part for the data. Unless otherwise indicated, all drawing numbers apply to equipment manufacturer and all type numbers apply to part manufacturer.

#### 6-4. LIST OF MANUFACTURERS.

Table 6-3 lists the manufacturers of parts used in the equipment. The table incl udes the manufacturer's code used in table 6-2 to identify the manufacturers.

#### 6-5. STOCK NUMBER IDENTIFICATION.

Allowance parts list (APL) issued by the Elec tronics Supply Office (ESO) include federal stock. numbers and source maintenance and recoverability codes. Therefore, reference should be made to the APL prepared for the equipment for stock numbering information.

#### 6-6. NOTES.

The following notes provide information as referenced in table 6-2.

1. GOVERNMENT FURNISHED EQUIPMENT.

ORGINAL

6-1.1

# TABLE 6-1. LIST OF UNITS

UNIT	QTY	NAME OF UNIT	DESIGNATION	COLLOQUIAL NAME	PAGE
NO					
1	1	Radio Receiver-	RT-695A/PRC-41	Receiver-Transmitter	6-3
		Transmitter			
2	3	Storage Battery	BB-451/U	Dc Power Supply	6-38
3	1	Power Supply	PP-3700/PRC-41	AC Power Supply	6-39
4	1	Antenna	AS-1405/PRC-41	Dirt, Antenna	6-42
5	1	Antenna	AS-1404/PRC-41	Omni, Antenna	6-43
6	1	Mounting	MT-2976/PRC-41	Mounting, Rec, & Xmtr	6-43
7	1	Battery Adapter		Battery Adapter	6-45
8	1	Power, Electric	CX-8687/PRC-41	Cable Assembly	6-45
		Cable Assembly			
9	1	Power, Electric	CX-8686/PRC-41	Cable Assembly	6-45
		Cable Assembly			
10	1	Radio Frequency	CG-55G/U	Cable Assembly	6-45
		Cable Assembly			
11	1	Special Purpose	CX-8688/PRC-41	Cable Assembly	6-45
		Electrical			
		Cable Assembly			
12	1	Mast	AB-777/PRC-41	Ant. Mast	6-46
13	1	Antenna Mast		Ant. Mast Adapter	6-46
		Adapter			
14	1	Mounting	MT-2977/PRC-41	Mount, Ac Power	6-47
				Supply	
15	1	Handset	H-33E/PT	Handset	6-47
16	1	Electrical Equipment		Harness	6-47
		Harness			o (7
17	1	Bracket Assembly		Bracket Assembly	6-47

ORGINAL

6-2

UNIT	QTY	NAME OF UNIT	DESIGNATION	COLLOQUIAL NAME	PAGE
NO					
18	1	Spare Parts Kit		Spare Parts Kit	6-48
19	1	Maintenance Kit		Maintenance Kit	6-48
20	1	Tool Kit		Tool Kit	6-49
21	3	Guy Rope Assembly		Guy Rope Assembly	6-49
22	3	Guy Stake		Guy Stake	6-49
23	1	Directional Antenna		Dirt. Ant. Case	6-49
		Case			
24	1	Radio Set Case	CY-3883/PRC-41	Radio Set Case	6-50
25	1	Electronic Equipment Case	CY-3885/PRC-41	Accessory Kit Case	6-50
26	1	Equipment Repair		Equipment Repair	6-50
		Parts		Parts	
27	1	Special Purpose	CX-10831/PRC-41A	Cable Assembly	6-50
		Electrical			
		Cable Assembly			

## TABLE 6-2. MAINTENANCE PARTS LIST

# RADIO RECEIVER-TRANSMITTER RT-695A/PRC-41

REF DESIG	NOTES	NAME AND DESCRIPTION	FIG NO
1		RADIO RECEIVER-TRANSMITTER: 225.0 to 399.99 Mhz freq range, 1 band, 1750 channels; 26.5 Vdc; 4-5/8 in by 11-3/8 in by 13-3/4 in o/a; mfr 13499 part no 787-6067-001	5-37

#### DC POWER SUPPLY MODULE

1A1	POWER SUPPLY: electronic type rectification full wave; 190 Vdc	5-38
	at 150 MA, 26.5 Vdc,-6.8 Vdc at 7mS; 1.437 in by 4 in by 4.375 in; mfr 13499 part no 528-0084-015	
1A1C1	CAPACITOR, FXD, PPR, DIEL: 1 UF, ± 20%, 400V; mfr 56285 part no 18s4864	5-38

ORIGINAL

6-3

# TABLE 6-2. (Continued)

DC POWER	<u>R SUPPLY N</u>	10DULE	
REF	NOTES	NAME AND DESCRIPTION	FIG
NOTES			NO
1A1C2		CAPACITOR, FXD, PPR DIEL: same as 1A1C1	5-38
1A1C3		CAPACITOR, FXD, PPR DIEL: 3UF, ±20%, 400V; mfr 56289	5-38
		part no 18S4865	
1A1C4		CAPACITOR, FXD, ELCTLT: 15UF, ±20%, 20V, mfr 56289	5-39
		part no 150D156X0020B2	0.00
141CR1		SEMICOND DEVICE: MIL-E-11/1143 type 1N649	5-39
1A1CR2		SEMICOND DEVICE: same as 1A1CR1	5-39
1A1CR3		SEMICOND DEVICE. same as 1A1CR1	5-39
		SEMICOND DEVICE: same as 1A1CP1	5-30
		SEMICOND DEVICE: same as TATOAT SEMICOND DEVICE: mfr 07699 port po 1015010	5-39
		TEDMINAL mfr 01662 port no DTMT12M	5-39
		TERMINAL. IIII 91003 partio RTMITIZM	5-39
		TERIVIINAL. Same as TATET	5-39
1A1E12			5.00
TATE13		CAP, XSTR: al, 11/16-24 Inti trid, 3/4 in dia, 7/16 in dia,	5-39
		7/16 in ig, mfr 13499 part no 548-7003-002	5.00
1A1E14		BASE, XSTR: al; 11/16-24 ext thd, 0.265 in lg, 0.391 in lg o/a,	5-39
		mfr 13499 part no 548-7002-002	
1A1H1		SCREW EXTERNALLY RELIEVED BODY: alloy sti, zinc pi;	5-38
		hex skt cap screw, 8-32 NC-2 thd, 3/8 in Ig,, mfr 13499	
		part no 544-8109-002	
1A1H2		NUT, PL,, RD,CAP: al, chromate dip, 0.21875 in w,, 0 625in dia,	5-39
		mfr 13499 part no 548-7024-002	
1A1H3		WASHER: glass cloth, silicone rbr, 0.086 in ID, 0 750 in OD,	5-39
		0.050 in thk; mfr 13499 part no 547-7024-002	
1A1J1		JACK: mfr 8291 part no SKT10WHITE	5-38
1A1J2		JACK: same as 1A1J1	5-38
1A1J3		JACK: same as 1A1J1	5-38
1A1L1		REACTOR: mfr 70674 part no A12008	5-38
1A1MP1		INSULATOR, BSHG: plstc; 5/16 in dia by 0.150 in lg, 0.177 in ID	5-38
		undercut to 0 096 in mfr 13499 part no 548-7004-002	
1A1MP2		CHASSIS, ELEC EQPT: al alloy, chromate dip finish, 1 624 in by	5-38
		1.914 in by 4 413 in mfr 13499 part no 548-7030-004	
1A1P1		CONNECTOR: mfr 71468 part no DE9PC7	5-39
1A1Q1		TRANSISTOR: MIL-S-19500/180 (SIGC) type no 2N1486	5-39
1A1Q2		TRANSISTOR: same as 1A1Q1	5-39
1A1R1		RESISTOR, FXD, CMPSN: MIL-R-11 type RC20GF390K	5-39
1A1R2		RESISTOR, FXD, VW 1.2K +5% 2.5 W. mfr 91637	5-39
		part no RSM2C12000J	5-39
1A1R3		RESISTOR, FXD, WW: same as 1A1R2	5-39
1A1R4		RESISTOR, FXD, CMPSN: same as 1A1R1	5-39
1A1R5		RESISTOR, FXD, CMPSN: MIL-R-11 type RC20GF474K	5-39
1A1T1		TRANSFORMER, PWR, STEP-DOWN AND STEP-UP	5-38
· · · · ·		3 pri and 1 sec winding, 26 Vdc at 800 cps supply, 6 8V, 180V.	
		1-1/8 in by 1-5/16 in by 2-3/8 in. mfr 13499 part no 548-7028-003	

1ST AND	2ND IF AMPL	IFIER MODULE	
REF DESIG	NOTES	NAME AND DESCRIPTION	FIG NO
1A2		AMPLIFIER INTMD FREQ: mfr 13499 part no 528-0085-03	5-37
1A2C1		CAPACITOR, FXD, CER DIEL: 10PF ±1/4PF, 500V, mfr 72982	5-75
1A2C2		CAPACITOR, VAR, GLASS DIEL: : tubular, piston type, 0 8PF to	5-40
1A2C3		8.5PF, 1000V, mfr 73899 part no VC20GY CAPACITOR, FXD, CER DIEL: 0.75PF +5%, 500V, mfr 78488	5-40
14004		part no GA-75UUFPORM5PCT	5 40
1A2C4		CAPACITOR, FAD, CER DIEL: MIL 1998 CC22CH100D	5-40
14205		CAPACITOR, VAR GLASS DIEL. Salite as TAZOZ	5-40 5-40
1A2C7		CAPACITOR, FXD, CER DIEL: same as 1A2C6	5-40
1A2C8		CAPACITOR, VAR, GLASS DIEL: same as 1A2C2	5-40
1A2C9		CAPACITOR, FXD, CER DIEL: MIL type CC22CH070D	5-40
1A2C1		CAPACITOR, FXD, CER DIEL: same as 1A2C3	5-40
1A2C1		NOT USED	
1A2C1		CAPACITOR, VAR, GLASS DIEL: same as 1A2C2	5-40
1A2C1		CAPACITOR, FXD, CER DIEL: same as 1A2C6	5-40
1A2C1		CAPACITOR, FXD, CER DIEL: same as 1A2C6	5-40
1A2C1		CAPACITOR, VAR, GLASS DIEL: same as 1A2C2	5-40
1A2C1		CAPACITOR, FXD, CER DIEL: same as 1A2C9	5-40
1A2C1		CAPACITOR, FXD, CER DIEL: same as 1A2C3	5-40
1A2C1		CAPACITOR, FXD, CER DIEL: MIL type CC22CJ030C	5-40
1A2C1		CAPACITOR, FXD, CER DIEL: same as 1A2C6	5-40
1A2C2		CAPACITOR, VAR, GLASS DIEL: same as 1A2C2	5-40
1A2C2		CAPACITOR, FXD, CER DIEL: same as 1A2C14	5-40
1A2C2 2		CAPACITOR, FXD, CER DIEL: 47PF <u>+</u> 5% 500V, mfr 72982	5-75
1A2C2		part no 338026C0H0470J CAPACITOR, FXD, CER DIEL: same as 1A2C6	5-40
3			5-40
4			5-40
1A2C2 5		CAPACITOR, FXD, CER DIEL: $5.6PF \pm 5\%$ , 50V, mtr 78488	5-40
1A2C2		CAPACITOR, FXD, MICA DIELMIL type CM05F181G03	5-40
0 1A2C2		CAPACITOR, FXD, CER DIEL: same as 1A2C25	5-40
1A2C2		CAPACITOR, FXD, MICA DIEL: MIL type CM05F221G03	5-40
1A2C2		CAPACITOR, FXD, MICA DIEL: MIL type CM06F102G03	5-40

9 1A2C3	CAPACITOR, FXD, CER DIEL: same as 1A2C6	5-78
0 1A2C3	CAPACITOR, FXD, CER DIEL: 22PF <u>+</u> 5%, 500V; mfr 72982	5-75
1A2C3	part no 331026C0H0220J CAPACITOR, FXD, CER DIEL: 100PF <u>+</u> 2% 500V mfr 72982	5-75
2	part no 338026U2J0101G	5 75
3	Dart no VC9GWY	5-75
1A2C3 4	CAPACITOR, FXD, CER DIEL: same as 1A2C32	5-75
1A2C3 5	CAPACITOR, FXD, CER DIEL: same as 1A2C6	5-76
1A2C3 6	CAPACITOR, FXD, CER DIEL: MIL type CK12AX102M	5-86
1A2C3 7	CAPACITOR, FXD, CER DIEL: same as 1A2C36	5-76
1A2C3 8	CAPACITOR, FXD, MICA DIEL: MIL type CM05F181J03	5-76
1A2C3 9	CAPACITOR, FXD, CER DIEL: 20PF <u>+</u> 2%, 500V; mfr 72982	5-76
1A2C4	CAPACITOR, FXD, CER DIEL: same as 1A2C25	5-40
1A2C4	CAPACITOR, FXD, CER DIEL: .same as 1A2C6	5-40
ORIGINAL		

6-5

## TABLE 6-2. (Continued)

### 1ST AND 2ND IF AMPLIFIER MODULE

REF DESIG	NOT ES	NAME AND DESCRIPTION	FIG NO
1A2C42 1A2C43 1A2C44		CAPACITOR, FXD, CER DIEL.same as 1A2C6 CAPACITOR, FXD, CER DIEL same as 1A2C32 CAPACITOR, FXD, CER DIEL: 43PF, <u>+</u> 5%, 500V; mfr 72982	5-78 5-78 5-78
1A2C45 1A2C46 1A2C47		CAPACITOR, FXD, CER DIEL same as 1A2C36 CAPACITOR, FXD, CER DIEL same as 1A2C32 CAPACITOR, FXD, ELCTLT MIL type CS13BF105M	5-78 5-78 5-78
1A2C48 1A2C49 1A2C50 1A2C51 1A2C52 1A2C53 1A2C53 1A2C55 1A2C55 1A2C57 1A2C58 1A2C59 1A2C60		CAPACITOR, FXD, CER DIEL same as 1A2C36 CAPACITOR, FXD, CER DIEL same as 1A2C36 CAPACITOR, FXD, CER DIEL same as 1A2C32 CAPACITOR, FXD, CER DIEL same as 1A2C18 CAPACITOR, FXD, CER DIEL same as 1A2C25 CAPACITOR, FXD, CER DIEL same as 1A2C36 CAPACITOR, FXD, CER DIEL same as 1A2C36 CAPACITOR, FXD, CER DIEL same as 1A2C4 CAPACITOR, FXD, CER DIEL same as 1A2C6 NOT USED NOT USED	5-78 5-75 5-40 5-40 5-75 5-78 5-78 5-40 5-40
1A2C60 1A2C61 1A2C62 1A2C63 1A2C64 1A2C65 1A2C66 1A2C67		CAPACITOR, FXD, CER DIEL same as 1A2C36 CAPACITOR, FXD, CER DIEL same as 1A2C36 CAPACITOR, FXD, CER DIEL same as 1A2C6 CAPACITOR, FXD, CER DIEL same as 1A2C6 CAPACITOR, FXD, CER DIEL same as 1A2C6 CAPACITOR, FXD, CER DIEL MIL type CC22LH270J NOT USED	5-40 5-76 5-75 5-75 5-75 5-40
1A2C74 1A2C76 1A2CR1 1A2CR1 1A2CR2 1A2CR3 1A2H1		CAPACITOR, FXD, CER DIEL same as 1A2C9 SEMICOND DEVICE: MIL-S-19500/188 type 1N251 SEMICOND DEVICE. same as 1A2CR1 SEMICOND DEVICE : same as 1A2CR1 WASHER, FLAT brs, cad, 0.099 IN ID, 0.187 in OD, 0 020 in thk; mfr 13499 part no 504-0705-003	5-40 5-78 5-75 5-40 5-76
1A2H2 1A2H3		SCREW SHOULDER, NO 1 CRES, passivite finish 0.127 in hex 0.270 in dia 3.546 in Ig, mfr 13499 part no 544-8222-002 WASHER, FLAT brs, 0.188 in ID, 0 3125 in OD, 0 020 in thk,	5-75 5-75
1A2H4		NUT, PL, HEX brs, sil pl, 1/4 in hex; 8-32 UNC-2B thd, 1/16 in thk, mfr 13499 part no 544-8678-002	5-75
1A2H5		CAP, RIVET CRES, Passivate finish, 0.027 in ID, 0.218 in OD, 0.029 in thk, mfr 13499 part no 502-1512-002	5-78
1A2H6		WASHER, SPR TNSN: ph brz, 0 281 in ID, 0.370 in OD, 0.010 in thk, mfr 13499 part no 544-8634-002	5-40
1A2H7		NUT, SLUG ADJUSTING brs, bright alloy plate, 0 216 in ID, 0 281 in OD, 0.250 in w, mfr 13499 part no 540-8601-002	5-75
1A2J1 1A2J2 1A2J3		JACK, TIP mfr 98291 part no SKT2BC JACK, TIP same as 1A2J1 JACK, TIP same as 1A2J1	5-75 5-78 5-75

## NAVSHIPS 0967-872-5020 TM-03816B-35/2 TM11-5820-510-35-1

# TABLE 6-2. (Continued)

### 1ST AND 2ND IF AMPLIFIER MODULE

REF	NOTES	NAME AND DESCRIPTION	FIG
DESIG			NÖ
1A2K1		RELAY, AMT: mfr 01526 part no 3SAF1263	5-75
1A2K2		RELAY same as 1A2K1	5-75
1A2K3		RELAY contact arrangement 2BRE 1V max rated voltage 100UA	5-75
17 121 10		resistive current rating cont arrangement 1B DC 28V max rated	010
		voltage 1 amp resistive current rating mfr 13499 part no	
		544-8688-002	
14211		COIL RE 25 turns, single laver wound type, no 30 AWG enamel	5-40
		insulated wire 0.072 obms DC res mfr 13499 part no 544-8671-002	0 40
14212		COIL REsame as 1A2L1	5-40
14213		COIL REsame as 1A2L1	5-40
14214		COIL REsame as 1A2L1	5-40
14215		COIL REsame as 1A2L1	5-40
14216		COIL REsame as 1A2L1	5-40
14217		COIL, RESIDE as TAZET	5-75
14218		COIL REMIL C-15305 type LT4K041	5-76
10210		COIL RE23 turns single laver wound type no 32 AWG	5-77
		polyurethane insulated wire 17.1 MHz mln self-resonant freq	511
		$0.262 \text{ obm } DC \text{ reg. mfr} 13400 \text{ part no } 544_8703_003$	
142110		COIL RE 22 turns single laver wound type no 32 AWG	5-77
		polyurethane inculated wire 18 1 MHz mm self-resonant freq	5-11
		10220  obm DC res: mfr 13/00 part no 5/1-870/-003	
102111		COIL RE 21 turns single laver wound type no 32 AWG	5-77
		polyurathana insulated wire: 10.1 MHz, min self-resonant freq	5-11
		10213  obm DC reg mfr 13400 part no 544-8705-003	
110112		COIL BE 10 turns, single layer wound type, no 32 AWG	5-77
		polyurethane insulated wire 20.1 MHz min' self-resonant freq	5-11
		10196  obm DC resulting marking solution of $12,100$ part no $544.8706.003$	
1 1 2 1 2		COIL = DE 18 turns single lower wound type no 22 AM/C	5 77
TAZETS		polyurathana insulated wire 21.1 MHz, min self-resonant freq	5-77
		10196  obm DC resulting mfr 13400 part no 544-8707-003	
110111		COll PE 17 turns single layer wound type no 32 AW/C	5-77
		polyurethane inculated wire 22.1 MHz min self-resonant freq	5-11
		10190 obm DC reg mfr 13400 part no 544-8708-003	
1421 15		COIL RE 16 turns single laver wound type no 32 AWG	5-77
		polyurethane inculated wire 23.1 MHz mln self-resonant freq	5-11
		10180  ohm  DC reg mfr 13/00 part no 5/1-8700-003	
142116		COIL RE 15 turns single laver wound type no 32 AWG	5-77
		polyurethane insulated wire 2/ 1 MHz min self-resonant freq	5-11
		10164  ohm DC res. mfr  13499  part no  544-8710-003	
1421 17		COIL RE 1/1 turns single layer wound type no 32 AW/G	5-77
		polyurethane insulated wire 25.1 MHz min self-resonant free.	577
		0.164  ohm DC res: mfr  13499  part no  544-8711-003	
1421 18		COll RE 13 turns single laver wound type no 32 AWG	5-77
		polyurethane insulated wire 26.1 MHz min self-resonant freq	5-11
		0.147  obm DC reg. mfr 13/00 part no 5/1-8712-003	
1421 19		COll RE 53 turns single laver wound type in 32 AWG	5-40
		enamel insulated wire: 0.182 ohm DC res. mfr 13499	0 40
		nart no	
		544-6872-002	
1421.20		COIL RE same as 1421 12	5-40
1421 21		COIL REsame as 1A2L12	5-40
1421 22		COIL RE same as 1A2L12	5-40
	1		

## NAVSHIPS 0967-872-5020 TM-03816B-35/2 TM11-5820-510-35-1

Table 6-2

## 1ST AND 2ND IF AMPLIFIER MODULE

## TABLE 6-2. (Continued)

REF	NOTES	NAME AND DESCRIPTION	FIG
DESIG			NO
1A2L23		COIL, RFsame as 1A2L19	5-40
1A2L24 1A2L25		COIL, RESAME AS TAZLIZ	5-40 5-75
1A2L25		BEAD mfr 02114 part no 56-590-65-3B	5-75
1A2L27		BEAD: same as 1A2L26	5-75
1A2L28		BEAD: same as 1A2L26	5-75
1A2L29		BEAD: same as 1A2L26	5-75
1A2L30		COIL, RF: 0.4/UH, mfr 82142 part no 4412-1M/5	5-75
TAZL3T thru		NOTUSED	
1A2L49			
1A2L50		COIL, RF: 27UH; mfr 99800 part no 1537-48	5-40
1A2MP1		BEARING mfr 83086 part no SFR155PPK25-7	5-75
1A2MP2		NUT DIAINI HEVI of chamical finish 4 40 LINC 2D tigd	F 70
TAZIVIPS		10.343 in lq, mfr 13499 part no 540-9036-003	5-76
1A2MP4		NUT.PLAIN, HEX al. chemical finish, 4-40 UNC-2B thd.	5-78
		0.312 in lg; mfr 13499 part no 540-9035-003	
1A2MP5		SPACER, SLV al, 0.035 in thk by 0.187 in OD by 0.187 in lg;	5-78
142MP6		MIT 13499 PAR NO 541-5977-002 SPRING HELICAL EXTENSION CRESS 0.187 in by 0.411 in	5-75
		in o/a dim: mfr 13499 part no 544-8633-002	5-75
1A2MP7		NUT, PL, HEX al, chemical finish, 4-40 UNC-2B thd, 0.218 in Ig,	5-78
		mfr 13499 part no 540-9032-003	
1A2MP8		CAM, CONTROL brs, 0.187 in by 0.796 in by 0.828 in 0/a dim; mfr 13499 part no 544-8629-002	5-75
1A2MP9		CAM. CONTROL: brs. 0.187 in by 0.796 in by 0.796 in $o/a \dim$	5-75
		mfr 13499 part no 544-8630-002	
1A2MP10		COUPLING, SHAFT, FLEX. Insert type, 0.906 in by 0.250 in	5-75
142MP11		0/a dlm, mfr 13499 part no 546-4872-002 SHAFT STR: CRES Passivate finish 0.1873 in dia 3.454 in	5-75
			5-75
		mfr 13499 part no 546-4875-002	
1A2MP12		CAM, CONTROL- brs, 0.187 in by 0.625 in by 0.796 in o/a dim;	5-75
142MP13		MIT 13499 part no 544-8636-002 CAM CONTROL - brs: 0.187 in by 0.593 in by 0.781 in 0/2 dim	5-75
		mfr 13499 part no 544-8637-002	5-75
1A2MP14		SHAFT, STR: CRES, Passivate finish, 0.1873 in dia, 3.454 in	5-75
		lg,	
1A2MP15		MIT 13499 part no 545-4676-002 SHAFT STR: CRES Passivate finish 0 1873 in dia 3 454 in	5-75
			010
		mfr 13499 part no 546-4877-002	
1A2MP16		COVER, AMPL al; 0.032 in by 3 234 in by 4.421 in o/a dim;	5-75
1A2MP17		100 + 3499 part 10 544-0009-002 COVER AMPL al: 0.032 in by 3 234 in by 4 421 in 0/a dim	5-75
.,		mfr 13499 part no 544-8670-002	0.10
1A2MP18		BEARING mfr 83086 part no SR144PK28-7	5-75
1A2MP19		RING mfr 79136 part no 5100-12C	5-75
TAZIVIPZU		dim: mfr 13499 part no 502-6005-002	5-75
1A2MP21		POST, FOLLOWER: CRES, PASSIVATE finish, 0.1245 in ala.	5-75
		1.850	
		in Ig, mtr13499 part no 544-8638-002	

## 1ST AND 2ND IF AMPLIFIER MODULE

REF	NOTES	NAME AND DESCRIPTION	FIG
DESIG			NO
1A2MP22		BASE, AMPL: CRES; incl 3 nuts; 1 in by 1.522 In by 2.218 in o/a dim: mfr 13499 part no 544-8676-002	5-75
12AMP23		RING: mfr 79136 part no 5133-12C	5-75
1A2MP24		POST, FOLLOWER: CRES, passivate finish, 0.1245 in dia, 2.890	5-75
1A2MP25		BASE, AMPL CRES, incl 6 nuts; 1.344 in by 2.156 in by 2.562 in o/a dim, mfr 13499 part no 544-8677-002	5-75
1A2MP26		CORE mfr 92054 part no 52-3811	5-75
1A2MP27		SCREW, ADJUSTABLE CORE brs sil pl 0.025 in ID, 0.089 in OD,	5-75
1A2MP28		INSULATOR, DISK. silicone rbr; 25/64 in dia by 1/16 in thk, mfr 13499 part no 546-6075-002	5-75
1A2MP29		NOT USED	5-76
		part no 540-9043-003	570
1A2MP31		PLATE, COIL MOUNTING: al, chromate dip, 0 032 in by 1.343 in	5-77
1A2MP32		NUT, SLV al, chemical; 4-40 UNC-2B trd, 0 500 in lg; mfr 13499 part no 540-9041-003	5-77
1A2MP33		NUT, PL, HEX: al, chromate dip, 4-40 UNC-2B tied, 0 187in w across flats, 0 687 in lg o/a, mfr 13499 part no 540-9047-003	5-77
1A2MP34 1A2MP35		RING mfr 79136 part no 5100-27C COVER, AMPL: bra, sil pl; 0.032 in thk. 1.453 in dia, 2 352 in lg,	5-40 5-40
1A2MP36		SPACER, SLV al, chromate dip, 0.093 in by 0.156 in by 0.562 in mfr 13499 part no 545-7138-002	5-76
1A2MP37		PLATE, XTAL HOLDER: al, chromate dip, 0.032 in thk, 1.3125 in by	5-76
1A2MP38		1 328 in mfr 13499 part no 544-8651-002 BRACKET, CONN, RCPT, ELEC: al, chromate dip, 0 250 in by	5-78
1A2MP39		0.344 in by 0.531 in; mfr 13499 part no 544-8657-002 AMPLIFIER SUBASSEMBLY: 3 370 in by 4.253 in by 4.432 in o/a	5-75
1A2MP40		DAMPENER, XTAL, NO 2 rbr sheet; 1-1/4 in by 1-1/4 in mfr	5-76
1A2MP41		SHIELD, RF brs, si1 pl finish, 0 0320 in by 1.933 in by 2.281 in	5-75
1A2MP42		PLATE, ELEC SHIELD: sil pl cop cont stripe, sil pl brs plate;	5-75
1A2MP43		POST, ELEC-MECH EQPT CRES, PASSIVATE finish; 0.086 in by	5-75
1A2P1		CONNECTOR: mfr 71468 part no DBM13W3PC27; includes 3	5-75
		CONNECTOR: mfr 71468 part no DM 53740-5000 (p/o 1A2P1) qty	
1A2Q1		TRANSISTOR: MIL-S-19500/80A(SIGC) type 3N35	5-40
1A2Q2		TRANSISTOR: mfr 07688 part no 2N915	5-40
1A2Q3		TRANSISTOR same as 1A2Q1	5-75
1A2Q4		TRANSISTOR Same as 1A2Q1	5-75 5-75
1A2Q5		TRANSISTOR: MIL-T-19500/69B(NAVY) type 2N338	5-78
1A2Q7		TRANSISTORsame as 1A2Q6	5-78

Table

#### NAVSHIPS 0967-872-5020 TM-03816B-35/2 TM11-5820-510-35-1



# TABLE 6-2. (Continued)

### . 1ST AND 2ND IF AMPLIFIER MODULE

REF	NOTES	NAME AND DESCRIPTION	FIG
			INU 5-40
14200		DESISTOR, FAD, CIVIFSIN. WILLERTT TYPE RC07GF135R	5 40
1/202		DESISTOR, FXD, CMPSN. MILEN-TT type NOUTOF 272N	5 40
14203		DESISTOR, FAD, CIVIESIN. Sallie as IAZAZ	5 40
		RESISTOR, FAD CMFSN. MILER-11 UPP RC07GF275R	5-40
		RESISTOR, FAD, CIVIPSIN. IVIIL-R-11 LYPE RCU/GF1U2R	5-40
		RESISTOR, FXD, CIVIPSIN: IVIIL-R-TTTYPE RCU/GF4/TK	5-40
1AZR7			5-40
1A2R8		RESISTOR, FXD, CMPSN: MIL-R-11 type RC07GF221K	5-40
1A2R9		RESISTOR, FXD, CMPSN: MIL-R-11 type RC07GF104K	5-75
1A2R10		RESISTOR, FXD, CMPSN: MIL-R-11 type RC0/GF682K	5-40
1A2R11		RESISTOR, FXD, CMPSN: same as 1A2R1	5-75
1A2R12		RESISTOR, FXD, CMPSN: MIL-R-11 type RC07GF472K	5-75
1A2R13		RESISTOR, FXD, CMPSN: same as 1A2R12	5-75
1A2R14		RESISTOR, FXD, CMPSN: MIL-R-11 type RC07GF103K	5-75
1A2R15		RESISTOR, FXD, CMPSN: same as 1A2R5	5-40
1A2R16		RESISTOR, FXD, CMPSN: MIL-R-11 type RC07GF152K	5-78
1A2R17		RESISTOR, FXD, CMPSN: MIL-R-11 type RC07GF333K	5-78
1A2R18		RESISTOR, FXD, CMPSN: same as 1A2R5	5-78
1A2R19		RESISTOR, FXD, CMPSN: same as 1A2R12	5-78
1A2R20		RESISTOR, FXD, CMPSN: same as 1A2R1	5-75
1A2R21		RESISTOR, FXD, CMPSN: same as 1A2R12	5-75
1A2R22		RESISTOR, FXD, CMPSN: MIL-R-11 type RC07GF182K	5-75
1A2R23		RESISTOR, FXD, CMPSN: same as 1A2R14	5-75
1A2R24		RESISTOR, FXD, CMPSN: same as 1A2R1	5-76
1A2R25		RESISTOR FXD CMPSN: same as 1A2R12	5-76
1A2R26		RESISTOR FXD CMPSN: same as 1A2R22	5-76
1A2R27		RESISTOR FXD CMPSN: same as 1A2R14	5-76
1A2R28		RESISTOR FXD CMPSN: MIL-R-11 type RC20GE120K	5-75
142R20		RESISTOR FXD CMPSN: same as 142R5	5-76
1420230		RESISTOR FXD, OMDSN: same as 1A2R5	5-78
1420231		PESISTOR, FXD, CIMPON, Same as FAZING PESISTOR FYD CMPSN: MIL P-11 type PC07CE303K	5-78
142022		PESISTOR, FXD, CMPSN. MIL-IN-TH type NO07GI 335N	5 79
142022		DESISTOR, FAD, CIVIFON. Salle as TAZAZ	570
14203		RESISTOR, FAD, CIVIFSIN. IVIIL-R-11 LYPE RCU/GF332R	570
142034		RESISTOR, FAD, CIVIFSIN SAILLE AS TAZRY	5-70
142022		RESISTOR, FAD, CIVIESIN. MILER-II UPE ROUTOFUIR	5-40
1420027		RESISTOR, FAD, CIVIFSIN. Sallie as TAZR14	5-40
142537		RESISTOR, FAD, CIMPSIN. Same as 142PL	5-76
1A2R38			5-75
1A2R39			5-40
1A2R40		RESISTOR, FXD, CMPSN: MIL-R-11 type RC07GF121K	5-75
1A251		PRINTED CIRCUTT PLATE: pisto board; 0.062 in by 1.325 in by	5-77
44000		1.343 in board dim; mfr 13499 part no 544-8695-003	5 70
1A2S2		PRINTED CIRCUIT PLATE same as 1A2S1	5-76
1A2S3		PRINTED CIRCUIT PLATE pistc board, 0.062 in by 1.297 in by	5-78
		3.343 in o/a dim, mfr 13499 part no 544-8697-003	
1A2Y1		CRYSTAL UNIT, QTZ: 2.90000 MHz, mfr 00136 part no 290-9303-00	5-78
1A2Y2		CRYSTAL UNIT, QTZ 3.00000 MHz, mfr 00136 part no 290-9604-00	5-78
1A2Y3		CRYSTAL UNIT, QTZ 3.10000 MHz, mfr 00136 part no 290-9605-00	5-78
1A2Y4		CRYSTAL UNIT, QTZ 3.20000 MHz, mfr 00136 part no 290-9606-00	5-78
1A2Y5		CRYSTAL UNIT, QTZ: 3.30000 MHz, mfr 00136 part no 290-9607-00	5-78

### TM-03816B-35/2 TM11-5820-510-35-1

## TABLE 6-2. (Continued)

## 1ST AND 2ND IF AMPLIFIER MODULE

REF DESIG	NOTES	NAME AND DESCRIPTION	FIG NO
1A2Y6		CRYSTAL UNIT, QTZ :3.40000 MHz, mfr 00136 part no 290-9608-00	5-78
1A2Y7		CRYSTAL UNIT, QTZ: 3 50000 MHz, mfr 00136 part no 290-9609-00	5-78
1A2Y8		CRYSTAL UNIT, QTZ: 3.60000 MHz; mfr 00136 part no 290-9610-00	5-78
1A2Y9		CRYSTAL UNIT, QTZ: 3.70000 MHz; mfr 00136 part no 290-9611-00	5-78
1A2Y10		CRYSTAL UNIT, QTZ. 3.80000 MHz, mfr 00136 part no 290-9612-00	5-78
1A2Y11		CRYSTAL UNIT, QTZ: 17.10000 MHz; mfr 94217 part no M04496	5-76
1A2Y12		CRYSTAL UNIT, QTZ: 18.10000 MHz, mfr 94217 part no M04495	5-76
1A2Y13		CRYSTAL UNIT, QTZ: 19.10000 MHz; mfr 94217 part no M04494	5-76
1A2Y14		CRYSTAL UNIT, QTZ: 20.10000 MHz, mfr 94217 part no M04493	5-76
1A2Y15		CRYSTAL UNIT, QTZ: 21.10000 MHz; mfr 94217 part no M04492	5-76
1A2Y16		CRYSTAL UNIT, QTZ: 22.10000 MHz, mfr 94217 part no M04491	5-76
1A2Y17		CRYSTAL UNIT, QTZ: 23.10000 MHz, mfr 94217 part no M04490	5-76
1A2Y18		CRYSTAL UNIT, QTZ: 24.10000 MHz, mfr 94217 part no M04489	5-76
1A2Y19		CRYSTAL UNIT, QTZ: 25.10000 MHz, mfr 94217 part no M04488	5-76
1A2Y20		CRYSTAL UNIT_OT7_26 10000 MHz_mfr 94217 part no M04487	5-76

### 3RD IF AND SQUELCH MODULE

REF DESIG	NOTES	NAME AND DESCRIPTION	FIG NO
1A3		AMPLIFIER, INTMD FREQ 500 Hz operating freq, 47 Hz band w at 6 db down, 50	5-37
		4 437 in: mfr 13499 part no 528-0372-017	
1A3C1		CAPACITOR, FXD, CER DIE L MIL type CK12AX102M	5-42
1A3C2		NOT USED	
1A3C2		CAPACITOR, FXD, ELCTLT MIL type CS13BE225M	5-42
1A3C4		CAPACITOR, FXD, ELCTLT same as 1A3C3	5-42
1A3C5		CAPACITOR, FXD, MICA DIEL MIL type CM05FD221J03	5-42
1A3C6		CAPACITOR, FXD, ELCTLT- same as 1A3C3	5-42
1A3C7		CAPACITOR, FXD, ELCTLT MIL type CS13BH104M	5-42
1A3C8		CAPACITOR, FXD, ELCTLT same as 1A3C3	5-42
1A3C9		CAPACITOR, FXD, MICA DIEL same as 1A3C5	5-42
1A3C10		CAPACITOR, FXD, CER DIEL. same as 1A3C1	5-42
1A3C11		CAPACITOR, FXD, CER DIEL same as 1A3C3	5-42
1A3C12		CAPACITOR, FXD, ELCTLT MIL type CS13BE106M	5-42
1A3C13		CAPACITOR, FXD, ELCTLT MIL type CS13BF475M	5-41
1A3C14		CAPACITOR, FXD, CER DIEL: same as 1A3C1	5-41
1A3C15		CAPACITOR, FXD, ELCTLT same as 1A3C3	5-41
1A3C16		CAPACITOR, FXD, ELCTLT same as 1A3C3	5-41
1A3C17		CAPACITOR, FXD, ELCTLT MIL type CS13BE156M	5-41
1A3C18		CAPACITOR, FXD, CER DIEL same as 1A3C1	5-41
1A3C19		NOT USED	
1A3C20		CAPACITOR, FXD, ELCTLT: MIL type CS13BF105M	5-41
1A2C21		CAPACITOR, FXD, CER DIEL MIL type CK14AX103M	5-42
1A2C22		CAPACITOR, FXD, ELCTLT same as 1A3C3	5-41
1A3CR1		SEMICOND DEVICE: JAN type 1N3024B	5-41
1A3CR2		SEMICOND DEVICE: JAN type JAN1N483B	5-41
1A2CR3		SEMICOND DEVICE same as 1A3CR2	5-41
1A3CR4		SEMICOND DEVICE: same as 1A3CR2	5-42
1A3CR5		SEMICOND DEVICE. same as 1A3CR2	5-41

ORIGINAL

6-11

#### NAVSHIPS 0967-872-5020 TM-03816B-35/2 TM11- 5820-510-35-1

# TABLE 6-2. (Continued)

## 3RD IF AND SQUELCH MODULE

REF DESIG	NOTES	NAME AND DESCRIPTION	FIG NO
1A3CR		NOT USED	
6 1A3CR		NOT USED	
/ 1A3CR		SEMICOND DEVICE: same as 1A3CR2	5-41
1A3CR		SEMICOND DEVICE: JAN type JAN1N751A	5-41
9 1A3E1		PRINTED CIRCUIT BOARD: plstc; 0.062 in by 3.070 in by 3.437 in;	5-42
1A3E2		PRINTED CIRCUIT BOARD: plstc; 0.062 in by 3.070 in by 3.162 in;	5-41
1A3FL1 1A3J1 1A3K1 1A3L1 1A3L2 1A3L3 1A3L4 1A3L4		FILTER: mfr 81815 part no TL80D110C1 JACK, TIP: WHT; mfr 98291 part no SKT41WHT RELAY: mfr 01526 part no 3SAK1029 COIL, RF: MS type MS90539-08 COIL, RF: MS type MS90539-15 COIL, RF: same as 1A3L2 COIL, RF: MS type MS90541-11 COVER, CHASSIS: al; 0.032 in by 3.218 in by 4.406 in; mfr 13499 part no 548-7371-003	5-42 5-41 5-42 5-42 5-42 5-42 5-41 5-41
1A3MP2		COVER, CHASSIS: al; 0.032 in by 3.218 in by 4 406 in mfr 13499	5-42
1A3MP3		SCREW, SHOULDER, NO 2 CRES, passivate finish; 8-32 UNC-2A thd, 0 127 in hex, 0.270 in dia, 3.584 in Ig, mfr 13499 part no	5-41
1A3P1 1A3Q1 1A3Q2 1A3Q3 1A3Q4 1A3Q5 1A3Q6 1A3Q7 1A3Q8 1A3Q9		CONNECTOR: mfr 71468 part no DAM15PA160C37 TRANSISTOR: JAN type 3N35 TRANSISTOR: same as 1A3Q1 TRANSISTOR: JAN type JAN2N2219A TRANSISTOR: JAN type JAN2N21013 TRANSISTOR: same as 1A3Q3 TRANSISTOR: JAN type JAN2N2905A TRANSISTOR: same as 1A3Q6 TRANSISTOR: same as 1A3Q4 NOT USED	5-41 5-42 5-42 5-41 5-42 5-42 5-42 5-41 5-41
1A3Q10 1A3R1 1A3R2 1A3R3 1A3R3		TRANSISTOR: same as 1A3Q6 RESISTOR, FXD, CMPSN: MIL type RC07GF123K RESISTOR, FXD, CMPSN: MIL type RC07GF472K RESISTOR, FXD, CMPSN: MIL type RC07GF221K RESISTOR, FXD, CMPSN: MIL type RC07GF272K	5-41 5-42 5-42 5-42 5-42
1A3R5 1A3R6 1A3R7 1A3R8		RESISTOR, FXD, CMPSN: same as 1A3R2 RESISTOR, FXD, CMPSN: same as 1A3R1 RESISTOR, FXD, CMPSN: same as 1A3R1 RESISTOR, FXD, CMPSN: same as 1A3R2	5-42 5-42 5-42
1A3R10 1A3R10 1A3R11 1A3R12 1A3R13 1A3R14 1A3R15 1A3R16		RESISTOR, FXD, CMPSN: same as 1A3R4 RESISTOR, FXD, CMPSN: same as 1A3R2 RESISTOR, FXD, CMPSN: MIL type RC07GF682K RESISTOR, FXD, CMPSN: same as 1A3R12 RESISTOR, FXD, CMPSN: MIL type RC07GF101K RESISTOR, FXD, CMPSN: MIL type RC07GF683K RESISTOR, FXD, CMPSN: MIL type RC07GF182K	5-42 5-42 5-42 5-42 5-42 5-41 5-42

### 3RD IF AND SQUELCH MODULE

REF DESIG	NOTES	NAME AND DESCRIPTION	FIG NO
440047			5 40
1A3R17		RESISTOR, FXD, CMPSN: MIL type RC07GF104K	5-42
1A3R18		RESISTOR, FXD, CMPSN: MIL type RC07GF102K	5-42
1A3R19		RESISTOR, FXD, CMPSN: MIL type RCU/GF222K	5-41
1A3R20		RESISTOR, FXD, CMPSN: same as 1A3R18	5-42
1A3R21		RESISTOR, FXD, CMPSN: MIL type RCU/GF153K	5-42
1A3R22		RESISTOR, FXD, CMPSN: same as 1A3R3	5-42
1A3R23		RESISTOR, FXD, CMPSN: same as 1A3R2	5-41
1A3R24		RESISTOR, FXD, CMPSN: MIL type RC0/GF103K	5-41
1A3R25		RESISTOR, FXD, CMPSN: same as 1A3R24	5-41
1A3R26		RESISTOR, VAR, WW: MIL type R112C2L203	5-42
1A3R27		RESISTOR, FXD, CMPSN: MIL type RC07GF151K	5-42
1A3R28		RESISTOR, FXD, CMPSN: MIL type RC20GF152K	5-41
1A3R29		NOT USED	
1A3R30		RESISTOR, FXD, CMPSN: MIL type RC07GF822K	5-41
1A3R31		RESISTOR, VAR, WW: MIL type RT12C2L502	5-41
1A3R32		RESISTOR, FXD, CMPSN: MIL type RC07GF271K	5-41
1A3R33		RESISTOR, FXD, CMPSN: MIL type RC20GF561K	5-41
1A3R34		NOT USED	
1A3R35		RESISTOR, FXD, CMPSN: same as 1A3R2	5-41
1A3R36		RESISTOR, FXD, CMPSN: MIL type RC07GF823K	5-41
1A3R37		NOT USED	
1A3R38		RESISTOR, FXD, CMPSN: same as 1A3R1	5-42
1A3R39		NOT USED	
1A3R40		RESISTOR, FXD, CMPSN: same as 1A3R33	5-41
1A3R41		RESISTOR, FXD, CMPSN: same as 1A3R4	5-42
1A3R42		RESISTOR, FXD, CMPSN: MIL type RC07GF223K	5-41
1A3R43		RESISTOR, FXD, CMPSN: same as 1A3R16	5-42
1A3R44		RESISTOR, FXD, CMPSN: same as 1A3R18	5-41
1A3R45		RESISTOR, FXD, CMPSN: MIL type RC07GF332K	5-41
1A3R46		RESISTOR, VAR, WW same as 1A3R31	5-41
1A3RT1		RESISTOR, THRM: 100 ohms, 10%; mfr 10646 part no 997F18	5-41
AUDIO AM	PLIFIER		
1A4		AMPLIFIER, AF: 0.5 W pwr output; mfr 13499 part no 528-0087-015	5-37
1A4C1		CAPACITOR, FXD, ELCTLT: 2.2UF, ±20%, 35V; mfr 56289	5-43
		part no 150D225X0035B2	
1A4C2		CAPACITOR, FXD, ELCTLT: 47UF, +20%, 20V; mfr 56289	5-43
		part no 150D476X0020R2	0.0
14403		CAPACITOR EXD ELCTLT: 15LIE $\pm 20\%$ 20V/ mfr 56289	5-43
171400		part no 150D156Y0020B2	0 40
10404		Part 10 150D 150A0020D2	E 42
17404		100000, 100, 100, 100000, 100, 100, 300, 100, 000000, 100, 1	0-40
1 4 4 0 5			E 40
1A4C5		UAPAUTUK, FXD, ELUTET: Same as 1A402	5-43
TA4C6			
thru			
1A4C10			
ORIGINAL			

## NAVSHIPS 0967-872-5020 TM-03816B-35/2 TM11- 5820-510-35-1

# TABLE 6-2. (Continued)

#### AUDIO AMPLIFIER

REF DESIG	NOTES	NAME AND DESCRIPTION	FIG NO
1A4C11		CAPACITOR, FXD, ELCTLT: 22UF, ±20%, 35V; mfr 56289 part no 150D226X0035R2	5-43
1A4C12 1A4C13		NOT USED	
1A4C14 1A4H1		CAPACITOR, FXD, ELCILI: same as 1A4C11 SCREW, SHOULDER, NO 1 CRES, passivate finish; 8-32 UNC-2A thd, 0.127 in hex, 0.270 in dia, 3.546 in lg; mfr 13499	5-43 5-43
1A4J1 1A4J2		JACK, TIP: WHT; mfr 98291 part no SKT41WHITE JACK, TIP: same as 1A4J1	5-43 5-43
1A4L1		COIL, RF: MS type MS16225-5	5~43
1A4L2		COIL, RF: 50MH; mfr 13499 part no MP206-14B	5-43
1A4MP1		COVER, AMPL: al, chromate dip finish, 0.032 in by 3.209 in by 3.437 in; mfr 13499 part no 548-7001-002	5-43
TA4MPZ		3.250 in, incl 33 terminals and 4 transistor holders; mfr 13499	5-43
1A4P1		CONNECTOR: dielectric; straight shape; 0.421 in by 0.484 in by	5-43
1A4Q1		TRANSISTOR: MIL-T-19500/69B (NAVY) type 2N338	5-43
1A4Q2		TRANSISTOR: MIL-T-19500/207 (SIGC) type 2N1481	5-43
1A4Q3		TRANSISTOR: MIL-T-19500/74 (NAVY) type 2N656	5-43
1A4Q4		IRANSISTOR: same as 1A4Q3	5-43
1A4R1 1A4R2		RESISTOR, FXD, CMPSN: MIL-R-11 type RC0/GF103K	5-43
1A4R2		NOT LISED	5-45
1A4R4		RESISTOR EXD CMPSN <sup>•</sup> MII -R-11 type RC07GE222K	5-43
1A4R5		RESISTOR, FXD, CMPSN: MIL-R-11 type RC07GF471K	5-43
1A4R6		RESISTOR, FXD, CMPSN: MIL-R-11 type RC07GF562K	5-43
1A4R7		RESISTOR, FXD, CMPSN: MIL-R-11 type RC07GF102K	5-43
1A4R8		RESISTOR, FXD, CMPSN: MIL-R-11 type RC07GF472K	5-43
1A4R9		RESISTOR, FXD, CMPSN: MIL-R-11 type RC07GF331K	5-43
1A4R10		NOT USED	5-43
1A4R11 1Δ4R12		NOTUSED	
1A4R13		NOT USED	
1A4R14		RESISTOR, FXD, CMPSN: MIL-R-11 type RC07GF221K	5-43
1A4R15		NOT USED	
1A4R16		NOT USED	
1A4R17		RESISTOR, FXD, CMPSN: same as 1A4R2	5-43
1A4R18		RESISTOR, FXD, CMPSN: same as 1A4R2	5-43
1A4R19		RESISTOR, VAR, WW 10K, $\pm 5\%$ , 1W; mfr 80294 part no 224L1-103	5-43
1A4R20 1A4R21		RESISTOR, VAR, WW. Same as 1A4K19 RESISTOR EXD CMPSN: MIL-R-11 type RC20GE101K	5-43
1A4R22		RESISTOR FXD CMPSN: MIL-R-11 type RC020GF101R	5-43
1A4RT1		RESISTOR, THRM 1K at 25 deg C: 1.8K at 25 deg C, designed	5-43
		for ac, dc; mfr 10646 part no 997F17	
1A4T1		TRANSFORMER mfr 70674 part no A11874	5-43
1A4T2		TRANSFORMER mfr 70674 part no A11873	5-43

ORIGINAL

6-14

### SPECTRUM GENERATOR MODULE

REF DESIG	NOTES	NAME AND DESCRIPTION	FIG NO
1A5 1A5C1		SPECTRUM GENERATOR: mfr 13499 part no 528-0373-007 CAPACITOR, FXD, CER DIEL: 1000PF, GMV, 500V, mfr 72082 part no 2465000W5T0102P	5-37 5-79
1A5C2 1A5C3 1A5C4		CAPACITOR, FXD, CER DIEL: same as 1A5C1 CAPACITOR, FXD, CER DIEL: MIL-C-20D type CC22CH060C CAPACITOR, FXD, CER DIEL: 1000PF, M20%P100%, 500V mfr	5-79 5-79 5-79
1A5C5 1A5C6		CAPACITOR, FXD, CER DIEL: MIL-C-20D type CC20CH040C CAPACITOR, FXD, MICA DIEL: 470PF, <u>+</u> 5%, 500V; mfr 72136 part no DM15E471500W/V4CR	5-79 5-79
1A5C7		CAPACITOR, FXD, CER DIEL: same as 1A5C4	5-79
1A5C9 1A5C10		CAPACITOR, FXD, MICA DIEL: MIL type CM05E200J03 CAPACITOR, VAR, GLASS DIEL 0.8PF to 4.5PF; mfr 73899 part no VC21GY	5-79 5-79
1A5C11		CAPACITOR, FXD, CER DIEL: 2.0PF to 1/4PF, 500V, mfr 72982	5-79
1A5C12 1A5C13		CAPACITOR, FXD, CER DIEL: same as 1A5C1 CAPACITOR, FXD, CER DIEL: 20PF, <u>+</u> 5%, 500V, mfr 72982 part no 331026 C0H0200 J	5-79 5-79
1A5C14 1A5C15 1A5C16 1A5C17 1A5C18		CAPACITOR, FXD, CER DIEL: same as 1A5C1 CAPACITOR, VAR, GLASS DIEL: same as 1A5C10 CAPACITOR, FXD, CER DIEL: same as 1A5C11 CAPACITOR, FXD, CER DIEL: same as 1A5C1 CAPACITOR, FXD, CER DIEL: same as 1A5C1 CAPACITOR, FXD, CER DIEL 400PF, M0%P100%, 360V, mfr	5-79 5-79 5-79 5-79 5-79 5-79
1A5C19 1A5C20 1A5C21		CAPACITOR, FXD, CER DIEL: same as 1A5C13 CAPACITOR, VAR, GLASS DIEL: same as 1A5C10 CAPACITOR, FXD, CER DIEL: 1.5PF to 1/4PF, 500V, mfr 72982	5-79 5-79 5-79
1A5C22 1A5C23		CAPACITOR, FXD, MICA DIEL: MIL type CM05E510J03 CAPACITOR, FXD, CER DIEL: same as 1A5C1	5-79 5-79
1A5C25 1A5C25 1A5C26 1A5C21 1A5C28 1A5C29 1A5C30		CAPACITOR, FXD, CER DIEL: same as 1A5C18 CAPACITOR, FXD, CER DIEL: same as 1A5C18 CAPACITOR, FXD, CER DIEL: same as 1A5C1 CAPACITOR, FXD, CER DIEL: same as 1A5C1 CAPACITOR, FXD, CER DIEL: same as 1A5C18 CAPACITOR, FXD, CER DIEL: 470PF, 20%, 500V, mfr 56289	5-79 5-79 5-79 5-79 5-79 5-79 5-79
1A5C31 1A5C32 1A5C33 1A5C34 1A5C35		NOT USED NOT USED CAPACITOR, FXD, CER DIEL: same as 1A5C1 NOT USED NOT USED	5-79
1A5C36 thru 1A5C41 1A5C42 1A5C43		CAPACITOR, FXD, CER DIEL: same as 1A5C18 NOT USED NOT USED	5-79

### SPECTRUM GENERATOR MODULE

REF	NOTES	NAME AND DESCRIPTION	FIG NO
DESIG			
1A5C44		NOT USED	F 70
1A5C45		LAPACITOR, FXD, MICA DIEL: Same as TA5C6	5-79
		NUT USED SEMICOND DEV/ICE MIL S 10500/188 type 1N251	5 70
		CONTACT ELEC: ben/lium con gold-plated finish: 0.234 in	5-79
		by 0.499 in by 1.531 in $o/a$ dim: mfr 13499 part no 539-5340-003	5-13
1A5E2		CONNECTOR: mfr 71468 part no DM53740-5000	5-79
1A5E3		CONTACT, ELEC bervilium cop, gold-plated cont surface, 0.484	5-79
		in dia by 0.040 in h o/a dim, mfr 13499 part no 544-7455-003	
1A5E4		CONTACT, ELEC cop, gold plated; 1-1/16 dia by 1/4 in h;	5-79
		mfr 13499 part no 544-8407-002	
1A5E5		ROTOR, ELEC SWITCH: mfr 13499 part no 549-3809-002	5-79
1A5E6		PRINTED CIRCUIT BOARD: plstc, cop clad, 2.312 in dia by 0.031 in	5-79
		thk o/a dim; mfr 13499 part no 549-3787-003	F 70
1A5E7		CONTACT, ELEC beryllium cop, goldplated finish, 0.562 in by	5-79
1450		0.087 IN DY 0.087 IN MILE 13499 PAIL NO 548-7893-003	5 70
		STATOR SOLDERED: 0.687 in by 1.125 in by 1.750 in 0/2 dim	5-79
IAJE3		mfr 13499 part no 544-8481-002	5-13
1A5E10		ROTOR. SOLDERED: 1.125 in dia by 0.562 in $o/a$ dim. mfr 13499	5-79
		part no 544-8482-002	
1A5E11		CONTACT, ELEC: beryllium cop, gold plated finish; 0.531 in	5-79
		dia by 0.050 in h o/a dim, mfr 13499 part no 547-0797-003	
1A5E12		NOT USED	
1A5E13		COIL ASSEMBLY, RF: 18 RF coils mounted on printed circuit	5-79
1 1 5 5 1 1		DOATO; MIL 13499 PAIL NO 549-3810-003	
		NOTUSED	
1A5E16		PRINTED CIRCUIT BOARD: plstc. cop clad: 0.031 in by 2.109 in	5-79
17 10 2 10		by 2.289 in o/a dim. mfr 13499 part no 549-3779-004	010
1A5H1		WASHER, FINISH cop, alloy-plated finish; 0.091 in ID countersunk	5-79
		82 deg. 0.212 in OD, 0.062 in thk, mfr 13499 part no 545-6590-002	
1A5H2		SCREW, EXTERNALLY RELIEVED BODY alloy stl, zinc-plated;	5-79
		hex skt cap screw 8-32 NC-2 thd, 3/8 in lg; mfr 13499 part no	
		544-8109-002	F 70
TASH3		NUT, PL, HEX CRES, passivate linish; 0.363 in nex,0.062 in thk,	5-79
14544		WASHER RECESSED: all chromate din 0.187 in ID 0.438 in OD	5-79
170114		0.050 in the mfr 13499 part no 544-8444-002	5-13
1A5H5		NUT. SLV CRES, passivate finish: 0.375 in hex: 0.187 in lg.	5-79
		0.2499 in dia, mfr 13499 part no 544-8446-002	
1A5J1		JACK, TIP BRN, mfr 98291 part no SKT-41BROWN	5-79
1A5J2		JACK, TIP same as 1A5J1 p/o 1A5MP24	5-79
1A5L1		COIL, RF 9 turns; single layer wound, no 26 AWG, polyurethane	5-79
		Insulation, 40 ohms DC res; color coded red and orn; mfr 13499	
14510		part no 549-3839-004	F 70
TAOLZ		insulation 40 obms DC rest color coded red, mfr 13400 part po	5-79
		549-3838-004	

# TABLE 6-2. (Continued)

## SPECTRUM GENERATOR MODULE

REF	NOTE	NAME AND DESCRIPTION	FIG NO
DESIG	S		
14513		COIL RE: 8 turns: single laver wound no 26 AWG: polyurethane	5-79
INOLO		insulation: 40 obes DC res color coded bro and wht: mfr 13499	070
		nation 540 2927 004	
1 / 51 /		COLL BE's Sturge single lover wound no 26 AWC polyurothone	F 70
TA3L4		COL, KF. 8 turns, single layer wound, ho 28 AWG, polyurethane	5-79
		Insulation, 40 onms DC res, color coded brn and vio, mr 13499	
		part no 549-3836-004	
1A5L5		COIL, RF: 7 turns, single layer wound, no 26 AWG; polyurethane	5-79
		insulation; 40 ohms DC res; color coded brn and blu, mfr 13499	
		part no 549-3835-004	
1A5L6		COIL, RF: 7 turns, single layer wound, no 26 AWG, polyurethane	5-79
		insulation. 40 ohms DC res. color coded brn and grn. mfr 13499	
		part no 549-3834-004	
1A5I 7		COIL RE: 7 turns: single layer wound: no 26 AWG, polyurethane	5-79
17 10 27		insulation 40 obes DC res color coded bro and vel mfr 13499	010
		not no 549-3833-004	
1/15/8		COLL BE 6 turns: single layer wound no 26 AWG polyurethane	5-70
IAJEO		insulation 40 obms DC ros: color coded bro and orn mfr 12400	5-13
		not action, 40 on this DC res, color coded bin and on, nin 15439	
11510			F 70
TASL9		COIL, RF: 9 turns, single layer wound; no 26 AWG; polyurethane	5-79
		Insulation, 40 onms DC res, color coded brn and red, mir 13499	
		part no 549-3831-004	
1A5L10		COIL, RF:8 turns, single layer wound, type, no 26 AVVG, polyurethane	5-79
		insulation, 40 ohms DC res, color coded brn, mfr	
		13499 part no 549-3830-004	
1A5L11		COIL, RF :8 turns single layer wound, no 26 AWG; polyurethane	5-79
		insulation, 40 ohms DC res, color coded wht, mfr 13499 part no	
		549-3829-004	
1A5L12		COIL, RF: 7 turns single layer wound type, no 26 AWG,	5-79
		polyurethane insulation, 40 ohms DC res; color coded vio; mfr	
		13499 part no 549-3828-004	
1A5L13		COIL, RF: 7 turns single laver wound, no 26 AWG, polyurethane	5-79
		insulation 40 ohms DC res color coded blu mfr 13499	
		nart no 549-3827-004	
1A5I 14		COll RF: 7 turns: single laver wound no 26 AWG polyurethane	5-79
		insulation: 40 ohms DC res. color coded arn. mfr 13499	010
		nati no 5/9-3826-00/	
1/5/15		COLL BE: 6 turns: single layer wound no 26 AWG: polyurethane	5-70
IAJEIJ		incultion: 0 turns, single layer would, ho 20 AWG, polyderiane	5-75
		nort no 540 2825 004	
4 4 5 1 4 6		COLL DE Stores circle lever would be 20 MMC achurethese	F 70
TASLIG		COIL, RF 6 turns, single layer wound, no 26 AWG, polyurethane	5-79
		Insulation, 40 online DC les, color coded om, mit 13499	
		part no 549-3824-004	
1A5L17		COIL, RF: 6 turns, single layer wound, no 26 AVVG, polyurethane	5-79
		Insulation, 40 ohms DC res; color coded red; mfr 13499	
		part no 549-3823-004	
1A5L18		COIL, RF: 6 turns, single layer wound, no 26 AWG, polyurethane	5-79
		insulation, 40 ohms DC res, color coded brn; mfr 13499	
		part no 549-3822-004	
1A5L19		COIL, RF: MS type MS18130-5	5-79
1A5L20		NOTUSED	

## NAVSHIPS 0967-872-5020 TM-03816B-35/2 TM11-5820-510-35-1

### AN/PRC-41A PARTS LIST

# TABLE 6-2. (Continued)

### SPECTRUM GENERATOR MODULE

REF DESIG	NOTE S	NAME AND DESCRIPTION	FIG NO
1A5L21 1A5L22		COIL, RF: MIL-C-15305 type LT4K030 COIL, RF: same as 1A5L21	5-79 5-79
thru 1A5L25 1A5L26 1A5L27		NOT USED CHOKE, BIFILAR: 0.125 in dia by 0.484 in lg o/a dim, mfr 13499	5-79
1A5L28		CHOKE, BIFILAR: same as 1A5L27	5-79
1A5L29 1A5L30 1A5L31		NOT USED COIL, RF: same as 1A5L21 NOT USED	5-79
1A5L32 1A5L33 1A5L34 1A5L35		COIL, RF: same as 1A5L21 COIL, RF: same as 1A5L21 NOT USED NOT USED	5-79 5-79
1A5L36 1A5L37 thru		COIL, RF: same as 1A5L21 NOT USED	5-79
1A5L41 1A5L42 1A5L43 thru		BEAD: mfr 02114 part no 56-590-65-3B BEAD: same as 1A5L42	5-79 5-79
1A5L61 1A5L62 thru		NOT USED	
1A5L65 1A5L66 1A5L67 1A5L68		COIL, RF: MS type MS16225-7 NOT USED	5-79
1A5L68 1A5L69 1A6I70 1A5L71 thru		COIL, RF: MIL-C-15305 type LT4K027 COIL, RF: same as 1A5I69 BEAD: same as 1A5L42	5-79 5-79 5-79
1A5L74 1A5MP1 1A5MP2 1A5MP3 1A5MP4 1A5MP5 1A5MP6		RING: mfr 78189 part no 213-141216-00-2303 BEARING: mfr 83086 part no SR156PPK28-7 BEARING: mfr 83086 part no SFR156PPK28-7 RING: MS type MS16633-1018 RING: MS type MS16632-1018 COUPLING HAL, SHAFT: sst, passivate finish, 0.3125 in OD, 0.106 in thk flange, 0.289 in Ig o/a, 0.420 in Ig of bore, mfr 13499 part no 540-8152-002	5-79 5-79 5-79 5-79 5-79 5-79 5-79
1A5MP7		COLLAR, SHAFT: al chromate dipped; 0.228 in ID, 0.375 in OD,	5-79
1A5MP8		POST, ELEC-MECH EQPT: al, chromate dipped, hex cross sectional shape, 0.750 in lg of post, 0.187 in w across flats; 2-56	5-79
1A5MP9		INSERT, FLEX COUPLING: plstc, 0.938 in dia by 0.185 in lg, mfr 13499 part no 548-7038-002	5-79

# TABLE 6-2. (Continued)

### SPECTRUM GENERATOR MODULE

REF	NOTE	NAME AND DESCRIPTION	FIG NO
DESIG	S		
1A5MP10		HOUSING, BEARING: CRES, passivate finish; 0.234 in ID to	5-79
1A5MP11		part no 544-8408-002 COLLAR, SHAFT, NO 2: brs, 0.228 in ID, 0.375 in OD, 0.164 in w,	5-79
1A5MP12		POST, CHASSIS, NO 1: brs, sil pl, 0.186 in by 0.186 in by 2.938 in,	5-79
1A5MP13		GEAR, SPUR-CRES; 0.531 in OD by 0.265 in lg o/a dim, mfr 13/49 part no 5/1/3/19/002	5-79
1A5MP14		GEAR, SPUR: ph brz, 0.781 in dia by 0.156 in lg o/a dim, mfr 13499 part no 544-8445-002	5-79
1A5MP15		INSULATOR, PLATE: plstc, 5/8 in by 1-1/16 in, mfr 13499 part no 547-0785-002	5-79
1A5MP16		COVER, GENERATOR SET: al; 0.032 in by 2.359 in by 2.921 in o/a dim mfr 13499 part no 544-8470-002	5-79
1A5MP17		COVER, GENERATOR: al; 0.125 in by 3.109 in by 3.625 in o/a dim, mfr 13499 part no 549-3791-003	5-79
1A5MP18		COVER, GENERATOR SET: al, 0.375 in by 3.109 in by 3.625 in o/a dam: mfr 13499 part no 548-7752-003	5-79
1A5MP19		COVER, GENERATOR: al, chromate dip, 0.422 in by 2.329 in by 3 156 in: mfr 13499 part no 548-7751-003	5-79
1A5MP20		COVER, GENERATOR: al, 0.040 in by 2.359 in by 2.437 in o/a dim, mfr 13499 part no 549-3802-003 c/o 145 l1	5-79
1A5MP21		PLATE, MOUNTING, SPECTRUM: al, 0.489 in by 2.438 in by 4.437 in o/a dim; mfr 13499 part no 544-8514-004, c/o 1A5MP22,,	5-79
1A5MP22 1A5MP23		BEARING: mfr 70417 part no F347-4 MILL6085A, p/o 1A5MP21 PIN, LOCATING: brs, chemical polish, 0.1249 in ID, 0.1867 in OD,	5-79 5-79
1A5MP24		0.344 in Ig, mfr 13499 part no 544-0277-002, p/o 1A5MP21 PLATE, RETAINING, BEARING: al, 0.452 in by 1.871 in by	5-79
1A5MP25		SHAFT, STR, OSC: CRES, passivate finish; 0.1873 in dia, 2.093 in	5-79
1A5MP26		COLLAR, PRINTED CIRCUIT BOARD: al, chromate dip; 7/16 in	5-79
1A5MP27		INSULATOR, PLATE: plstc, 0.136 in ID, 0.250 in OD, 0.125 in thk,	5-79
1A5MP28		PLATE, CHASSIS: brs, 0.406 in by 2.491 in by 2.352 in o/a dim, mfr 13499 part no 548-7681-003	5-79
1A5MP29		GEARSHAFT, SPUR: brs, 0.968 in dia by 3.203 in Ig o/a dim, mfr 13499 part no 544-8451-002	5-79
1A5MP30 1A5MP31		CHASSIS, ELEC EQPT: mfr 13499 part no 549-3817-004 PLATE, COVER: tees, sil pl, 3/4 in by 27/32 in; mfr 13499	5-79 5-79
1A5MP32		PLATE, ELEC SHIELD: brs, 0.381 in by 2.359 in by 2.796 in	5-79
1A5MP33		GEAR, SPUR: ph brz, 1.031 in dia by 0.265 in lg o/a dim, mfr 13499 part no 544-8453-002	5-79
1A5P1 1A5R1 1A5R2		CONNECTOR: mfr 71468 part no DAM11W1PC27 RESISTOR, FXD, CMPSN: MIL-R-11 type RC42GF333K RESISTOR, FXD, CMPSN: same as 1A5R1	5-79 5-79 5-79

### SPECTRUM GENERATOR MODULE

REF	NOTE	NAME AND DESCRIPTION	FIG NO
1A5R3 1A5R4 1A5R5 1A5R6 1A5R6 1A5R7 1A5R8	0	RESISTOR, FXD, CMPSN: MIL-R-11 type RC07GF104K RESISTOR, FXD, CMPSN: MIL-R-11 type RC07GF103K RESISTOR, FXD, CMPSN: MIL-R-11 type RC07GF681K RESISTOR, FXD, CMPSN: MIL-R-11 type RC07GF153K RESISTOR, FXD, CMPSN: MIL-R-11 type RC07GF331K NOT USED	5-79 5-79 5-79 5-79 5-79 5-79
1A5R9 1A5R10 1A5R11 1A5R12		NOT USED RESISTOR, FXD, CMPSN: same as 1A5R5 RESISTOR, FXD, CMPSN: MIL-R-11 type RC07GF223K	5-79 5-79
1A5K12 1A5R13 1A5R14 1A5R15 1A5V1 1A5V2 1A5V3 1A5V4 1A5V4 1A5XV1 1A5Y2 1A5Y1 1A5Y2 1A5Y3 1A5Y4 1A5Y5 1A5Y6 1A5Y7 1A5Y10 1A5Y10 1A5Y11 1A5Y12 1A5Y13 1A5Y14 1A5Y15 1A5Y16 1A5Y17 1A5Y18		NOT USED RESISTOR, FXD, CMPSN: MIL-R-11 type RC07GF224K RESISTOR, FXD, CMPSN: MIL-R-11 type RC07GF101K RESISTOR, FXD, CMPSN: MIL-R-11 type RC07GF471K ELECTRON TUBE: MIL-E-1 type 7077 ELECTRON TUBE: same as 1A5V1 ELECTRON TUBE: same as 1A5V3 SOCKET: mfr 04435 part no 86-001 SOCKET: same as 1A5XV1 CRYSTAL UNIT, QTZ: 66.66667 MHz, mfr 00136 part no 289-1986-00 CRYSTAL UNIT, QTZ: 70.0000 MHz; mfr 00136 part no 289-1987-00 CRYSTAL UNIT, QTZ: 76.6667 MHz, mfr 00136 part no 289-1987-00 CRYSTAL UNIT, QTZ: 76.6667 MHz; mfr 00136 part no 289-1989-00 CRYSTAL UNIT, QTZ: 76.6667 MHz; mfr 00136 part no 289-1989-00 CRYSTAL UNIT, QTZ: 80.00000 MHz; mfr 00136 part no 289-1998-00 CRYSTAL UNIT, QTZ: 80.00000 MHz; mfr 00136 part no 289-1998-00 CRYSTAL UNIT, QTZ: 80.00000 MHz; mfr 00136 part no 289-1998-00 CRYSTAL UNIT, QTZ: 80.00000 MHz; mfr 00136 part no 289-1998-00 CRYSTAL UNIT, QTZ: 72.50000 MHz; mfr 00136 part no 289-1993-00 CRYSTAL UNIT, QTZ: 72.50000 MHz, mfr 00136 part no 289-1993-00 CRYSTAL UNIT, QTZ: 72.50000 MHz, mfr 00136 part no 289-209-00 CRYSTAL UNIT, QTZ: 75.00000 MHz; mfr 00136 part no 289-2008-00 CRYSTAL UNIT, QTZ: 77.50000 MHz; mfr 00136 part no 289-2009-00 CRYSTAL UNIT, QTZ: 77.50000 MHz; mfr 00136 part no 289-2010-00 CRYSTAL UNIT, QTZ: 77.50000 MHz; mfr 00136 part no 289-2010-00 CRYSTAL UNIT, QTZ: 85.00000 MHz; mfr 00136 part no 289-2010-00 CRYSTAL UNIT, QTZ: 87.50000 MHz; mfr 00136 part no 289-2011-00 CRYSTAL UNIT, QTZ: 87.50000 MHz; mfr 00136 part no 289-2012-00 CRYSTAL UNIT, QTZ: 87.50000 MHz; mfr 00136 part no 289-2012-00 CRYSTAL UNIT, QTZ: 87.50000 MHz; mfr 00136 part no 289-2012-00 CRYSTAL UNIT, QTZ: 87.50000 MHz; mfr 00136 part no 289-2012-00 CRYSTAL UNIT, QTZ: 87.50000 MHz; mfr 00136 part no 289-2013-00 CRYSTAL UNIT, QTZ: 87.50000 MHz; mfr 00136 part no 289-2014-00	5-79 5-79 5-79 5-79 5-79 5-79 5-79 5-79

### MODULATOR MODULE

146	MODULATOR RAD XMTR 5W max power output: 300 to 3500 Hz	5-37
17.0	80 obms input, 3000 obms output: 24 5V and 190V operating power	0.01
	of onins input, sooo onins odiput, 24.5 v and 190 v operating power,	
	1.437 in by 3.437 in by 3 562 in; mfr 13499 part no 528-0089-017	
1A6C1	CAPACITOR, FXD, ELCTLT: MIL type CSR13F565KL	5-44
1A6C2	CAPACITOR, FXD, ELCTLT: MIL type CSR13B566KL	5-44
1A6C3	CAPACITOR, FXD, PR-MYLAR DIEL: MIL type CQ09A1KF103K3	5-44
1A6C4	CAPACITOR, FXD, CER DIELL: MIL type CK14AX103M	5-44
1A6F1	PRINTED CIRCUIT BOARD: plstc: 0.062 in by 1.750 in by 3 in	5-44
171011	incl 54 tubeless: mfr 13499 part no 548-7014-005	
	1 101 04 (ubeless, 1111 10400 part 10 040-7014-000	
		1

### TABLE 6-2. (Continued)

### MODULATOR MODULE

REF	NOTE	NAME AND DESCRIPTION	FIG NO
DESIG	S		
1A6H1		WASHER glass cloth, silicone rbr; 0.086 in ID, 0.750 in OD, 0.050 in the mfr 13499 part no 547-2288-003	5-44
1A6H2		SCREW, SHOULDER, NO 2 CRES, passivate finish 8-32 UNC-2A thd, 0.127 in hex, 0.270 in dia 3.584 in lg, mfr 13499 part no	5-44
1A6H3		544-8223-002 SPACER, SLV al, chromate dip 0.031 in wall, 0.125 in lg, 0.156 in OD; mfr 13499 part no 541-5949-002	5-44
1A6J1		JACK, TIP: WHT, mfr 98291 part no SKT41WHITE	5-44
1A6J2		JACK, TIP same as 1A6J1	5-44
1A6MP1		BASE, X STR al, 11/16-24 ext thd, 0.265 in lg, 0.391 in lg o/a, mfr 13499 part no 548-7002-002	5-44
1A6MP2		CAP, XSTR al; 11/16-24 intl thd, 3/4 in dia by 7/16 in lg, mfr 13499 part no 548-7003-002	5-44
1A6MP3		INSULATOR, BUSHING: plstc, 0.313 in dia by 0.150 in lg; 0.177 in ID undercut to 0.096 m; mfr 13499 part no 548-7004-002	5-44
1A6MP4		COVER, MODULATOR al, chromate dip finish; 0.032 in by 3.209 in by 3.27 in mfr 13499 part no 548-7000-003	5-44
1A6P1		CONNECTOR mfr 71468 part no DEMM9P	5-44
1A6Q1		TRANSISTOR MIL-S-19500/180(SIGC) type 2N1486	5-44
1A6Q2		TRANSISTOR: same as 1A6Q1	5-44
1A6Q3		TRANSISTOR: same as 1A6Q1	5-44
1A6R1		RESISTOR, FXD, CMPSN MIL-R-11 type RC07GF101K	5-44
1A6R2		RESISTOR, VAR, WW: MIL-R-27208 type RT12C2L102	5-44
1A6R3		RESISTOR, FXD, CMPSN: MIL-R-11 type RC20GF682K	5-44
1A5R4		RESISTOR, FXD, CMPSN MIL-R-11 type RC07GF821K	5-44
1A6R5		RESISTOR, FXD, CMPSN MIL-R-11 type RC07GF470K	5-44
1A6R6		RESISTOR, FXD, WW: MIL-R-26 type RW69V431	5-44
1A6R7		RESISTOR, FXD, CMPSN same as 1A6R5	5-44
1A6R8		RESISTOR, FXD, CMPSN MIL-R-11 type RC20GF272K	5-44
1A6R9		RESISTOR, FXD, WW MIL-R-26 type RW69V8R2	5-44
1A5R10		RESISTOR, FXD, CMPSN MIL-R-11 type RC0/GF154K	5-44
1A6R11		RESISTOR, FXD, CMPSN-MIL-R-11 type RC0/GF153K	5-44
		REDIDIOR, FAD, UNIPON: Sallie as IAOK4	5-44 5-44
INOKII		トロンシーン (10040) 1000 (10000) 1000 (10000) 1000 (10000) 1000 (10000) 1000 (10000) 1000 (10000) 1000 (10000) 1000	5-44
1A6T1		TPANSEOPMEP mfr 73386 part no 38802	5-11
146T2		TRANSFORMER mfr $80008$ part no $F15279$	5-44
17012			5-44

## GUARD RECEIVER MODULE

1A7 1A7C1 1A7C2 1A7C3	RECEIVER, RADIO: mfr 13499 part no 528-0090-005 CAPACITOR, VAR, GLASS DIEL: 0.8PF to 8.0PF, 1000V, mfr 73899 part no VCJ705 CAPACITOR, FXD, CER DIEL: MIL type CC22CH100D NOT USED	5-37 5-48 5-48
1A7C4	CAPACITOR, FXD, CER DIEL: 7.0PF to 1/4PF, 500V; mfr 72982	5-48
1A7C5	part no 331026C0H0709C CAPACITOR, FXD, MICA DIEL: 1000PF, <u>+</u> 5%, 100V, mfr 72136 part no DM15E102J0100WV4CR	5-48

### **GUARD RECEIVER MODULE**

REF DESIG	NOTE S	NAME AND DESCRIPTION	FIG NO
147C6		CAPACITOR VAR GLASS DIFL: same as 1A7C1	5-48
1A7C7		CAPACITOR, FXD, CER DIEL: 6.0PF to 1/2PF, 500V, mfr	5-48
1A7C8		CAPACITOR, FXD, CER DIEL: 1000PF, GMV, 500V; mfr	5-48
1A7C9		CAPACITOR, FXD, CER DIEL: 2PF to 1/2PF, 500V, mfr 72982	5-48
1A7C10		CAPACITOR, FXD, MICA DIEL:: same as 1A7C5	5-48
1A7C11		CAPACITOR, FXD, CER DIEL: same as 1A7C8	5-48
147012		CAPACITOR, VAR, GLASS DIEL. Same as TA701	0-40 5 49
IA/CI3		part no 331026C0H0100D	J-40
1A7C14		CAPACITOR, FXD, CER DIEL: 0.5PF to 1/4PF, 500V, mfr 72982 part no 331026C0K0508C	5-48
1A7C15		CAPACITOR, FXD, CER DIEL: 7.0PF to 1/2PF, 500V, mfr 72982	5-48
1A7C16		CAPACITOR VAR GLASS DIEL same as 1A7C1	5-48
1A7C17		CAPACITOR, FXD, CER DIEL: same as 1A7C9	5-48
1A7C18		CAPACITOR, FXD, MICA DIEL: MIL type CM05E470J03	5-48
1A7C19		CAPACITOR, FXD, CER DIEL: MIL type CK14AX103M	5-48
1A7C20		CAPACITOR, FXD, CER DIEL; same as 1A7C19	5-48
1A7C21		CAPACITOR, FXD, CER DIEL: 18PF, <u>+</u> 10% 500V, mfr 72982	5-48
147022		CAPACITOR VAR GLASS DIEL same as 1A7C1	5-48
1A7C23		CAPACITOR EXD CER DIEL same as 1A7C13	5-48
1A7C24		CAPACITOR EXD CER DIEL same as 1A7C9	5-48
1A7C25		CAPACITOR FXD CER DIEL: MIL-C-20D type CC20T.1060D	5-48
1A7C26		CAPACITOR, FXD, CER DIEL: 1500PF, <u>+</u> 10%, 500V, mfr 56289	5-48
147027		CAPACITOR EXD MICA DIEL: MIL type CM05E200.103	5-48
147C28		CAPACITOR FXD, CFR DIEL: Mile type Omode200000	5-48
147C29		CAPACITOR EXD. CER DIEL: same as 1A7C19	5-48
1A7C30		CAPACITOR EXD CER DIEL same as 1A7C19	5-49
1A7C31		NOT USED	0.0
1A7C32		CAPACITOR, FXD, CER DIEL: 10PF to 1/2PF, 500V, mfr 72982	5-45
147033		CAPACITOR EXD CER DIEL: same as 1A7C19	5-49
1A7C34		CAPACITOR, FXD, CER DIEL: 27PF, <u>+</u> 10%, 500V, mfr 71590	5-45
147025			5_15
147030		CAPACITOR, FAD, CER DIEL, Sallie as TA7C10	5-45 5-46
147030		CAPACITOR, FAD, CER DIEL, Same as 147019	0-40 5 46
147037		CAPACITOR, FAD, CER DIEL, Sallie as 147010	5-40
147030		$\Box$ CAPACITOR FXD CER DIEL Same as $147013$	5-40
1470/0		CAPACITOR FXD CER DIFL same as 14700	5-45
147040		CAPACITOR EXD CER DIEL: same as 14700	5-45
147041		CAPACITOR FXD CER DIFL same as 147013	5-40
147042		CAPACITOR FXD CER DIFL: same as 147021	5-46
1A7C44		CAPACITOR FXD CFR DIFL: same as 1A7C8	5-45

#### GUARD RECIEVER MODULE

REF DESIG	NOTE S	NAME AND DESCRIPTION	FIG NO
1A7C45 1A7C46 1A7C47 1A7C48 1A7C49 1A7C50 1A7C51 1A7C52		CAPACITOR, FXD, CER DIEL: same as 1A7C8 CAPACITOR, FXD, CER DIEL: same as 1A7C19 CAPACITOR, FXD, CER DIEL: same as 1A7C21 CAPACITOR, FXD, CER DIEL: same as 1A7C19 CAPACITOR, FXD, CER DIEL: same as 1A7C9 CAPACITOR, FXD, CER DIEL: same as 1A7C8 CAPACITOR, FXD, CER DIEL: same as 1A7C19 CAPACITOR, FXD, CER DIEL: same as 1A7C19 CAPACITOR, FXD, CER DIEL: same as 1A7C19	5-46 5-46 5-46 5-46 5-48 5-45 5-45 5-46
1A7C53 1A7C54 1A7C55		NOT USED CAPACITOR, FXD, CER DIEL: same as 1A7C21 CAPACITOR, FXD, ELCTLT: 22UF, <u>+</u> 20%, 35V, mfr 56289	5-46 5-48
1A7C56		part no 150D226X0035R2 CAPACITOR, FXD, ELCTLT: 2.2UF, <u>+</u> 20%, 20V, mfr 56289	5-45
1A7C57		part no 150D225X0020A2 CAPACITOR, FXD, ELCTLT: 10UF, <u>+</u> 20%, 20V, mfr 56289 part no 150D106X0020B2	5-45
1A7C58 1A7C59		CAPACITOR, FXD, CER DIEL: CK12AX102M CAPACITOR, FXD, ELCTLT: 0.10UF, <u>+</u> 20%, 35V, mfr 56289	5-45 5-49
1A7C60		CAPACITOR, FXD, ELCTLT: 1.0UF, <u>+</u> 20%, 35V, mfr 56289	5-49
1A7C61 1A7C62 1A7C63 1A7C64 1A7C65 1A7C66 1A7C67 1A7C68 1A7C69 1A7C70 1A7C71 1A7C72		part no 150D105X0035A2 CAPACITOR, FXD, ELCTLT: same as 1A7C56 CAPACITOR, FXD, CER DIEL: same as 1A7C58 CAPACITOR, FXD, CER DIEL: same as 1A7C8 CAPACITOR, FXD, CER DIEL: same as 1A7C9 CAPACITOR, FXD, CER DIEL: same as 1A7C19 CAPACITOR, FXD, CER DIEL: same as 1A7C19 CAPACITOR, FXD, MICA DIEL: MIL type CM05FIOIJ03 CAPACITOR, FXD, CER DIEL: same as 1A7C19 CAPACITOR, FXD, CER DIEL: same as 1A7C19 CAPACITOR, FXD, CER DIEL: same as 1A7C58 CAPACITOR, FXD, CER DIEL: same as 1A7C58 CAPACITOR, FXD, CER DIEL: same as 1A7C58 CAPACITOR, FXD, CER DIEL: same as 1A7C26 CAPACITOR, FXD, ELCTLT: 4.7UF, ±20%, 35V, mfr 56289 part no 150D475X0035B2	5-49 5-45 5-48 5-48 5-45 5-45 5-48 5-48 5-48
1A7C73 1A7C74 1A7C75 1A7C76 1A7C77 1A7C78 *1A7CR1 *1A7CR1 *1A7CR1 *1A7CR1 *1A7CR1 1A7CR1 1A7CR2 1A7CR3 1A7CR3 1A7CR4		CAPACITOR, FXD, ELCTLT: same as 1A7C72 CAPACITOR, FXD, ELCTLT: same as 1A7C72 CAPACITOR, FXD, CER DIEL: MIL type CM05D181J03 CAPACITOR, FXD, CER DIEL: same as 1A7C32 CAPACITOR, FXD, CER DIEL: same as 1A7C32 CAPACITOR, FXD, CER DIEL: same as 1A7C58 CAPACITOR, FXD, CER DIEL: same as 1A7C56 SEMICOND DEVICE: MIL-S-19500/127 (NAVY) type 1N747A SEMICOND DEVICE: MIL-S-19500/127 (NAVY) type 1N749A SEMICOND DEVICE: MIL-S-19500/127 (NAVY) type 1N751A SEMICOND DEVICE: MIL-S-19500/127 (NAVY) type 1N753A SEMICOND DEVICE: MIL-S-19500/127 (NAVY) type 1N754A SEMICOND DEVICE: MIL-S-19500/127 (NAVY) type 1N754A SEMICOND DEVICE: MIL-S-19500/127 (NAVY) type 1N753A SEMICOND DEVICE: MIL-E-1 type 1N457 SEMICOND DEVICE: Same as 1A7CR3	5-48 5-47 5-47 5-47 5-45 5-49 5-49 5-49 5-49 5-49 5-49 5-49

## NAVSHIPS 0967-872-5020 TM-03816B-35/2 TM11-5820-510-35-1

# TABLE 6-2. (Continued)

### GUARD RECEIVER MODULE

REF	NOTE	NAME AND DESCRIPTION	FIG NO
DESIG	S		
147CR5		SEMICOND DEV/ICE: MIL-S-19500/115 (NAV/V) type 1N3027B	5-48
1A7CR6		SEMICOND DEVICE: mfr 03877 part no 1N816	5-45
1A7CR7		SEMICOND DEVICE: MIL-D-19500/188 type 1N251	5-48
1A7FL1		FILTER: mfr 00136 part no 2B5	5-45
1A7FL2		FILTER: mfr 01121 part no SMFB2A2	5-48
1A7H1		NUT, PL, HEX: al, chromate dip; 0.1875 in hex; 4-40 UNC-2B thd,	5-47
14742		0.643 IN IG, MIT 13499 PAR NO 548-7153-002	5 45
IATIZ		0.615 in lg. mfr 13499 part no 548-7171-002	5-45
1A7H3		SCREW, SHOULDER: CRES, passivate finish 0.125 in ID, 0.250 in	5-48
		OD, 3.578 in lg; mfr 13499 part no 548-7159-002	- 10
1A7J1		JACK, TIP: BRN: mfr 98291 part no SKT-41BROWN	5-48
1A7J2		JACK, TIP: RED: mir 98291 part no SK1-10RED	5-45
IA/LI		mfr 13499 part no 548-7156-002	5-47
1A7L2		COIL, RF: single layer wound type; 3 turns of no 18 AWG wire,	5-47
		mfr 13499 part no 548-7157-002	5 47
1A7L3			5-47
1A7L4 1A7L5		COIL, RF: Same as TA7L2	5-47
14716		COIL, RF: single laver wound type: 21 turns of no 36 AWG wire	5-47
IATEO		mfr 13499 part no 548-7155-002	5 47
1A7L7		COIL, RF: same as 1A7L2	5-47
1A7L8		COIL, RF: MIL type LT4K030	5-47
1A7L9		NOT USED	- 17
1A7L10		COIL, RF: single layer wound type; 7 turns of no 26 AWG wire;	5-47
1A7I 11		COll RF same as 1A7I 6	5-47
1A7L12		COIL, RF: same as 1A7L6	5-46
1A7L13		COIL, RF: same as 1A7L6	5-46
1A7L14		COIL, RF: same as 1A7L6	5-46
1A7L15		COIL, RF: same as 1A7L6	5-46
1A7L16		COIL, RF: single layer wound type; 15 turns of no 36 AWG wire;	5-45
1A7I 17		$COII  RE \cdot MII  type I T4K047$	5-47
1A7L18		COIL, RF: same as 1A7L17	5-45
1A7L19		COIL, RF same as 1A7L17	5-45
1A7L20		COIL, RF: same as 1A7L17	5-49
1A7L21		COIL, RF: same as 1A7L5	5-47
1A7L22		COIL, RF: same as 1A7L5	5-47
1A7MP1		GROMMET: mfr 93106 part no GB2156	5-48
1A7MP2		COVER, RCVR, NO 1: al, chromate dip, 0.063 in thk, 3.218 in dia,	5-45
1A7MP3		COVER, RCVR NO 2: al, chromate dip; 0.063 in thk, 3.218 in dia, 4.406 in lg: mfr 13499 part no 548-7578-003	5-48
1A7P1		CONNECTOR: mfr 71468 part no DAM11W1PC27	5-45
1A7Q1		TRANSISTOR, mfr 07688 part no 2N917	5-48
1A7Q2		TRANSISTOR. Same as 1A7Q1	5-48
1A7Q3		TRANSISTOR: same as 1A7Q1	5-48
1A7Q4		TRANSISTOR: same as 1A7Q1	5-48

#### GUARD RECIEVER MODULE

REF DESIG	NOTE S	NAME AND DESCRIPTION	FIG NO
1A7Q5 1A7Q6 thru		TRANSISTOR: MIL-S-19500/80A(SIGC) type 3N35 TRANSISTOR: same as 1A7Q5	5-48 5-46
thru 1A7Q9 1A7Q10 1A7Q11 1A7Q12 1A7Q13 1A7Q14 1A7Q15 1A7R1 1A7R2 1A7R3 1A7R4 1A7R5 1A7R6 1A7R7 1A7R8 1A7R7 1A7R8 1A7R7 1A7R8 1A7R10 1A7R10 1A7R10 1A7R11 1A7R12 1A7R13 1A7R14 1A7R15 1A7R16 1A7R17 1A7R18 1A7R16 1A7R17 1A7R20 1A7R21 1A7R23 1A7R23 1A7R23 1A7R24 1A7R25 1A7R26 1A7R27 1A7R28 1A7R28		TRANSISTOR: mfr 07688 part no. 2N706 TRANSISTOR: MIL-S-19500/111(SIGC) type 2N329A TRANSISTOR: MIL-S-19500/20(SIGC) type 2N706 TRANSISTOR: same as 1A7Q12 TRANSISTOR: same as 1A7Q12 TRANSISTOR: same as 1A7Q11 RESISTOR, FXD, CMPSN: MIL-R-11 type RC07GF103K RESISTOR, FXD, CMPSN: MIL-R-11 type RC07GF102K RESISTOR, FXD, CMPSN: MIL-R-11 type RC07GF102K RESISTOR, FXD, CMPSN: MIL-R-11 type RC07GF102K RESISTOR, FXD, CMPSN: same as 1A7R1 RESISTOR, FXD, CMPSN: same as 1A7R2 RESISTOR, FXD, CMPSN: same as 1A7R2 RESISTOR, FXD, CMPSN: MIL-R-11 type RC07GF122K RESISTOR, FXD, CMPSN: Same as 1A7R2 RESISTOR, FXD, CMPSN: MIL-R-11 type RC07GF152K RESISTOR, FXD, CMPSN: MIL-R-11 type RC07GF152K RESISTOR, FXD, CMPSN: MIL-R-11 type RC07GF152K RESISTOR, FXD, CMPSN: MIL-R-11 type RC07GF162K RESISTOR, FXD, CMPSN: MIL-R-11 type RC07GF163K RESISTOR, FXD, CMPSN: MIL-R-11 type RC07GF163K RESISTOR, FXD, CMPSN: Same as 1A7R3 RESISTOR, FXD, CMPSN: Same as 1A7R3 RESISTOR, FXD, CMPSN: Same as 1A7R4 RESISTOR, FXD, CMPSN: MIL-R-11 type RC07GF472K RESISTOR, FXD, CMPSN: Same as 1A7R1 RESISTOR, FXD, CMPSN: Same as 1A7R2 RESISTOR, FXD, CMPSN: Same as 1A7R1 RESISTOR, FXD, CMPSN: Same as 1A7R1 RESISTOR, FXD, CMPSN: Same as 1A7R1 RESISTOR, FXD, CMPSN: Same as 1A7R3 RESISTOR, FXD, CMPSN: Same as 1A7R3 RESISTOR, FXD, CMPSN: Same as 1A7R1 RESISTOR, FXD, CMPSN: Same as 1A7R1 RESISTOR, FXD, CMPSN: Same as 1A7R1 RESISTOR, FXD, CMPSN: Same as 1A7R3 RESISTOR, FXD, CMPSN: Same as 1A7R1 RESISTOR, FXD, CMPSN: Same as 1A7R3 RESISTOR, FXD, CMPSN: Same as 1	5-45 5-45 5-45 5-45 5-45 5-45 5-45 5-47 5-47
1A7R30 1A7R31 1A7R32		RESISTOR, FXD, CMPSN: same as 1A/R14 RESISTOR, FXD, CMPSN: same as 1A7R18 RESISTOR, FXD, CMPSN: same as 1A7R7	5-46 5-46 5-46
1A7R33 1A7R34 1A7R35 1A7R36 1A7R37 1A7R38		NOT USED RESISTOR, FXD, CMPSN: same as 1A7R3 RESISTOR, FXD, CMPSN: same as 1A7R16 RESISTOR, FXD, CMPSN: same as 1A7R14 RESISTOR, FXD, CMPSN: same as 1A7R18 RESISTOR, FXD, CMPSN: same as 1A7R27	5-46 5-46 5-46 5-46 5-46

ORIGINAL

Table 6-2

#### **GUARD RECEIVER MODULE**

REF	NOTE	NAME AND DESCRIPTION	FIG NO
DESIG	5		
1A7R39		RESISTOR, FXD, CMPSN: same as 1A7R27	5-49
1A7R40		RESISTOR, FXD, CMPSN: same as 1A7R3	5-46
1A7R41		RESISTOR, FXD, CMPSN: same as 1A7R16	5-46
1A7R42		RESISTOR, FXD, CMPSN: same as 1A7R14	5-46
1A7R43		RESISTOR, FXD, CMPSN: same as 1A7R18	5-46
1A7R44		RESISTOR, FXD, CMPSN: same as 1A7R7	5-46
1A/R45		RESISTOR, FXD, CMPSN: same as 1A/R2	5-45
1A/R46		RESISTOR, VAR, WW: 10K, <u>+</u> 10%, 0.8W; mfr 80294 part no	5-45
1A7R47		RESISTOR, FXD, CMPSN: MIL-R-11 type RC07GF154K	5-49
1A7R48		RESISTOR, FXD, CMPSN: same as 1Å7R3	5-45
1A7R49		RESISTOR, FXD, CMPSN: MIL-R-11 type RC07GF221K	5-45
1A7R50		RESISTOR, FXD, CMPSN: same as 1A7R1	5-49
1A/R51		RESISTOR, FXD, CMPSN.: same as 1A/R3	5-45
*1A/R52		RESISTOR, FXD, CMPSN: MIL-R-11 type RC0/GF4/2K	5-45
1A/KJZ *1A7D52		RESISTOR, FAD, CIVIPSIN. MILER-TT LYPE RCU/GF302K	5-45 5-45
*147R53		RESISTOR, FXD, CMFSN, MIL-R-11 type RC07GF392R	5-45
*1A7R53		RESISTOR FXD CMPSN: MIL R-11 type RC07GF392K	5-45
*1A7R53		RESISTOR, FXD, CMPSN: MIL-R-11 type RC07GF272K	5-45
*1A7R53		RESISTOR, FXD, CMPSN: MIL-R-11 type RC07GF272K	5-45
1A7R54		RESISTOR, FXD, CMPSN: same as 1A7R1	5-45
1A7R55		RESISTOR, FXD, CMPSN: same as 1A7R49	5-45
1A7R56		RESISTOR, FXD, CMPSN: same as 1A7R3	5-45
1A7R57		RESISTOR, FXD, CMPSN: same as 1A7R1	5-49
1A7R58		RESISTOR, FXD, CMPSN: same as 1A7R16	5-49
1A/R59		RESISTOR, FXD, CMPSN: MIL-R-11 type RC0/GF823K	5-45
1A7R60		RESISTOR, FXD, CMPSN: same as 1A7R16	5-45
147062		RESISTOR, FAD, CMPSIN. Same as 1A7R1	5-45 5-49
147R63		NOT LISED	5-49
*1A7R64		RESISTOR FXD CMPSN MIL-R-11 type RC07GF823K	5-49
1A7R64		RESISTOR, FXD, CMPSN: MIL-R-11 type RC07GF124K	5-49
1A7R64		RESISTOR, FXD, CMPSN: MIL-R-11 type RC07GF154K	5-49
*1A7R64		RESISTOR, FXD, CMPSN: MIL-R-11 type RC07GF104K	5-49
1A7R65		RESISTOR, FXD, CMPSN: same as 1Å7R2	5-47
1A7R66		RESISTOR, FXD, CMPSN: same as 1A7R21	5-47
1A7R67		RESISTOR, FXD, CMPSN: same as 1A7R14	5-45
1A/R68		RESISTOR, FXD, CMPSN: same as 1A7R14	5-45
1A7R09		NOT LISED	5-45
1A7TB2		TERMINAL BOARD, plete: incl 10 terminals, 0.062 in by 0.703 m	5-48
		by 1.687 in board dim, mfr 13499 part no 548-7181-003	0 40
1A7TB3		TERMINAL BOARD plstc; incl 5 terminals, 0.062 in by 1 in by	5-45
		1.125 in board dim, mfr 13499 part <i>no</i> 548-7150-003	<b>_</b>
1A71B4		I ERMINAL BOARD: plstc, glass cloth, 1/16 in by 1-5/32 m	5-45
1471		Dy 2-9/04 in Incl 23 terminals, mir 13499 part no 548-7581-003	5 47
		ON TOTAL UNIT, QTZ TTT.2200 HIC, HIII 00100 PAIL HO 290-9602-	5-47
*Chosen per	r requirem	ents	

# TABLE 6-2. (Continued)

### RF AND POWER AMPLIFIER MODULE

REF	NOTE	NAME AND DESCRIPTION	FIG NO
DESIG	S		
1A8		AMPLIFIER, RF: 225 to 400 MHz freq range, mfr 13499 part no 528-0091-015	5-37
1A8C1		CAPACITOR, FXD, CER DIEL: 470PF, <u>+</u> 20%, 500V; mfr 56289 part no 40C204A5	5-52
1A8C2		CAPACITOR, FXD, CER DIEL: 400PF, GMV at room temp, 360V. mfr 72982 part no 2404032W5P0401P	5-52
1A8C3 1A8C4		CAPÁCITOR, FXD, CER DIEL: same as 1A8C2 CAPACITOR, FXD, CER DIEL: same as 1A8C2	5-52 5-50
1A8C5		CAPACITOR, FXD, CER DIEL: same as 1A8C1	5-52
1A8C6 1A8C7		CAPACITOR, FXD, CER DIEL: MIL-C-20B type CC22CK1R5D CAPACITOR, VAR, GLASS DIEL 0.75PF to 3PF, 500V; mfr 73899	5-52 5-50
1A8C8		CAPACITOR, FXD, CER DIEL: same as 1A8C2	5-52
1A8C10		CAPACITOR, FXD, CER DIEL. same as TAOCT CAPACITOR, FXD, CER DIEL: same as 1A8C2	5-52 5-50
1A8C11		CAPACITOR, FXD, CER DIEL: same as 1A8C6	5-52
1A8C12		CAPACITOR, VAR, GLASS DIEL: same as 1A8C7	5-50 5-50
1A8C14		CAPACITOR includes grid plate Collins Radio Company	5-52
		part no 548-7077-002, insulator Collins Radio Company	
		part no 548-7102-003, bushing Collins Radio Company part	
1A8C15		CAPACITOR, FXD, CER DIEL: MIL-C-20B type CC22CH200J	5-52
1A8C16		CAPACITOR, FXD, CER DIEL: same as IA8C2	5-50
1A8C17 1A8C18		CAPACITOR, FXD, CER DIEL: MIL-C-20D type CC220K020D CAPACITOR VAR GLASS DIEL: same as 1A8C7	5-52 5-80
1A8C19		CAPACITOR, FXD, CER DIEL: same as 1A8C2	5-52
1A8C20		CAPACITOR includes grid plate Collins Radio Company	5-50
		part no 548-7077-002, insulator Collins Radio Company part	
		part no 548-7078-002	
1A8C21		CAPACITOR, VAR, GLASS DIEL: same as 1A8C7	5-50
1A8C22 1A8C23		CAPACITOR; Includes 1A8MP29 and 1A8MP30 and 1A8MP32	5-51 5-80
1A8C24		CAPACITOR, FXD, CER DIEL: same as 1A8C2	5-50
1A8C25		CAPACITOR. includes grid plate Collins Radio Company	5-50
		no 548-8570-003, insulator Collins Radio Company part no 548-7101-003, bushing Collins Radio Company part no	
1A8C26		CAPACITOR, FXD, CER DIEL: same as 1A8C1	5-51
1A8C27		CAPACITOR, VAR, AIR DIEL: single sect, 1.2PF to 5PF; mfr	5-50
148028		97137 part no 875001 CAPACITOR includes 1A8MP20 and 1A8MP30 and 1A8MP32	5-51
1A8C29		CAPACITOR, FXD, CER DIEL: same as 1A8C1	5-51
1A8C30		CAPACITOR, VAR, GLASS DIEL: 0.5PF t0 2PF. 700V; includes	5-80
1A8C31		CAPACITOR, FXD, CER DIEL: same as 1A8C2	5-50

## NAVSHIPS 0967-872-5020 TM-03816B-35/2 TM11-5820-510-35-1

## TABLE 6-2. (Continued)

### RF AND POWER AMPLIFIER MODULE

REF DESIG	NOTE S	NAME AND DESCRIPTION	FIG NO
1A8C32 1A8C33 1A8C34 1A8C35 1A8C36 1A8C37 1A8C38 1A8C39 1A8C40 1A8C41 1A8E1		CAPACITOR, FXD, CER DIEL: same as 1A8C2 CAPACITOR, FXD, CER DIEL. same as 1A8C2 CAPACITOR, FXD, CER DIEL: MIL-C-20D type CC22TJ040D CAPACITOR, FXD, CER DIEL: same as 1A8C17 CONTACT, ELEC: cop, gold pl; 1-1/16 in dia by 1/4 in h,	5-50 5-80 5-52 5-52 5-52 5-51 5-52 5-80 5-50 5-51 5-51
1A8E2		mfr 13499 part no 544-8407-002 ADAPTER, HEATER: 0.422 in dia by 0.234 ln lg; mfr 13499	5-50
1A8H1		part no 540-8750-002 SCREW, SHOULDER, NO 2 CRES, passivate finish 8-32 UNC-2A thd, 0.127 in hex, 0.270 in dia, 3 584 in lg; mfr 13499 part no	5-51
1A8H2		WASHER, FINISHING: cop, alloy-plated finish, 0.091 in ID, csk 82 deg. 0.212 in OD, 0.062 in thk; mfr 13499	5-52
1A8H3		WASHER, FLAT: cop, bright alloy 0.125 in ID, 0.250 in OD,	5-51
1A8H4		NUT, SPCL: al, chromate dip, 4-40 UNC-2B thd, 5/16 in by 2/8 in mfr 12409 part no 5/48 7558 002	5-51
1A8H5		SETSCREW: stl, cd pl; fluted multiple spline cut point 2-56	5-51
1A8H6		SCREW, MACH: nylon, slotted filh, 4-40 UNC-2A thd, 3/8 in Ig, mfr 13499 part no 330-2248-00	5-51
1A8H7		NUT, HEX: CRES, passivate finish, 7/16-32 UNC-2B thd, 0.5625 In hey 0.062 in the mr 13499 part no 544-8410-002	5-80
1A8J1 1A8J2 thru 1A8J11		JACK TIP: WHT: mfr 98291 part no. SKT5BCWHITE JACK, TIP: same as 1A8J1	5-80 5-80
1A8K1		RELAY, AMT: 2C cont arrangement; 2 amp at 28 Vdc or 115 Vac; 550 ohms dc resistance, single layer wound; mfr 01526 part no 3SAE2053A2	5-80
1A8L1 1A8L2 1A8L3 1A8L4 1A8L5 1A8L6 1A8L7 1A8L8 1A8L9 1A8L9 1A8L10 1A8L11 1A8L12 1A8L13		COIL, RF: MIL type LT5K027 COIL, RF: MIL type LT4K030 COIL, RF MIL type LT4K029 COIL, RF: same as 1A8L2 COIL, RF: same as 1A8L3 COIL, RF: same as 1A8L2 COIL, RF: same as 1A8L2	5-50 5-50 5-52 5-52 5-52 5-52 5-52 5-50 5-51 5-50 5-51 5-50 5-51 5-50 5-52
# TABLE 6-2. (Continued)

### RF AND POWER AMPLIFIER MODULE

REF	NOTE	NAME AND DESCRIPTION	FIG NO
DESIG	S		
1A8L14 1A8L15		COIL, RF: same as 1A8L13 COIL RF: same as 1A8L13	5-52 5-52
1A8L16		COIL, RF: same as 1A8L2	5-50
1A8L17		COIL, RF: same as 1A8L2	5-50
1A8MP1		RING, plstc: 0.364 in ID, 0.504 in OD, 0.070 in thk, mfr 78189 part no 213-141216-00-2303	5-51
1A8MP2		BEARING: mfr 40920 part no S5632CHHP28L02	5-51
		BEARING: MIT 40920 PAR NO 55632FCHHP37L01	5-80 5-80
		in dia by 0.189 in Ig, mfr 13499 part no 540-8152-002	5-00
1A8MP5		INSERT, FLEX: COUPLING pistc: 0.938 in dia by 0.185 in Ig; mfr 13499 part no 548-7038-002	5-80
1A8MP6		COLLAR, SHAFT: al, chromate dip; 0.135 in by 0.228 in by 0.375 in: mfr 13499 part no 544-7442-002	5-51
1A8MP7		ARM, CAM: plstc; 0.187 in by 0.250.in by 1.094 in; mfr 13499	5-51
1A8MP8		HUB, GROUNDING: brs, gold and rhodium pl; 0.187 in ID, 0.270	5-51
1A8MP9		STATOR, SWITCH: ceramic Insulator, sil pl brs cons; 0.532 in	5-52
		Interchangeable with old MP9	
1A8MP10		ROTOR ASSY: 0.535 in by 0.593 in by 1.186 in o/a approx; mfr 13400 part po 544	5-52
1A8MP11		CONTACT, ELEC: cop "/gold-plated finish, 12 slots spaced 30 deg	5-52
1A8MP12		HOUSING, BEARING: CRES; 7/16-32 ext thd; 0.234 in Ig o/a,	5-80
		0.313 in ID undercut to 0.234 in dia, mfr 13499 part no 544-8408- 002	
1A8MP13		SPRING, RETAINING, CONT: CRES, passivate finish, 0.406 in ID, 0.430 in OD, 0.024 in thk: mfr 13499 part no 544-8520-002	5-51
1A8MP14		SHAFT: gold-plated brs shaft; 0.250 In dia by 3.500 in Ig, mfr 13499	5-52
1A8MP15		INSULATOR, BLOCK: plstc; 0.625 in by 0.656 in by 0.843 in, incl	5-50
1A8MP16		COVER, AMPL: al, chromate dip finish. 0.032 in by 3.190 in by	5-80
1A8MP17		COVER, AMPL: al, chromate dip, 0.032 in by 3.190 in by	5-80
1A8MP18		COVER, AMPL: al, chromate dip finish; 0.032 in by 3.190 in	5-80
1A8MP19		COVER, AMPL: al, 0.032 in by 3.190 in by 4.359 in mfr 13499	5-80
1A8MP20		part no 548-7063-002 INSULATOR, BLOCK: plstc; 0.468 in by 0.656 in by 0.688 in	5-52
1A8MP21		incl contacts, terminals and clips; mfr 13499 part no 548-7054-002 GEARSHAFT, SPUR: plstc, 36 teeth; 0.791 in dia by 0.531 in lg;	5-52
		mfr 13499 part no 548-7064-002	
1A8MP22		GEARSHAFT, SPUR: CRES; 58 teeth, 1.250 in dia by 0.719 in lg;	5-80
1A8MP23		GEAR, SPUR: plstc; 0.187 in ID, 1.250 in OD, 0.250 in Ig, 58 teeth, mfr 13499 part no 548-7066-002	5-51

### RF AND POWER AMPLIFIER MODULE

REF	NOTE	NAME AND DESCRIPTION	FIG NO
DESIG	S		
1A8MP24		CAM, CONTROL: brs, gold and rhodium pl, 0.437 in dia by 0.250 in lo. mfr 13499 part no 548-7073-002	5-51
1A8MP25		CLIP ASSY, ELECTRON TUBE. 0.718 in by 0.750 in by 1.281 in approx o/a dim mfr 13499 part no 548-7076-002	5-50
1A8MP26		SPACER, SLV, Long al, chromate dip, 0.120 in by 0.250 in by 0.420 in by 0.250 in by	5-51
1A8MP27		SPACER, SLV, SHORT. al, chromate dip; 0.120 in by 0.125 in by 0.250 in mfr 13499 part no 548-7081-002	5-51
1A8MP28		PLATE, GEAR: al, chromate dip finish 0.312 in by 2.219 in by 3.3188 in mfr 13499 part no 548-7104-003	5-80
1A8MP29		ROTOR ASSY, CAPACITOR 0.531 in by 0.937 in by 1 218 m approx o/a, mfr 13499 part no 548-7105-003, p/o 1A8C22,	5-51
1A8MP30		STATOR ASSY, GRID-0.998 in by 1 656 in by 2 501 in approx,	5-51
1A8MP31		13499 part no 648-7115-004, p/o 1A8C22, 1A8C28 STATOR ASSY, PLATE 0.?65 in by 1.781 in by 2.671 in approx: mfr 13499 part no 548-7116-004	5-51
1A8MP32		SHAFT ASSY, CAPACITOR sil pl brs shaft and rotor sly, coin sil cont sly; 0.437 in dia by 3.347 in lg; mfr. 13499 part no 548-	5-51
1A8MP33		PLATE, BOTTOM: al; 0.187 in by 3.375 in by 4.375 m, Incl	5-51
1A8MP34		PLATE, TOP al, 0.187 in by 3.375 in by 4.375 in mfr 13499	5-51
1A8MP35		SPACER, SLV, AMT: al, chromate dip, 0.089 in by 0.140 in by 0.250 in mfr 13499 part no 548-7057-002	5-51
1A8MP36		CLAMP, TUBE: sil pl brs; 0.234 in by 0.375 in by 0.672 in mfr 13499 part no 548-7237-002	5-50
1A8MP37		BLOCK, MTG CONN: brs, sil pl, 0.138 in by 0.375 in by 0.906 in mfr 13499 part no 548-7430-002	6-50
1A8MP38		INSERT, SCREW THD al, chromate dip finish 0.250 in w across flats by 0.128 in lg 2-56 thd mfr 13499 part no 548-7061-002	5-51
1A8MP39		SPACER, SLV al, chromate dip, 0.031 in wall, 0.156 in OD, 0.312 in lg. mfr 13499 part no 541-5955-002	5-50
1A8MP40		SPACER, SLV al, chromate dip; 0.031 in wall, 0.156 in OD, 0.375 in lg. mfr 13499 part no 541-5957-002	5-80
1A8MP41		GEAR, SPUR al, 58 teeth; 0.187 in ID, 0.250 in OD, 0.250 in Ig, mfr 13499 part no 548-7067-002	5-52
1A8MP42		BRACKET al, chromate dip finish, 1-3/8 in by 1-11/32 in by 6-1/8 in mfr 13499 part no 544-6904-002	5-51
1A8P1 P2 thru P11		CONNECTOR mfr 71468 part no DBM21WIPC27 NOT USED	5-80
1A8P12 1A8Q1		CONNECTOR: mfr 98278 part no 052-0471 TRANSISTOR MIL-S-19500/207(SIGC) type 2N1481	5-80 5-80
1A8R1 1A8R2		RESISTOR, FXD, CMPSN: MIL-R-11 type RC07GF105K RESISTOR, FXD, CMPSN: MIL-R-11 type RC07GF820K	5-52 5-50
1A8R3 1A8R4		RESISTOR, FXD, CMPSN :MIL-R-11 type RC20GF105K RESISTOR_EXD_CMPSN: same as 1A8R2	5-52 5-52
1A8R5 1A8R6		RESISTOR, FXD, CMPSN: same as 1A8R1 RESISTOR, FXD, CMPSN: MIL-R-11 type RC07GF101K	5-50 5-60

### RF AND POWER AMPLIFIER MODULE

REF DESIG	NOTE S	NAME AND DESCRIPTION	FIG NO
1A8R7 1A8R8 1A8R9 1A8R10 1A8R11 1A8R12 1A8R13 1A8R14 1A8R15 1A8R16 1A8R17 1A8R18		RESISTOR, FXD, CMPSN: same as 1A8R1 RESISTOR, FXD, CMPSN: MIL-R-11 type RC20GF471K RESISTOR, FXD, CMPSN: Same as 1A8R1 RESISTOR, FXD, CMPSN: MIL-R-11 type RC20GF151K RESISTOR, FXD, CMPSN: same as 1A8R1 RESISTOR, FXD, CMPSN: same as 1A8R8 RESISTOR, FXD, CMPSN: same as 1A8R8 RESISTOR, FXD, CMPSN: same as 1A8R1 RESISTOR, FXD, CMPSN: MIL-R-11 type RC20GF330K RESISTOR, FXD, CMPSN: Same as 1A8R1 RESISTOR, FXD, CMPSN: MIL-R-11 type RC20GF331K RESISTOR, FXD, CMPSN: MIL-R-11 type RC20GF681K RESISTOR, FXD, CMPSN: MIL-R-11 type RC20GF681K RESISTOR, FXD, WW: 10 ohms, ±1%, 2.5W, mfr 91637 part no RSM2C10R00F	5-52 5-52 5-50 5-50 5-50 5-50 5-50 5-50
1A8R19 1A8R20 1A8R21 1A8R22 1A8R23 1A8R24 1A8R25 1A8R26 1A8R26 1A8TB1 1A8V1 1A8V2 1A8V2		RESISTOR, FXD, WW: MIL-R-26 type RW67V272 RESISTOR, FXD, CMPSN: MIL-R-11 type RC07GF472K RESISTOR, FXD, CMPSN: MIL-R-11 type RC07GF103K RESISTOR, FXD, CMPSN: MIL-R-11 type RC07GF332K RESISTOR, FXD, CMPSN: MIL-R-11 type RC32GF222K RESISTOR, FXD, WW: MIL-R-26 type RW67V182 RESISTOR, FXD, CMPSN: MIL-R-11 type RC42GF472K RESISTOR, FXD, CMPSN: MIL-R-11 type RC42GF472K RESISTOR, FXD, CMPSN: same as 1A8R25 TERMINAL BOARD: plstc, 0.062 in by 0.938 in by 1.313 in; incl 4 terminals, mfr 13499 part no 548-7034-002 ELECTRON TUBE: MIL-E-1 type 7077 ELECTRON TUBE: same as 1A8V1 ELECTRON TUBE: Same as 1A8V1	5-80 5-80 5-80 5-80 5-80 5-80 5-80 5-80
1A8V3 1A8V4 1A8V5 1A8XV1 1A8XV2		ELECTRON TUBE: MIL-E-1 type 7554 ELECTRON TUBE: same as 1A8V3 ELECTRON TUBE: MIL-E-1C type 6442 SOCKET: mfr 04435 part no 86-001 SOCKET: mfr 04435 part no 86-071	5-52 5-50 5-52 5-52 5-52 5-52

RECEIVER AND TRANSMITTER SUBASSEMBLY

1A9	RECEIVER-TRANSMITTER SUBASSEMBLY: mfr 13499	5-37
1A9A1	AMPLIFIER, AC VOLTAGE: 28 Vdc supply voltage, 400 to 550 ohms input impedance 220-j 66 ohms output impedance, 5 ma-nom load current, 5 to 29K Hz freq response, mfr code 13499 part no 767-1964-001	5-82
1A9A2	AMPLIFIER, AC VOLTAGE same as 1A9A1	5-82
1A9C1	CAPACITOR, FXD, ELCTLT: (tantalum) 33UF, <u>+</u> 20%,35V, code 56289 part no 150D336X0035S2	5-83
1A9C2	CAPACITOR, FXD, ELCTLT: same as 1A9C1	5-81
1A9C3	CAPACITOR, FXD, MICA DIEL: MIL type CM05E394J03	5-83
1A9C4	CAPACITOR, FXD, MICA DIEL: MIL type CM05E200J03	5-83
1A9C5	CAPACITOR, FXD, ELCTLT: MIL type CK14AX103M	5-83
1A9C6	CAPACITOR, FXD, CER DIEL: MIL type CK12AX102M	5-82
1A9CR1	SEMICOND DEVICE: mfr 80131 part no 1N916	5-83

### RECEIVER AND TRANSMITTER SUBASSEMBLY

REF DESIG	NOTE S	NAME AND DESCRIPTION	FIG NO
1A9CR2 CR3	4	SEMICOND DEVICE: JEDEC type 1N1358A NOT USED	5-81 5-81
1A9CR4 1A9CR5 1A9CR6		SEMICOND DEVICE: MIL-E-1/1143 type 1N647 SEMICOND DEVICE: same as 1A9CR4 SEMICOND DEVICE: same as 1A9CR4	5-81 5-83 5-83
1A9CR7 1A9CR8 1A9CR9		SEMICOND DEVICE: same as 1A9CR4 SEMICOND DEVICE: mfr 81483 part no 69-1020-10Z6-3 SEMICOND DEVICE: same as 1A9CR4	5-54 5-81 5-54
1A9CR10 1A9DS1 1A9DS2		SEMICOND DEVICE: same as 1A9CR4 LAMP: MIL type MS25237-327 LAMP: same as 1A9DS1	5-54 5-81 5-81
1A9E1		CONTACT ASSY, ELEC: 0.342 in by 1.125 in by 2.062 in o/a dim, mfr 13499 part no 548-7516-003;c/o 1A9J1, 1A9J2, 1A9J3 ELISE, MIL, type E024/501//4AS	5-81 5-81
1A9F2 1A9FL1		FUSE: same as 1A9F1 FILTER, RAD INTRF: 5500PF, GMV, 200V; mfr 01121	5-81 5-81
1A9FL2 thru		FILTER, RAD INTRF: same as 1A9FL1	5-81
1A9FL7 1A9FL8		FILTER, HIGH PASS-LOW PASS: 200 to 400 MHz passband, 10W	5-53
1A9FL9		pwr rating in passband; 0.750 in by 1.125 in by 3.875 in o/a dim excl connectors; mfr 13499 part no 241-0467-00 FILTER. RAD INTRF: same as 1A9FL1	5-81
thru 1A9FL12		WASHEP: glass cloth silicone rbr: 0.086 in ID. 0.750 in OD	5-81
1A9H2		0.0050 in thk, mfr 13499 part no 547-2288-003 NOT USED	5-01
1A9H3 1A9H4		0.005 in thk, mfr 13499 part no 553-5050-003 NOT USED	5-81
1A9H5 1A9H6		SCREW, MACH: sst; Phillips recessed fil kid; 8-32NC-2A thd 1 in lg; mfr 13499 part no 548-7531-002 NOT USED	5-81
1A9H7		WASHER, STOP: al, 0.080 in by 0.342 in by 0.571 in o/a dim; mfr 13499 part no 548-7498-002	5-83
1A9H8		WASHER, NM: plstc, 0.88 in ID, 0.150 in OD, mfr 76854 part no 15517	5-83
1A9H9		SHIM: plstc; 0.010 in thk by 1.031 in dia o/a dim; mfr 13499 part no 548-7350-002	5-83
1A9J1		CONTACT, ELEC: sil-pl cop cont surface; 0.187 in dia by 0.342 in lg o/a dim; mfr 13499 part no 548-7447-002, p/o 1A9E1	5-81
1A9J2 1A9J3 1A9J4		CONTACT, ELEC: same as 1A9J1 CONTACT, ELEC: same as 1A9J1 NOT USED	5-81 5-81
1A9J5 1A9J6		CONNECTOR: mfr 71468 part no DEF9SC7A101 CONNECTOR: mfr 71468 part no DBMF13W3SIC27	5-81 5-81
1A9J7 1A9J8		CONNECTOR: nill / 1406 part to DAF 1557 ATOT CONNECTOR: same as 1A9J5	5-81 5-81

Table 6-2

## TABLE 6-2. (Continued)

### RECEIVER AND TRANSMITTER SUBASSEMBLY

REF DESIG	NOTE S	NAME AND DESCRIPTION	FIG NO
1A9J9 1A9J10 1A9J11 1A9J12 1A9J13 1A9J13 1A9J14 1A9J15 1A9J16 1A9J17 thru		CONNECTOR: 71468 part no DAMF11W1S1C27 CONNECTOR: same as 1A9J5 CONNECTOR: same as 1A9J9 CONNECTOR: mfr 71468 part no DBMF21W151C27 CONNECTOR: mfr 71468 part no DBMF21W151C27 CONNECTOR: mfr 77820 part no 371-7038-010 CONNECTOR: mfr 77820 part no 3012 CONNECTOR: mfr 94375 part no RF0752 CONNECTOR: same as 1A9J15	5-81 5-81 5-81 5-83 5-83 5-83 5-83 5-83 5-83 5-83 5-83
1A9J21 1A9J22 1A9J23 1A9J24 1A9J25 1A9J26 1A9K1 1A9K2		NOT USED NOT USED ADAPTER: mfr 94375 part no RF02105 JACK, TIP: WHT; mfr 98291 part no SKT10WHITE CONNECTOR-mfr 98278 part no 053-0577 RELAY mfr 01526 part no 3SAE2049A2 NOT USED	5-81 5-81 5-81 5-81
1A9K3 1A9K4 1A9K5		RELAY: same as 1A9K1 RELAY: same as 1A9K1 NOT USED	5-81 5-81
1A9K6 1A9L1		RELAY mfr 09026 part no BR7X300D2S3-26V REACTOR: mfr 07388 part no 12532B	5-84 5-81
1A9L3 1A9L4 1A9L5 1A9L6 1A9L7 1A9MP1		COIL, RF: MIL type LT4K032 COIL, RF: MIL type LT4K030 SOLENOID: mfr 81840 part no A38992-001 SOLENOID: same as 1A9L6 CHASSIS, RECEIVER-TRANSMITTER: al, 0.697 in by 3.656 in by 10.250 in act of a diameter of 548, 7403, 005	5-83 5-83 5-83 5-83 5-83 5-81
1A9MP2		FILTER BOX: brs, 0.781 in by 1.125 in 2.062 in o/a dim, fr 13/490 part no 548-7510.003	5-81
1A9MP3		CHASSIS, ELEC EQPT: al, 3 015 in by 3.656 in by 11 187 in o/a dim,	5-81
1A9MP4		mfr 13499 part no 548-7233-005 CHASSIS, ELEC EQPT: al, 2.937 in by 3.656 in by 11 187 in o/a dim,	5-81
1A9MP5		mtr 13499 part no 548-7235-005 BASE, XSTR: al, 11/16-24 ext thd, 0.265 in lg, 0.391 in lg o/a; mtr 12400 part no 548-7002 002	5-81
1A9MP6		CAP, XSTR: al, 11/16-24 nit thd, 3/4 in dia by 7/16 in lg, mfr 13499 part no 548-7003-002	5-81
1A9MP7 1A9MP8		NOT USED SPACER. SLV: al, chromate dipped, 0.152 in ID, 0.250 in OD, 0.250 in g, mfr 12400 part no 541 6021 002	5-81
1A9MP9		NUT, PL, HEX: al, 4-40 thd, 0.250 in hex 0.250 in lg,	5-81
1A94P10		SHAFT, SWITCH: 0.324 in by 0.703 in by 1 281 in o/a dim,	5-81
1A9MP11		COVER, SWITCH: al, chromate dip 0.063 in thk, 1.206 in by 1 518 in mfr 13499 part no 548-7347-002	5-83

### RECEIVER AND TRANSMITTER SUBASSEMBLY

REF	NOTE	NAME AND DESCRIPTION	FIG NO
DESIG	5		
1A9MP12		BLOCK, ELEC EQPT: sil pl finish; 0.971 in by 1.264 in by 1 908	5-83
1A9MP13		BLOCK, ELEC EQPT. brz, incl 4 terminals 0.905 in by 1.264 in	5-83
		by 1 908 in o/a dim, excl terminals, mfr 13499 part no 548-7536-	
1A9MP14		SPRING, SWITCH cop, 0.163 in by 0.218 in by 0.915 in o/a dim,	5-83
1A9MP15		mfr 13499 part no 548-7542-003 SPRING CLIP beryllium cop sil pl: 0 253 in by 0 260 in by	5-83
		0.412 in mfr 13499 part no 548-7540-002	5 00
1A9MP16		0.428 in lg; mfr 13499 part no 548-7349-002	5-83
1A9MP17		NOT USED	
1A9MP19			
1A9MP20		COVER mfr 95712 part no 583-3	5-83
1A9MP21		BEARING: mfr 43334 part no 77NM1220ZD5J	5-81
1A9MP23		BEARING mfr 96881 part no 3L3F	5-81
1A9MP24		RING: mfr 79136 part no X5133-18MD	5-81
1A9MP25 1A9MP26		RING mfr 79136 part no 5555-18MD	5-81
1A9MP27		POST, PIVOT, THREADED: CRES, passivate finish; 0.3125 in	5-81
		hex; 0.1868 in dia: 0.578 in la: mfr 13499 part no 548-7388-002	
1A9MP28		BRACKET, SOCKET: al, chromate dip; 0.900 in thk, 1.047 in dia,	5-81
1001020		2 4375 in Ig, mfr 13499 part no 548-7391-002	5-81
		part no 548-7393-003	5-01
1A9MP30		CABINET, ELEC, EQPT: 9.718 in by 13.983 in by 21.625 in mfr 13499 part no 548-9396-004	5-81
1A9MP31		HANDLE, BOW: al; 0.937 in by 2.750 in by 4.656 in o/a dim,	5-81
1A9MP32		CLAMP. WINDOW: al. chemical film finish. 0.100 in by 0.437 in	5-81
14014022		by 1 718 in mfr 13499 part no 548-7978-004	5 Q1
TASIMF 33		mfr 13499 part no 548-7450-002	5-61
1A9MP34		GEARSHAFT, SPUR: delrin, 0.790 in dia by 1.875 in lg o/a dim,	5-81
1A9MP35		GEAR, SPUR: CRES, 1.428 in dia by 0.125 in thk o/a dim,	5-81
1A9MP36		mfr 13499 part no 543-7462-002 GEAR, SPUR: al. 1.541 in dia by 0.375 ln lg o/a dim.	5-81
14014027		mfr 13499 part no 548-7455-002	E 91
TA9IVIP37		mfr 13499 part no 548-7457-002	10-C
1A9MP38		GEAR, SPUR: CRES, 0.458 in dia by 0.375 in Ig o/a dim;	5-81
1A9MP39		GEAR, SPUR: CRES; 1 428 in dia by 0.125 in thk o/a dim,	5-81
1A9MP40		mtr 13499 part no 548-7452-002 GEARSHAFT, SPUR: CRES, 0.291 in dia by 2.562 in lo o/a dim.	5-81
		mfr 13499 part no 548-7463-002	5.04
TA9MP41		GEARSHAFT, SPUR: deirin, 0.291 in dia by 2.312 in ig 0/a dim, mfr 13499 part no 548-7464-002	5-81

# TABLE 6-2. (Continued)

### RECEIVER AND TRANSMITTER SUBASSEMBLY

REF	NOTE	NAME AND DESCRIPTION	FIG NO
DESIG	S		
1A9MP42		SHAFT, STR, SPECTRUM: CRES, passivate finish, 0.1872 in dia, 2.125 in lg. mfr 13499 part no 548-7460-002	5-81
1A9MP43 1A9MP44		NOT USED GEAR, SPUR, IDLER, NO 3; al, anodize finish, 0.275 in by 0.500 in	5-81
1A9MP45		SPACER, SLV: al, chemical film finish, 0.250 in OD by 0.187 in Ig, mfr 13499 part no 541-6019-002	5-81
1A9MP46 1A9MP47		NOT USED GEARSHAFT, SPUR: delrin, 1.291 in 1.875 in Ig o/a dim, mfr 12400 part no 548 7521 002	5-81
1A9MP48		CLAMP, COUPLER: SST, passivate finish 0.187 in w, 0.562 in	5-81
1A9MP49		COUPLING HALF, SHAFT: CRES, 0.937 in dia by 0.353 in lg o/a	5-81
1A9MP50		COUPLING HALF, SHAFT: CRES, 0.937 in dia by 0.328 in Ig o/a	5-81
1A9MP51		COUPLING HALF, SHAFT: CRES, 0.937 in dia by 0.303 in Ig o/a	5-81
1A9MP52		PANEL, FRONT: al 4.440 in by 4.552 in by 11.352 in mfr 13499	5-81
1A9MP53		HOUSING, GEAR: al, 3.525 in by 3.656 in by 8.431 in o/a dim,	5-81
1A9MP54		SHAFT, STRAIGHT-DIAL, CONTROL: CRES, passivate finish, 0.1238 in dia 3.594 in lar mfr 13499 part no 548-7486-002	5-81
1A9MP55		SHAFT, STR, NO 4: CRES, passivate finish, 0.1872 in dia, 3.594	5-81
1A9MP56		GEAR, SPUR: al; 0.583 in dia by 0.312 in lg o/a dim, mfr 13499 part no 548-7489-002	5-81
1A9MP57		GEAR, HELICAL: al, 0.906 in dia by 0.312 in Ig o/a dim, mfr 13499 part no 548-7491-002	5-81
1A9MP58		GEAR, HELICAL: al, 1.484 in dia by 0.312 in Ig o/a dim, mfr 13499 part no 548-7492-002	5-81
1A9MP59		GEAR, BEVEL: CRES, 0.562 in dia by 0.375 in lg o/a dim, mfr 13499 part no 548-7503-002	5-81
1A9MP60		GEAR, BEVEL: CRES, 0.562 in dia by 0.375 in lg o/a dim, mfr 13499 part no 548-7505-002	5-81
1A9MP61		GEAR, BEVEL: NO 3.CRES, passivate finish, 16 teeth, 20°	5-81
		angle, 32 diametral pitch; 0.187 in ID, 0.594 in OD, 0.375 in Ig, mr 13499 part no 548-7504-002	
1A9MP62		DIAL ASSY: 1.437 in dia by 1.562 in Ig o/a dim, mfr 13499 part no 548-7507-002	5-81
1A9MP63		DIAL ASSY: 1.437 in dia by 0.421 in Ig o/a dim, mfr 13499 part no 548-7508-002	5-81
1A9MP64		DIAL ASSY: 1.437 in dia by 1.264 in Ig o/a dim, mfr 13499 part no 548-7509-002	5-81
1A9MP65		GEAR CLUSTER, BEVEL-SPUR: two complements of 16 and 36 teeth, 0.790 in dia by 0.468 in Ig o/a dim, mfr 13499	5-81
1A9MP66		part no 548-7510-002 SHAFT, STR DIAL: CRES, passivate finish 0.1872 in dia, 5.031 in lg, mfr 13499 part no 548-7511-002	5-81

### RECEIVER AND TRANSMITTER SUBASSEMBLY

REF	NOTE	NAME AND DESCRIPTION	FIG NO
DESIG	S		
1A9MP67		SHAFT, STR, DIAL, CONTROL, TENTHS: CRES, passivate	5-81
		finish; 0.1872 in dia, 3.453 in lg; mfr 13499 part no 548-7501-002	
1A9MP68		SHAFT, STR, DIAL, CONTROL, TENTHS: CRES, passivate finish;	5-81
		0.1872 in dia, 4.094 in lg; mfr 13499 part no 548-7500-002	
1A9MP69		GEAR, SPUR, COUNTER, NO 7: al, anodize finish;5/16 in by	5-81
		13/32 m; mfr 13499 part no 548-7495-002	
1A9MP70		GEAR, SPUR, COUNTER, NO 2: al, anodize finish;39 teeth, 20°	5-81
		pressure angle, 48 diametral pitch; 5/16 in by 13/32 in mfr 13499	
4404074		part no 548-7490-002	5.04
1A9MP71		SPIDER: mfr 13499 part no 990-0511-00	5-81
1A9MP72		SPIDER: mfr 13499 part no 990-0512-00	5-81
1A9MP73		SHAFT: STR, DIAL, CONTROL, UNITS: CRES, passivate finish;	5-81
		0.1872 in dia, 3.0781 in ig, mtr 13499 part no 548-7502-002	5.04
TA9MP74		STUD, EXTENSION: al; 0.312 in w across flats by 0.824 in ig	5-81
		0/a ultii, iiiii 13499 pait no 346-7409-002 SHAFT, STR, INDUT, CRES, passivote finish, 0.1973 in die	E 01
TA9IVIP75		5 TAFT, STR, INPUT. CRES, passivale IIIIIsT, 0.1072 III 01a, 1 1975 in lay mfr 12400 port po 549 7471 002	5-01
1 A 0 M D 7 6		RUTTON CABLE: plete: 4-40NC-28 int thd, 0.187 in la by	5-81
TABIVIE 70		0.250 in bey mfr 13/00 part no 5/1-5177-002	5-61
149MP77		SPRING HELICAL EXTENSION: stl: 0.187 in dia by 0.781 in	5-81
		$\ln 0/a \dim m fr 13499 \text{ part no 340-1018-00}$	001
1A9MP78		NOT LISED	
1A9MP79		NOT USED	
1A9MP80		COLLAR. STOP: al. chromate dip. 0.156 in by 0.426 in by 0.618 in	5-81
		0.187 in dia bore: mfr 13499 part no 548-7480-002	
1A9MP81		COLLAR, STOP: al, 0.226 in by 0.426 in by 0.618 in 0.187 in dia	5-81
		bore;	
		mfr 13499 part no 548-7526-003	
1A9MP82		SHAFT, STR, SPUR GEAR: CRES, passivate 0.1872 in dia, 1 766	5-81
		m lg, mfr 13499 part no 548-7488-002	
1A9MP83		GEAR, SPUR: al, anodize finish, 48 teeth, 20° pressure angle,	5-81
		48 diametral pitch; 3/8 in by 13/32 in; mtr 13499 part no	
4404504		548-7479-002	5.04
1A9MP84		SHAFT, STR: CRES, passivate finish; 0.1872 in dia, 2.573 in ig;	5-81
		MIT 13499 part no 548-7459-002	F 01
TA9IVIP65		diametral pitch, 2/9 in by 12/22 in mfr 12/00 port po	5-01
		5499 part no	
140MP86		SHAFT STR SPECTRUM IDLER: CRES passivate finish: 0.1872	5-81
TABIVIE 00		in dia 1.875 in la: mfr 13499 part no 548-7461-002	5-61
1A9MP87		GEAR SPUR INPUT CRES passivate finish 40 teeth 20° pres	5-81
		angle, 48 diametral pitch: 3/8 in by 13/32 in mfr 13499	
		part no 548-7472-002	
1A9MP88		WINDOW, DIAL: clear plstc; 0.060 in thk: 1.312 in by 1.719 in	5-81
		by 1.844 in mfr 13499 part no 548-7514-003	
1A9MP89		BUSHING, SLY, GEARSHAFT: ph brz, chemical polish; 0.8760 in	5-81
		ID, 1.000 in OD, 0.375 in lg, mfr 13499 part no 548-7397-002	
1A9MP90		SPRING, HELICAL, COMPRESSION: CRES, passivate finish;	5-81
		0.0220	
		l in by 0.171 in by 0.406 in mfr 13499 part no 544-7906-002	

# TABLE 6-2. (Continued)

### RECEIVER AND TRANSMITTER SUBASSEMBLY

REF	NOTES	NAME AND DESCRIPTION	FIG
DESIG			NO
1A9MP91		BALL: mfr 43334 part no 3-16A1BWBALL	5-81
1A9MP92		BRACKET, STAR WHEEL: al, chemical film finish;0.125 in by	5-81
		1.156 in by 1.390 in, mfr 13499 part no 544-6940-003	
1A9MP93		RETAINER, SPR: al, anodized finish; 0.296 in by 0.359 in by 0.654 in;	5-81
		mfr 13499 part no 544-7864-002	
1A9MP94		STAR WHEEL: CRES, passivate finish: 0.281 in by 0.406 in by	5-81
		1.093 in. mfr 13499 part no 554-6939-003	
1A9MP95		GASKET: MIL type ÅN6227-3	5-81
1A9MP96		COVER: mfr 13499 part no 372-2179-00	5-81
1A9MP97		COVER. ELEC CONN: same as 1A9MP96	5-81
1A9MP98		COLLAR, SHAFT:al, chemical film finish, 0.190 in by 0.234 in	5-81
		by 0.375 in: mfr 13499 part no 548-7832-00	
1A9MP99		BEARING, mfr 43334 part no 77NM1020Z05J	5-81
1A901		KNOB. VOLUME al. blk anodize: 0.609 in ID. 0.750 in OD.	5-81
		0.875 in lg: mfr 13499 part no 548-7389-002	
1A902		KNOB-SQUELCH: al. blk anodize. 0.468 in ID. 0.531 0.563 in lg.	5-81
		mfr 13499 part no 548-7390-002	
1A903		KNOB: al, anodized finish; 0.750 in dia by 0.875 in lg;	5-81
		mfr 13499 part no 554-6973-003	
1A904		KNOB: setscrew type, al body, 1.125 in dia by 0.750 in lg o/a dim:	5-81
		mfr 13499 part no 549-6077-002	
1A905		KNOB: same as 1A904	5-81
1A906		KNOB: same as 1A904	5-81
1A9P1		NOT USED	
1A9P2		NOT USED	
1A9P3		CONNECTOR: mfr 98291 part no UG1461/U	5-83
1A9P4		CONNECTOR: same as 1Å9P3	5-83
1A9P5		CONNECTOR: same as 1A9P3	5-83
1A9P6		CONNECTOR :mfr 94375 part no RF-0721-50	5-83
1A9P7		NOT USED	
1A9P8		CONNECTOR: same as 1A9P6	
1A9P9		CONNECTOR: same as 1A9P3	5-53
1A9P10		CONNECTOR: same as 1A9P6	5-53
1A9Q1		TRANSISTOR:MIL-S-19500/180(SIGC) type 2N1486	5-54
1A9R1		RESISTOR, FXD, WW: MIL-R-26 type RŴ69V101	5-81
1A9R2		RESISTOR, FXD, WW: 0.50 ohms ±1%, 2 5W, mfr 91637	5-81
		part no RSM2COR500F	
1A9R3		RESISTOR, FXD, WW: MIL-R-26 type RW30V250	5-81
1A9R4		RESISTOR, VAR, CMPSN: MIL-R-94B type RV5NAYSD103D	5-54
1A9R5		RESISTOR, VAR, CMPSN: 50K, ±20%, 1/2W; mfr 13499	5-54
		part no 380-6952-00	
1A9R6		RESISTOR, FXD, WW: same as 1A9R3	5-81
1A9R7		RESISTOR, FXD, CMPSN: MIL:-R-11 type RC20GF391K	5-83
1A9R8		RESISTOR, FXD, CMPSN: MIL:-R-11 type RC07GF474K	5-83
1A9R9		RESISTOR, FXD, WW same as 1A9R1	5-80
1A9R10		RESISTOR, FXD, CMPSN: MIL type RC07GF151K	5-83
1APR11		RESISTOR, FXD, FILM MIL-R-10509 type RN55D2612F	5-82
1A9R12		RESISTOR, FXD, CMPSN: MIL-C-5 type RC07GF103K	5-82
1A9R13		RESISTOR, FXD, FILM: MIL-R-10509 type RN55D6812F	5-82

## TABLE 6-2. (Continued)

### RECEIVER AND TRANSMITTER SUBASSEMBLY

	NOTEO		
REF	NOTES	NAME AND DESCRIPTION	FIG
DESIG			NO
DESIG			NO
1A9R14		RESISTOR, FXD, FILM: same as 1A9R11	5-82
1A9R15		RESISTOR, FED, FILM: same as 1A9R13	5-82
1A9S1		SWITCH, RF, XMSN LINE: 6 coupling positions, 59	5-83
		ohms impedance characteristic 26 5 Vdc 2 in by 2-1/4	
		In by 3 in mfr 13499 part no 548-7537-005	
149514		SWITCH SECTION: mfr 13499 part no 269-2273-00	5-83
440040			5 00
1A921B		SWITCH SECTION: mfr 13499 part no 269-2274-00	5-83
1A9S2		NOT USED	
1A9S3		SWITCH- mfr 76854 part no 211576F1C	5-54
1A9TB1		TERMINAL BOARD. Incl five solder stud terminals:	5-83
-		0.032 ln by 1.116 ln by 1.428 ln board dim: mfr 13499	
		part no 548-7687-003	
1492051		IGHT mfr 08817 part no 855878	5-81
1407000			5 01
TA9XD52		LIGHT: same as TA9XDS1	5-81
1A9XF1		FUSEHOLDER mfr 71400 part no HKPEHLQRWZ	5-81
149XF2		FUSEHOLDER same as IA9XE1	5-81
17.07.1 Z			0.01

RECEIVER-TRANSMITTER CASE CY-3884/PRC-41

1A10	CASE, RECEIVER - TRANSMITTER:al, enamel finish;	5-37
	4 in by 10.750 in by 12.875 in 0/a; mfr 13499 part no	
1A10H1	SCREW, MACH SST, passivate finish; slotted pan hd;	5-55
	10-32 NF-2A	
	part no 548-7362-002 trd, 3/4 I1A9R14 lg; mfr 13499	
1A10H2	SCREW, MACH SST, passivate finish: 6-32 NC-2A	5-55
	tied, 7/16 in lg; mfr 13499 part no 549-6246-002	
1A10H3	NUT STRIP CRES, passivate finish; 3/8 in by 11/16 ln,	5-55
1410.11	CONNECTOR RCPT FLEC: 1 sil male cont: 0.452 in	5-55
17(1001	w across flats by 0 781 in Ig excl cont, mfr 13499	0.00
	part no 548-7367-003	
1A10J2	CONNECTOR, RCP1, ELEC: same as 1A10J1	5-55 5-55
1A10MP	3/16 in dia. mfr 13499 part no 018-1291-00	5-55
1		
1A10MP	CASE, RECEIVER-TRANSMITTER: al, 4.420 in by	5-55
2	no 548-7223-005	
1A10MP	PLATE, RETAINING, ELEC CONN: 0.718 in by 3.125	5-55
3	in by 10 in o/a dim, mfr 13499 part no 548-7369-004	F
	5EAL, RBR, RND 5ECT. SIIICON RBR, 0.240 IN ID, 0.450 in OD, 0.405 in la: mfr 13499 part no 548-7363-	5-55
-	002	
1A10MP	PIN, LOCATING: CRES, passivate finish; 0 563 in hex,	5-55
5	0.02 U 422 In dia, 0 791 in ig, mfr 13499 part no 548-7368-	
1A10MP	STRIKE, CATCH: stl; 0.187 in by 0.562 in by 0.625 in	5-55
6	o/a dim; mfr 13499 part no 549-6247-002	
1A10MP	GASKET: MIL-P-5516 type AN6227-7	5-55
1		

### STORAGE BATTERY BB-451/U

2	NOTE	BATTERY STORAGE sil zing alkaline 24 volt	5-35
~	NOTE		0.00
	1	nominal 25 amp hour nominal: rechargeable supplied	
		normal, 20 amp nour normal, reenargeable, supplied	
		in a dry charge condition: contained in a watertight	
		In a dry charge condition, contained in a waterlight	
		fiber case; weight is 14.2 pounds; 7-11/16 in by 11-	
		9/16 in by $4-9/64$ in mfr 13499 part no 522-2527-004	
		$3/10$ in by $4^{-3}/04$ in this 13433 part to $322^{-2}327^{-0}04$ .	

Table 6-2

# **TABLE 6-2. (Continued)**

POWER SUPP	<u>LY PP-3700/PRC-</u>	41	
REF	NOTE	NAME AND DESCRIPTION	FIG
DESI	S		NO
G			
3		POWER SUPPLY: 26 5 Vdc, 4 amp output, 115 to 230	5-36
		Vac, 50 to 400 Hz single; 4-3/16 in w by 7-3/4 in h by 12-	
004		in Ig, mfr 13499 part no 522-2528-005	5.04
301		CAPACITOR, FXD, ELCTLT: 6600F,M15%P75%, 50V,	5-61
202			F F7
362		CAPACITOR, FXD, ELGTET: 330F, ±20%, 35V, MIT	5-57
		00209 port no 150D226V002562	
303		CARACITOR EXD EL CTLT: come as 3C2	5-57
303		$CAPACITOR FXD ELECTET: same as 302CAPACITOR FXD ELECTET: 4711E \pm 20\% 351/ mfr$	5-57
504		56289	5-57
		nart no 150D476X0035S2	
305		CAPACITOR FXD FLCTLT 100UF +20% 20V mfr	5-57
		56289	0.01
		part no 150D107X0020S2	
3C6		CAPACITOR, FXD, ELCTLT: same as 3C5	5-57
3C7		CAPACITOR, FXD, ELCTLT: same as 3C2	5-57
3C8		CAPACITOR, FXD, ELCTLT: same as 3C2	5-57
3C9		CAPACITOR, FXD, ELCTLT: same as 3C4	5-57
3C10		CAPACITOR, FXD, PPR DIEL. 0.1UF, <u>+</u> 20%, 400V, mfr	5-57
		00656 part no 931-4552-00	
3CR1		SEMICOND DEVICE: MIL-S-19500/134(SIGC) type	5-57
3CR2		SEMICOND DEVICE: same as 3CR1	5-57
3CR3		SEMICOND DEVICE: same as 3CR1	5-57
3CR4		SEMICOND DEVICE: Same as 3CR1	5-57
		SEMICOND DEVICE: MIL-E-1/1143 type 1N647	5-51
		SEMICOND DEVICE: Same as 30Kb	5-57 5-57
3017		11/2000B	5-57
3088			5-57
30110		1N965B	5-57
3CR9		SEMICOND DEVICE: same as 3CR5	5-57
3E1		CLIP: mfr 71400 part no 4548	5-59
3E2		CLIP. ELEC same as 3E1	5-59
3E3		CONTACT, ELEC: sil, 0.187 in dia by 13/16 in lg, mfr	5-59
		13499	
		part no 549-1692-002, p/o 3P1, 3P2, 3P3	
3E4		CONTACT, ELEC: coin sil, gold pl fininsh, 0.187 in dia by	5-59
		1 in lg, mfr 13499 part no 756-3162-003; Effective on MCN	
		no 2013E4 TERMINAL, LUG: brs;0.515 in dia mtg hole,	
		accomodates 0.125 in dia cndct, mtr 13499 part no 548-	
254		$\frac{1}{547-002}$	E E7
0F1 2F2		FUSE INIL TYPE FUZAZOUV T- T- ZAS	5-57
3FZ 2U1		COEW MACH MODIFIED NO 1: SST passivate finish:	5 50
3111		stid nan hd 10-32 NE-24 tied 0.750 in la mfr 13499 nart	5-59
		no 548-7362-002	
3H2		RING: mfr 91314 part no 340-0643-00	5-59
3H3		NUT.PL, HEX; CRES; passivate 5/8-18 UNF-2B tied. 0	5-59
		688 in w across flgts, 0.125 in thk. mfr 13499 part no 548-	
		7419-002H4	
3H4		WASHER: glgss cloth, silicone rbr, 0 086 in ID, 0.750 in	5-57
		OD,	
		0 0050 in thk, mfr 13499 part no 547-2288-003	
3H5		WASHER: glgss cloth, silicone rbr coated, 0.500 in ID, 1	5-58
0.10		3/5 in OD, 0.0050 in thk, mtr 13499 part no 553-5038-003	
3H6		WASHER: al; 0 515 in ID, 1 250 in OD, 0.125 in thk, mfr	5-58
		13499 part no 553-5039-003	

### AN/PRC-41A PARTS LIST

## TABLE 6-2. (Continued)

POWER SUPPLY PP-3700/PRC-41			
REF DESIG	NOTES	NAME AND DESCRIPTION	FIG NO
3H7		WASHER plstc; 0.510 in ID, 0 672 in OD, 0.125 in thk, mfr 13499 part no 553-5041-003	5-58
3J1 3J2 3K1 3L1		CONNECTOR: MIL-C-5015 type MS3102R16S1P CONNECTOR: MIL-C-5015 type MS3102R14S7S RELAY: mfr 09026 part no BR7X300D2S3-26V REACTOR: mfr 98055 part no TS3722	5-57 5-57 5-57 5-57
3MP1		CHASSIS, ELEC EQPT: al; 3.592 in by 5 in by 8-3/16 in; mfr 13499 part no 548-7551-005	5-57
		mfr 13499 part no 548-7424-004 COVER PWR SUPPLY: al: 0.624 in by 4.156 in by 10.937 in o/a:	5-57
3MP3		mfr 13499 part no 548-7422-004 COVER PWR SUPPLY: al. chromate dip finish: 0.625 in by 4.156	5-57
3MP4		in by 10.937 in; mfr 13499 part no 548-7423-004 SHELL, ELEC CONN: plstc; 0.750 in dia by 47/64 in lg;	5-59
3MP5 3MP6		mfr 13499 part no 548-7420-003; p/o 3P1, 3P2 and 3P3 RING mfr 79136 part no 5133-9C SPACER, SLV, NM: neoprene rbr; 0.406 in dia 0.437 in Ig,	5-59
3MP6		mfr 13499 part no 548-7238-002; p/o 3P1, 3P2, and 3P3 SEAL, CONT 0.423 in dia by 0.393 in lg; mfr 13499	5-59
3MP7 3MP8 3MP9		Dust no 756-3161-002; Effective on MCN 201 DUST CAP: mfr 02660 part no 9760-16-291 CAP: mfr 02660 part no 9760-14-291 RETAINER, CAPACITOR al, 0.719 in by 0.750 in by 2-3/32 in	5-57 5-57 5-61
3MP10 3MP11 3MP12 3MP13		GROMMET: mfr 75543 part no 911 GROMMET: mfr 75543 part no 911 GROMMET: mfr 75543 part no 901 HOLDER: mfr 13499 part no 352-9970-00 BASE, XSTR: al: 11/16-24 ext thd, 0.625 in dia 25/64 in lg o/a:	5-59 5-57 5-57 5-58
3MP14		mfr 13499 part no 548-7002-002 CAP, XSTR al; 11/16-24 int thd; 3/4 in dia by 7/16 in Ig,	5-58
3MP15		Infr 13499 part no 548-7003-002 INSULATOR, BSHG: plstc, 5/16 in dia by 0.150 in lg, 0.177 in ID undercut to 0.096 in mfr 13499 part no 548-7004-003	5-58
3MP16 3P1 3P2 3Q1 3Q2 3Q3 3Q4 3R1 3R2 3R3 3R4 3R5 3R6		NOT USED CONNECTOR: c/o 3E3 and 3MP4, 3MP5 and 3MP6 CONNECTOR: same as 3P1 CONNECTOR: same as 3P1 TRANSISTOR MIL-S-19500/182(NAVY) type 2N1893 TRANSISTOR MIL-S-19500/80A(SIGC) type 3N35 TRANSISTOR: mfr 07688 part no 2N2110 TRANSISTOR: JAN type JAN2N1613 RESISTOR, FXD, WW: MIL-R-26 type RW67V471 RESISTOR, FXD, WW: MIL-R-26 type RW69V150 RESISTOR, FXD, WW: MIL-R-26 type RW69V150 RESISTOR, FXD, WW: MIL-R-26 type RW69V150 RESISTOR, FXD, WW: MIL-R-11 type RC32GF331K RESISTOR, FXD, CMPSN: MIL-R-11 type RC32GF821K RESISTOR, FXD, CMPSN: MIL-R-94B type RV5LAYSB102B	5-59 5-57 5-59 5-57 5-58 5-58 5-57 5-60 5-57 5-57 5-57 5-57 5-57

Table 6-2

# TABLE 6-2. (Continued)

	POWER SUPPL	_Y PP-3700/PR	C-41	
l	REF DESIG	NOTES	NAME AND DESCRIPTION	FIG NO
	3K7 3R8 3R9 3R10 3R11 3R12 3R13 3R14 3R15 3R16 3R17 3R18 3R19 3R20 3R21 3R22 3R23 3T1 3XF1 3XF2		RESISTOR, FXD, CMPSN: MIL-R-11 type RC326562X RESISTOR, FXD, CMPSN: MIL-R-11 type RC206F393K RESISTOR, FXD, CMPSN: MIL-R-11 type RC206F273K RESISTOR, FXD, CMPSN: MIL-R-11 type RC206F273K RESISTOR, FXD, CMPSN: MIL-R-11 type RC326F472K RESISTOR, FXD, WW: MIL-R-26 type RW69V661 RESISTOR, FXD, WW: MIL-R-26 type RW69V661 RESISTOR, FXD, WW: same as 3R16 RESISTOR, FXD, WW: same as 3R16 RESISTOR, FXD, CMPSN: MIL-R-11 type RC206F102K RESISTOR, FXD, CMPSN: MIL-R-11 type RC206F40K RESISTOR, FXD, CMPSN: MIL-R-11 type RC206F470K RESISTOR, FXD, CMPSN: MIL-R-11 type RC206F470K TRANSFOR, FXD, CMPSN: MIL-R-11 type RC206F470K TRANSFORMER: mfr 95089 part no TS3682 FUSEHOLDER: mfr 71400 part no HKPEHLQRWZ FUSEHOLDER: same as 3XF1	5-61 5-57 5-57 5-57 5-57 5-57 5-57 5-57 5-60 5-60 5-60 5-60 5-61 5-57 5-58 5-58 5-58 5-58

## NAVSHIPS 0967-872-5020 TM-03816B-35/2 TM11-5820-510-35-1

## AN/PRC-41A PARTS LIST

# TABLE 6-2. (Continued)

DIRECTIONAL	ANTENNA	AS-1405/PRC-41

REF	NOTES	NAME AND DESCRIPTION	FIG
DESIG		ANTENNA: array "yagi": 225 to 400 MHz, moyable rotating:	NU 5-36
-		mfr 13499 part no 522-2529-005	0.00
4E1		ANTENNA ELEMENT: mfr 13499 part no 548-7313-003; c/o 4MP1,	5-62
		4MP5, 4MP6, 4MP7 and 4H2	
4E2		ANTENNA ELEMENT: 3/8 in dia by 10.656 in lg, approx, mfr 13499	5-62
450		part no 548-7314-003; c/o 4MP2, 4MP5, 4MP6, 4MP7, and 4H1, 4H2	5.00
4E3		ANTENNA ELEMENT: 3/8 in dia in by 9.468 in Ig, approx; mfr 13499	5-62
1=1		Part no 548-7315-003; C/O 4MP3, 4MP5, 4MP6, 4MP7 and 4H1, 4H2 ANTENINA ELEMENT: 2/9 in dia by 9.274 in la approx: mfr 12400	5 62
464		part no 548-7316-003 $c/o$ 4MP4 4MP5 4MP6 4MP7 and 4H1 4H2	5-02
4E5		ANTENNA ELEMENT: 0.312 in dia by 7.656 in lg: mfr 13499	5-62
		part no 548-7308-003; c/o 4MP8, 4MP9	0 0-
4E6		ANTENNA ELEMENT: 0.312 in dia by 6.750 in lg; mfr 13499	5-62
		part no 548-7309-003; c/o 4MP8, 4MP10	
4E7		ANTENNA ELEMENT: 0.312 in dia by 5.875 in lg; mfr 13499	5-62
400		part no 548-7310-003, c/o 4MP11, 4MP12	F 60
4⊏0		ANTENNA ELEMENT. 0.312 III 01a by 5.250 III 19, IIII 1.3499	2-02
4F9		ANTENNA ELEMENT: 0.312 in dia by 4.656 in lg. mfr 13499	5-62
120		part n0 548-7312-003: c/o 4MP11, 4MP14	0.02
4E10		CONTACT STRIP, ELEC: cop, sil pl; 0.010 in by 2-5/16 in by	5-62
		29-5/8 in; mfr 13499 part no 548-7307-003	
4H1		NUT, SLV: CRES; 0.375 in dia 0.563 in lg; mfr 13499 part no	5-62
4110		548-7289-002 DNL_CDD_MC_type_MC40502.400	F 00
		NSEPT I APGE: hrs: 0.625 in dia by 1.1/2 in la linclinia: mfr	5-62 5-62
4115		13499 part no 548-7356-002	5-02
4H4		INSERT SMALL thrs: $0.562$ in dia by 1-1/2 in lot include:	5-62
		mfr 13499 part no 548-7357-002	0 0-
4MP1		ELEMENT SECTION, ANT: brs tube; 0.089 in ID, 0.205 in OD,	5-62
		7-5/32 in lg; mfr 13499 part no 548-7291-002	
4MP2		ELEMENT SECTION, ANT: brs tube; 0.089 in ID, 0.205 in OD,	5-62
41402		5 IN IG; MIT 13499 PAR NO 548-7292-002	F 60
4111-3		3-13/16 in lg. mfr 13499 part no 548-7293-002	0-02
4MP4		ELEMENT SECTION. ANT: brs tube: 0.089 in ID. 0.205 in OD.	5-62
		2-23/32 in lg; mfr 13499 part no 548-7294-002	• •-
4MP5		ELEMENT SECTION, ANT brs tubing; 5/16 in dia by 6-1/8 in lg;	5-62
		mfr 13499 part no 548-7287-002	
4MP6		FERRULE, ELEMENT: brs, chrome pl, 0.270 in dia by 15/64 in lg;	5-62
		MIT 13499 part no 548-7288-002 CAD ELEC: bro 0.250 in dia by 1/2 in lay mfr 12400 part no	F 60
		548- 7290-002	0-02
4MP8		PLUG. ELEMENT: brass: 0.312 in dia by 0.250 in lo: mfr 13499	5-62
		part no 548-7300-002	
4MP9		ELEMENT SECTION, ANTENNA: brs tubing;; 0.248 in ID, 0.312 in OD,	5-62
		7-17/32 in lg; mfr 13499 part no 548-7295-002	
4MP10		ELEMENT SECTION ANT: brs tubing, 0.248 in ID, 0.312 in OD,	5-62
		o-5/6 in ig, mir 13499 part no 548-7296-002	

## TABLE 6-2. (Continued)

# DIRECTIONAL ANTENNA AS-1405/PRC-41

DIRECTION	IAL AN I ENN/	A AS-1405/PRC-41	
REF	NOTES	NAME AND DESCRIPTION	FIG
4MP11		PLUG, ELEMENT: Brs, 0.250 in dia by 0.250 in ig; mfr 13499	5-62
		part no 548-7301-002	
4MP12		ELEMENT SECTION, ANT: brs, tubing; 0.186 in ID, 0.250 in OD,	5-62
		5-7/8 in lg: mfr 13499 part no 548-7297-002	
4MP13		FLEMENT SECTION ANT brs tubing 0.186 in ID 0.250 in OD	5-62
		5-1/8 in lg: mfr 13499 part no 548-7298-002	0 0-
4MP14		ELEMENT SECTION ANTENNA: hrs tubing: 0.186 in ID, 0.250 in OD	5-62
		4.17/32 in lg mfr 13400 not no 548-7200.002	0.02
		NOT LICED	
			F 00
4IVIP16		SUPPORT, ANT: 1.500 in by 5.781 in by 24.562 ig approx 0/a dim,	5-62
		mtr 13499 part no 548-7284-002	
4MP17		COVER, ELEC CONN: mfr 95712 part no 583-3	5-62
4MP18		LINE SECTION, RADIO FREQUENCY TRANSMISSION: brs; 5/32 in	5-62
		by 0.343 lg by 29.749 in mfr 13499 part no 548-7285-002	
4MPI9		BOOM. ANT, 0.875 in by 3.500 in by 30.718 in approx o/a	5-62
		dim: mfr 13499 part no 548-7318-004: c/o 4E10 and 4H3, 4H4 and	
		4MP15 4MP16 4MP17 4MP18 and 4W1	
4\N/1		CABLE ASSY RE: type RG-141A/LL coay cable terminated $w/1$ cont:	5-62
- <b>V</b> V I		d = 1/2 in la $d/a$ : mfr 12/100 not no 5/19 7296 002	0.02
		$40^{-1/2}$ in g 0/a, the 13435 part to 340 <sup>-1/2</sup> 00002	E CO
			5-62
		CONNECTOR: mir 13499 part no 357-9994-00, p/0 4001	5-62

## ANTENNA AS-1404/PRC-41

5	ANTENNA: stud type; 225 to 400 MHz; fixed; mfr 13499 part no 522-2530-003	5-35
---	--	------

### MOUNTING MT-2976/PRC-41

6	MOUNTING; al, item attached to mtg by locking jaws in front and conn rcpt at rear; 5-9/16 in h by 11-1/4 lg w by 16-7/32 in lg;	5-36
	mfr 13499 part no 522-2531-005	
6C1	CAPACITOR, FXD, ELCTLT:1300UF,M10% P100%, 50V mfr 56289	5-64
	part no D34393	
6C2	CAPACITOR, FXD, ELCTLT: same as 6C1	5-64
6C3	CAPACITOR, FXD, ELCTLT: 1,675 UF, M10% P100%, 50 V;	5-64
	mfr 56289 part no D36741	
6C4	CAPACITOR, FXD, ELCTLT: same as 6C3	5-64
6C5	CAPACITOR, FXD, ELCTLT: same as 6C3	5-64
6C6	CAPACITOR, FXD, PPR DIEL: 1.0 UF 20%, 100 V; mfr 56289	5-64
	part no 186P10501S15	
6C7	CAPACITOR, FXD, PPR DIEL: same as 6C6	5-64
6CR1	SEMICOND DEVICE: MIL-S-19500/134 (SIGC) type 1N249B	5-64
6E1	CONTACT, ELEC: sil; 0.210 in dia by 1.000 in Ig o/a dim; mfr 13499 part no	5-64
	548-7206-002; c/o 6P1, 6P2, and 6P3	

### TABLE 6-2. (Continued)

MOUNTING MT-2976 /PRC-41			
REF DESIG	NOTES	NAME AND DESCRIPTION	FIG NO
6H1		NUT, SLV: al; 1.000 in dia by 1.000 in lg o/a dim; mfr 13499 part no 548-7406-002	5-63
6H2		THUMBSCREW: CRES; 0.750 in dia by 3.250 in Ig o/a dim; mfr 13499 part no 548-7216-002	5-63
6H3		BUSHING, MACH THD: CRES; 1.000 in w across flgts by 0.375 in lg	5-64
6H4		BUSHING, MACH THD: CRES; 1.000 in w across flgts by 0.375 in lg	5-64
6H5		NUT, PL, HEX: CRES, 1.000 in w across flgts by 0.187 in lg o/a	5-64
6H6		POST, ELEC-MECH EQPT: al, 0.187 in w across flgts by 1.375 in	5-64
6H7		WASHER, FLAT: rd shape; CRES, passivated finish, 0.120 in ID,	5-64
6H8		RING: mfr 79136 part no 5133-12MD: p/o 6P1, 6P2, and 6P3	5-64
6H10		POST, ELEC-MECH EQPT: al, 6-32 UNC-2B thd, 5/16 in hex,	5-64 5-64
6H11		POST, ELEC-MECH EQPT: al; 6-32 UNC-2B thd, 5/16 in hex,	5-64
6H12		NUT, CONT: CRES; 1.000 in w across flgts by 0.468 in Ig o/a dim;	5-64
6H13		INSULATOR, BSHG. plstc; 0.500 in w across flgts by 0.625 in lg a/a dim: mr 12409 part no 548 7207 002 p/0 6P1 6P2 and 6P2	5-64
6L1 6MP1		COIL, RF: mfr 13499 part no 240-0021-00 CLAMP, RIM CLENCHING: al; 0.375 in by 0.953 in by 1.062 in o/a	5-64 5-63
6MP2		dim; mfr 13499 part no 548-7215-002 PIN, CLEVIS: CRES, 0.375 in dia by 0.656 in lg o/a dim,	5-63
6MP3		mfr 13499 part no 548-7214-002 STRAP, RETAINING: al; 0.396 in by 1.500 in by 4.404 in o/a dim;	5-64
6MP4		mfr 13499 part no 548-7411-003 COVER, FLTR: al; 0.531 in by 4.718 in by 11.480 in o/a dim,	5-63
6MP5		mfr 13499 part no 548-7221-004 RETAINER, CAPACITOR: al; 1.312 in by 2.109 in by 4.500 in o/a	5-64
6MP6		dim; mfr 13499 part no 548-7410-003 RETAINER, CAPACITOR: al; 1.312 in by 2.109 in by 4.500 in o/a	5-64
6MP7		dim, mfr 13499 part no 548-7408-003 TRAY, MOUNT: al bracket CRES tray; 6.187 in by 11.600 in by	5-63
6MP8		14.937 in o/a dim, mfr 13499 part no 548-7218-004 CAP: mfr 02660 part no 9760-14-291	5-63
6P1		ADAPTER, PWR SUPPLY: sil cons, pistc bshg; 0.500 in w across figts by 1 in Ig, mfr 13499 part no 548-7208-002, c/0 6E1 and 6H8	5-63
6P2		ADAPTER, PWR SUPPLY: same as 6P1	5-63 5-63
6P4		CONNECTOR, RCPT, ELEC: MIL-C-5015 type MS3102R14S7P	5-64
6R2		RESISTOR, FXD, WW: MIL-K-26 type RW29VR56 RESISTOR, FXD, WW: same as 6R1	5-64 5-64
6R3 6R4		RESISTOR, FXD, WW: same as 6R1 RESISTOR, FXD, WW: same as 6R1	5-64 5-64

## **BATTERY ADAPTER W-9**

REF DESIG	NOTES	NAME AND DESCRIPTION	FIG NO
7		BATTERY ADAPTER, ASSY, ELEC: 27.250 in Ig o/a approx; mfr 13499 part no 548-7568-002	5-36
7E1		TERMINAL, LUG: cop, 0.750 in by 0.750 in by 1.750 in; mfr 13499 part no 548-7566-002	5-63
7J1		NOT USED	
7J2		NOT USED	
7J3		CONNECTOR mfr 71468 part no CA3101E14S7SME	5-65

# POWER ELECTRIC CABLE ASSEMBLY CX-8687/PRC-41

REF DESIG	NOTES	NAME AND DESCRIPTION	FIG NO
8		CABLE ASSY, PWR, ELECTRIC. 3 conductors, no 16 AWG terminated ea end w cone; 50 ft lg o/a, mfr 13499 part no 522-2533- 002	5-36
8P1 8P2		CONNECTOR: MIL type MS3106A16S1S CONNECTOR: mfr 74545 part no 7057	5-66 5-66

# POWER ELECTRIC CABLE ASSEMBLY CX-8686/PRC-41

REF DESIG	NOTES	NAME AND DESCRIPTION	FIG NO
9		CABLE ASSY, PWR, ELECTRIC: 2 conductors no 14 AWG terminated ea end w conn; 20 ft lg o/a; mfr 13499 part no 522-2534-002	5-56
9P1 9P2		CONNECTOR: mfr 71468 part no CA06R14S7P CONNECTOR. mfr 71468 part no CA06R14S7S	5-66 5-66

# RADIO FREQUENCY CABLE ASSEMBLY CG-55G/U

REF DESIG	NOTES	NAME AND DESCRIPTION	FIG NO
10		CABLE ASSY, RF: 50 ohms coax; stranded cop cndct, 20 ft lg o/a; terminated each end w/ cone; mfr 13499 part no 522-2535-003	5-36
10P1		CONNECTOR: MIL type MS91236	5-66
10P2		CONNECTOR: same as 10P1	5-66
10MP1		COVER, ELEC, CONN: mfr 95712 part no 9756-1	5-66
10MP2		COVER, ELEC, CONN: same as 10MP1	5-66

### SPECIAL PURPOSE ELECTRICAL CABLE ASSEMBLY CX-8688/PRC-41

REF DESIG	NOTES	NAME AND DESCRIPTION	FIG NO
11		CABLE ASSY, SP, ELEC: 10 cndct no 26 AWG: terminated ea end	5-36
11P1 11P2		CONNECTOR: MIL type MS91481 CONNECTOR: same as 11P1	5-66 5-66

ORGINIAL

### **TABLE 6-2. (Continued)**

## MAST AB-777/PRC-41

REF	NOTES	NAME AND DESCRIPTION	FIG
DESIG			NO
12		MAST: al; 28.250 in closed max h, 78.250 in when nested, mfr	5-36
		13499	
4014		part no 522-2537-004	F 07
12H1		mfr 13409 part po 549 7277 002	5-67
12H2		NIT SLV LARGE: CRES chemically blk: 1 625 in dia 1 625 in la:	5-67
12112		mfr 13499 part no 548-7276-003	5-07
12MP1		CAP MAST: al alloy grn enamel finish: 2 687 in dia by 2 in lg	5-67
		mfr 13499 part no 548-7279-00	0 01
12MP2		CAP, TUBE: al; 1.390 in dia by 0.375 in lg; mfr 13499	5-67
		part no 548-7274-002	
12MP3		SPIKE ASSY, ANT ANCHOR: 0.875 in dia by 7.875 in lg; mfr 13499	5-67
		part no 548-7266-002; c/o 12MP4, 12MP5, and 12MP6	
12MP4		PIN, SPR: MIL type MS16562-217	
12MP5		SPIKE, ANT ANCHOR: CRES; 0.500 in dia by 7.875 in Ig; mfr	5-67
		13499 nort no 549 7004 000	
12MD6		PARTING 548-7284-002   PLISHING SPIKE: CRES: 0.427 in ID 0.975 in OD 1.275 in lat	5.67
12IVIE0		$3/4_14$ ovt the $3/8$ in let mfr $13/400$ part pp $5/8_7265_002$	5-07
12MP7		MAST SECTION: al 1 250 in dia by 25 625 in la: mfr 13499	5-67
		part no 548-7269-003	0.01
12MP8		MAST SECTION: al: 1.225 in ID. 1.375 in OD. 24 in lg: mfr 13499	5-67
		part no 548-7270-002	
12MP9		WEDGE, SMALL: plstc, 1 219 in ID, 1.333 in OD, 0.500 in lg;	5-67
		mfr 13499 part no 548-7273-002	
12MP10		WEDGE, LARGE: plstc; 1 375inID, 1.489inOD, 0.500 in thk;	5-67
4014044		mfr 13499 part no 548-7272-002	5.07
12MP11		MAST SECTION: al; 1.385 IN ID, 1.498INOD, 22.500 IN IG;	5-67
12MD12		1111 13433 Pail 110 348-727 1-002   DI LIG MAST: CRES: 1 300india by 0 375inla: mfr 13400	5-67
		part no 5/8-7275-002	5-07
1	1	part 10 070 1210-002	

## ANTENNA MAST ADAPTER

REF	NOTES	NAME AND DESCRIPTION	FIG
DESIG			NO
13		ADJUSTMENT, ELEVATION: mfr 13499 part no 548-7335-004	5-36
13H1		THUMBSCREW: CRES; 0 750india by 0.219 in h head; 10-32 tied,	5-68
		0.750 in lg; 1.625 in lg o/a; mfr 13499 part no 548-7332-002	
13H2		WASHER, FLAT: CRES, passivate finish, 0 203 in ID, 0.375 in OD,	5-68
		0.006 in thk; m\$r 13499 part no 547-2314-003	
13MP1		PIN: MIL type MS16562-190	5-68
13MP2		PIN: MIL type MS16562-192	5-68
13MP3		PIN: MIL type MS16562-223	5-68
13MP4		JOINT SECTION, LONG: 1.875 in dia by 6.625 in lg, approx, mfr	5-68
		13499	
		part no 548-7324-003	
13MP5		JOINT SECTION, SHORT: 1.875 in dia by 4.062 in lg, approx;	5-68
		mfr 13499 part no 548-7324-002	
13MP6		SHAFT, STR: CRES, 0.312 in dia by 1.375 in lg, 5/16-24 ext thd,	5-68
		0.750 in lg; mfr 13499 part no 548-7330-002	

## TABLE 6-2. (Continued)

## ANTENNA MAST ADAPTER

REF	NOTES	NAME AND DESCRIPTION	FIG
DESIG			NO
13MP7		PIN, STR, HDLS: CRES; 0.312 in dia by 0.484 in lg; mfr 13499 part no 548-7329-002	5-88
13MP8		CLAMP, LOOP: al; accommodates 1-3/4 in dia material, mfr 13499 part no 548-7334-002	5-68
1301		KNOB: cast iron, grn enamel finish; screw on type, star shape; 2 in dia by 0.750 in thk; mfr 13499 part no 548-7331-002	5-68

## MOUNTING MT-2977/PRC-41

14	MOUNTING al; item attached to mtg by two quick release clgmps; 3/8 in h by 5-1/4 in w by 10-3/4 in lg; mfr 13499 part no 522-2539- 003	5-36
14MP1	BRACKET, SHELF: . al, chromate dlp; 0.125 in by 0.281 in by 0.812	5-69
14MP2	RING, RETAINING: al, chromate dip; 0.125 in by 2.188 in by 3.563	5-69
	mfr 13499 part no 548-7204-004	

### HANDSET H-33E/PT

15	HANDSET: mfr 97101 part no 1212	5-35

# ELECTRICAL EQUIPMENT HARNESS

16		ELECTRICAL EQUIPMENT HARNESS:	5-35
	NOTE 1	FRAME: mfr 13499 part no 015-1630-010	5-70
	-	STRAP: mfr 13499 part no 011-0110-00	5-70
		STRAP. mfr 13499 part no 021-0190-00	5-70
		BILLET: mfr 13499 part no 021-0192-00	5-70
		CARRIER: mfr 13499 part no 021-0253-00	5-70
		SHOULDER, STRAP ASSY: incl strap support and 2 plates, mfr	5-70
		13499 part no 548-7583-003	
		FRAME ASSY: 4.906 in by 7.875 in by 10.686 in mfr 13499	5-70
		part no 548-7590-003	

# BRACKET ASSEMBLY

17	BRACKET ASSY, CONN: 1-7/8 in by 4 937 in by 8-5/8 in	6-36
17H1 17J1 17MP1	mfr 13499 part no 548-7413-003 NUT, PL, RND: sil pl; 5/8-24; mfr 94375 part no 02105-2 ADAPTER: MIL-C-71 type UG30DU BRACKET: al; 1-7/32 in by 4-3/16 in by 8.625 in;mfr 13499 part no 548-7414-004	5-71 5-71 5-71

### TABLE 6-2. (Continued)

## SPARE A PARTS A KIT

REF DESIG	NOTES	NAME AND DESCRIPTION	FIG NO
18		SPARE PARTS KIT: c/o HOLDER ASSY, TOP, mfr 13499 part no 554-6868-004, HOLDER ASSY, BOTTOM, mfr 13499 part no 554-6867-004, 1 Electron Tube MIL type 7554, 1 Electron Tube MIL type 7077, 1 Electron Tube, MIL type 6442, 2 Lamps, MS type MS25237-328 and 6 Fuses, MIL type F02A250V1 1-2AS	

## MAINTENANCE KIT

19	MAINTENANCE ACCESSORY KIT: incl 8 cables, 3 test adapters and 1 bag: mfr 13499 part no 548-7559-00	3-56
10001	ADADTED TEST: $1.2\%$ in by $1.1\%$ in by $2.216$ in mfr $12400$	E 70
19001	part no 548-7545-003; c/o 19E1 19H1 19H2, 19H3, 19J1,	5-73
	19MP1, 19MP2, and 19MP3	
19CP2	ADAPTER: mfr 13499 part no 357-9918-00	5-72
19CP3 19F1	ADAPTER: mfr 13499 part no 357-9919-00 CONTACT: cop_sil PL: 0 250 in by 0 500 in by 1 753 m; mfr	5-72 5-73
1021	13499	0.10
19H1	part no 548-73B1-002, p/o 19CP1 SCREW. MACH: MIL type MS35217-55.p/o 19CP1	5-73
19H2	SCREW, MACH: MIL type MS35216-14; p/o 19CP1	5-73
19H3 19J1	CONNECTOR: MIL-C-5015 type MS3102R14S7P; p/o 19CP1	5-73 5-73
19J2	ADAPTER: MIL-C-71 type UG29B/U	5-73
	elec	5-75
	cont; mdt on plstc plgte, mfr 13499 part no 548-7544-003, p/0	
19MP2	NUT: al, chromate dipped, open end type, headless, 4-40 NC-	5-73
	2B thd, 0.250 in W across figts, 0.750 in ig, mfr 13499 part no 540-9176-003. p/o 19CP1	
19MP3	RING: mfr 78189 part no 213-070408-2303; p/0 19CP1	5-73
19001	178B/U	5-72
	coax cable terminated w/conn ea end, 18 ft lg o/a, mfr 13499	
19W1J1	CONNECTOR: mfr 71468 part no DBM21W1S	5-72
19W1P1 19W2	CONNECTOR: mfr /1468 part no DBM21W1PC27 CABLE ASSY: SPG, ELEC: stranded cop conductors no 22	5-72 5-72
	AWG, PC 178 P/L cable terminated as and w/ connectors: 18 ft la o/a;	
	mfr 13499 part no 548-7571-003	
19W2J1 19W2P1	CONNECTOR: mfr 71468 part no DBM13W3SC27 CONNECTOR: mfr 71468 part no DBM13W3PC27	5-72 5-72
19W3	CABLE ASSY: SPCL, ELEC: stranded no 22 AWG cndct RG-	5-72
	cable terminated ea end w/ con; 18 ft lg o/a, mfr 13499	
19\\/3.11	part no 548-7572-003 CONNECTOR: mfr 71468 part no DAM11W1SC27	5-72
19W3P1	CONNECTOR: mfr 71468 part no DAM11W1PC27	5-72
19W4	ea end	5-72
10\/// 11	w/ connector: 18 ft lg o/a; mfr 13499 part no 548-7573-003	5 70
1377431		5-12

Table 6-2

### TABLE 6-2. (Continued)

MAINIENAN	ICE KII		
REF	NOTES	NAME AND DESCRIPTION	FIG
DESIG			NO
19W4P1		CONNECTOR: mfr 71468 part no DA15PC7	5-72
19W5		CABLE ASSY SPCL ELEC: stranded no 22 AWG conductor terminated	5-72
		ea end w/connector: 18 ft lg $o/a$ : mfr 13499 part no 548-7574-003	0.2
19W5.I1		CONNECTOR: mfr 71468 part no DE9PC7	5-72
19W/5P1		CONNECTOR: mfr 71468 part no DE9SC7A101	5-72
19\//6		CABLE ASSY: mfr 98291 part no 53-0164-061	5-72
10//7		CABLE ASSY mfr 08201 part no 53-0163-061	5-72
10\//8		$CABLE ASSY RE: type PG_303/LL coay cable terminated each$	5-72
1900		and w/conp 18 in $\ln n/2$ mfr 13/00 part no 5/8-7570-003	5-72
		end w/conin 10 in 19 0/a, nin 13499 part no 340-7370-005	
		TOOL KIT. ELEC: 12 electronic tools and 1 bruch analoged	E 26
20		in action duck has mer 12400 part no 549 7529 002	5-30
		In collon duck bag, fill 13499 part no 346-7536-005	
		$EX_{1} = 10026$ part no $C_{1} = 1002$	
		RET. IIII 70270 part 10 GT 1003 CODEWDDWED: mfr CE014 part no CO	
		SUREWDRIVER: IIII 00814 part no 024 0400 00	
		ALIGNMENT TOOL: MIT 13499 part no 024-0426-00	
		SCREWDRIVER: mfr /9061 part no P111	
		KEY: mfr 13499 part no 024-0019-00	
		KEY: mfr 13499 part no 024-0730-00	
		KEY: mfr 13499 part no 024-2900-00	
		KEY: mfr 13499 part no 024-0178-00	
		SCREWDRIVER: mfr 79061 part no A116-3	
		WRENCH, SPANNER: CRES; 0.065 in thk, 1.250 in OD, 4 375	
		in Ig; mtr 13499 part no 548-7431-002	
		BRUSH PAINT: mfr 13499 part no 024-0371-00	
		ALIGNMENT TOOL, ELECTRONIC EQPT: plstc handle and tip,	
		w/hex on end, mfr 13499 part no 548-9286-002	
		ALIGNMENT TOOL: mfr 81815 part no 25C104	

### GUY ROPE ASSEMBLY

21	GUY ROPE ASSY mfr 13499 part no 012-5103-00	5-36

# GUY STAKE

22	STAKE GUY: . CRES, chemically blacken, 1 in by 1 in by	5-36
	10 in; mfr 13499 part no 548-7338-003	

### DIRECTIONAL ANTENNA CASE

23	COVER ASSY ANT: 1.280 in by 19.718 in by 31.531 in	5-36
23MP1	GROMMET: mfr 83014 part no H322-3-1	5-74
23MP2	PLUNGER: mfr 13499 part no 015-2243-00	5-74
23MP3	right cover, mfr 13499 part no 548-7428-005	5-74

REF DESIG	NOTES	NAME AND DESCRIPTION	FIG NO
23MP4		COVER, ANT: plstc; 0.781 in by 19.718 in by 31.531 in; left cover: mfr 13499 part no 548-7429-005	5-74
23MP5		ROD, HANDLE: al, 0.620 in ID, 0.750 in OD, 5.125 in Ig; mfr 13499 part no 548-7426-002	5-74

RADIO SET CASE CY-3883/PRC-41

24	CASE.: mfr 13499 part no 021-0207-00	1-2

ELECTRONIC EQUIPMENT CASE CY-3885/PRC-41

25	CASE:. mfr 13499 part no 021-0208-00	1-3

EQUIPMENT REPAIR PARTS (WHEN ORDERED)

ELECTRON TUBE: MIL-E-1 type 7554 (qty 1) ELECTRON TUBE: MIL-E-1C type 6442 (qty 1) LAMP: MIL-E-6363 type MS25237-327 (qty 2)	
LAMP: MIL-L-6363 type MS25237-327 (gty 2)	
FUSE: MIL-F- 15160 type F02A250V3AS (qty 2)	

SPECIAL

# PURPOSE ELECTRICAL CABLE ASSEMBLY CX-10831/PRC-41A

27	CABLE ASSY, SPCL, ELEC: 2 ft long; mfr 13499 part no 767-1975-001	5-36
27PI	CONNECTOR: MS type MS3116J14-19P	5-84
27P2	CONNECTOR: mfr 77820 part no PT06W16-26S	5-84

# **TABLE 6-3 LIST OF MANUFACTURERS**

MFR CODE	NAME AND ADDRESS	MFR CODE	NAME AND ADDRESS
00136	Mc Coy Electronics C0.	10646	Carborundum Co., The
	Watts-Chestnut St.		P.O. Box 337
	Mt. Holly Springs, PA 17065		Niagara Falis, NY 14302
00656	Aerovox Corp	13499	Collins Radio Co
00000	740 Belleville Ave.	10400	5225 C Ave. N.E.
	New Bedford, MA 02745		Cedar Rapids,IA
01121	Allen-Bradley Co.	40920	MPB Corp.
	1201 5. 200 5t Milwaukoo WI 53204		Keepe NH 03/31
	Willwadkee, WI 55204		Reene, NIT 03431
01526	General Electric Co	43334	New Departure-Hyatt
			Bearings Dlv.
	Specialty Control Dept		General Motors Corp.
	P.U. BOX 812 Waynashara VA 22080		Hayes Ave. Sondusky OH 44870
	Waynesboro, VA 22960		Sandusky, OH 44070
02114	Ferroxcube Corp	56289	Sprague Electric Co.
	Mt. Marion Rd		North Adams, MA 01247
	Saugerties, NY 12477	05044	
		65814	VVIIIIams, J. H. and CU.
02660	Amphenol Corp		Buffalo NY 14207
02000	2801 S 25th Ave		2011010,111 1201
	Broadview, IL 60153		
00077		70276	Allen Mfg. Co
03877	I ransitron Electronic Corp		BOX 570
	Wakefield MA 01880		
		70417	Chrysler Corp.
04405			Amplex Div.
04435	Jettron Products, Inc.		6501 Horpor Ave
	Hanover N.1 07936		Detroit MI 48211
06001	General Electric Co	70674	ADC Products
	Electronic Capacitor and		Div. of Magnetic Controls
	Battery Dept		00. 6405 Cambridge St
	P O Box 158		Minneapolis MN 55426
	Irmo, SC 29063		
		71400	Bussmann Mfg.
07388	l orotel, Inc.		Div of McGraw-Edison Co.
	Grandylew MO 64030		St Louis MO 63017
07688	Joint Electron Device	71468	ITT Cannon Electric, Inc
	Engineering Council		3208 Humbolt St.
	Washington, DC 20443		Los Angeles, CA 90031
08817	Epec Inducstries Inc	71590	Globe-Umon Inc
	New Bedford Industrial Park		Centralgb Dlv
	New Bedford MA 02745		P.O. Box 591
00000	Debeeek Fleetreniss Com		Milwaukee, WI 53201
09026	Babcock Electronics Corp Relays Div	72136	Electro Motive Mfg. Co. Inc.
		12100	The South Park and John
			Streets
	P.O. Box 1499		
	Costa Mesa, CA 92626		Willimantic, CT 06226

# TABLE 6-3. LIST OF MANUFACTURERS (Continued )

MFR CODE	NAME AND ADDRESS	MFR CODE	NAME AND ADDRESS
72982	Erie Technological Products, Inc.	80294	Bourns, Inc
	644 W. 12th St.		200 Columbia Ave.
	Elle, PA 16512		Riverside, CA 92507
73386	Freed Transformer Co, Inc	81349	Military Specifications
	1736 Weirfield St.		Promulgated by Standardization Dlv.
	Brooklyn, NY 11227		Directorate of Logistic Services DSA
73899	JFD Electronics Corp	81483	International Rectifier Corp
	15th At 62nd St.		233 Kansas St.
	Brooklyn, NY 11219		El Segundo, CA 90245
74545	Hubbell Harvey Inc	81815	Communication Coil Co
	State St. and Bostwick Ave	01010	2839 N. Narragansett Ave.
	Bridgeport, CT 06602		Chicago, IL 60634
75543	Lavelle Rubber Co	81840	ledex Inc
10010	424 N. Wood		123 Webster St
	Chicago, IL 60622		Dayton, OH 45402
76854	Oak Mfg Co	82142	Airco Speer Electronic Components
10001	Div of Oak Electro/Netics Corp	02112	4 Hoover Ave.
	S Main		Du Bois, PA 15801
	Crystal Lake, IL 60014	83014	Hartwell Corp
77820	Bendix Corp ,The	00011	9035 Venice Blvd.
	Electrical Components Div.		Los Angeles, CA 90034
	Sidney NY 13838	83086	New Hampshire Ball Bearings Inc
		00000	Peterborough, NH 03458
78189	Illinois Tool Works, Inc	04044	Louis Caring and Mrg. Co.
	Shakeprool DIV. 91314	91314	2652 W North Ave
	Elgm, IL 60126		Chicago, IL 60647
70400	Staalingle Carbon Co		
70400	Stackpole Carbon Co. St. Marvs. PA 15857		
		91637	Dale Electronics, Inc
			P O Box 609
79061	Vaco Products Co		
	510 N. Dearborn 91663	91663	ArmelElectronics, Inc.
	Chicago, IL 60610		1601 75th St.
			Bergen, NJ 07047
79136	Waldes Kohinoor, Inc		
	Long Island City, NY	92054	Radio Cores, Inc
			5757 W 95th St
			Oak Lawn, IL 60453
80008	Electro Engineering Works	00100	
	401 Preda St.	93106	Accurate Products Co., Inc.
	San Leandro, CA 94577		
00124	Electrome Industrian Assoc	04217	Midland Adhenive and Chemical Com
00131	2001 Eve St in W	94217	2604 Goodrich Ave
	Washington DC 20006		Ferndale, MI 48220
1	1		

# TABLE 6-3. LIST OF MANUFACTURERS (Continued)

MFR CODE	NAME AND DESCRIPTION	MFR CODE	NAME AND DESCRIPTION
94375	Automatic Metal Products Corp 315-323 Berry St Brooklyn, NY 11211	97101	Audiosears Corp. South St. Stamford, NY 12167
95088	Transomc, Inc 808 16th St Bakersfield, CA 93301	97137	TRW Electronic Components Div. Chicago, IL 60607
95712	Bendix Corp, The Microwave Devices Div. Hurricane Rd. Franklin, IN 46131	98055	Whaling City Marine Co., Inc 56 Prospect New Bedford, MA 02740
96881	Thomson Industries, Inc	98278	Microdot, Inc 220 Pasadena Ave. S. Pasadena, CA 91030
	1029 Plgndome Rd Manhasset NY 11030		
		98291	Sealectro Corp. 225 Hoyt Mamaroneck NY 10544
96906	Military Standards Promulgated by Standardization Dly		
	Directorate of Logistic Services DSA	99800	Delevan Electronics Corp. 270 Quaker Rd. East Aurora, NY 14052

ORIGINAL

6-53, 6-54

### USER ACTIVITY TECHNICAL MANUAL COMMENT SHEET NAVSHIPS 4914 (10-62) (COG 1 - 11-DIGIT STOCK NUMBER: 0105-503-9850)

NAVSHIPS NO.

VOLUME NO.

(Fold on dotted. line on reverse side, staple, and mail to NAVSEC)

PROBLEM AREA:

Ship or Station

Fold

-----

Department of Navy Naval Ship Systems Command Washington, D.C. 20360 OFFICIAL BUSINESS

Postage and Fees Paid NAVY DEPARTMENT

\_\_\_\_\_

Commander, Naval Ship Engineering Center Technical Support Branch Department of the Navy Washington, D.C. 20360

Fold

RECOMMENDED CHANGES TO EQUIPMENT TECHNICAL PUBLICATIONS		
	SOMETHING WRONG WITH PUBLICATION	
THENJOT DOWN THE DOPE ABOUT IT ON THIS FORM. CAREFULLY TEAR IT OUT, FOLD IT AND DROP IT IN THE MAIL.		
PUBLICATION NUMBER	PUBLICATION DATE PUBLICATION TITLE	
BE EXACT PIN-POINT WHERE IT IS	IN THIS SPACE, TELL WHAT IS WRONG	
PRINTED NAME, GRADE OR TITLE AND TELE	PHONE NUMBER SIGN HERE	
DA 1 JUL 79 2028-2 PRE ARE	EVIOUS EDITIONS P.SIF YOUR OUTFIT WANTS TO KNOW ABOUT YOUR COBSOLETE. RECOMMENDATION MAKE A CARBON COPY OF THIS AND GIVE IT TO YOUR HEADQUARTERS.	

# THE METRIC SYSTEM AND EQUIVALENTS

#### **'NEAR MEASURE**

. Centimeter = 10 Millimeters = 0.01 Meters = 0.3937 Inches

- 1 Meter = 100 Centimeters = 1000 Millimeters = 39.37 Inches
- 1 Kilometer = 1000 Meters = 0.621 Miles

### **VEIGHTS**

Gram = 0.001 Kilograms = 1000 Milligrams = 0.035 Ounces 1 Kilogram = 1000 Grams = 2.2 lb.

1 Metric Ton = 1000 Kilograms = 1 Megagram = 1.1 Short Tons

### LIQUID MEASURE

1 Milliliter = 0.001 Liters = 0.0338 Fluid Ounces

1 Liter = 1000 Milliliters = 33.82 Fluid Ounces

### APPROXIMATE CONVERSION FACTORS

TO CHANCE	10	
		MULTIPLT BT
Inches	Centimeters	2.540
Feet	Meters	0.305
Yards	Meters	0.914
Miles	Kilometers	1.609
Square Inches	Square Centimeters	6.451
Square Feet	Square Meters	0.093
Square Yards	Square Meters	0.836
Square Miles	Square Kilometers	2.590
Acres	Square Hectometers	0.405
Cubic Feet	Cubic Meters	0.028
Cubic Yards	Cubic Meters	0.765
Fluid Ounces	Milliliters	
nts	Liters	0.473
arts	Liters	0.946
allons	Liters	3.785
Ounces	Grams	28.349
Pounds	Kilograms	0.454
Short Tons.	Metric Tons	0 907
Pound-Feet	Newton-Meters	1 356
Pounds per Square Inch	Kilonascals	6 895
Miles per Gellon	Kilometers per Liter	0.425
Miles per Hour	Kilometers per Hour	1 609
since per nour	Infometers per fibur	1.005
TO CHANGE	то	MULTIPLY BY
<b>TO CHANGE</b> Centimeters	<b>TO</b> Inches	<b>MULTIPLY BY</b> 0.394
<b>TO CHANGE</b> Centimeters Meters	TO Inches Feet	MULTIPLY BY 0.394 3.280
TO CHANGE Centimeters Meters. Meters.	TO Inches Feet Yards	MULTIPLY BY 0.394 3.280 1.094
TO CHANGE Centimeters Meters. Meters. Kilometers	TO Inches Feet Yards Miles	MULTIPLY BY 0.394 3.280 1.094 0.621
TO CHANGE Centimeters Meters Kilometers Square Centimeters	TO Inches Feet Yards Miles Souare Inches	MULTIPLY BY 0.394 3.280 1.094 0.621 0.155
TO CHANGE Centimeters Meters Meters Kilometers Square Centimeters Square Meters	IO   Inches   Feet   Yards   Miles   Square Inches   Square Feet	MULTIPLY BY 0.394 3.280 1.094 0.621 0.155 10.764
<b>TO CHANGE</b> Centimeters   Meters.   Meters.   Kilometers   Square Centimeters   Square Meters.   Square Meters.	IO   Inches   Feet   Yards   Miles   Square Inches   Square Feet   Souare Yards	MULTIPLY BY 
<b>TO CHANGE</b> Centimeters   Meters.   Meters.   Kilometers   Square Centimeters   Square Meters.   Square Meters.   Square Meters.   Square Meters.   Square Kilometers	IO   Inches   Feet   Yards   Miles   Square Inches   Square Feet   Square Yards   Sourre Miles	MULTIPLY BY 
<b>TO CHANGE</b> Centimeters   Meters.   Meters.   Square Centimeters   Square Meters.   Square Meters.   Square Meters.   Square Meters.   Square Meters.   Square Hectometers.   Square Hectometers.	IOInchesFeetYardsMilesSquare InchesSquare FeetSquare YardsSquare MilesAcres	MULTIPLY BY 0.394 3.280 1.094 0.621 0.155 10.764 1.196 0.386 
TO CHANGE   Centimeters   Meters.   Meters.   Kilometers   Square Centimeters   Square Meters.   Square Meters.   Square Meters.   Square Hectometers   Cubic Meters.	IOInchesFeetYardsMilesSquare InchesSquare FeetSquare YardsSquare MilesAcresCubic Feet	MULTIPLY BY 
TO CHANGE   Centimeters   Meters.   Meters.   Milometers   Square Centimeters   Square Meters.   Square Kilometers.   Square Hectometers   Cubic Meters.   Cubic Meters.	IOInchesFeetYardsMilesSquare InchesSquare FeetSquare YardsSquare MilesAcresCubic FeetCubic Yards	MULTIPLY BY 
TO CHANGE   Centimeters   Meters   Meters   Kilometers   Square Centimeters   Square Meters   Square Meters   Square Meters   Square Kilometers   Square Hectometers   Square Hectometers   Cubic Meters   Cubic Meters   Milliliters	IOInchesFeetYardsMilesSquare InchesSquare FeetSquare YardsSquare MilesAcresCubic FeetCubic YardsFluid Ounces	MULTIPLY BY 
TO CHANGE   Centimeters   Meters.   Meters.   Kilometers   Square Centimeters   Square Meters.   Square Meters.   Square Meters.   Square Meters.   Square Hectometers.   Square Hectometers   Cubic Meters   Cubic Meters   Milliliters   Liters	TO Inches Feet Yards Miles Square Inches Square Feet Square Yards Square Miles Acres Cubic Feet Cubic Feet Cubic Yards Fluid Ounces Pints	MULTIPLY BY 
TO CHANGE   Centimeters   Meters.   Meters.   Kilometers   Square Centimeters   Square Meters.   Square Meters.   Square Meters.   Square Meters.   Square Hectometers   Square Hectometers   Cubic Meters   Milliliters   Liters.	TO Inches Feet Yards Miles Square Inches Square Feet Square Yards Square Miles Acres Cubic Feet Cubic Feet Cubic Yards Fluid Ounces Pints Ouarts	MULTIPLY BY 
TO CHANGE   Centimeters   Meters   Meters   Kilometers   Square Centimeters   Square Meters   Square Meters   Square Meters   Square Meters   Square Hectometers   Cubic Meters   Cubic Meters   Milliliters   Liters   Liters	IOInchesFeetYardsMilesSquare InchesSquare FeetSquare YardsSquare MilesAcresCubic FeetCubic YardsFluid OuncesPintsQuartsCallons	MULTIPLY BY 0.394 3.280 1.094 0.621 0.155 10.764 1.196 0.386 2.471 35.315 1.308 0.034 2.113 1.057 0.264
TO CHANGE   Centimeters   Meters.   Meters.   Kilometers   Square Centimeters   Square Meters.   Square Hectometers.   Cubic Meters.   Cubic Meters.   Milliliters   Liters.   Liters.   ms	IOInchesFeetYardsMilesSquare InchesSquare FeetSquare YardsSquare MilesAcresCubic FeetCubic YardsFluid OuncesPintsQuartsGallonsOunces	MULTIPLY BY 0.394 3.280 1.094 0.621 0.155 10.764 1.196 0.386 2.471 35.315 1.308 0.034 2.113 1.057 0.264 0.025
TO CHANGE   Centimeters   Meters.   Meters.   Kilometers   Square Centimeters   Square Meters.   Square Hectometers.   Cubic Meters.   Cubic Meters.   Milliliters   Liters.   iters.   ms.   ograms	IOInchesFeetYardsMilesSquare InchesSquare FeetSquare YardsSquare MilesAcresCubic FeetCubic YardsFluid OuncesPintsQuartsGallonsOuncesPounde	MULTIPLY BY 0.394 
TO CHANGE Centimeters Meters. Meters. Kilometers Square Centimeters Square Meters Square Meters Square Meters Square Kilometers Square Hectometers Cubic Meters Cubic Meters Cubic Meters Milliliters Liters. Liters. Square Same Metric Three	IOInchesFeetYardsMilesSquare InchesSquare FeetSquare YardsSquare MilesAcresCubic FeetCubic YardsFluid OuncesPintsQuartsGallonsOuncesPoundsShort Tong	MULTIPLY BY 0.394 
TO CHANGE   Centimeters   Meters.   Meters.   Kilometers   Square Centimeters   Square Meters.   Square Hectometers   Cubic Meters   Cubic Meters   Liters.   Liters.   .ograms   Metric Tons.   Newton-Meters	IOInchesFeetYardsMilesSquare InchesSquare FeetSquare YardsSquare MilesAcresCubic FeetCubic YardsFluid OuncesPintsQuartsGallonsOuncesPoundsShort TonsPounds	MULTIPLY BY 0.394 3.280 1.094 0.621 0.155 10.764 1.196 0.386 2.471 35.315 1.308 0.034 0.264 0.035 2.205 1.102 0.728
TO CHANGE Centimeters	IOInchesFeetYardsMilesSquare InchesSquare FeetSquare YardsSquare MilesAcresCubic FeetCubic YardsFluid OuncesPintsQuartsGallonsOuncesPoundsShort TonsPoundsPoundsPounds	MULTIPLY BY 
TO CHANGE Centimeters Meters Meters Square Centimeters Square Meters Square Meters Square Meters Square Meters Square Hectometers Cubic Meters Cubic Meters Milliliters Liters Liters Square Salar Metric Tons Newton-Meters Kilopascals	IOInchesFeetYardsMilesSquare InchesSquare FeetSquare YardsSquare MilesAcresCubic FeetCubic YardsFluid OuncesPintsQuartsGallonsOuncesPoundsShort TonsPounds per Square Inch	MULTIPLY BY 
TO CHANGE Centimeters	IOInchesFeetYardsMilesSquare InchesSquare FeetSquare YardsSquare MilesAcresCubic FeetCubic YardsFluid OuncesPintsQuartsGallonsOuncesPoundsShort TonsPounds per Square InchMiles per Gallon	MULTIPLY BY 0.394 

### SQUARE MEASURE

1 Sq. Centimeter = 100 Sq. Millimeters = 0.155 Sq. Inches

- 1 Sq. Meter = 10,000 Sq. Centimeters = 10.76 Sq. Feet
- 1 Sq. Kilometer = 1,000,000 Sq. Meters = 0.386 Sq. Miles

### **CUBIC MEASURE**

1 Cu. Centimeter = 1000 Cu. Millimeters = 0.06 Cu. Inches 1 Cu. Meter = 1,000,000 Cu. Centimeters = 35.31 Cu. Feet

### TEMPERATURE

 $5/9(^{\circ}F - 32) = ^{\circ}C$ 

212° Fahrenheit is evuivalent to 100° Celsius

90° Fahrenheit is equivalent to 32.2° Celsius

32° Fahrenheit is equivalent to 0° Celsius

 $9/5C^{\circ} + 32 = {}^{\circ}F$ 



PIN: 015745